

Te Whare Wānanga o Otāgo

Invariant Natural Killer T Cells in Chronic Lymphocytic Leukaemia

Robert Weinkove

Submitted in fulfilment of the requirements of the degree of Doctor of Philosophy

November 2012

Malaghan Institute of Medical Research & Department of Pathology and Molecular Medicine, Wellington School of Medicine, University of Otago "It was afterwards...when the remedies had already been discovered, that men began to discuss the reasons for them: the Art of Medicine was not a discovery following upon reasoning, but after the discovery of the remedy, the reason for it was sought out"

Aulus Cornelius Celcus, De Medicina, ca. 30AD¹

Abstract

Invariant natural killer T (iNKT) cells recognise glycolipid antigens, such as the synthetic ligand alpha galactosylceramide (α -GalCer), in the context of CD1d. Upon recognition of α -GalCer on CD1d, iNKT cells become activated and rapidly produce cytokines. Recruitment of iNKT cells with α -GalCer results in potent antitumour activity in some pre-clinical cancer models. Effector mechanisms include direct cytotoxicity of iNKT cells against CD1d-expressing tumours in the presence of α -GalCer, transactivation of natural killer (NK) cells by iNKT cell-derived interferon gamma (IFN- γ), and iNKT cell-induced maturation and activation of dendritic cells (DCs), leading to enhanced T cell responses to DC-presented peptides.

Chronic lymphocytic leukaemia (CLL) is a clonal malignancy of B lymphocytes. Chemotherapy induces remission in most patients, but most eventually relapse. Allogeneic stem cell transplantation can be curative, but is not available to most patients due to advanced age or co-morbidities. Thus, CLL is an attractive candidate for cancer immunotherapy. This thesis aims to assess the iNKT cell/CD1d axis of patients, and to explore the possibility of exploiting iNKT cells for the immunotherapy of CLL.

Peripheral blood mononuclear cells (PBMCs) were isolated from patients with CLL, and from healthy age-matched controls, and analysed by flow cytometry. Absolute number and phenotype of circulating iNKT cells was similar in patients and controls, although patients exhibited a relative reduction in iNKT cells due to expansion of other T cell populations. iNKT cell frequency did not correlate with disease stage or with subsequent progression-free survival in patients with untreated CLL. Expression of CD1d on dendritic cells and monocytes from patients with CLL was similar to controls.

The cytokine profile of patient iNKT cells was similar to that of controls. *In vitro* proliferation of invariant natural killer T (iNKT) cells from patients with CLL was preserved, and iNKT cell lines generated from patients exhibited cytokine and cytotoxicity profiles similar to those from healthy controls, producing both Th1- and Th2-type cytokines, and lysing a target cell in a CD1d- and α -GalCer-dependent manner. Lysis of autologous CLL cells by iNKT cell lines was inefficient.

In vitro vaccine recall responses were enhanced by α -GalCer in PBMCs

containing high frequencies of iNKT cells. The treatment of leukaemic cells with α -GalCer enhanced their ability to stimulate proliferation of allogeneic PBMCs from healthy donors *in vitro*, largely due to iNKT cell proliferation. Leukaemic cells treated with α -GalCer induced proliferation of autologous iNKT cells, and α -GalCer treatment also led to enhanced proliferation of 'conventional' T cells.

These results indicate that the iNKT cell/CD1d axis is largely intact in patients with CLL and suggest that if low iNKT cell frequencies can be overcome, the adjuvant activity of iNKT cells might be exploited in cellular immunotherapy of CLL, for example by employing α -GalCer-pulsed leukaemic cells as a whole tumour vaccine.

To my wife Teresa and son Henry.

Acknowledgements

I would like to thank my PhD supervisors, Professor Franca Ronchese, Associate Professor Ian Hermans and Associate Professor John Carter for their guidance and support. I wish to thank the Malaghan Institute of Medical Research, in particular the Malaghan Family and Professor Graham Le Gros, for unstinting administrative, financial, social and educational support, and for the use of the Institute's first class facilities throughout my research. The administrative, IT and support staff at The Malaghan Institute provided an exceptionally well-organised working environment alongside a helpful attitude and good humour. Apii deserves special mention for her indefatigable hard work keeping the Institute spotless, and for her eternally cheerful and friendly nature.

Funding for this research was provided by the NZ Lottery Grants Board, the Genesis Oncology Trust, the Leukaemia and Blood Foundation and from a Genzyme New Investigators Scholarship awarded by the Haematology Society of Australia and New Zealand (HSANZ), with additional funding for materials from the Infinity Foundation. I thank each of these organisations for their generous support. I am particularly grateful to the HSANZ for the opportunity to speak at their meetings and for the award of the 2010 Albert Baikie Memorial Medal, both of which have provided encouragement during an occasionally difficult research journey.

Thank you to the examiners of this thesis for finding the time to read it and for their considered and helpful suggestions, and to the convener, Professor Delahunt. The University of Otago Wellington provided administrative support, a quiet and pleasant library to work in, and the kind award of their 2012 Clinical Research PhD prize.

I thank the National Institutes of Health Tetramer Core Facility for providing the CD1d monomer, and Vincenzo Cerundolo for provision of the CD1dtransfected and untransfected C1R cell lines.

Thank you to my collaborators at the Malaghan Institute, particularly to Evelyn Hyde for her assistance with experiments, to both Evelyn and Mark Yang for technical assistance, to Rachel Perret for showing me which way up to hold a pipette, and to Nori Enomoto for inspiring me to improve my figures. I am grateful to Kylie Price and Brigitta Mester for flow cytometry support, and particularly for coming in at weekends to perform sterile flow cytometric cell sorting for me. Thanks to my 'office mates', Troels Petersen and Martin Hunn for providing an excellent 'sounding board' for ideas.

Particular thanks go to Collin Brooks for assistance with some of the experiments, for providing editorial advice on the thesis, and for his social and moral support. Fortunately our low points rarely coincided, so we were able to support each other until the next gig, match or cycle ride—I look forward to doing so again! Thanks too to Peter Ferguson for enthusiasm, stimulating exchanges of ideas and for sheer commitment to the dinner jacket.

I wish to thank the team at Wellington Blood & Cancer Centre, for providing a clinical session to keep me in (medical) practice. I would particularly like to mention Glennis White for helpful and insteresting discussions regarding clinical flow cytometry, and Marina Dzhelali for advice on the preparation of my ethics committee application.

None of the research presented in this thesis would have been possible without the generous donation of time and blood by numerous patients and healthy donors, some of whom travelled a long way to participate in the study. I am grateful to all of you for supporting medical research.

Finally, thanks go to my wonderful family. I am grateful to my parents for moral support and understanding, and for not putting me off science forever. Particular thanks go to my lovely wife Teresa, for not only tolerating, but actively supporting a busy, distracted husband and father during the nights and weekends spent in preparation of this thesis – I love you.

Table of Contents

Ab	strac	t	vi
Ac	know	ledgements	x
Ta	ble of	f Contents	xi
Lis	st of T	Tables	xv
Lis	st of I	igures x	vii
Lis	st of A	Abbreviations x	ix
1	Intro 1.1	oduction Chronic lymphocytic leukaemia 1 1 Definition and Epidemiology	1 1 1
		1.1.1 Definition and Epidemiology 1.1.2 Diagnosis 1.1.3 Pathophysiology	1 2 3
	1.2	1.1.4 Prognostic factors 1.1.5 1.1.5 Treatment 1.1.5 Invariant natural killer T cells 1.1.1.1	11 15 21
		1.2.1 Antigen presentation by CD1 molecules	21 22
		1.2.3 CD1d ligands 1.2.4 Cytokine production 1.2.4 Cytokine production 1.2.5 Cytotoxicity	28 32 34
		1.2.6Interactions with other cells of the immune system1.2.7Role in health and disease	35 38
	1.3	1.2.8 Directing the iNKT cell response 4 Cellular immunotherapy 4 1.3.1 Immunological control of CLI 4	42 43
		1.3.1 Ininitial ological control of CLL 1 1.3.2 Strategies for cellular immunotherapy of CLL 1 1.3.3 Incorporating iNKT cells into immunotherapy 1	46 55
	1.4 1.5	Conclusions	51 52
2	Metl 2.1	nods General Materials and Methods	53 63

		2.1.1 Culture media and buffers	63
		2.1.2 C1R cell lines	64
		2.1.3 Cell counting	64
		2.1.4 Culture conditions	64
	2.2	Patients and ethical considerations	64
	2.3	Venepuncture and PBMC cryopreservation	64
	2.4	B cell depletion and enrichment	65
	2.5	Loading and tetramerization of CD1d monomer	65
	2.6	Immunofluorescent staining of cell surface markers for flow cytometry	66
	2.7	Immunofluorescent staining of intracellular cytokines, FoxP3, granzyme B	
		and ZAP-70 for flow cytometry	69
	2.8	Flow cytometric analysis	69
	2.9	IFN-v ELISpot assay	72
	2.10	iNKT cell proliferation assays	72
	2.11	Generation and maintenance of iNKT cell lines	73
	2.12	Cytokine bead array analysis	73
	2.13	Cytotoxicity assays	74
	2.10	2 13 1 Lactate dehydrogenase release assay	74
		2.13.1 Electric denyerogenuse release assay	74
	2 14	IFN-y a-GalCer and resignimod effects on CLL cell surface markers	75
	2.11	Vaccine recall responses	75
	2.10	2 15 1 Thymidine incorporation	75
		2 15 2 CFSF dilution	76
	2 16	In vitro alloresponse to CLL cells	76
	2.10	Autologous responses to CLL cells	76
	2.17	Statistical analyses	70
	2.10		//
3	Freq	uency and immunophenotype of iNKT cells in patients with CLL	79
	3.1	Introduction	79
	3.2	Aims	80
	3.3	Results	80
		3.3.1 Patient and control characteristics	80
		3.3.2 Immunomagnetic depletion of B cells	83
		3.3.3 Flow cytometric identification of invariant natural killer T cells	86
		3.3.4 Enumeration of circulating iNKT cells in patients and controls	86
		3.3.5 Analysis of T cell subsets	90
		3.3.6 Immunophenotype of iNKT cells	95
		3.3.7 iNKT cell numbers do not correlate with clinical outcome	95
	3.4	Summary	100
4	CD1	d expression on PBMC subsets	101
1	4.1	Introduction	101
	4.2	Aims	101
	43	Results	102
	1.0	4.3.1 Relative CD1d expression levels on PBMC subsets	102
		4.3.2 Myeloid dendritic cell numbers and CD1d expression	102
		4.3.3 Monocyte numbers and CD1d expression	105
	лл		100
	4.4	Summary	109

5	Fun	ction of	iNKT cells from patients with CLL	111
	5.1	Introdu	uction	111
	5.2	Aims .		112
	5.3	Results	s	112
		5.3.1	Cytokine production by iNKT cells	112
		5.3.2	Granzyme B expression by iNKT cells	115
		5.3.3	Proliferative capacity of iNKT cells	117
		5.3.4	Immunophenotype of iNKT cells after expansion	121
		5.3.5	Effect of leukaemic cells on iNKT cell proliferation	122
		5.3.6	Generation of iNKT cell lines	126
		5.3.7	Cytokine profile of iNKT cell lines	129
		5.3.8	Cytotoxicity of iNKT cell lines	129
	5.4	Summ	ary	136
6	Арр	lication	of iNKT cells to CLL immunotherapy	137
	6.1	Introdu	uction	137
	6.2	Aims .		138
	6.3	Results	3	138
		6.3.1	α -GalCer-induced cytokine release in whole PBMC cultures	138
		6.3.2	Effect of α-GalCer on APC co-stimulatory markers	139
		6.3.3	Activation of CLL cells by patient-derived iNKT cell lines is not α-	
			GalCer-dependent	144
		6.3.4	Lysis of autologous CLL cells by patient-derived iNKT cell lines	144
		6.3.5	Expression of MHC Class I, but not CD1d, is upregulated by CLL	
			cell exposure to IFN- γ	149
		6.3.6	Effect of α -GalCer on vaccine recall responses	149
		6.3.7	Pulsing CLL cells with α -GalCer induces allogeneic iNKT cell expan-	
			sion	152
		6.3.8	Autologous α-GalCer-pulsed CLL cells induce proliferation of iNKT	
			cells	156
		6.3.9	α-GalCer treatment of CLL cells leads to increased proliferation of	
			allogeneic PBMCs	156
		6.3.10	Effect of α -GalCer treatment is additional to that of TLR 7/8 ligation	156
		6.3.11	Increased allogeneic PBMC proliferation in response to α-GalCer-	
			pulsed CLL cells is primarily due to iNKT cell proliferation	161
		6.3.12	Autologous α-GalCer-pulsed CLL cells induce proliferation of con-	
			ventional T cells	161
	6.4	Summ	ary	165
7	Disc	ussion		167
	7.1	Introdu	uction	167
	7.2	Summ	ary and discussion of results	167
		7.2.1	Numbers and immunophenotype of iNKT cells	167
		7.2.2	Relationship between iNKT cell numbers and progression-free sur-	
			vival	173
		7.2.3	CD1d expression	175
		7.2.4	Function of INKT cells	177
		7.2.5	Potential roles of iNKT cells in immunotherapy	186

	7.2.6	Strengths and limitations of this study	194
7.3	Cellula	r immunotherapy of CLL	197
	7.3.1	Factors associated with successful cellular immunotherapy	197
	7.3.2	Potential role of iNKT cell engagement in CLL immunotherapy	203
	7.3.3	Limitations of iNKT cell-directed immunotherapies	205
	7.3.4	Proposal for a cellular immunotherapy of CLL	207
7.4	Conclu	sions and future directions	212
Referen	nces		215
Append	lices		271
A Stu	dy recru	itment materials	271
B Part	t <mark>icipant</mark>	information sheets	275
C Part	ticipant	questionnaire	283
D Con	sent for	rms	285
E Pub	lication	S	291

List of Tables

1.1	Diagnostic criteria for CLL	1
1.2	Immunophenotypic CLL scoring system	3
1.3	Murine models of CLL	.2
1.4	Rai clinical staging system 1	.3
1.5	Binet clinical staging system 1	.3
1.6	Cytogenetic findings in CLL	.4
1.7	Treatment indications 1	.5
1.8	CD1 isoforms	:3
1.9	Nomenclature of iNKT cells	:6
1.10	Comparison of the phenotype of T cells and iNKT cells 2	:9
1.11	CD1d-bound lipids 3	0
1.12	Microbial-derived CD1d ligands	;1
1.13	Cytokines and chemokines produced by iNKT cells	:2
1.14	Infections exacerbated by iNKT deficiency 3	69
1.15	Methods of generating T cell anti-tumour immunity 4	-8
1.16	Tumour-associated antigens in CLL	-9
1.17	Clinical trials of α -GalCer-pulsed DC	8
2.1	Fluorescent-labelled antibodies	57
2.2	Flow cytometer configuration	'0
2.3	Immunofluorescence panels for flow cytometry 7	'1
3.1	Control and patient characteristics	32
3.2	Immunomagnetic depletion of leukemic B cells	33
71	Circulating iNKT cell numbers reported in healthy donors	50
7.1	Circulating iNKT cell numbers reported in nationts with cancer	77
7.2	Reported immunophenotype of circulating iNKT calls in healthy donors	7/I
7.5	Sources of error in quantitative flow extemptry	7Q
7.4	Methods of polyalonal iNKT call line generation	0
7.5	Methods of polyciolidi liver cell ille generation	אי 10
7.0	Feators associated with guagessful cellular immunotherapy	γU λ4
1.1	ractors associated with successful centular minimunotherapy	14

List of Figures

1.1	CLL pathogenesis	5
1.2	Comparison of peptide and glycolipid presentation	24
1.3	Nomenclature of NKT cells	27
1.4	Structure of α -GalCer	32
1.5	Structure of selected α-GalCer derivatives	33
1.6	Interaction between DC, iNKT and T cells	36
1.7	Factors influencing the outcome of iNKT cell stimulation	43
3.1	Age and peripheral blood counts of study participants	81
3.2	Immunomagnetic depletion of CLL cells	84
3.3	Effect of immunomagnetic B cell depletion on PBMC subsets	85
3.4	Effect of immunomagnetic B cell depletion on APC and T cell activation	87
3.5	Flow cytometric gating to identify iNKT cells	88
3.6	Comparison of CD1d tetramer and 6B11 antibody clone for iNKT cell identi-	0.0
~ -	fication	89
3.7	Relative and absolute iNKT cell numbers in patients and controls	91
3.8	iNKT cell frequency stratified by stage, lymphocyte count, ZAP-70 status, and	
_	prior treatment	92
3.9	Relative proportions of T cell subsets in controls and patients with CLL	93
3.10	Absolute numbers of T cell subsets in controls and patients with CLL	94
3.11	Immunophenotype of circulating iNKT cells	96
3.12	Correlation between iNKT cell numbers and iNKT cell CD4 status	98
3.13	Progression-free survival does not correlate with iNKT cell numbers	99
4.1	Expression of CD1d on PBMC from patients with CLL	103
4.2	Numbers of mDC in patients and controls	104
4.3	Expression of CD1d on patient and control mDC	106
4.4	Absolute numbers of monocyte subsets	107
4.5	CD1d expression on monocyte subsets	108
5.1	α-GalCer-induced IFN-γ ELISpot assay	113
5.2	IFN- γ ELISpot production by patient and control PBMCs in response to α -GalCer	114
5.3	Intracellular cytokine staining of iNKT cells	116
5.4	Differential cytokine production by CD4+ and CD4- subsets of iNKT cells	117
5.5	Cytokine production by T and iNKT cells in patients and controls	118
5.6	Intracellular granzyme B expression of iNKT cells	119
5.7	Optimising cytokine conditions for iNKT cell proliferation	120

5.8	iNKT cell proliferation from patients and controls	121
5.9	Immunophenotype of iNKT cells after expansion	123
5.10	Effect of IL2, IL7 and IL15 on iNKT subsets	125
5.11	Proliferation of iNKT cells in the presence of leukaemic cells	126
5.12	Effect of addition of autologous CLL cells on iNKT cell expansion	127
5.13	Effect of feeder cells on iNKT cell expansion	128
5.14	Flow cytometric cell sorting of iNKT cells	129
5.15	Purity and numbers of iNKT cell lines	130
5.16	Phenotype of iNKT cell lines	131
5.17	Intracellular staining of iNKT cell lines	132
5.18	Cytokine production by iNKT cell lines	133
5.19	Phenotype of iNKT cell lines	134
5.20	Cytotoxicity of iNKT cell lines	135
6.1	Detection of cytokines in supernatants of cultured PBMCs	140
6.2	α -GalCer induced cytokine production in patient and control PBMCs	141
6.3	Effect of α -GalCer on monocyte activation in PBMC cultures	142
6.4	Effect of addition of α -GalCer on B cells	143
6.5	Activation of CLL cells by iNKT cell lines is α -GalCer-independent	145
6.6	Expression of CD40 on CLL cells and CD40L on iNKT cells	146
6.7	VITAL assay to assess iNKT cell target killing in whole PBMCs	147
6.8	Lysis of CLL cells by patient-derived iNKT cell lines	148
6.9	Effect of IFN- γ on CLL MHC Class I and CD1d expression	150
6.10	Effect of α -GalCer on vaccine recall responses	151
6.11	Immunophenotype of divided cells after α -GalCer-enhanced vaccine recall re-	1 = 0
6 1 0	sponse	153
6.12	Influenza vaccine recall responses with and without α -GalCer	154
6.13	Proliferation of allogeneic iNKT cells in response to α -GalCer pulsed CLL cells	155
6.14	α -GalCer-pulsed CLL cells support proliferation of autologous iNKT cells	157
6.15	α -GalCer treatment of CLL cells leads to increased proliferation of allogeneic	1 = 0
<i>с</i>		158
6.16	Upregulation of CD86 on CLL cells by resignimod	159
6.17	Resignimed, but not α -GalCer, leads to upregulation of MHC Class I and II on	1.00
6 1 0		160
6.18	Effect of α -GalCer on allogeneic PBMC proliferation in response to CLL cells is	1 (0
6.10		162
6.19	Increased allogeneic PBMC proliferation in response to α -GalCer treatment of CLL with the law to DMZT with a set for a set of the	100
6.00	CLL cells is due to iNKT cell, not conventional' T cell, proliferation	163
6.20	α -GalCer-pulsed CLL cells may support proliferation of 'conventional' T cells \cdot .	164
7.1	A whole tumour cell vaccination strategy for CLL	209
7.2	Bationale of tumour vaccination strategy	210
,		210

List of Abbreviations

α-GalCer Allo-SCT Auto-SCT ADCC AEC AIHA ANOVA	alpha galactosylceramide allogeneic stem cell transplantation autologous stem cell transplantation antibody-dependent cellular cytotoxicity 3-amino-9-ethylcarbazole autoimmune haemolytic anaemia analysis of variance
APC	allophycocyanin
APC	antigen-presenting cell
APRIL	a proliferation-inducing ligand
ATG	anti-thymocyte globulin
BAFF BCG	B-cell activation factor belonging to the TNF family Bacille Calmette-Guerin
BCR	B cell receptor
BDCA-1	Blood Dendritic Cell Antigen 1
$\beta_2 \mathbf{M}$	beta 2 microglobulin
Btk	Bruton's tyrosine kinase
C1R	C1R cell line (HLA-A and -B negative human B cell line)
C1R-CD1d	CD1d-transfected C1R cell line
CAR	chimeric antigen receptor
CCR7	C-C chemokine receptor type 7
CDK	cyclin-dependent kinase
CDR	complementarity determining region
CFSE	carboxyfluorescein succinimidyl ester
cIMDM	complete Iscove's Modified Dulbecco's Medium
CLL	chronic lymphocytic leukaemia
CMV	cytomegalovirus
CTLA-4	Cytotoxic T Lymphocyte Antigen-4
CIO	CellTracker [™] Orange (5-(and-6)-(((4-chloromethyl)benzoyl)- amino)tetramethylrhodamine)
DAPI	4',6-diamidino-2-phenylindole
DC	dendritic cell
DiOC6	dihexyloxacarbocyanine iodide

DLI	donor lymphocyte infusion
DMSO	dimethyl sulfoxide
DPBS	Dulbecco's phosphate buffered saline
EDTA	ethylenediaminetetraacetic acid
ELISpot	enzyme-linked immunosorbent spot
Fab	antigen-binding fragment
FACS	fluorescence-activated cell sorting
FasL	Fas ligand
Fc	crystallizable fragment
FCR	fludarabine, cyclophosphamide and rituximab
FCS	foetal calf serum
FISH	fluorescent in-situ hybridisation
FITC	fluorescein isothiocyanate
FMO	fluorescence minus one
Foxp3	forkhead box P3
GM-CSF	granulocyte-macrophage colony stimulating factor
GVHD	graft versus host disease
GVL	graft versus leukaemia
HEPES	4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid
HLA	human leukocyte antigen
HMB-PP	(E)-4-hydroxy-3-methyl-but-2-enyl pyrophosphate
HRP	horseradish peroxidase
ICS	intracellular cytokine staining
IFN-γ	interferon gamma
Ig	immunoglobulin
IgG	immunoglobulin G
IgH	immunoglobulin heavy chain
IgVH	immunoglobulin heavy chain variable region
IL	interleukin
IMDM	Iscove's Modified Dulbecco's Medium
IMiD	immunomodulatory drug
iNKT	invariant natural killer T
IPP	isopentyl pyrophosphate
KIR	killer Ig-like receptor
LDH	lactate dehydrogenase
LDT	lymphocyte doubling time

Lin 1	lineage cocktail 1
mAb	monoclonal antibody
MAIT	mucosal-associated invariant T cells
MART-1	melanoma antigen recognised by T cells-1
MBL	monoclonal B cell lymphocytosis
mDC	myeloid dendritic cell
MDSC	myeloid-derived suppressor cell
MHC	major histocompatibility complex
MLR	mixed lymphocyte reaction
moDC	monocyte-derived dendritic cell
MRD	minimal residual disease
NIH	National Institutes of Health
NK	natural killer
NKT	natural killer T
NOD	non-obese diabetic
NY-ESO-1	New York esophageal squamous cell carcinoma-1
NZB	New Zealand Black
OVA	ovalbumin
PBMC	peripheral blood mononuclear cell
PBS	phosphate-buffered saline
PCR	polymerase chain reaction
PD-1	Programmed Cell Death-1
pDC	plasmacytoid dendritic cell
PE	phycoerythrin
PerCP	peridinin-chlorophyll protein complex
PHA	phytohaemagglutinin
PI	propidium iodide
PI3K	phosphatidylinositol-3-kinase
PIM	phosphatidylinositol mannoside
PMA	phorbol 12-myristate 13-acetate
RIC-allo	reduced intensity conditioning allogeneic stem cell transplantion
RPMI	Roswell Park Memorial Institute medium
RT-PCR	reverse transcription quantitative polymerase chain reaction
scFv	single-chain variable fragment
SCID	severe combined immunodeficiency
SEM	standard error of the mean
SFU	spot-forming unit

Signi Suitace Igni	N	surface IgM
--------------------	---	-------------

- SLL small lymphocytic lymphoma
- **SLVL** splenic lymphoma with villous lymphocytes
- **Syk** spleen tyrosine kinase
- TAA tumour-associated antigen
- **TACI** transmembrane activator, calcium-modulator, and cyclophilin ligand interactor
- **TAM** tumour-associated macrophage
- **TCR** T cell receptor
- **TGF-***β* transforming growth factor beta
- Th1 T helper 1
- Th2 T helper 2
- TLR Toll-like receptor
- **TNF-** α tumour necrosis factor alpha
- TRAIL tumour necrosis factor-related apoptosis-inducing ligand
- **TREC** T cell receptor gene rearrangement excision circle
- T_{reg} regulatory T cell
- **VITAL** *in vivo/in vitro* technique for assessing lysis
- WT1 Wilms' tumour 1
- ZAP-70 zeta chain-associated protein 70

Chapter 1

Introduction

1.1 Chronic lymphocytic leukaemia

1.1.1 Definition and Epidemiology

CHRONIC LYMPHOCYTIC LEUKAEMIA (CLL) is an indolent B cell malignancy, characterised by an infiltration of mature clonal B lymphocytes in the peripheral blood and bone marrow. The clinical features of CLL include a lymphocytosis, lymphadenopathy, hepatosplenomegaly, constitutional symptoms, propensity to infection, bone marrow failure due to replacement of the haematopoietic compartment with malignant lymphocytes and a predisposition to autoimmune cytopenias.

CLL is distinguished from other indolent B cell malignancies by a characteristic immunophenotype,² which allows the diagnosis to be established by flow cytometric analysis of the peripheral blood in the majority of cases.³ Diagnostic criteria for CLL are listed in table 1.1.

Many healthy individuals harbour a circulating clonal B cell population with an immunophenotype identical to that of CLL, but do not meet the numeric criteria for CLL (the B lymphocyte count is $< 5.0 \times 10^9$ /L). This condition is known as monoclonal B cell lymphocytosis (MBL). MBL rises in frequency with age, being identified in over 5% of healthy individuals over 60 years of age,^{6,7} and can be considered pre-malignant: progression to CLL requiring treatment occurs at a rate of 1 - 2% per year.⁶

All of the following criteria should be met:
\geq 5 ×10 ⁹ /L B lymphocytes in peripheral blood
\leq 55% of lymphocytes show atypical morphology
Low surface IgM or IgD expression with κ or λ restriction
B cell surface antigen expression (CD19, CD23)
CD5 expression

Table 1.1 Diagnostic criteria for CLL^{4,5}

Small lymphocytic lymphoma (SLL) is an indolent B-cell lymphoma with similar immunophenotypic and cytogenetic characteristics to CLL, and is considered its lymphomatous counterpart. Many patients with CLL develop significant lymphadenopathy during the course of their disease, and many patients initially diagnosed with SLL will enter a leukaemic phase (with > 5×10^9 /L malignant B cells in the peripheral blood) and be recategorised as having CLL. The term CLL/SLL is often used to emphasise the overlap between these two clinical entities.

CLL is the most common leukaemia in the Western World. It is primarily a disease of the elderly, with a median age at diagnosis of around 70 years.^{8,9} CLL has a strong association with age, the relative risk in an individual 70 – 79 years old being 352 times that of an individual < 40 years of age.⁸ There is a male predisposition, nearly 60% of cases occurring among males.⁸

The incidence of CLL varies with ethnicity, age-adjusted incidence ratios being approximately 0.75 and 0.23 for Americans of Black or Asian descent compared to White Americans, respectively.¹⁰ The incidence of CLL in the Far East is much lower than in Western countries, although CLL rates appear to be rising in more recent birth cohorts.¹¹ Asian migrants to the United States retain their lower risk of CLL, implying that the differences in rates by ethnicity are largely genetic or are due to antenatal or early childhood environmental exposures.¹²

A familial predisposition to CLL can be identified: A first-degree relative of a patient with CLL has a relative risk of ~8.5 for developing CLL, and of ~1.9 for the development of other indolent B cell malignancies.¹³ In keeping with this, first-degree relatives of patients with CLL also have an increased chance of developing MBL (from 3.5% to 13.5%).¹⁴ A number of single nucleotide polymorphisms have been associated with increased risk of CLL, including polymorphisms of the genes for the cell cycle regulator cyclin H, the apoptotis-associated protein caspase 8, the chemokine receptor CCR7 and the interleukin IL-16.¹⁵

Specific environmental factors appear to predispose to CLL. A history of benzene exposure is associated with an increased risk of CLL, ¹⁶ as is an occupational history including crop growing and horticulture, textile bleaching and dyeing, or plastics manufacturing. ^{17,18}

1.1.2 Diagnosis

Lymphocytosis, often noted incidentally on a routine blood count, is usually the first indication of CLL. The blood film shows a monomorphic population of small mature lymphocytes with scanty cytoplasm and clumped chromatin. Prolymphocytes, with a prominent nucleolus, are present in a minority of cases. CLL cells frequently become disrupted during preparation of the blood film, leading to characteristic 'smudge' cells on routine microscopy. The propensity of CLL cells to form these 'smudge' cells has been linked to low expression of the cytoskeletal protein vimentin in some patients. A smudge cell percentage of 30% appears to identify a subset of patients with inferior prognosis.¹⁹

The surface immunophenotype of CLL cells resembles that of normal mature antigenexperienced B lymphocytes with the exception of aberrant CD5 expression: Alongside

Surface marker	Score, according to	Score, according to surface expression		
Burrace marker	0	1		
SmIg *	Strong	Weak		
CD5	Negative	Positive		
CD23	Negative	Positive		
FMC7 [†]	Positive	Negative		
CD79b	Strong	Weak		

Table 1.2 Immunophenotypic CLL scoring system Scores are summed for expression of each surface marker. A score \geq 4 strongly supports a diagnosis of CLL.²⁰

* Surface membrane immunoglobulin

[†] Antibody clone directed against an epitope of CD20

CD5, CLL cells express B cell markers including CD19 and CD23, and express low levels of surface IgM (sIgM) (with or without IgD), CD79b and the CD20 epitope FMC7. These markers have been combined to create a simple flow cytometric scoring system, which has been prospectively validated and is widely used for the diagnosis of CLL (table 1.2).²⁰

Peripheral blood flow cytometry is usually sufficient to secure a diagnosis of CLL. However, in those with a CLL score of \leq 3, or where SLL is suspected, bone marrow and/or lymph node biopsy may be necessary. Where performed, the bone marrow aspirate is hypercellular, with increased numbers of mature lymphocytes, typically constituting > 30% of nucleated cells. On bone marrow trephine histology, CLL infiltrates may be interstitial, nodular or diffuse. Immunohistochemistry can support the flow cytometric diagnosis, CLL cells typically being positive for CD5, CD20 (weak), CD23, CD43 and CD79a, and negative for CD10 and cyclin D1.²¹ Lymph node histology in CLL and SLL shows a diffuse infiltration with monomorphic small round lymphocytes. Ill-defined pale nodular regions, known as pseudofollicles, can be appreciated at low magnification. Immunohistochemical findings in the lymph node are as in the bone marrow.²²

1.1.3 Pathophysiology

1.1.3.1 Cell of origin

The malignant cells in CLL are clonal B lymphocytes. Clonality can be demonstrated both by light chain restriction,²³ and by molecular evidence of clonal immunoglobulin heavy chain variable region (IgVH) gene usage. The emergence of this clonal B cell population precedes the clinical development of CLL: in a prospective cohort study, 44 of 45 patients (98%) who subsequently developed CLL had a detectable B cell clone (a MBL) in the peripheral blood at an average of 32 months before diagnosis.²⁴ These MBL clones frequently exhibit genomic aberrations typical of CLL, such as trisomy 12 and 13q deletion.⁶

Since aberrant CD5 expression is a hallmark of CLL, normal CD5 expressing B cells have

been proposed as the cell of origin for the leukaemia. CD5 is expressed by only 3% of circulating B cells in healthy humans, but is more frequently expressed by B cells in the marginal and mantle zones of lymphatic tissue and in foetal spleen.²⁵ However, CLL cells do not share other typical immunophenotypic characteristics with marginal zone or mantle zone B cells. Indeed, gene expression profiling indicates that CLL cells are most closely related to memory B cells,²⁶ a finding which would be consistent with their expression of the memory markers such as CD27.

Several lines of evidence indicate that the leukaemic cells in CLL derive from antigenexperienced B cells. In common with other antigen-experienced B cells, clonal IgM CLL cells are capable of isotype class switching, secreting IgG and IgA upon CD40 ligation *in vitro*.²⁷ The leukaemic cells of CLL also express a mannosylated form of sIgM, which is associated with antigen exposure.²⁸

Studies of the immunoglobulin gene provide evidence of antigen exposure in CLL cells. The use of VH genes CLL is biased: Certain VH genes are preferentially expressed in CLL, including VH4-34, VH3-07, VH1-69 in Western countries.²⁹ Interestingly, VH usage varies by ethnicitity: compared to cases of CLL from Western countries, cases in Japan, China and Iran use VH4 genes more frequently, and VH1 genes less frequently.³⁰ Usage of the D and JH genes is also non-random, varying significantly depending on the VH gene used.^{29,31} The length and composition of the CDR3 (complementarity-determining region) varies with VH usage, implying that selection has taken place,²⁹ stereotyped CDR3 sequences accounting for 20% – 30% of cases of CLL.^{32–34}

Somatic hypermutation is evident in a proportion of CLL cases, with around 50% of cases demonstrating $\geq 2\%$ mutation from the germ line sequence. Somatic hypermutation is more frequent in CLL cells than in normal circulating CD5+ or CD5- B cells.²⁹ Finally, gene expression profiling indicates that CLL cells are most closely related to memory B cells.²⁶

How, then, can the characteristic aberrant CD5 expression of CLL B cells be reconciled with a memory B cell origin? The expression of CD5 on B cells can be induced by chronic antigen stimulation: normal human B cells express CD5 *in vitro* after stimulation with phorbol 12-myristate 13-acetate (PMA), ²⁵ after cross-linking of sIgM, ³⁵ or following chronic stimulation by autoantigens. ³⁶

Other features of CLL cells also suggest that they are chronically antigen-stimulated *in vivo*. Leukaemic cells directly isolated from patients with CLL express activation markers such as CD23, CD25, CD69 and CD71.³⁷ CLL cells express low levels of sIgM and the IgM-associated heterodimer, CD79a/CD79b,³⁷ consistent with sIgM downregulation secondary to antigen exposure, and CLL cells spontaneously re-upregulate sIgM upon *in vitro* culture.³⁸

CD5 itself may play a major role in generation of the CLL cell phenotype: CD5 is phosphorylated upon engagement of the B cell receptor (BCR), leading to upregulation of IL-10, transforming growth factor beta (TGF- β) and surface activation markers including CD54, and down-regulation of co-stimulatory antigens such as CD80 and CD86,³⁹ all phenotypic features of CLL.

Together these lines of evidence suggest that the leukaemic clone in CLL is an antigenexperienced memory B lymphocyte, which upregulates CD5 and CD23 due to chronic



Figure 1.1 Model of CLL pathogenesis, adapted from Klein and Dalla-Favera⁴⁸

antigen stimulation and due to microenvironmental factors. As will be discussed, cases of CLL with IgVH genes that have not undergone somatic hypermutation tend to produce polyspecific autoantibodies, suggesting that the cell of origin in 'unmutated' CLL is a B cell that has undergone T cell-independent activation. In contrast, cases with mutated IgVH genes tend to produce antibodies of narrower specificity, and in some cases are class switched, implying that their cell of origin has been activated in a T cell-dependent manner.⁴⁰

1.1.3.2 Nature of the antigen

If the leukaemic cells are antigen-experienced, what is the antigen? It has long been recognised that around half of CLL cases produce autoantibodies, which are frequently polyspecific.^{41,42} This could be considered consistent with the CD5 positivity of CLL, as CD5+ B cells are often considered characteristic producers of 'natural antibodies': In the mouse, CD5+ B cells are termed B-1a cells and produce low affinity, polyreactive IgM natural antibodies which may be directed against microorganisms or against autoantigens.⁴³

Antibodies from cases of CLL with both unmutated and mutated IgVH genes can be autoreactive. Antibodies cloned from cases of unmutated CLL are frequently polyreactive, whereas those cloned from cases of mutated CLL exhibit more restricted reactivity. However, when nonpolyreactive antibodies cloned from mutated CLL cases were reverted to their germline sequence, they became polyreactive, ⁴⁴ suggesting that for both mutated and unmutated CLL forms, the cell of origin is a self-reactive B cell.

Specific antigens identified as targets of CLL cell BCRs include cytoskeletal proteins such as non-muscle myosin heavy chain IIa⁴⁵ and vimentin,⁴⁶ *Streptococcus pneumoniae* polysaccharides,⁴⁶ and oxidised lipids.^{46,47}

A schematic model of the pathogenesis of CLL is given in figure 1.1.⁴⁸ In this model, naïve

Chapter 1. Introduction

B cells are stimulated by antigen, and depending on the nature of the antigen, undergo activation in a germinal centre with CD4+ T cell help (T-dependent activation), which results in somatic hypermutation and affinity maturation, or outside of the germinal centre without T cell help (T-independent activation), in which case there is no somatic hypermutation. Because the BCR is directed against a common antigen (autoantigen, oxidised lipid or common bacterial antigen), these B cells are subjected to chronic antigenic stimulation. Genetic changes, such as 13q deletion take place and drive clonal proliferation of the B cell, which may result in monoclonal B cell lymphocytosis or early CLL. Eventually, the acquisition of further genetic changes, such as TP53 mutation/deletion or ATM mutation/deletion may result in transformation and the clinical phenotype of CLL.

1.1.3.3 CLL kinetics and apoptosis

Anti-apoptotic proteins are markedly overexpressed in CLL, including Bcl-2⁴⁹ and the caspase inhibitor, survivin.⁵⁰ This may appear difficult to reconcile with the high spontaneous apoptosis rates observed when CLL cells are cultured *in vitro*. However, the spontaneous apoptosis of CLL cells can be inhibited by co-culture with stromal cells,^{51,52} IL-4,⁵³ or by cross-linking of surface IgM.⁵⁴ Thus, factors in the tumour microenvironment contribute to a resistance to apoptosis in CLL cells *in vivo*.

Despite *in vivo* resistance to apoptosis and typically slow clinical progression, CLL cannot be considered a static condition: CLL cells undergo measureable degrees of 'birth' and 'death', as determined by *in vivo* incorporation of deuterated water into DNA. In one study, birth rates ranged from 0.1 - 1.0% of circulating CLL cells daily. Death rates were in a similar range, the difference between birth and death rates correlating with changes in the lymphocyte count.⁵⁵

A relatively small proportion of circulating CLL cells actively proliferate, around 99.5% of the circulating population being in the G0/G1 phase of the cell cycle.⁵⁶ It has been proposed that the majority of CLL cell proliferation takes place within pseudofollicles in secondary lymphoid tissue,⁵⁷ where CLL cells interact with activated CD4+ T cells and upregulate CD38.⁵⁸ Consistent with this hypothesis, the majority of the dividing circulating CLL cells are CD38+, perhaps representing a population that has recently emerged from lymphoid tissue.⁵⁹

Together, the above findings give rise to the prevailing view that CLL is an active process involving birth and death of a substantial number of leukaemic cells daily, and that leukaemic cells require appropriate environmental signals in order to survive and proliferate.⁶⁰ There is therefore considerable interest in developing therapeutic strategies that deprive CLL cells of their supporting microenvironmental stimuli.

1.1.3.4 Clonal evolution

Clonal evolution of CLL after diagnosis has been observed: In longitutinal series, a quarter of patients acquire a new cytogenetic abnormality over five years. 13q- is most commonly acquired, although acquisition of trisomy 12, 17p- and 11q- have also been observed.^{61,62} Clonal evolution can also be observed in individual patients at a single time point: in one

study, three out of four patients exhibited differential copy number aberrations between the circulating CD38+ and CD38- CLL cell subsets, one of which was at the TP53 locus, and another of which was at the ATM locus.⁶³

Clinical progression of the malignant clone is also well recognised in CLL: by 10 years from diagnosis, around 15% of patients will experience transformation to diffuse large B cell lymphoma (DLBCL), a phenomenon known as Richter's transformation. The risk of Richter's transformation is higher in patients without the good-prognosis 13q- deletion.⁶⁴ In approximately 80% of cases of Richter's transformation, DLBCL arises from the CLL clone, as determined by IgVH analysis. The remaining cases represent a new B cell clone. Richter's transformation is frequently associated with acquisition of a TP53 mutation or 17p-.

1.1.3.5 Immune dysfunction

Immune dysfunction has long been recognised as a feature of CLL,⁶⁵ and includes predisposition to autoimmune cytopenias and to infections.

Autoimmunity is a common feature of CLL,⁶⁶ and is usually directed against cells of myeloid origin. Autoimmune haemolytic anaemia (AIHA) is seen in 10 - 25% of patients during the course of their disease and autoimmune thrombocytopenia in 2%. Autoimmune neutropenia and pure red cell aplasia are less frequently seen. Despite the potential of CLL cells to produce autoreactive monoclonal IgM antibodies, the autoimmune cytopenias observed in CLL are usually due to polyclonal IgG antibodies produced by non-clonal B cells.⁶⁷

In keeping with the notion that autoimmune cytopenias are not directly mediated by the leukaemic clone, successful treatment of CLL with radiotherapy⁶⁸ or with chemotherapy such as alkylating agents or fludarabine can precipitate rather than ameliorate overt AIHA.⁶⁹ Interestingly, chemotherapy with a combination of fludarabine and the alkylating agent, cyclophosphamide, results in a lower rate of AIHA than that seen with fludarabine therapy alone.⁶⁹

One intriguing explanation for the phenomenon of autoimmune cytopenias in CLL is the possibility that leukaemic cells themselves are able to present certain autoantigens to CD4+ T cells. Although CLL cells are poor antigen presenters *in vitro*, they are numerous, and CLL cells from patients with anti-Rh related AIHA have been shown to present Rh autoantigens to autologous T helper cells.⁷⁰

Numerous specific defects of immunological function have been described in CLL, affecting innate immunity, humoral immunity, antigen presenting cell function, T cell function, and levels of circulating cytokines levels. Patients with CLL are particularly prone to bacterial infections including pneumonia, septicaemia, cellulitis and urinary tract infections.⁷¹

Innate immunity Dysfunction of innate immunity has been reported in CLL. Despite increased numbers of circulating NK cells, their capacity to lyse target cells *in vitro* is reduced.^{72,73} In some patients with CLL, defects of neutrophil migration, chemotaxis and

Chapter 1. Introduction

oxidative burst generation are observed, and correlate with a history of recurrent infections.⁷⁴ Reduced serum levels of complement proteins have been described in patients with CLL, and are lower in patients with more advanced clinical stages.⁷⁵

Humoral immunity Hypogammaglobulinaemia is a common clinical feature of CLL, and is associated with an increased risk of bacterial infections in patients with CLL.^{65,76} T cells activated *in vitro* in the presence of CLL cells upregulate CD30, exhibit reduced CD40 ligand expression and impair immunoglobulin class switching in non-malignant B cells.⁷⁷

Antigen Presenting Cells The numbers of myeloid dendritic cells (mDCs) and plasmacytoid dendritic cells (pDCs) are reduced as a proportion of circulating nucleated cells in patients with CLL,⁷⁸ as would be expected due to the large numbers of circulating leukaemic cells. Dysfunction of mDCs has also been described in CLL: One study suggested that CLL DCs exhibit normal function but possess an abnormal cytokine profile, with increased IL-10 and reduced IFN- γ production.⁷⁹ Another study demonstrated reduced expression of the co-stimulatory molecule CD80, reduced stimulatory ability in allogeneic mixed lymphocyte reactions, and reduced IL-12 secretion by mDCs from patients with CLL.⁸⁰ Soluble CLL cell factors emulated these defects in DCs generated *in vitro* from healthy donors, suggesting that soluble factors from the CLL cells themselves are responsible for DC dysfunction.⁸⁰ In keeping with this observation, functional monocyte-derived dendritic cells (moDCs) can be generated from patients provided there is no contamination with CLL cells.^{81,82}

T cells Patients with CLL have markedly increased numbers of T cells.^{83,84} This T cell expansion has been attributed to interactions between B-cell activation factor belonging to the TNF family (BAFF) and a proliferation-inducing ligand (APRIL) expressed on the maligant B cells, and transmembrane activator, calcium-modulator, and cyclophilin ligand interactor (TACI) expressed on T cells.⁸⁵

T cell expansion may be a critical component in the pathogenesis of CLL: a xenograft model suggests that T cells are necessary for the maintenance of the leukaemic B cell population.⁸⁶ In light of this, is interesting to note that two of the most effective treatments for CLL, fludarabine and alemtuzumab, both cause profound and prolonged depletion of T cells.

Despite their increased overall number, numerous abnormalities and functional defects of T cells have been described in patients with CLL. These include:

- Numeric abnormalities, such as inversion of the CD4/CD8 ratio, with CD8+ T cells being relatively increased compared to the CD4+ subset, ^{83,84} and oligoclonal T cell expansions within both the CD4+⁸⁷ and the CD8+⁸⁸ subsets of T cells. In one instance, a large clonal expansion of CD8+ T cells from a patient with CLL had specificity for the leukemic clone.⁸⁹
- Immunophenotypic abnormalities, such as reduced T cell expression of CD40 ligand⁹⁰ (the supernatant from activated CLL cells inhibits activation-induced upreg-

ulation of CD40 ligand on healthy donor T cells⁹¹). Reduced expression of CD28 and the ζ chain of CD3 on T cells is seen in patients with CLL, and is more marked in advanced than in early clinical stages.⁹² Increased T cell Cytotoxic T Lymphocyte Antigen-4 (CTLA-4, or CD152) expression has been observed on T cells from patients with CLL. CTLA-4 binds to the co-stimulatory molecules CD80 and CD86, inhibiting T cell activation, and *in vitro*, CTLA-4 blockade significantly enhances T cell proliferative responses to autologous, CD40-activated CLL cells.⁹³

- Functional abnormalities, such as impaired *in vitro* T cell expansion in response to mitogenic stimuli, ⁹⁴ and increased T cell IL-4 production. ⁹⁵ The IL-4 receptor is expressed by CLL cells, and there is evidence that IL-4 may protect CLL cells from apoptosis. ⁹⁶ Defective immunological synapse formation between T cells and antigen-presenting cells (APCs) has been described in patients with CLL. This appears to be an effect of the malignant clone, as healthy donor T cells which have previously been cultured with CLL cells fail to generate normal immunological synapses, and effect which is dependent upon cell:cell contact. ⁹⁷
- An increased frequency of circulating regulatory T cell (T_{reg}) cells is seen in early stage CLL, higher T_{reg} numbers being associated with a shorter time to treatment.⁹⁸ Higher frequencies of circulating T_{reg} s are associated with reduced CD8+ IFN- γ enzyme-linked immunosorbent spots (ELISpots) in response to CLL tumour-associated antigens (TAAs).⁹⁹ A possible explanation for the increased T_{reg} numbers is that CLL cells express high levels of CD200, and ligation of the CD200 receptor on APCs leads to enhanced induction of Tregs.¹⁰⁰
- The gene expression profile of CD4+ and CD8+ T cells is abnormal in patients with CLL. CD4+ T cells from CLL patients show changes in the JNK and p38 MAPK pathways, which are involved in regulating Th1 and Th2 differentiation. CD8+ T cells from CLL patients exhibit changes in the expression of genes relating to the cytoskeleton and intracellular trafficking, potentially accounting for reduced cytotoxicity. CLL cells are able to induce similar gene expression changes in healthy donor T cells, in a manner which requires cell-cell contact.¹⁰¹

Cytokines and other soluble factors Abnormalities of plasma and PBMC culture supernatant cytokine levels have been observed in patients with CLL. These include higher plasma TGF- β levels in patients with CLL than in healthy controls¹⁰²—CLL cells secrete TGF- β upon *in vitro* culture, and TGF- β can inhibit DC migration and T cell stimulatory activity.¹⁰³

Patients with CLL have raised serum IL-6 and IL-10 levels, higher circulating levels of these cytokines being associated with an inferior prognosis.¹⁰⁴ IL-6 from CLL cells may result in blunting of T cell activation, and a Th2 skewing of T cell responses.⁹¹ IL-10 inhibits the antigen-presenting capacity of monocytes to abrogate antigen-specific T cell proliferation.¹⁰⁵

The leukaemic cells in CLL express CD25 (the IL-2 receptor alpha), and may have the capacity to both absorb IL-2 directly, ¹⁰⁶ and to secrete soluble CD25¹⁰⁷ which acts as a decoy receptor for IL-2, potentially depriving T cells of an activating and proliferative signal. CLL cells can also release soluble CD40, CD80 and CD83. The soluble forms of these

APC co-stimulatory factors can bind to their ligands on T cells and mediate immunosuppression. $^{108-110}$

1.1.3.6 Murine models of CLL

A number of murine models for CLL have been developed, which as well as enabling *in vivo* research, provide some insights into the pathogenic mechanisms underlying CLL. Selected murine models are summarised in table 1.3, and some discussed in more detail.

The New Zealand Black (NZB) mouse spontaneously develops an age-associated monoclonal expansion of autoreactive CD5+ B-1 cells which morphologically and immunophenotypically resembles CLL.^{111,112} This predisposition to a CLL-like condition has been attributed to a germline point mutation close to the pre-mir-16-1 micro-RNA gene in NZB mice.¹¹³ Genomic alterations affecting miR-16 expression are present in the around half of CLL patients (those with the 13q deletion). Consistent with a role for miR-16 in the pathogenesis of CLL, mice harbouring a deletion of the murine homologue of the 13q14minimal deleted region observed in many humans with CLL (MDR mice) also develop an indolent monoclonal B cell disorder resembling CLL.¹¹⁴

 $E\mu$ -miR-29 transgenic mice, in which overexpression of the micro-RNA miR-29 is governed by an IgVH promoter, spontaneously develop an expansion of CD5+ B cells, which is clonal in a proportion of cases. 20% of $E\mu$ -miR-29 mice develop a fatal leukaemia at two years of age. Consistent with a role for this micro-RNA gene, miR-29 is overexpressed in patients with indolent, but not with aggressive CLL. The reporters of this model suggest that miR-29 overexpression causes B cell expansion, but that secondary events are necessary for the phenotype of aggressive CLL.¹¹⁵

The TCL1 oncogene is expressed in CLL, splenic lymphoma with villous lymphocytes (SIVL) and in T-cell prolymphocytic leukaemia, but not in normal B cells. $E\mu$ -TCL1 transgenic mice, which overexpress TCL1, accumulate polyclonal CD5+ B cells from around 5 months of age, which then become oligo- or monoclonal. Evidence of disease with leukocytosis, lymphadenopathy and splenomegaly appears at 13 – 18 months.^{116,117} Morphologically, the leukaemia in TCL1 transgenic mice resembles prolymphocytic CLL. Together with the finding that among patients with CLL, the highest levels of TCL1 expression are observed in cases with unmutated IgVH genes,¹¹⁸ has led to the conclusion that TCL1 transgenic mice are a model of aggressive, rather than indolent, CLL.¹¹⁷

Insertion of the simian virus 40 T (SV40 T) oncogene into the immunoglobulin heavy chain (IgH) results in mice which develop a monoclonal expansion of CD5+ B cells. In this model, IgVH usage was biased, and both unmutated and mutated IgVH genes were observed in difference mice, resembling the spectrum of IgVH features in patients with CLL. The leukaemia incidence was increased further by concomitant deletion of p53.¹¹⁹

Finally, xenograft models of CLL have been developed. In one example, $Rag2^{-/-} \gamma c^{-/-}$ mice, which lack B cells, T cells and natural killer (NK) cells were injected with the stable human CLL line, MEC1. The transferred MEC1 CLL cells localise to lymph nodes, spleen and peripheral blood, as well as lung, kidney and the peritoneal cavity.¹²⁰

It is notable that three of the murine models of CLL listed in table 1.3 (the miR-29 transgenic, the MDR knockout and the New Zealand Black mouse) rely solely or largely on enhanced or reduced expression of micro RNAs. In all three of these models, penetrance is incomplete, the lymphoproliferative disorder may be described as indolent and the leukaemia passes through an initial polyclonal expansion before becoming clonal. These models may be suitable for studies of CLL pathogenesis or prevention.

Models that utilise oncogenes (SV40 T, TCL1) tend to exhibit higher penetrance, an earlier age of onset of disease and are more clinically aggressive. These models may lend themselves better to the investigation of therapeutic strategies for CLL.

1.1.4 Prognostic factors

1.1.4.1 Clinical staging systems

Useful prognostic information can be derived from readily-available clinical and laboratory data. Advanced patient age, male sex, higher absolute lymphocyte count, greater extent of lymphadenopathy and raised serum beta 2 microglobulin (β_2 M) levels are all associated with inferior prognosis on multivariate analysis.¹²⁴ The clinically-observed lymphocyte doubling time (LDT) is a strong predictor for overall survival, a doubling time of ≤ 12 months being associated with a median survival of three years versus >8 years for a LDT of >12 months.¹²⁵

Two clinical staging systems are in routine use for CLL: the Rai (see table 1.4) and Binet (see table 1.5) systems. Current clinical guidelines recommend the use of both systems to give a combined score.⁴

A prognostic nomogram incorporating age, sex, lymphocyte count, $\beta_2 M$, extend of lymphadenopathy and clinical stage has been developed, and is highly predictive of overall survival duration.¹²⁴

1.1.4.2 Cytogenetics

Approximately 80% of cases of CLL harbour a genomic abnormality detectable by interphase fluorescent in-situ hybridisation (FISH). The most common abnormalities identified by FISH, together with their frequencies and prognostic implications are listed in table 1.6.

In addition to these cytogenetic abnormalities, a proportion of patients harbour mutations of the TP53 or ATM genes, as determined by gene sequencing, without deletion of 17p or 11q, respectively. Patients with these cytogenetically-silent defects of TP53 or ATM have similar prognoses to those with 17p or 11q deletion, respectively.

Although the two most unfavourable genomic abnormalities (11q deletion and 17p deletion) are found more frequently in unmutated CLL cases, they remain independent prognostic factors on multivariate analysis, even after taking IgVH mutation status into account.¹³⁰

Background strain(s)	Transgene	Earliest disease manifestations	Special characteristics	Reference(
cBALB/c x FVB/N	TRAFDN/Bcl2	Splenomegaly and lymphadenopathy by 6m	Clonal; 20% survival at 14m	122
FVB/N	miR-29	Splenomegaly from 12m	Clonal in 60% of cases; 20% develop leukaemia by 24m	115
cC56BL/6	SV40 T	Splenomegaly and lymphadenopathy from 4 – 6m	Clonal; biased igVH usage; unmutated and mutated IgVH genes seen; median survival 6m	119
FVB/N	Deletion of mIR-15a/16-1 and DLEU2	Peripheral blood from 15 – 18m	Clonal; 42% penetrance	114
cB6C3	TCL1	Peripheral blood from 6 months; splenomegaly from 8m	Clonal; prolymphocytic morphology; model of aggressive/unmutated CLL	116
NZB	None (point mutation near mIR-16)	Peripheral blood and spleen from 4 – 6m	Clonal; associated with autoimmune haemolytic anaemia	111,112
cC57/Bl6	TNFSF13 (APRIL)	Mesenteric lymph nodes and Peyer's patches	Clonality not reported	123
Rag2 ^{-/-} γc ^{-/-}	None (xenograft model)	Spleen, liver, peripheral blood, kidney, liver, peritoneal cavity	Death at day 40 post-transfer of human MEC1 CLLs	120

Table 1.3 Selected murine models of CLL Adapted from 121

Stage	Criteria	5 year overall survival (%)	
0	Absolute lymphocytosis > 15×10^9 /L	98	
Ι	as stage 0 + lymphadenopathy	95	
Ш	as stage 0 + hepato- or splenomegaly +/- lymphadenopathy	88	
III	as stage 0 + anaemia (Hb <110 g/L) +/- organomegaly +/- lymphadenopathy	< 5	
IV	as stage 0 + thrombocytopenia (< 100 x10 ⁹ /L) +/- organomegaly +/- lymphadenopathy	0	

 Table 1.4 Rai clinical staging system¹²⁶

Table 1.5 Binet clinical staging system¹²⁷

Stage	Criteria	Median survival (months)
А	0, 1 or 2 lymph node areas *	> 120
В	3, 4 or 5 lymph node areas *	61
С	Any areas with Hb < 100 g/L or platelets < 100 x10 ⁹ /L	32

* one lymph node area = lymph nodes >1cm in neck, axilla, groin or spleen or liver enlargement

Cytogenetic finding	Frequency	Prognostic implications	Median overall survival (months)	Candidate gene(s) affected
13q deletion	55%	Better prognosis. More frequent in mutated IgVH CLL	133	miR-15a and miR-16-1 micro- RNAs ¹²⁹
Trisomy 12	10%	Standard prognosis	111	Unknown
11q deletion	15%	Inferior prognosis. Rapid progression & bulky adenopathy. More frequent in unmutated IgVH CLL	79	ATM
17p deletion	7%	Poorest prognosis. More frequent in unmutated IgVH CLL	32	TP53

Table 1.6 Principal cytogenetic changes in CLL Adapted from Zenz et al.¹²⁸

1.1.4.3 IgVH mutation status

Somatic hypermutation of the BCR is a normal part of the process of affinity maturation of antigen-stimulated B cells. Analysis of the clonal IgVH sequence from patients with CLL indicates that in around half of cases, somatic hypermutation has taken place (defined as $\geq 2\%$ deviation from the germ line sequence).¹³¹ At diagnosis, patients with CLL harbouring a mutated IgVH have a superior prognosis, with a median survival 310 months compared to 119 months in those with an unmutated IgVH.¹³²

1.1.4.4 Expression of ZAP-70

Determining IgVH mutation status requires access to high-throughput DNA sequencing, facilities for which are not routinely available in many hospital laboratories. Intracellular levels of the zeta chain-associated protein 70 (ZAP-70) within CLL cells correlates with IgVH mutational status: patients with raised intracellular ZAP-70 are more likely to have an unmutated IgVH gene and to have a poorer prognosis.¹³³ ZAP-70 can be readily measured by intracellular flow cytometry in most haematology laboratories. However, the widespread adoption of ZAP-70 as a prognostic marker has been limited by difficulties in standardisation of this quantitative flow cytometric assay.¹³⁴ Some of these difficulties may be overcome by using RT-PCR to determine the level of ZAP-70 transcripts within CLL cells.¹³⁵
Table 1.7 Indications for treatment of CLL⁴

At least ONE of:

Progressive marrow failure

Massive or symptomatic splenomegaly

Massive lymphadenopathy (at least 10 cm in longest diameter), or progressive or symptomatic lymphadenopathy

Progressive lymphocytosis with an increase of more than 50% over a two month period, or a lymphocyte doubling time of <6 months*

Autoimmune anaemia or thrombocytopenia that is poorly responsive to corticosteroids or other standard therapy

Constitutional symptoms (unintended weight loss of $\geq 10\%$ within previous 6 months; significant fatigue, fevers $> 38^{\circ}$ C for ≥ 2 weeks without evidence of infection; night sweats for > 1 month without evidence of infection

It is recommended that in patients with an initial lymphocyte count of $< 30 \times 10^9$ /L, LDT is not used as the sole criterion for initiation of treatment. Other factors contributing to lymphadenopathy or lymphocytosis (eg infection) should be excluded.

1.1.4.5 CD38 expression

CD38 is frequently expressed on CLL cells, and correlates with inferior prognosis independently of IgVH mutational status.¹³⁶ Unlike IgVH mutation status, CD38 expression is dynamic, changing with time and depending upon contact with CD4+ T cells.⁵⁸

1.1.4.6 Others

Many additional prognostic markers for CLL have been proposed, but are not in widespread clinical use. These include serum biomarkers, proliferative indices, telomere length, immunophenotypic characteristics, gene expression profiles, epigenetic characteristics and microRNA expression, and have been reviewed elsewhere.¹³⁷

1.1.5 Treatment

1.1.5.1 Treatment indications

Randomised controlled trials have demonstrated that treatment of early, asymptomatic CLL with alkylating agents does not prolong survival, ¹³⁸ leading to recommendations that evidence of CLL progression or symptoms be established before treatment is initiated.⁴ Established indications for treatment are listed in table 1.7.

1.1.5.2 Choice of treatment

Chemotherapy is the mainstay of therapy for CLL. Combined chemotherapy using the purine analogue fludarabine with the alkylating agent cyclophosphamide leads to better response and progression free survival rates than either the alkylating agent chlorambucil or fludarabine alone.¹³⁹ Although CD20 expression on CLL cells is weak, the humanised anti-CD20 monoclonal antibody improves outcome further: The German CLL8 study group reported a benefit in overall survival for patients with CLL with the addition of rituximab to fludarabine and cyclophosphamide.¹⁴⁰ This establishes fludarabine, cyclo-phosphamide and rituximab (FCR) as the current standard of care for first-line treatment of CLL.

Although prospective randomised trials of 'risk-adapted' therapy for CLL are lacking, laboratory and clinical data indicate that patients with deletion of 17p or with mutations of p53 are resistant to standard FCR chemoimmunotherapy.¹⁴⁰ Some authors advocate alternative first-line treatment in these high-risk individuals, typically using the anti-CD52 antibody alemtuzumab with or without high-dose corticosteroids, followed by early allogeneic stem cell transplantation (allo-SCT) for eligible patients.⁵

1.1.5.3 Monoclonal antibodies

Most chemotherapeutic agents induce apoptosis of target cells via a pathway involving the p53 and ATM proteins. Monoclonal antibodies have the potential to induce CLL cell death in a manner independent of this pathway, so may have a particular role in the treatment of CLL with 17p deletions (affecting the p53 gene), or with 11q deletions (affecting ATM). Indeed, clinical experience suggests that the application of either rituximab (anti-CD20) or alemtuzumab (anti-CD52) can overcome the adverse prognostic influence of the 11q deletion in CLL.¹⁴¹ The use of rituximab in the first-line treatment of CLL is now well established.

Alemtuzumab, a humanised anti-CD52 monoclonal antibody, is licensed for relapsed and refractory CLL, leading to partial response rates of around 50% in this setting.^{142,143} As well as targeting CLL cells, alemtuzumab leads to a profound T cell depletion, requiring routine antimicrobial prophylaxis. In addition, high rates of cytomegalovirus (CMV) reactivation demand routine CMV monitoring during alemtuzumab therapy and prompt treatment with appropriate antiviral agents.^{144,145} The rate of infectious complications may be lower when low-dose subcutaneous alemtuzumab is used.¹⁴⁶

Alternative monoclonal antibodies are in clinical trials. Ofatumumab is an anti-CD20 antibody that binds to different CD20 epitopes to rituximab, and has an enhanced ability to engage complement-mediated lysis of CLL cells *in vitro*.¹⁴⁷ Ofatumumab appears to be effective in combination with fludarabine and cyclophosphamide in treatment-naïve patients,¹⁴⁸ and has activity as monotherapy in fludarabine-refractory CLL.¹⁴⁹ An anti-CD23 monoclonal antibody, lumiliximab, has modest single-agent activity; the results of phase 3 clinical trials in combination with FCR are awaited.¹⁵⁰

1.1.5.4 Corticosteroids

Like monoclonal antibodies, corticosteroids have the capacity to induce CLL cell apoptosis independently of the p53 and ATM proteins, so have a particular role in the management of chemorefractory CLL.¹⁵¹ The combination of high dose methylprednisolone with rituximab or alemtuzumab results in high response rates in patients refractory to fludarabine.¹⁵²

1.1.5.5 Haematopoietic stem cell transplantation

Haematopoietic stem cell transplantation entails myelosuppressive and/or immunosuppressive chemo- or radiotherapy ('conditioning'), followed by an infusion of autologous or tissue type-matched allogeneic bone marrow or CD34+ rich peripheral blood mononuclear cells, which reconstitutes the host's haematopoietic and immune system compartments.

Autologous Autologous stem cell transplantation (auto-SCT) can be performed in CLL with low treatment-related mortality (<10%). Response rates are high, although no randomised trials have been performed.¹⁵³ However, a survival plateau is not observed after auto-SCT, suggesting that it is not a curative procedure in CLL. Moreover, a high incidence of myelodysplasia after auto-SCT for CLL has been observed.¹⁵⁴ Together, these factors limit the utility of auto-SCT for CLL, and indications for auto-SCT have not been clearly established in this disease.¹⁵³

Allogeneic Unlike auto-SCT, allogeneic stem cell transplantation (allo-SCT) offers the possibility of a graft versus leukaemia (GVL) effect by the grafted donor immune system, and may result in long-term cure. Levels of minimal residual disease (MRD), as assessed by polymerase chain reaction (PCR) for the disease-specific rearranged IgH, are lower after allo-SCT than after auto-SCT. Moreover, the decline in MRD levels occurs later after allo-SCT than after auto-SCT, suggesting that this is primarily an immunological effect rather than an effect of the conditioning regimen.¹⁵⁵

Registry data and clinical experience provide further evidence for a GVL effect in CLL: the presence of chronic graft versus host disease (GVHD) is associated with reduced risk of relapse; T cell depletion of the graft results in higher relapse risks; and treatment of relapse with infusion of additional lymphocytes from the donor (donor lymphocyte infusion, or DLI) can result in clinical responses.¹⁵⁴ Consistent with these clinical observations of a GVL effect, cytotoxic T cell lines with specificity for minor histocompatibility antigens and TAAs can be identified after allo-SCT for CLL, their presence correlating with clinical response.¹⁵⁶

Allo-SCT carries substantial morbidity and mortality, due primarily to toxicities of the conditioning regimen, infection, and donor lymphocyte immunity directed against normal host tissues (GVHD). Older age and the presence of co-morbidities are strong predictive factors for mortality following allogeneic stem cell transplantation. Therefore, because

Chapter 1. Introduction

of the high median age of patients with CLL, the role of myeloablative allo-SCT has been limited to all but the very youngest patients with refractory or aggressive disease.

The reduced treatment-related morbidity and mortality associated with less intensive conditioning regimens, which do not irreversibly ablate host haematopoiesis, have resulted in significant interest in reduced intensity conditioning allogeneic stem cell transplantion (RIC-allo) for CLL. A five year overall survival of 50%, and five-year disease-free survival of 39% has been reported for RIC-allo in fludarabine-refractory patients.¹⁵⁷

The European Blood and Marrow Transplantation working group has drawn up consensus guidelines on the indications for allograft in CLL, concluding that allo-SCT is a reasonable option for younger patients who are fludarabine-refractory (primary resistance, or relapse within 12 - 24 months of treatment), and for patients with p53 abnormalities including 17p deletion.¹⁵⁸

1.1.5.6 Other therapies

A number of other medical treatments are licensed for use in CLL, or are at advanced stages in clinical trials.

Bendamustine Bendamustine is an alkylating agent with a purine-like group, and has efficacy against CLL either alone or in combination with rituximab. At present, there is insufficient evidence to recommend bendamustine over FCR for the routine first-line treatment of CLL, but bendamustine-containing combinations have a useful role in refractory CLL, in older patients and in patients with renal impairment, in whom high response rates have been reported.¹⁵⁹

Lenalidomide Lenalidomide, a derivative of thalidomide, is a member of the class of drugs called immunomodulatory drugs (IMiDs), and is in routine clinical use in the treatment of the mature B cell malignancy, myeloma. Lenalidomide can reverse the defective T cell immunological synapse formation observed in CLL.⁹⁷ Despite its value in myeloma, the use of lenalidomide in CLL is less well established, because of modest response rates as monotherapy in refractory disease, ¹⁶⁰ and a phenomenon of 'tumour flare', which can be fatal.¹⁶¹

Cyclin-dependent kinase inhibitors Cyclin-dependent kinases (CDKs) regulate cell cycle progression, and CDK inhibition can induce apoptosis of tumour cells. Flavopiridol, a CDK inhibitor, induces apoptosis of CLL cells *in vitro* independently of the p53 pathway,¹⁶² and may therefore play a particular role in the treatment of *TP53* mutated (or 17p deleted) cases of CLL. Flavopiridol monotherapy results in modest response rates in refractory CLL,¹⁶³ including some durable responses in patients with bulky disease. The CDK inhibitor dinaciclib sensitizes primary CLL cells to vinblastine,¹⁶⁴ and is currently in clinical trials.

BCL2 inhibitors The antiapoptotic (pro-survival) protein Bcl-2 (encoded by *BCL2*) is over-expressed in CLL. Oblimersen is an antisense oligonucleotide directed against *BCL2* mRNA. In a phase II trial, addition of oblimersen to fludarabine and cyclophosphamide for the re-treatment of patients with CLL resulted in modestly improved complete response rates.¹⁶⁵ Activation of BH3-only proteins such as Bim, Bad and Noxa leads to the inactivation of pro-survival proteins such as Bcl-2. The BH3 mimetic ABT-737 shows *in vitro* activity against primary CLL cells, and *in vivo* activity against lymphomas and solid malignancies.^{166,167} In a phase 1 clinical trial, the BH3 mimetic navitoclax (ABT-263) showed a 35% partial response rate in relapsed or refractory CLL, including responses among patients carrying the 17p deletion.¹⁶⁸

Inhibitors of kinases involved in B cell receptor signalling As discussed in section 1.1.3.1, activation of the B cell receptor (BCR) contributes to the pathophysiology of CLL. Bruton's tyrosine kinase (Btk), mutation of which leads to Bruton's agammaglobulinaemia, is required for BCR signalling. The Btk inhibitor ibrutinib (PCI-32765) blocks BCR signalling in human B cells,¹⁶⁹ induces primary CLL cell apoptosis,¹⁷⁰ and has been associated with overall response rates of over 70% in patients with CLL, including those with adverse cytogenetic features.¹⁷¹ The phosphatidylinositol-3-kinase (PI3K) inhibitor GS-1101 (formerly CAL-101) inhibits B cell receptor signalling in primary CLL cells, and shows promising activity in combination with ofatumumab, especially in patients with bulky lymphadenopathy.^{172,173} Spleen tyrosine kinase (Syk) is essential for BCR signalling. In a phase I study, the Syk inhibitor fostamatinib resulted in a partial response in six of eleven treated patients with CLL.¹⁷⁴

T cell redirection Redirection of T cells has resulted in promising clinical responses in CLL using either T cells transfected with chimeric antigen receptors specific for CD19 or a bispecific antibody directed against CD3 and CD19 (blinatumumab). These treatment strategies will be discussed in section 1.3.2.

1.1.5.7 Effect of CLL treatments on immune function

Myeloid cells Many of the treatments used for CLL cause a dose-dependent myelosuppression. Grade 3 or 4 febrile neutropenia occurs in 34% of patients treated with FCR chemotherapy.¹⁴⁰ In contrast to this, monocyte numbers are relatively preserved after FCR chemotherapy.¹⁷⁵

Bendamustine has both purine analog and alkylator properties, and when given with rituximab was associated with grade 3 or 4 neutropenia in 37% of patients with relapsed indolent B cell lymphomas.¹⁷⁶

Immunotherapy with alemtuzumab and corticosteroids is associated with grade 3 or 4 neutropenia in the majority of patients.¹⁷⁷ In a randomized trial comparing rituximab and alemtuzumab in combination with fludarabine and cyclophosphamide, rates of grade 3 and 4 neutropenia were significantly higher in the alemtuzumab arm.¹⁷⁸

The PI3K inhibotir GS-1101 has been associated with grade 3 or 4 neutropenia in 24% of patients, while neutropenia has been reported in < 10% of ibrutinib-treated patients¹⁷⁴

Antigen presenting cells may also be affected by CLL cell treatment: circulating mDCs express CD52,¹⁷⁹ and alemtuzumab results in depletion of circulating DCs.¹⁸⁰

T cells First-line chemoimmunotherapy for CLL with FCR results in a profound decline in circulating T cell numbers, including CD4+, CD8+ and $\gamma\delta$ T cells. The decline in circulating T cells is prolonged, absolute numbers remaining low three years after chemoimmunotherapy.¹⁷⁵ Recent fludarabine treatment is associated with poor antibody responses to subsequent vaccination, suggesting that CD4+ T cell help to B cells is functionally impaired.¹⁸¹ Numbers of circulating T_{reg} cells are disproportionately reduced following fludarabine treatment, suggesting that T_{reg} cells may be particularly sensitive to fludarabine.¹⁸²

The effect of bendamustine on T cell subsets is not well-documented, but CMV reactivation following treatment has been reported.^{176,183}

Like fludarabine, the anti-CD52 monoclonal antibody alemtuzumab is also associated with profound T cell depletion.¹⁸⁴ Infections are common after alemtuzumab, CMV reactivation occurring in 23% of CLL patients treated with alemtuzumab and corticosteroids.¹⁷⁷ When given in combination with fludarabine and cyclophosphamide, alemtuzumab results in lower CD4+ T cell counts than rituximab, and higher rates of infection.¹⁷⁸ The effect of alemtuzumab on T cells is prolonged, lasting at least three years, although CD8+ T cell recovery occurs earlier than CD4+ T cell recovery.¹⁸⁵

In contrast to the prolonged T cell depletion that follows fludarabine and alemtuzumabcontaining treatments, the treatment of CLL with lenalidomide is associated with a very transient reduction in T cell numbers, lasting only three weeks.¹⁸⁶ In patients with CLL treated with lenalidomide, greater T cell activation as determined by CD69 upregulation, and greater CLL cell activation as determined by CD80 upregulation, are both associated with more severe clinical manifestations of cytokine release syndrome.¹⁸⁶ This is consistent with the *in vitro* finding that lenalidomide leads to normalisation of immunological synapse formation between T cells and autologous CLL cells.⁹⁷

NK cells NK cells decline after FCR chemotherapy, although are relatively preserved compared to T cells.¹⁷⁵ Compared with healthy donors, *in vitro* NK cell cytotoxic function in the presence of IL-2 is impaired in CLL patients before FCR chemotherapy and at the end of treatment, but normalises by 12 months after FCR chemotherapy.¹⁷⁵

In contrast with the relative sparing of NK cell numbers and activity, alemtuzumab induces a marked and prolonged depletion of NK cells, which express CD52.¹⁸⁴

B cells As would be expected, the anti-CD20 monoclonal antibody rituximab leads to a rapid depletion of B cells, both as monotherapy and in the context of FCR chemoimmunotherapy.^{175,187} In contrast to T cell depletion, recovery of normal B cell numbers after FCR chemotherapy is relatively rapid, B cell numbers normalising within 12 months of treatment.¹⁷⁵ Depletion of B cells with rituximab is associated with impairmed antibody responses to subsequently-administered vaccinations in patients with B cell malignancies.^{181,188}

The anti-CD20 monoclonal antibody of a umumab also leads to rapid and profound B cell depletion.¹⁸⁹ B cells express CD52, and alemtuzumab induces a depletion of normal B cells lasting at least 9 months.¹⁸⁴

The kinase inhibitors showing promising activity in CLL, such as the Btk inhibitor ibrutinib, might be expected to cause depletion of normal B cells, but preliminary reports of ibrutinib have not indicated normal B cell depletion or significant hypogammaglobulinaemia.¹⁷⁴ Published data on the effects of PI3K and other kinase inhibitors on normal B cells are lacking.

1.2 Invariant natural killer T cells

1.2.1 Antigen presentation by CD1 molecules

'Conventional' T cells express a T cell receptor (TCR) that recognises a specific peptide in the context of a specific major histocompatibility complex (MHC)—also known in humans as human leukocyte antigen (HLA). CD4+ T cells recognise peptides presented by MHC Class II, and CD8+ cells recognise peptides on MHC Class I. The recognition of specific peptides on specific MHC molecules by variably recombined TCRs lends enormous range to the adaptive immune response.

More recently, the existence of subsets of T cells that are not restricted by MHC has been demonstrated. Such T cells possess functional T cell receptors that recognise antigens in the context of nonclassical MHC molecules such including CD1, HLA-E, ¹⁹⁰ and MR1.¹⁹¹

The CD1 molecules are conserved β_2 M-associated MHC class I-like molecules that present lipids rather than peptides.¹⁹² Unlike MHC, CD1 molecules are minimally polymorphic.¹⁹³ Indeed, CD1 molecules are so conserved that mouse CD1d molecules can present glycolipid antigens to human CD1d-restricted T cells and vice versa.

Hydrophobic pockets in CD1 molecules accommodate the fatty acid tails of various selfand microbial-derived glycolipids, lipopeptides and phospholipids. The polar heads of these lipid antigens emerge from the surface of the CD1 molecule, and are recognised by the TCR of CD1-restricted T cells.¹⁹⁴

CD1 molecules have been divided into two groups on the basis of DNA sequence homology: Group 1 CD1 molecules include CD1a, CD1b, CD1c and CD1e; CD1d is a group 2 molecule. Group 1 CD1-restricted T cells recognise a number of mycobacterial cell wall components, and may be important in the generation of immunity to mycobacteria.^{195,196} Group 2 CD1-restricted (i.e. CD1d-restricted) T cells recognise glyco- and phospholipids.

Chapter 1. Introduction

Usage of CD1 genes varies between species: Humans possess and express all five CD1 genes; cattle express only the group 1 CD1 molecules CD1a, CD1b and CD1e with no CD1c or CD1d;¹⁹⁷ mice do not express any group 1 CD1 molecules, possessing only two copies of the CD1d gene. A transgenic mouse which expresses the human group 1 CD1 molecules has been developed, and may prove a useful tool for the analysis of these T cells.¹⁹⁶

The tissue expression, known antigens and reported responding T cells of the various CD1 isoforms are summarised in table 1.8.

1.2.2 iNKT cells

The archetypal lipid-reactive T cell is the iNKT cell, which recognises glycolipids, such as the synthetic α -GalCer, in the context of CD1d. Of all CD1-reactive T cells, iNKT cells have been studied in most detail for several reasons. In some mouse strains, iNKT cells constitute a very high proportion of circulating T cells (comprising up to 50% of liver T cells),²⁰⁸ and are potent and rapid producers of cytokines. Potent and specific synthetic iNKT cell agonists are available, including α -GalCer, and can be used for iNKT cell stimulation, or for iNKT cell detection. Finally, iNKT cells express a canonical ('invariant') TCR α chain, enabling their detection and isolation using specific anti-TCR antibodies.²⁰⁵

The presentation of peptide and glycolipid antigens to conventional CD4+ and CD8+ T cells and to iNKT cells is illustrated in figure 1.2.

1.2.2.1 Nomenclature

The nomenclature of invariant natural killer T (iNKT) cells has evolved in a piecemeal fashion, leading to some inconsistencies. The origin of the term natural killer T (NKT) arose from their initial description in the C57BL/6 mouse strain, where a population of T cells expressing intermediate levels of TCR alongside the murine NK cell marker, NK1.1, were identified. These cells attracted interest, as they were noted to be potent producers of T helper 1 (Th1) and T helper 2 (Th2) cytokines, and to be restricted by CD1d rather than by MHC. Following recognition that the vast majority of these NKT cells use an invariant TCR α chain (V α 14J α 18 in mice), the term iNKT cells was used.²⁰⁹ However, it should be emphasised that despite their expression of certain molecules classically associated with natural killer cells, iNKT cells are unequivocally thymic-derived $\alpha\beta$ T cells.

In humans, the majority of circulating T cells that co-express NK markers such as CD16, CD56 or CD161, the human homologue of NK1.1, are not iNKT cells. Thus, in human literature, the term 'NKT cell' has been used both to refer to conventional MHC-restricted T cells which co-express NK cell markers, and to the much narrower population of iNKT cells, which express an invariant TCR- α chain (V α 24J α 18 in humans), and which are restricted by CD1d.

A further distinction has been made between type I and type II NKT cells: Type I NKT cells, also termed iNKT cells, possess an invariant TCR- α chain, exhibit CD1d restriction and recognise α -GalCer. These iNKT cells are found in both mice and humans, although their frequency is substantially lower in humans than in certain mouse strains. iNKT cells

Table 1.8 CD1 is	otorms: tıssue expression, known antigei	ns, and responding T cell populations	
CD1 Isoform	Tissue expression	Known antigens	Responding T cell population(s)
CD1a	Langerhans cells, mDCs, Thymocytes	Self-lipids, ¹⁹⁸ Lipopeptides ¹⁹⁹	αβ T cells, including skin-homing Th22 cells, ¹⁹⁸ duodenal γδT cells ²⁰⁰
CD1b	DCs, thymocytes	Mycobacterial lipids and glycolipids (eg mycolic acids, lipoglycan lipoarabinomannan, phosphatidylinositol mannoside), self-lipids (eg ganglioside GM1)	αβ T cells, ²⁰¹ γδ T cells ²⁰²
CD1c	mDCs, Marginal zone B cells, Thymocytes	Phospholipids (including mycobacterial-derived)	αβ T cells, ²⁰³ γδT cells ²⁰⁰
	mDCs, B cells, epithelium,	Glycolipids (e.g. α-GalCer), Phosphotidylinositols ²⁰⁴	iNKT cells ²⁰⁵ (semi-invariant TCR)
CD1d	nepatocytes, vascutar smooth muscle, haematopoietic cells, thymocytes	Lysophospholipids, sulfatide	Type II NKT cells ²⁰⁶ (variable TCR usage)
		Phospholipids	γδT cells ²⁰⁰
CD1e	DCs (within Golgi and lysosomes)	Glycolipids which require intracellular processing e.g. phosphatidylinositol mannoside (PIM) ₄	None—involved in lipid processing ²⁰⁷

23

Figure 1.2 Comparison of peptide and glycolipid presentation (a) CD4+ T cells recognise peptides on MHC Class II. (b) CD8+ T cells recognise peptides on MHC Class I. (c) iNKT cells recognise glycolipids on CD1d



typically constitute a median of 0.05% of T cells in the blood of healthy humans, ^{210,211} but 1% of T cells in BALB/C or C57 mice. iNKT cells are widely distributed within the lymphoreticular compartment in the mouse, being detectable in blood, bone marrow, spleen, lymph nodes and thymus. iNKT cells are particularly enriched in the liver, where they constitute 25 – 50% of T cells in some mouse strains, ²⁰⁸ and 0.5% of T cells in humans. ²¹²

Type II NKT cells have also been referred to as 'non-classical' NKT cells. They are CD1d -restricted, but in contrast to iNKT cells (type I NKT cells), exhibit more varied TCR chain usage and show no reactivity to α -GalCer.²⁰⁹ Instead, type II NKT cells respond to sulfatide or lysosphingolipid antigens presented on CD1d, and are frequently immunosuppressive.²¹³ There is evidence for the presence of type II NKT cells in humans: A population of T cells with diverse TCR usage was found to bind to CD1d dimers loaded with lysophospholipids, but not to CD1d dimers loaded with α -GalCer. These cells were potent producers of the immunosuppressive cytokine IL-13.²⁰⁶ CD1d-reactive T cells with diverse TCR usage have been identified in human bone marrow and in the placenta, and suppress mixed lymphocyte reactions.^{214,215}

Finally, the heterogeneous population of NK-like T cells (that is, T cells which express NK markers, but which are not CD1d restricted and do not respond to α -GalCer) are sometimes referred to as 'type III NKT cells'.

This thesis primarily concerns type I NKT cells (CD1d-restricted T cells expressing the V α 24-J α 18 TCR- α chain and responding to α -GalCer, which I refer to throughout as iNKT cells for brevity. I use the term NK-like T cells to refer to T cells co-expressing classical NK markers (type III NKT cells). Note that by typical immunophenotypic criteria (CD3+CD16/56+ or CD3+CD161+), the NK-like T cell subset includes the majority of iNKT cells. Figure 1.3 illustrates the various NKT cell subsets, and table 1.9 summarises the synonyms, immunophenotype and antigen presentation systems for the various subsets.

1.2.2.2 Other invariant T cell populations

Invariant natural killer T cells are not the only T cell population showing limited diversity of TCR repertoire and a lack of MHC restriction. Two other notable populations include $V\gamma 9V\delta 2 +$ gammadelta T cells and mucosal-associated invariant T cells (MAIT) cells, both of which share some characteristics with iNKT cells.

Vγ9Vδ2+ gammadelta T cells Gammadelta ($\gamma\delta$) T cells constitute up to 10% of circulating nucleated cells in humans, and a higher proportion in some epithelial surfaces. A major subpopulation expresses the invariant Vγ9Vδ2 TCR, and can comprise up to 50% of circulating gammadelta T cells during certain infections. Vγ9Vδ2+ cells recognise phosphoantigens such as (E)-4-hydroxy-3-methyl-but-2-enyl pyrophosphate (HMB-PP), which lead to TCR-dependent activation. They are also indirectly activated by aminobisphosphonates, including pamidonate and zoledronate, via production of the phosphoantigen isopentyl pyrophosphate (IPP). Activated V γ 9V δ 2+ are able to produce both Th1-

Cell type	Synonyms	Immunophenotype (in humans)	Presenting molecule	Known antigens
iNKT	Type I NKT, NKT, Classical NKT	CD3+ V α 24J α 18+, usually V β 11+ α -GalCer loaded CD1d tetramer+, CD161+ (usually)	CD1d	α-GalCer and analogues, microbial glycolipids, iGb3
Type II NKT	Non-classical NKT	CD3+, Lysosphingolipid- or sulfatide-loaded CD1d tetramer+, Variable TCR usage	CD1d	Lysosphingolipids sulfatide
NK-like T	Type III NKT, NK T, NKT	CD3+, CD16+, CD56+ or CD161+, Variable TCR usage	MHC Class I or II *	Peptides *
Ц		CD3+, variable TCR usage	MHC Class I or II *	Peptides *
NK		CD3-, CD16+, CD56+ or CD161+	n⁄a	n/a
* NK-like T and	T cell subsets inclue	de subpopulations of other non-classical T cells, such	as vô T cells and T cells re	actrintad to aroun 1 (



Figure 1.3 Nomenclature of NKT cells Venn diagram illustrating the overlaps between T, NK, NK-like T, iNKT (also known as type I NKT) and type II NKT cells.

and Th2-type cytokines and are able to lyse some tumour cell lines in a granzyme- and NKG2D-dependent manner. 216

MAIT cells MAIT cells express the invariant TCR α chain iV α 7.2-J α 33 (iV α 19-J α 33 in mice), which pairs with a restricted repertoire of V β chains. These cells are found in the lamina propria of the gut, the liver and the blood, and are restricted by the MHC Class I-like molecule MR1, expressed on B cells. Like CD1, MR1 is highly conserved between species. The ligand for MR1 is not known, but as MAIT cells are activated by a wide range of gut microbes, is assumed to be a conserved microbial compound. MAIT cells express high levels of CD161 alongside CCR6, and rapidly produce IFN- γ , tumour necrosis factor alpha (TNF- α), IL-17 and granzymes upon stimulation. MAIT cells express NK receptors such as NKG2D, suggesting that NK signals may also play a role in their activation.²¹⁷

1.2.2.3 Subsets

A number of subsets within the iNKT cell compartment have been described. In humans and mice, iNKT cells can be divided on the basis of their CD4 expression. In humans, both CD4+ and CD4- iNKT cells are potent producers of IFN- γ , but CD4+ iNKT cells also produce Th2-type cytokines such as IL-4 and IL-13.^{218,219} The CD4- iNKT cell subset expresses more perforin.²²⁰ In mice, the gene expression profile of CD4+ iNKT cells differs from that of their CD4- counterparts.²²¹ In mice, the CD4-, but not the CD4+, subset of iNKT cells is able to reject an inoculated sarcoma.²²² In humans, but not in mice, a

small proportion of iNKT cells also express CD8, the functional significance of which is not clear.

The proportion of CD4+ iNKT cells changes with age: In ageing humans, as well as a reduction in the overall frequency of iNKT cells, a bias towards increased numbers of CD4+, Th2 cytokine-producing iNKT cells occurs.²²³

The majority of circulating human iNKT cells possess an effector memory phenotype: CD45RO+, CCR7-, CD62L-.²¹¹ Indeed, even iNKT cells from umbilical cord blood exhibit a memory phenotype, in contrast with conventional T cells, which are naïve in neonates. This suggests that iNKT cells are exposed to an naturally-occurring ligand *in utero*.²²⁴ The immunophenotypic and functional properties of iNKT cells and of naïve, central memory, and effector memory T cells are compared in table 1.10.

In humans, the majority of iNKT cells express the NK marker CD161 (the human homologue of NK1.1), whereas only 10% express the NK marker CD56.²¹⁸ There is evidence that the NK receptors expressed by iNKT cells, including CD161, are functional. In mice, blockade of the NK activating receptor NKG2D prevents a CD1d-dependent NKT-mediated hepatitis.²²⁵ Murine iNKT cells also express the NK inhibitory receptor NKG2A, blockade of which exacerbates an iNKT-dependent Con A-induced hepatitis.²²⁶ A proportion of human iNKT cells also express NKG2A.²¹⁹ In humans, ligation of CD161 (also known as NKR-P1A, a C-type lectin) enhances TCR-mediated iNKT cell activation.²²⁷ Finally, human iNKT cells preferentially proliferate in response to allogeneic rather than autologous DCs, a finding attributed to iNKT cell upregulation of activating killer Ig-like receptors (KIRs) in response to TCR signalling.²²⁸

A functional difference between the CD56+ and CD56- subsets of human iNKT cells has been described: CD56+ iNKT cells express higher levels of perforin, and exhibit greater cytotoxicity against myeloid leukaemia cells in the *absence* of α -GalCer. Cytotoxicity in the *presence* of α -GalCer was equivalent between the two subsets.²²⁹ This may indicate that the CD56+ subset of iNKT cells exhibits enhanced NK-receptor directed cytotoxicity, but that once a sufficiently strong TCR signal is present (for example following α -GalCertreatment of the target cells), both subsets are able to lyse the target.

1.2.3 CD1d ligands

First reported to specifically activate murine iNKT cells in a CD1d- and TCR-restricted fashion in 1997, α -GalCer, a synthetic glycolipid derived from the marine sponge *Agelas mauritianus*, is the archetypal iNKT cell agonist.²⁰⁵ The high frequency of iNKT cells, their conservation across species, and their effector memory phenotype, however, imply that naturally-occurring CD1d ligands exist, either endogenous or derived from ubiquitous microbes.

1.2.3.1 Endogenous

Activation of iNKT cells by CD1d-expressing APCs in the absence of exogenous glycolipids provides indirect evidence of the presentation of endogenous ('self') lipids by CD1d. Consistent with this, glycolipid and phospholipid fractions of cell extracts are able to

Table 1.10 (Comparison o	of the phenot	ype of T ce	ells and iNK	T cells				
T cell subset	CD45RA	CD45RO	CCR7	CD62L	Response to antigen (or anti-CD3)	Proliferative potential	Cytokines produced	Tissue distribution	Notes
Naïve	+	r	+	+	* -/+	+++++++++++++++++++++++++++++++++++++++	IL-2	Lymph nodes	Can differentiate into T _{CM} and T _{EM}
T_{CM}	I	+	+	+	+ +	+ +	IL-2	Lymph nodes	Can differentiate into T _{EM}
T_{EM}		+		-/+	+ + +	+	IFN- _Y IL-4 IL-5 (rapid)	Peripheral tissues	CD8+ T _{EM} express perforin. Differentiate into T _{EMRA}
$\mathrm{T}_{\mathrm{EMRA}}$	+	ı	ı	ŗ	+ + +	ŗ	IFN- _Y IL-4 IL-5 (rapid)	Peripheral tissues	Terminally differenti- ated
iNKT cells	I	+			++++++	+ +	IFN- _Y IL-4 IL-5 (rapid)	Peripheral tissues	
* Costi	mulatory sign	als (i.e. CD28	3) are requi	red					

Class	Molecules
Glycerophospholipids	Diacylglycerols
	Plasmalogens
	Lysophospholipids
	Cardiolipins
Sphingolipids	Sphingomyelins
	Glycosylated sphingolipids

Table 1.11 Lipids eluted from secreted CD1d from a transfected human lymphoblastoid cell line 230

stimulate iNKT cells when presented by plate-bound recombinant CD1d. Analysis of lipids eluted from CD1d purified from cell cultures reveals a range of polar lipids. Table 1.11 summarises the lipids eluted from a CD1d secreted by a CD1d-transfected human cell line in one study.²³⁰

Particular interest has focussed on a few specific potential endogenous ligands. Isoglobotrihexosylceramide (iGb3) was proposed as a major endogenous ligand for iNKT cells, but it is now clear that although iGb3 functions as a CD1d ligand, its expression is not required for iNKT cell development or maintenance.²³¹ The ganglioside GD3 is expressed on melanoma cells, and can activate iNKT cells in a CD1d-restricted fashion.²³² The myelinderived glycolipid sulfatide is recognised by a population of immunosuppressive type II iNKT cells, and administration of sulfatide can prevent experimental autoimmune encephalitis in a CD1d-dependent manner.²³³

1.2.3.2 Microbial

Some microbial-derived glycolipids are able to bind CD1d and to be recognised by iNKT cells. Table 1.12 lists selected microbial CD1d ligands; the role of iNKT cells in the protection against infection will be discussed later (section 1.2.7.1).

The glycolipid α -GalCer, the structure of which is illustrated in figure 1.4, is a potent iNKT cell agonist, but the mixed Th1- and Th2-type cytokine responses it elicits have prompted searches for alternative ligands capable of generating more selective responses. Modifications to the sphingosine base, the length or saturation of the acyl chains, the composition of the glycosidic linkage or to the structure of the carbohydrate moiety can all result in α -GalCer derivatives with iNKT cell stimulating activity.^{231,234,235} For instance, a glycolipid with a modified fatty acid chain, 7DW8-5, is reported to exhibit higher CD1d binding affinity, improved induction of iNKT cell cytokine production and enhanced adjuvant activity compared to α -GalCer.²³⁴

In the mouse, some α -GalCer derivatives, such as its sphingosine-truncated analog OCH (figure 1.5), exhibit greater Th2 stimulating capacity,²³⁶ whereas others, such as α -C-GalCer (figure 1.5), induce greater Th1 cytokine production.²³¹ Using human iNKT clones and a library of α -GalCer derivatives, the Th2-type cytokine IL13 was induced by weak ag-

Table 1.12 Microbial-derived CD10	1 ligands adapted from Venkataswamy & F	Orcelli ²³¹
Organism(s)	CD1d ligand	Activity
Mycobacterium bovis	PIM_4	IFN-y secretion from a subset of iNKT cells
Borrelia burgdorferi	α-galactosyl diacylglycerols	IFN- γ secretion from a subset of mouse iNKT cells and a human iNKT cell line
Sphingomonas spp Ehrlichia spp	α-glucuronosylceramides	Th1 and Th2 cytokine production from human and mouse iNKT cells
Leishmania donovani	Lipophosphoglycan	IFN- γ secretion from a fraction of mouse iNKT cells
Entamoeba histolytica	Lipopeptidophosphoglycan	IFN- γ secretion from mouse iNKT cells

1.2. Invariant natural killer T cells





Table 1.13 Cytokines and chemokines produced by iNKT cells^{240,241}

Th1-type	Th2-type	Other	Chemokines
IFN-γ	IL-4	IL-17	RANTES
TNF-α	IL-5	IL-21	Eotaxin
	IL-6	TGF-β	Lymphotoxin
	IL-10	GM-CSF	MIP-1a
	IL-13	IL-2	MIP-1β
		IL-3	

onists at modest concentrations or by strong agonists at very low concentrations, whereas IFN- γ induction and transactivation of NK cells required strong iNKT agonists at modest concentrations or weak agonists at high concentrations.²³⁷ This polarisation of cytokine response according to avidity of the agonist is also observed with 'conventional' T cells: CD4+ T cells are polarised towards production of Th2-type cytokines in the presence of weak agonists, while strong agonists induce Th1-type cytokines.^{238,239}

1.2.4 Cytokine production

In keeping with their effector memory phenotype, iNKT cells rapidly produce cytokines upon stimulation. Reported iNKT cell-derived cytokines are listed in table 1.13. Both CD4+ and CD4- iNKT cell subsets produce Th1-type cytokines, but the production of the Th2-type cytokines is largely restricted to the CD4+ iNKT cell subset.^{218,220}

A number of factors influence the cytokine profile of stimulated iNKT cells:



Figure 1.5 Structure of selected α -GalCer derivatives

- Relative proportion of iNKT cell subsets: Human CD4+ iNKT cells exhibit greater production of Th2-type cytokines than their CD4- counterparts.^{218,220} In mice, the NK1.1+ subset of iNKT cells produce less IL-4 and more IFN-γ than NK1.1- iNKT cells.²⁴¹
- **Organ of origin:** In mice, thymic iNKT cells produce more Th2 cytokines, and liver iNKT cells produce more Th1 cytokines than those from the periphery.²⁴¹
- Structure and concentration of iNKT cell agonist: The stimulation of iNKT cells with weak agonists, such as the truncated analogue of α -GalCer, OCH, or with very low concentrations of strong agonists, results in preferential production of Th2-type cytokines. In contrast, strong agonists such as α -C-GalCer, or high concentrations of weak agonists, tend to induce the production of Th1-type cytokines by iNKT cells. ^{231,237}
- Antigen presenting cell: Optimal IFN- γ production by iNKT cells requires both DCderived IL-12 and a CD40/CD40L interaction, ^{242,243} and TLR ligand-activated DCs are particularly strong inducers of IFN- γ production by iNKT cells.²⁴⁴ In contrast, the presentation of α -GalCer by non-professional APCs such as B cells or Schwann cells results in Th2-skewed iNKT cell cytokine production.^{245,246}
- **Exogenous cytokines:** Whether derived directly from DCs, or added to culture media, the presence of IL-12 favours IFN- γ production by iNKT cells. In contrast, IL-2 in combination with weak TCR stimulation favours the production of Th2-type cytokines by human iNKT cells.²⁴⁷
- **Metabolic factors:** Tryptophan catabolites skew iNKT cells toward a Th2 profile, while inhibition of indoleamine 2,3-dioxygenase, which degrades tryptophan, favours a Th1 profile.²⁴⁸

Interleukin-17-producing iNKT cells have been described in mice, and have been implicated in the induction of airway neutrophilia²⁴⁹ and collagen-induced arthritis²⁵⁰ in experimental models. However, human iNKT cells have not been conclusively shown to produce IL-17:²⁵¹ Although one study showed that human NK-like T cells can produce IL-17 upon stimulation, the cell population studied was not restricted to iNKT (type I NKT) cells.²⁵²

1.2.5 Cytotoxicity

iNKT cells can lyse target cells, including moDCs, upon recognition of α -GalCer on CD1d.²⁵³ The degree of cytotoxicity exhibited by iNKT cells correlates positively with target cell CD1d expression and iNKT cell agonist potency.²⁵⁴ As discussed above, iNKT cell cytotoxicity also appears to be modulated by NK receptor signals.^{225,226}

iNKT cells express granzyme B and perforin, can upregulate Fas ligand (FasL) and tumour necrosis factor-related apoptosis-inducing ligand (TRAIL), and can produce TNF- α . There is evidence for involvement of each of these cytotoxicity pathways in iNKT cell lysis of CD1d-expressing target cells.

Perforin and granzyme B are required for a-GalCer-induced iNKT cell lysis of primary

human myelomonocytic leukemia cells.²²⁹ Concanamycin A inhibits the lysis of target cell lines by human umbilical cord blood-derived iNKT cells, suggesting a perforin-dependent mechanism.²⁵⁵

Upon activation, iNKT cells upregulate FasL,²²⁰ which can induce apoptosis of target cells expressing the Fas receptor. α -GalCer-pulsed B cells are killed by iNKT cells in wild type but not in FasL-deficient mice, suggesting that this pathway plays a role in iNKT cell cytotoxicity.²⁵⁴ Both Fas- and FasL-deficient mice are resistant to α -GalCer induced hepatocellular damage.²⁵⁶

Human iNKT cells express TRAIL on activation. Co-culture of leukemic blasts from patients with acute myelomonocytic leukemia with TRAIL-expressing iNKT cells leads to leukemia cell apoptosis, an effect blocked by a TRAIL neutralising antibody.²⁵⁷

Finally, blocking studies indicate a contribution of TNF- α to the killing of target cells by human iNKT cells, ^{229,255} although both of these reports found that the major contributor to cytotoxicity was the perforin/granzyme B pathway.

To summarise, iNKT cells integrate signals from their TCR, their activating and inhibitory NK receptors and their environment to determine target cell killing. A potent TCR stimulus (i.e. high target cell CD1d expression in association with a strong iNKT cell agonist) and a balance of activating NK signals will favour target cell lysis. Several mechanisms of cytotoxicity may be involved, but the perforin/granzyme B and Fas/FasL systems appear to be the most important.

1.2.6 Interactions with other cells of the immune system

1.2.6.1 Dendritic cells

Myeloid dendritic cells express CD1d, and can present α -GalCer to iNKT cells. In response to this TCR stimulation, iNKT cells release cytokines including TNF- α and IFN- γ , and upregulate costimulatory molecules such as CD40L. This results in DC upregulation of the co-stimulatory markers CD80 and CD86, and production of IL-12, which can feed back on iNKT cells.^{242,258} The reciprocal interaction between iNKT cells and DCs in the presence of α -GalCer enhances responses to DC-presented protein antigens by 'conventional' T cells.^{258–260} A diagram of the interaction between DC, iNKT and T cells is presented in figure 1.6.

Under some circumstances, the interaction of iNKT cells with DCs can result in immunosuppression or induction of tolerance. The outcome of an interaction between DCs and iNKT cells appears to depend on a number of factors:

- Exposure to iNKT cell agonist: Both α -GalCer and iNKT cells are required to turn immature DCs into effective producers of IL-12. In the absence of α -GalCer, iNKT cells fail to induce full maturation of DCs.²⁶¹
- Contact of iNKT cells with non-professional APCs: Although iNKT cells can induce DC maturation and enhance Th1 responses when added to α -GalCer pulsed immature human moDCs, the presence of iNKT cells throughout the period of monocyte culture has the opposite effect: moDCs generated in the presence of iNKT cells



Figure 1.6 Interaction between DC, iNKT and T cells

have an impaired capacity to produce IL-12 and to elicit Th1-type T cell responses, whether α -GalCer is present or not. Thus, the α -GalCer/iNKT cell stimulus operates differentially on monocytes (DC precursors) and DCs themselves.²⁶²

• Route of α -GalCer administration: While α -GalCer presented exclusively by DCs can induce DC maturation, the systemic administration of α -GalCer (e.g. by intravenous injection), results in a skewing of DCs towards a tolerogenic phenotype in non-obese diabetic (NOD) mice.²⁶³

Thus, presentation of α -GalCer to iNKT cells by DCs results in DC maturation, upregulation of co-stimulatory markers, improved T cell responses to protein antigens, and enhanced anti-tumour immunity. In contrast, premature exposure of monocytes to iNKT cells and α -GalCer, exposure of DCs to iNKT cells in the absence of α -GalCer, and α -GalCer delivered in a non-directed fashion can all result in the generation of DCs with poor stimulatory capacity, or even tolerogenic properties. The timing and mode of delivery of α -GalCer are therefore critical if immune responses are to be augmented.

1.2.6.2 T cells

Activation of iNKT cells with α -GalCer results in the enhancement of peptide-specific CD4+ and CD8+ T cell responses.^{258,260} Much of this effect appears to be due to iNKT cell enhancement of DC maturation, as discussed above, because the transfer of DCs from α -GalCer- and peptide-treated mice leads to transfer of the enhanced immunity,²⁶⁰ and because CD40 expression on DCs is essential for α -GalCer enhancement of T cell responses.²⁵⁹

Under certain circumstances, iNKT cells may also be able to inhibit T cell responses. Culture of iNKT cells in the presence of TGF- β results in expression of forkhead box P3 (Foxp3) on a subset of iNKT cells. Similar to Foxp3+ T cells, these iNKT cells suppress T cell responses in a contact-dependent manner.²⁶⁴

 T_{reg} cells also interact with iNKT cells in a bi-directional fashion: T_{reg} cells can inhibit the proliferation, cytokine production and cytotoxicity of iNKT cells through a cell-cell contact mechanism. In the opposite direction, iNKT cells can promote T_{reg} generation by inducing tolerogenic DCs as outlined in section 1.2.6.1 above. Finally, iNKT cell-derived IL-2 can promote T_{reg} proliferation.²⁶⁵

1.2.6.3 B cells

iNKT cell effect on antibody production Co-administration of protein immunogens with α -GalCer results in higher antibody titres and improved protection against infection. Moreover, combining protein vaccination with α -GalCer results in IgG responses in MHC class II knockout mice, implying that iNKT cells can substitute for antigen specific CD4+ T cell help in the generation of B cell immunity.²⁶⁶ Chimeric mouse studies indicate that CD1d expression on B cells themselves is necessary for this effect.²⁶⁷

There is also evidence that iNKT cells can also provide help to B cells to produce antibodies directed against glycolipids themselves—this interaction requires CD40L, CD80/CD86 and IFN- γ .²⁶⁸

Role of B cells in maintenance of iNKT cells B cells express CD1d, and have been reported to be important for the maintenance of iNKT cells. Depletion of CD19+ B cells from human PBMCs resulted in impaired *in vitro* iNKT cell proliferation and reduced iNKT cell cytokine production, and α -GalCer-loaded B cells are able to induce iNKT cell proliferation.²⁶⁹ In pre-clinical studies, marginal zone B cells are required for the full activation of iNKT cells by α -GalCer *in vivo*, an effect which appears to be mediated by DCs.²⁷⁰

1.2.6.4 Natural killer cells

Systemic administration of α -GalCer to mice leads to activation and expansion of NK cells and enhancement of NK cytotoxicity.^{271,272} Indeed, NK cell-derived IFN- γ is required for the antimetastatic effect of α -GalCer in a preclinical melanoma model.²⁷¹

The transactivation of NK cells following α -GalCer administration appears to be cytokinemediated—it can be reduced by simultaneously blocking both IFN- γ and IL-12.²⁷³ Murine iNKT cells are also capable of IL-21 production, a cytokine with mixed effects on NK cell function, including enhanced NK cell maturation and cytotoxicity as well as early NK cell death.²⁷⁴ An immunotherapy strategy comprising α -GalCer-pulsed DCs followed by systemic IL-21 administration resulted in improved NK cell cytotoxicity and regression of established metastases in several mouse models of cancer.²⁷⁵

There is *in vitro* evidence that human NK and iNKT cells can co-operate in a similar fashion. The addition of *in vitro*-expanded iNKT cells and α -GalCer to PBMC cultures resulted in enhanced NK cell cytotoxicity, an effect which was augmented by IL-21. Transwell experiments demonstrated that the transactivation of human NK cells by iNKT cells was dependent on soluble factors.²⁷⁶

1.2.6.5 Myeloid cells

Administration of intravenous α -GalCer to mice leads to increased numbers of haematopoietic progenitor cells in the spleen and an increase in circulating mature neutrophils, an effect consistent with the production of IL-3 and granulocyte-macrophage colony stimulating factor (GM-CSF) by iNKT cells.²⁷⁷

The effects of iNKT cells on neutrophil recruitment to tissues can be variable: in murine studies, Th-17-producing iNKT cells contributed to airway infiltration by neutrophils in one model, ²⁷⁸ whereas IL-4 secreting iNKT cells suppressed neutrophil IFN- γ secretion and lung inflammation in another.²⁷⁹

The production of Th2 cytokines and eotaxin by *in vitro* stimulated iNKT cells is consistent with the finding that several mouse models of asthma indicate a role for iNKT cells in the eosinophil recruitment to the airways.²⁸⁰ Evidence of role for iNKT cells in human asthma is less clear, in part because methodological difficulties have complicated the detection of iNKT cells in human respiratory samples.²⁸¹

Myeloid cells with suppressive activity, such as tumour-associated macrophages (TAMs) and myeloid-derived suppressor cells (MDSCs) have been identified within tumours, and themselves express CD1d. TAMs, which support tumour cell growth through IL-6 production, can be killed by iNKT cells, leading to suppression of tumour growth in a neuroblastoma xenograft model.²⁸² iNKT cells can also control MDSC numbers: iNKT cell-deficient mice have increased numbers of MDSCs, and adoptive transfer of iNKT cells into $J\alpha 18^{-/-}$ (iNKT-cell deficient) mice led to a reduction of MDSC numbers and alleviated MDSC-mediated immunosuppression.²⁸³

1.2.7 Role in health and disease

1.2.7.1 Infection

A number of infections are exacerbated in iNKT cell-deficient mice (both in $CD1d^{-/-}$ and $J\alpha 18^{-/-}$ animals), suggesting that iNKT cells are of relevance in controlling the pathogenic organisms or the immune response. The impact of iNKT cell deficiency on selected infection models is summarised in table 1.14.

Indirect evidence suggests that iNKT cells may play a role in mycobacterial infections in humans. Circulating iNKT cells have an activated phenotype in patients with tuberculosis.²⁸⁵ Patients with both tuberculosis and leprosy have a higher proportion of CD4+ iNKT cells than healthy controls.²⁸⁶ Incubation of human monocytes with liposomes bearing *Mycobacterium leprae* cell wall antigens results in upregulation of CD1b and CD1d, and enhanced IFN- γ secretion within the NK-like T cell population.²⁸⁷

Infections may activate iNKT cells via several mechanisms. As discussed in section 1.2.3.2, microbial-derived iNKT cell agonists have been isolated, and may be presented to iNKT cells by APCs. Infections may result in changes of cellular glycolipid composition, resulting in altered expression of self glycolipids to iNKT cells. Finally, microbial danger signals acting on other immune system cells may lead to changes in circulating cytokines

le 1.1	14 Selected infections exacer	bated in iNKT deficient mice adapte	d from Kinjo et al ²⁸⁴	
-	Organism type	Species	Effect of iNKT defici	ency on infection
			CD1d -/-	$J\alpha 18^{-/-}$
	Parasitic	Leishmania major	I	Exacerbated
		Trypanosoma cruzi	Not exacerbated	Exacerbated
	Fungal	Cryotococcus neoformans	I	Exacerbated
	Bacterial	Borrelia burgdorferi	Exacerbated	I
		Pseudomonas aeruginosa	Exacerbated	I
		Streptococcus pneumoniae	I	Exacerbated
		Sphingomonas spp.	Exacerbated	Exacerbated
	Viral	Herpes simplex	Exacerbated (HSV-2)	Exacerbated (HSV-1)

or altered expression of co-stimulatory molecules on APCs, resulting secondary iNKT cell activation. ²⁸⁴

1.2.7.2 Allergy and autoimmunity

The effect of iNKT cells on allergy and autoimmunity varies according to the model used.

In some models, iNKT cell activation exacerbates allergy and autoimmunity. For instance, the systemic administration of α -GalCer abrogates oral tolerance in mice, an effect mediated by DCs in the lamina propria and involving the activation of small intestine iNKT cells.²⁸⁸ In some animal models of allergic asthma, iNKT cell-deficient mice are resistant to allergen challenge.²⁸⁹ A bacterial-induced autoimmune hepatitis is dependent upon iNKT cells.²⁹⁰ Deficiency of iNKT cells ameliorates collagen-induced arthritis, and CD1d blockade early in the course of the arthritis mitigates the disease.²⁹¹ Finally, in humans, activated iNKT cells are reported to account for a high proportion of the cellular infiltrate in contact dermatitis.²⁹²

In contrast, other studies indicate that iNKT cells can provide protection against allergy and autoimmunity. The Th2-polarising iNKT cell ligand OCH prevents the development of experimental autoimmune encephalitis in an IL-4 dependent manner.²⁹³ In non-obese diabetic mice, repeated intraperitoneal injections of α -GalCer suppress type 1 diabetes by recruiting tolerogenic DCs,²⁶³ and the adoptive transfer of iNKT cells suppresses diabetes, while CD1d knockout exacerbates it.²⁹⁴ The Th1-polarising iNKT cell agonist α -C-GalCer is protective against experimental autoimmune uveitis.²⁹⁵ Finally, in a spondyloarthritis model which involves gut and joint inflammation, deficiency of iNKT cells aggravated the disease.²⁹⁶

Studies of mouse models of GVHD are particularly informative: host iNKT cells are required for protection against acute GVHD.²⁹⁷ This tolerance requires host recognition of CD1d on donor marrow cells²⁹⁸ and involves IL-4 produced by host iNKT cells acting upon donor T_{reg} cells.²⁹⁹ In contrast, early administation of α -GalCer after mismatched transplant results in DC- and IL-12-dependent transactivation of NK and T cells, hyper-acute GVHD and early death.³⁰⁰ Interestingly, the α -GalCer variant C20:2, which unlike α -GalCer fails to induce NK cell transactivation, inhibits GVHD.³⁰⁰

1.2.7.3 Cancer

Experimental and observational evidence suggests that iNKT cells play a role in immune surveillance against spontaneous malignancies.

Mice deficient in iNKT cells have an increased susceptibility to methylcholanthreneinduced sarcoma, and tumour resistance can be restored by adoptive iNKT cell transfer.³⁰¹ In a mouse model of spontaneous prostate adenocarcinoma, the absence of iNKT cells was associated with earlier onset of cancer and shorter survival.³⁰²

Patients with advanced solid malignancies and acute myeloid leukaemia have reduced numbers of circulating iNKT cells compared to healthy controls.^{210,303–305} There is some

evidence that numerical and functional defects of iNKT cells are selective for this subset. Omental iNKT cell numbers are reduced in patients with colorectal carcinoma, while other omental T cell subsets are retained. ³⁰⁶ In advanced prostate cancer, reduced iNKT cell IFN- γ production has been observed, despite normal IFN- γ production by bulk T cells. ³⁰⁴ Finally, impaired iNKT cell production of IFN- γ has been reported in patients with progressive myeloma, despite normal IFN- γ responses to influenza and staphylococcal enterotoxin A, and despite normal circulating T and NK cell numbers. ³⁰⁷

The number of circulating iNKT cells may also be of prognostic relevance. In patients with advanced prostate cancer, the reduction in iNKT cell numbers compared to healthy controls was similar to the reduction seen in T cells and NK cells, but only iNKT cell numbers correlated with survival.³⁰⁸ In patients with acute myeloid leukaemia, although the number of bone marrow T and NK-like T cell populations correlated with survival on univariate analysis, only peripheral blood iNKT cell number predicted survival on multivariate analysis.³⁰⁵

There are several mechanisms by which iNKT cells might help to eradicate malignancy, including direct cytotoxicity against CD1d-expressing tumour cells, transactivation of other effector cell types, such as NK cells, and augmentation of 'conventional' T cell responses to TAAs by promotion of DC function. Each of these mechanisms has been demonstrated preclinically *in vivo*.

- **Direct cytotoxicity:** An α -GalCer-pulsed CD1d-transfected B16 melanoma cell line was lysed by iNKT cells *in vivo*, resulting in significant protection against meta-stasis.²⁵⁴ *In vivo* iNKT cell cytotoxicity against tumour cells is dependent on the expression of CD1d on the tumour and is strongly enhanced by the presence of α -GalCer.^{309,310}
- **Transactivation of NK cells:** The production of IFN- γ by NK cells is necessary for α -GalCer-induced protection against metastatic malignancy.²⁷¹ The combination of α -GalCer and IL-21 resulted in potent antitumour activity, which was dependent upon NK cells, perform and IFN- γ .²⁷²
- Activation of DCs: The administration of α -GalCer in conjunction with tumour antigen resulted in dendritic cell maturation and enhanced 'conventional' CD4+ and CD8+ T cell responses against subsequent tumour challenge.^{258,259,311}

In contrast, there is also evidence that iNKT cells can suppress immune responses to cancer, with deficiency of iNKT cells improving outcomes in some murine tumour models. For instance, responses of a 4T1 mammary carcinoma to radiotherapy and anti-CTLA-4 monoclonal antibody (mAb) were augmented in V α 14J α 18^{-/-} iNKT cell-deficient mice.³¹² Mice deficient in iNKT cells (both CD1d^{-/-} and J α 18^{-/-}) were more resistant to an inoculated CD1d-transfected T cell lymphoma than wild type mice. Following tumour inoculation, wild type mice had much higher levels of circulating IL-13 than the iNKT cell deficient mice, suggesting that iNKT cell-derived or -induced IL-13 may be involved in iNKT cell-mediated immunosuppression.³¹³

In some models, CD1d-dependent tumour tolerance appears to be mediated by type II NKT cells rather than iNKT cells: CD1d^{-/-}, but not wild type, mice rejected an immunogenic fibrosarcoma. J α 18^{-/-} mice also failed to reject the tumour, indicating that type II NKT cells were responsible for tumour tolerance in the wild type and J α 18^{-/-} mice.³¹⁴

Chapter 1. Introduction

The administration of sulfatide, a type II NKT cell agonist, resulted in an increase in the number of metastases from an inoculated colon carcinoma cell line, an effect which could be attributed to CD4+ type II NKT cells. Moreover, in this latter model, the immunosuppressive effect of sulfatide-mediated type II NKT cell stimulation was dominant over the tumour protective effect of α -GalCer-mediated iNKT cell stimulation.³¹⁵

1.2.8 Directing the iNKT cell response

The findings in models of infection, allergy, autoimmunity and cancer, outlined in sections 1.2.7.1 to 1.2.7.3, indicate that iNKT cells are capable of both immunostimulatory and immunosuppressive effects. This could be considered consistent with the capacity of iNKT cells to produce a broad range of cytokines, as discussed in section 1.2.4, and with their interactions with multiple cell types, as discussed in section 1.2.6.

Effective application of iNKT cell-directed immunotherapies requires an understanding of the factors which influence the outcome of iNKT cell stimulation. Considering the factors which influence the outcome of iNKT cell activation in different *in vivo* and *in vitro*, settings, common themes emerge.

Potency of iNKT cell ligand Activation of iNKT cells with potent synthetic iNKT cell ligands, such as α -GalCer or α -C-GalCer tends to favour the induction of Th1-type cytokine production by iNKT cells, and rejection of tumour. Activation with weaker iNKT cell ligands, such as endogenous glycolipids, or with stronger ligands at limiting concentrations, or with strong ligands at very low concentrations, tends to favour production of Th2-type cytokines by iNKT cells, or results in activation of type II NKT cells, which can result in tolerance.

Type of antigen-presenting cell The presentation of a potent iNKT cell ligand on CD1d by professional APCs, such as DCs, tends to favour iNKT cell-induced rejection of tumour. Presentation on non-professional APCs, such as monocytes, tends to favour tolerance.

Cytokine environment The presence of IL-12 during iNKT cell stimulation, which may be derived from mDCs during iNKT cell stimulation *in vivo*, favours Th1-type responses and tumour rejection. In contrast, the presence of IL-2 can favour the induction of tolerance.

Natural killer receptor signals Invariant natural killer T cells express functional NK receptors and, like NK cells, integrate signals from activating and inhibitory NK receptors. A balance of activating NK receptor signals, such as may be elicited by allogeneic cells or by tumour cells which have downregulated MHC Class I, will favour a Th1-type cytokine response and GVHD. In contrast, the presence of strong inhibitory NK receptor signals, for instance elicited by an autologous MHC Class I- or HLA-G-expressing tumour, would favour tolerance.



Figure 1.7 Factors influencing the outcome of iNKT cell stimulation

Phenotype of responding natural killer T cell The CD4+ subset of iNKT cells produces the potentially immunosuppressive Th2-type cytokines, whereas the CD4- subset does not. While iNKT cells can reject tumour, type II NKT cells (which are CD1d-restricted but are non-invariant, and which do not respond to α -GalCer), are associated with tolerance induction and tumour suppression.^{316,317}

To summarise, stimulation of iNKT cells with high concentrations of potent synthetic ligands such as α -GalCer presented by professional APCs, in combination with stimulation of activating NK receptors, and/or IL-12, favours a Th1-type cytokine response, eradication of malignancy and infection, and, in some models, autoimmunity.

In contrast, stimulation of iNKT cells (or type II NKT cells) with weak or endogenous ligands presented by non-professional APCs, in conjunction with an inhibitory NK signal, and the presence of suppressive cytokines, favours a Th2-type cytokine response and tolerance induction.

Figure 1.7 illustrates some stimuli which may influence iNKT cell cytokine production and effector responses.

1.3 Cellular immunotherapy

As opposed to humoral immunotherapy or conventional chemotherapy, cellular immunotherapies aim to elicit long-lasting T cell immunity against malignant cells, with the goal of eradicating or controlling the tumour.

Chapter 1. Introduction

In this section, I shall discuss the evidence that CLL is susceptible to immunological control. I shall then review strategies for adoptive immunotherapy and discuss the clinical experience of adoptive immunotherapy of CLL to date. Finally, I shall review methods of incorporating iNKT cell engagement into the adoptive immunotherapy of cancer.

1.3.1 Immunological control of CLL

A number of clinical observations suggest that CLL is susceptible to immunological control, and may therefore be a suitable candidate malignancy for adoptive immunotherapy.

Experience of allo-SCT for CLL indicates that allogeneic T cells are capable of eradicating CLL. Allo-SCT can result in durable clinical remissions ('cures'), and in allogeneic transplantation registry studies, T cell depletion of the graft is associated with higher relapse risk, and development of chronic GVHD is associated with lower relapse risk.¹⁵⁴ After allo-SCT, a graft-versus-leukaemia effect against CLL can be observed, with disease remission following the infusion of additional unselected donor lymphocytes to patients who have relapsed after allo-SCT.³¹⁸ Interestingly, the rates of long-term disease-free survival after syngeneic transplantation are higher than those observed after autologous transplantation, suggesting that T cells can control CLL cells even without the assistance of minor histocompatibility antigen or NK receptor mismatches.³¹⁹

Autologous T cells may also be able to eradicate CLL. Spontaneous remissions of CLL, although rare, are well documented, ^{320,321} and may be associated with the development of autoimmune cytopenias.³²² Leukaemia-reactive T cells can be detected in the peripheral blood of some patients with CLL, and can increase in number after administration of autologous tumour vaccines^{323,324} or a DC vaccine.³²⁵ Supporting the notion of control of leukaemia by autologous T cells, patient-derived activated T cells were able to protect against leukaemia development after autologous CLL challenge in a chimeric mouse model.³²⁶

Finally, early clinical experience with T cell redirection using bispecific antibodies against B cells and CD3 and with chimeric T cell receptors directed against B cells suggests that T cells can eradicate CLL, even in patients with a heavy disease burden.^{327,328}

Together, these findings imply that T cells have the potential to mediate long-lasting immunity against CLL, and that at least some patients with CLL have a T cell population capable of recognising their own leukaemic cells. Nonetheless, the fact that patients have developed CLL could be construed as *de facto* evidence that barriers to T cell control of autologous CLL cells exist.

There are many potential reasons that immunological control of CLL might fail, resulting in disease progression.

Recruitment of suppressive myeloid cells The principal of immunosuppression by myeloid cells within the tumour microenvironment is well-established in solid malignancies, where MDSCs, a heterogeneous population of myeloid cells, accumulate in tumours and lymph nodes and potently suppress T cell function.³²⁹ In CLL, the term 'nurse-like cells'

has been used to refer to myeloid-derived cells that attract CLL cells in culture and support CLL cell survival. Although derived from CD14+ precursors, these nurse-like cells have an immunophenotype distinct from monocytes.³³⁰ In addition, monocytes themselves can promote survival of CLL cells through release of soluble CD14.³³¹

Recruitment of regulatory T cells T_{reg} cells exert immunosuppression via both contactdependent mechanisms and through the release of soluble inhibitors. The number of Tregs is increased in CLL, and higher numbers are associated with a poorer prognosis.^{98,332} CLL-derived TGF- β or CLL-induced IL-10 may be responsible for T_{reg} recruitment.

Induction of dendritic cell dysfunction Numeric,⁷⁸ immunophenotypic and functional⁸⁰ defects have been reported in the circulating DC compartment of patients with CLL. These defects are dependent on CLL cells, as moDCs generated from the same patients in the absence of CLL cells have normal phenotype,³³³ and addition of allogeneic CLL cells to healthy donor moDCs results in abnormal DC maturation and function⁸⁰

Secretion of inhibitory factors Levels of the immunosuppressive cytokine IL-10 are increased in patients with CLL, and correlate inversely with survival, ¹⁰⁴ although the cell of origin is not clear. ³³⁴ Tumour cell-derived IL-6 has been proposed as a mediator of CLL immunosuppression; serum IL-6 levels are raised in CLL patients, and higher levels are associated with an inferior prognosis. ⁹¹ Soluble forms of the APC costimulatory molecules CD40, CD80 and CD83 exhibit immunosuppressive properties *in vitro*, presumably by binding their ligands on T cells. Levels of soluble CD40, CD80 and CD83 are raised in patients with CLL, and correlate inversely with prognosis. ^{108–110} CLL cells also produce TGF- β , ³³⁵ which suppresses effective DC presentation of tumour antigens. ¹⁰³

Diversion of immunostimulatory factors CLL cells frequently express CD25, the IL-2 receptor alpha, and may deprive T and NK cells of a proliferative signal by themselves absorbing $IL-2^{106}$

Resistance to natural killer cell killing HLA-G is a non-classical HLA class Ib antigen, and is tolerogenic: HLA-G expression on target cells inhibits NK cytotoxicity by binding to NK inhibitory receptors.³³⁶ HLA-G can also suppress CD8+ cytotoxic T cell responses, and when expressed on antigen presenting cells skews CD4+ responses towards tolerance. Expression of HLA-G in health is usually restricted to immune-privileged organs including the placenta, thymus, cornea and erythroblasts. CLL cells express surface HLA-G, and blocking antibodies to HLA-G enhance the susceptibility of CLL cells to NK cell killing.³³⁷ Additionally, soluble forms of ligands of the activating NK receptor NKG2D, can reduce NK cell cytotoxicity by inducing downregulation of NKG2D on NK cells. Levels of soluble NKG2D ligands are raised in patients with CLL and correlate inversely with prognosis.³³⁸

Restriction of T cell repertoire Although the diversity of the CD4+ T cell repertoire is well maintained in healthy humans up to 60 years of age, it declines dramatically beyond the age of 70,³³⁹ the median age of patients with CLL at diagnosis. Patients with CLL have evidence of oligoclonal T cell expansions, and exhibit a restricted T cell repertoire as determined by TCR V β spectratyping,^{326,340} even in comparison with agematched controls.⁸⁷ CLL treatment with alemtuzumab results in a further exacerbation of this CDR3 restriction, although this recovers over time.³⁴¹ Thus, there is a possibility that, because of advanced age, chemo- or immunotherapy, or the leukemia itself, patients with CLL do not possess a sufficiently diversified T cell repertoire to be able to counter their own malignant clone.

The majority of the mechanisms of immune evasion of CLL outlined above are dependent on presence of the leukemic clone. Therefore, depletion of leukaemic cells before immunotherapy may enhance immunological responses. This may be one reason that patients who are in remission before undergoing RIC-allo have a significantly better long-term disease-free survival than those who enter the procedure with active disease.³⁴²

1.3.2 Strategies for cellular immunotherapy of CLL

Allo-SCT is the only form of cellular immunotherapy currently in routine clinical use for the treatment of haematologic malignancies, but despite its proven efficacy, the application of allo-SCT remains limited due to high treatment-related morbidity and mortality, cost and a lack of suitable stem cell donors. Cellular immunotherapy strategies with the potential for lower toxicity are therefore attractive.

Many cellular immunotherapy strategies rely on a common series of events that mimic the normal induction of immunity: TAAs are taken up by APCs and presented to T cells. As a result, T cells specific for the TAAs are activated and expand, and directly or indirectly lyse the target tumour cells.

Direct vaccination of a patient with TAAs, such as with whole tumour cells, tumourderived proteins or peptides, or with tumour DNA or RNA (which is transcribed and/or translated to tumour protein) is simple, and may result in immunological responses against residual tumour within the patient. However, this strategy relies on intact APC and T cell function within the tumour-bearing host.

To overcome potential impairments of patient APC function, some groups have used DC vaccination, in which autologous or allogeneic moDCs are generated *in vitro* under optimal conditions, are pulsed with whole tumour cell, tumour peptide or protein or tumour DNA or RNA, and are subsequently administered as a cellular vaccine, to generate antitumour T cell responses *in vivo*.

As DC vaccination may not overcome defects of T cell activation and proliferation in patients with cancer, some groups go further by selecting and expanding tumour-reactive T cells *in vitro*, for subsequent reinfusion. Even this 'adoptive immunotherapy' approach has theoretical limitations, as it presupposes that patients with cancer possess a sufficient T cell repertoire to eradicate the tumour. Administration of allogeneic T cells, or autologous T cells bearing a synthetic TCR or a chimeric antigen receptor directed against a tumour antigen may overcome this limitation. Finally, patients' own T cells may be redirected against tumour using bispecific antibodies against T cell CD3 and a TAA.

A number of these cellular immunotherapy strategies are illustrated schematically in table 1.15. The next section will discuss each of these strategies, with specific reference to their use in the treatment of CLL.

1.3.2.1 Peptide vaccination

Studies of mRNA expression indicate that certain TAAs are expressed in the leukaemic cells of CLL, but not in normal cells. These are potential candidate antigens for peptide, protein, RNA or DNA vaccination strategies. Selected CLL-associated antigens reported in the literature are detailed in table 1.16. Individual or multiple peptide antigens can be administered as vaccines, although even when a single antigen is used, the phenomenon of antigen spreading can result in antitumour responses against tumour-specific peptides other than those vaccinated against.³⁴³

A phase I clinical trial peptide vaccination in CLL utilised a HLA-A2 restricted RHAMMderived epitope (the 'R3 peptide') together with GM-CSF and incomplete Freund's adjuvant. This vaccination strategy resulted in proliferation of antigen-specific CD8+ T cells, as detected by tetramer and by ELISpot. White cell counts were lower after the four vaccine doses than before vaccination in four out of six patients, although no patients met criteria for a clinical partial response.^{3,344}

Heteroclitic peptides are those that have been modified to bind to MHC with higher affinity, while leaving the TCR recognition residues intact. Heteroclitic peptides can lead to enhanced induction of T cell responses against both the heteroclitic and the native peptide. Cytotoxic lymphocytes generated using heteroclitic Ig-derived peptides exhibited enhanced cytotoxicity against primary CLL cells *in vitro*, compared to those generated using native peptide.³⁴⁵

1.3.2.2 Tumour cell vaccination

As B cells express MHC class II as well as MHC class I, and are effective, albeit nonprofessional APCs, CLL cells themselves may be employed as a vaccine. This approach has some appeal, as leukaemic cells are readily available from peripheral blood, and are sufficiently common that minimal or no purification steps are required.

Unmodified CLL cells In a phase I study, autologous irradiated leukemic cells were administered intradermally to patients with early CLL with or without the Bacille Calmette-Guerin (BCG) bacterium as an adjuvant. Reductions in the leukocyte count were observed in five of seventeen patients, and a prolongation of the lymphocyte doubling time in seven of seventeen patients. There was no evidence of an improvement in vaccine efficiency with the addition of BCG.³⁵⁶

Another clinical trial consisted of repeated intramuscular vaccinations with heat-shocked, ultraviolet light-treated and oxidised autologous whole blood (which included a high

T cell redirection	Transgenic T cell transfer	Adoptive T cell transfer	DC vaccination	Peptide or protein vaccination	Whole tumour cell vaccination	Allogeneic stem cell transplantation	Immunotherapeutic strategy
			In v	In vitro			Generation of TAA
1	I	In v	itro				APC uptake of TAA
I		itro		In 1	In vivo	In vivo	TAA presentation to T cells
	In vitro		In vivo	vivo			T cell expansion
In vivo	In vivo	In vivo					Tumour killing

Table
1.15
Methods c
of generating
÷
cell
anti-tumour
immunity

Table 1.16 Selected (CLL tumour-associ	ated antigens		
CLL-associated antigen	Frequency of expression	T cell responses	Reference(s)	Notes
RHAMM *	77%	3/6 patients showed RHAMM R3 peptide-specific ELISPOTs after restimulation. Expansion of R3-specific CD8+ T cells seen in 2/3 cases	346	Clinical trial of R3 peptide vaccination reported ^{3,344}
Fibromodulin	63 - 100%	Fibromodulin-specific CD8+ T cells detected by ELISPOT and tetramer after restimulation	346,347	
† bi	100%	Antigen differs for each patient. 67% of patients have Id-peptide reactive T cells	348,349	
hTERT ‡	76%	hTERT-specific cytotoxic T cells expanded from 6/7 telomerase CLL patients	350	A second study showed no hTERT expression in CLL ³⁴⁶
ROR1	100%	Anti-ROR1 antibodies detected in sera of 3/6 patients who received iv autologous CD40L-transfected CLL cells	351,352	
Ly9 (CD229)	100%	Expansion of CD229-specific T cells observed in 5/5 cases after 2x restimulations	353	
MDM2 [§]	78 – 85%	T cell responses demonstrated by ELISPOT and tetramer staining	354	
CD23	100%	T cell responses demonstrated by ELISPOT and tetramer staining	355	
 * receptor for hy * idiotype (speci * human telomen § human homolo 	lauronic acid media fic CLL cell BCR) rase reverse transcri sgue of murine doul	ited motility; CD168 iptase ble-minute 2		

49

proportion of CLL cells). Clinical partial responses were seen in 5 out of 18 patients with CLL. 357

CD40L-activated CLL cells Although they express MHC Class II, unmodified CLL cells are poor inducers of allogeneic T cell responses.³⁵⁸ However, CLL cells express CD40, and upon treatment with CD40 ligand (CD40L, or CD154), upregulate costimulatory and adhesion molecules, and become potent presenters of antigen to both CD4+ and CD8+ allogeneic T cells.³⁵⁹

The effect of CD40L-activated CLL cells may be somewhat different in the autologous setting: CD40L-activated CLL cells led to expansion of CD8+ cytolytic T cells in the allogeneic setting, but in the autologous setting resulted in expansion of a population of CD4+ Th1-like (IFN- γ producing) T cells.³⁶⁰ A separate study also showed impaired CD8+ responses to CD40L-activated CLL cells in the autologous setting.³⁶¹

An alternative approach to providing a CD40L signal to CLL cells is to induce CLL cells to express CD40L themselves. Transfection of human CLL cells with an adenoviral vector encoding mouse CD154 (CD40L) resulted in crosslinking of CD40 on the transduced, and on bystander CLL cells. This resulted in upregulation of costimulatory molecules on the CLL cells and improved allogeneic and autologous T cell responses to the leukaemic cells *in vitro*. ³⁶² Co-transfer of both CD40L and the additional accessory molecule OX40 ligand (OX40L) to CLL cells may further enhance and enable expansion of autologous cytotoxic CD8+ cells. ³⁶³

One concern regarding CD40 ligand-stimulated B cells is that CD40L treated CLL cells are resistant to apoptosis. However, despite increased resistance to chemotherapy-induced apoptosis, CD40L-treated, peptide-pulsed CLL cells remain sensitive to killing by autologous T cells.³⁶⁴

Phase I clinical trials of administration of autologous CD40 activated CLL cells have been reported. In one study, eleven patients were treated with autologous CLL cells transfected with mouse CD40L.³²⁴ This resulted in at least a doubling of T cell numbers and a transient reduction in CLL cell count in most patients. Interestingly, reductions in lymph node size were more marked and prolonged. An increase in T cell responses to autologous CLL cells was observed by ELISpot and mixed lymphocyte reaction (MLR). A further study using CLL cells transfected with human CD40L yielded similar results, including regression of lymphadenopathy and a prolongation of lymphocyte doubling time in around half of patients.³⁶⁵

In a clinical trial CD40L- and IL-2 transfected CLL cells were co-administered as a subcutaneous vaccine to nine patients with CLL. *In vitro* T cell responses against autologous CLL cells were observed in seven patients, but no reduction in circulating CLL cells was seen, and only three patients exhibited a transient reduction in lymphadenopathy.³⁶⁶ An increase in CD4+CD25+ T cells, a phenotype associated with T_{reg} activity, was seen after vaccination, raising the possibility that the IL-2 transfected CLL cells promoted T_{reg} proliferation.
B-7-, ICMA-1- and LFA-3- transfected CLL cells The transfection of CLL cells with a viral vector encoding the co-stimulatory molecules B7-1, ICAM-1 and LFA-3 resulted in a similar degree of augmentation of autologous T cell responses to transfection with CD40L *in vitro*.³⁶⁷

Toll-like receptor ligand-treated CLL cells CLL B cells express the Toll-like receptors (TLRs) TLR-1, TLR-2, TLR-6, TLR-7 and TLR-9. TLR stimulation can result in production of inflammatory cytokines including IL-6 and TNF-α, upregulation of costimulatory molecules such as CD86, enhanced immunogenicity in MLRs and can increase CLL cell susceptibility to cytotoxic chemotherapies. A case report describes regression of a cutaneous CLL deposit with topical imiquimod, a TLR-7/8 agonist.³⁵⁷ However, a subsequent phase I trial of the imidazoquinoline TLR-7 agonist 852A given intravenously as monotherapy to patients with relapsed CLL did not result in clinical responses.³⁶⁸ A note of caution should also be sounded, as TLR ligands can trigger CLL cell proliferation.^{369–371} Nonetheless, interest in TLR ligands in combination with chemotherapies, or as adjuvants to vaccination strategies, remains.

Epigenetically-modified CLL cells CLL cells treated with a combination of a demethylation and histone deacetylase inhibitor upregulated cancer testis antigens and costimulatory molecules, becoming more immunogenic in an allogeneic assay *in vitro*.³⁷²

Trioma vaccine 'Trioma' cells are produced by fusing tumour cells with an anti-CD64 mouse hybridoma. CD64 is the high affinity IgG receptor (FcγRI), and is expressed on macrophages and monocytes. Trioma CLL cells possess TAAs and express anti-CD64, which targets them to macrophages and monocytes. A CLL trioma vaccine was tested for T cell activating activity *in vitro*, and resulted in expansion of T cells against antigens overexpressed on the original leukemic cells.³⁷³ No clinical studies of this approach have been published to date.

1.3.2.3 Dendritic cell vaccination

As defects of DC function have been described in CLL, a rational approach may be to generate functional DCs *in vitro*, pulse these with tumour antigen, and deliver tumour-loaded DCs as a vaccine. In support of this approach, patient moDCs loaded with CLL cell lysates can stimulate proliferation of tumour-specific T cells *in vitro*.³⁷⁴

Various sources of TAAs for DCs have been studied, including apoptotic tumour cells themselves, whole cell lysates, peptide eluates or tumour RNA. A phase I trial used autologous moDC loaded with cell lysates, and resulted in reductions in CLL cell burden in five of twelve patients.^{356,375} A phase 1 study of vaccination of patients with low-grade B cell malignancies with autologous moDC loaded with eluted peptide included two patients with CLL: neither patient experienced a clinical response.³⁷⁶ *In vitro* studies suggest that in CLL, apoptotic tumor cells provide stronger autologous T cell responses than tumour cell lysate or RNA.³⁷⁷ An alternative strategy is to fuse tumour cells with DCs, although in one study, CLL/DC fusion hybrids exhibited a lower antigen presenting capability than DCs loaded with apoptotic CLL cells. 378

Despite occasional clinical responses, the adoption of autologous DC vaccination for CLL has been limited by difficulties in generating moDCs from CLL patients. Yields of moDC are often lower in CLL patients than in healthy donors.³⁷⁶ This may be in part because monocytes constitute a smaller proportion of circulating nucleated cells when circulating leukemic cells are frequent. In addition, CLL cells can directly interfere with moDC generation.⁸¹ Immunomagnetic selection of CD14+ monocytes can overcome these problems,³⁷⁹ and a clinical-scale process for this has been described.³⁸⁰

One way of overcoming practical difficulties in the generation of moDC from patients with CLL, and bypassing any defective moDC function would be to make use of allogeneic rather than autologous moDC as antigen presenters. A phase I clinical trial of DC vaccination with allogeneic moDCs, loaded with whole cell lysates or apoptotic bodies³²⁵ resulted in transient reductions in white cell counts in some patients, but no durable objective clinical responses.

1.3.2.4 Adoptive T cell transfer

Activated autologous T cells In one study, five of twenty patients with CLL had circulating T cells exhibiting *in vitro* cytotoxicity against autologous CLL cells. When these patients' T cells were activated by CD3/CD28 engagement and co-administered to mice alongside CLL cells in a xenograft model, CLL cells were eradicated. Unactivated T cells, and activated T cells from patients which did not show *in vitro* cytotoxicity, did not eradicate CLL.³²⁶

A phase I/II clinical trial of CD3/CD28 activated autologous T cells for CLL has been performed. Final results have not been published, but a preliminary abstract indicated that the treatment led to reductions in lymphadenopathy, but no reduction in circulating CLL cell numbers.³⁸¹

Allogeneic T cells The experience of allo-SCT indicates that allogeneic T cells have the capacity to eradicate leukaemia cells in CLL. In one notable case report, the administration of CD3/CD28-activated donor T cells resulted in a complete clinical response in a patient with persistent CLL after allo-SCT.³⁸²

The major disadvantages of using allogeneic T cells are the need for tissue-type matching and the risk of GVHD in the recipient. One approach that may overcome both of these risks is the generation of allo-restricted antigen-specific T cells.

MoDCs from healthy HLA-A*0201-negative donors were engineered to express HLA-A*0201 presenting a peptide from CD20 (a B cell antigen). These moDCs were cultured with autologous T cells. T cells specific for the CD20 peptide presented on HLA-A*0201 expanded, and could be detected and purified using a pentamer. *In vitro*, these allorestricted CD20-specific T cells were able to specifically lyse CLL lines from HLA-A*0201 patients. HLA-A*0201 negative cells and CD20 negative cells were not lysed.³⁸³

A potential disadvantage of this approach is that allo-restricted antigen-specific T cells generated in this way would be susceptible to rejection by the leukaemia-bearing host.

Transgenic TCR Transfection of autologous T cells with a transgenic TCR specific for a tumour antigen holds the promise of overcoming deficiencies in the patient's T cell repertoire. Preclinical and clinical studies suggest that this is a promising approach.

In a xenograft model, T cells from patients with myeloid leukaemias were transfected to express a TCR specific for the Wilms' tumour 1 (WT1) tumour antigen. These T cells were able to eradicate a human leukaemia cell line in NOD/severe combined immunodeficiency (SCID) mice.³⁸⁴

Lymphodepleting chemotherapy followed by adoptive transfer of autologous T cells transfected with a TCR specific for the tumour-associated antigen NY-ESO-1 led to objective clinical responses in five of eleven patients with metastatic melanoma and four of six patients with synovial cell carcinoma.³⁸⁵ Clinical responses have also been reported following adoptive transfer of autologous T cells bearing transgenic TCRs specific for the tumour-associated antigen MART-1 in patients with metastatic melanoma.³⁸⁶

A major disadvantage of the use of transgenic TCR T cells is that the marked polymorphism of the HLA locus means that the generation and validation of antigen-specific TCRs to cover a comprehensive range of tissue types and malignancies would be a large undertaking. There are also theoretical concerns that heterodimerisation of a transgenic TCR- α or β chain with a native TCR chain might generate a new TCR with undesired (for instance, autoreactive) specificity.³⁸⁷ Finally, tumour cells may downregulate MHC, reducing their susceptibility to TCR-transduced T cells.³⁸⁸

Chimeric antigen receptors Chimeric antigen receptors (CARs) comprise a single chain antibody fragment linked to the CD3 ζ chain of the T cell receptor. In some constructs, the CD3 ζ chain is also linked to costimulatory molecules such as CD28 or CD137 to provide a co-stimulatory signal. Upon recognition of the CAR cognate antigen, T cell signalling is triggered, and effector functions take place, thus CARs confer specificity without tissue-type restriction.

T cells transfected with a CAR against CD19 or against the CLL-associated antigen ROR1 can lyse autologous CLL cells *in vitro*.^{389,390} An anti-CD23 CAR showed promising efficacy *in vitro* and in a xenograft model *in vivo*.³⁹¹

The efficacy of CD19-directed CAR T cells for follicular lymphoma in a phase I clinical trial was been limited, perhaps owing to short persistence of the CAR-transfected T cells.³⁹² However, preliminary results from subsequent trial which incorporates T cell depletion before, and high dose IL-2 after, infusion of CAR T cells indicate that this treatment has the potential to control follicular lymphoma.³⁹³

Finally, a phase I clinical trial of CD19-directed CAR T cells as a treatment of CLL has shown positive preliminary results, with complete remission of a patient with refractory CLL and persistence of chimeric T cells six months after infusion.³²⁸

Chapter 1. Introduction

Risks of adoptive T cell transfer As with all treatments, adoptive T cell transfer is not without risk: in an animal study of adoptive T cell transfer combined with a long peptide vaccine, >1000 fold expansion of the transferred CD8+ T cells was observed. Subsequent repeated vaccination caused a cytokine storm and multi-organ failure³⁹⁴

In a phase I clinical trial of CAR T cells for CLL, one patient died of a sepsis-like syndrome, which began 20 hours after the infusion of autologous CAR T cells following cyclophosphamide chemotherapy.³⁹⁵ The death was not definitively attributed to CAR T cell infusion, but modifications to the infusion protocol were subsequently made.

Clinical trials of gene therapy for severe combined immunodeficiency have demonstrated the potential for retroviral vectors to cause malignancies, by causing over-expression of proto-oncogenes.³⁹⁶ One way around this is to transfect T cells with mRNA for a CAR. T cells transfected with the mRNA for an anti-CD19 were able to lyse target B cells up to 8 days after transfection *in vitro*.³⁹⁷

Finally, autoimmunity is a potential risk for all immunotherapies.³⁹⁸ On-target autoimmunity is particularly well described in clinical trials of transgenic TCR therapy for melanoma: vitiligo was frequently observed, 25% of patients required treatment for anterior uveitis (melanocytes are present in the eye), and around a third developed mild hearing loss (melanocytes exist in the inner ear).³⁹⁹ Similarly, clinical trials of CD19-directed CAR T cells result in a profound depletion of normal, polyclonal B cells,³⁹⁷ the longterm impacts of which are not yet known. Finally, off-target autoimmunity may occur, either through cross-reactivity of the transfected receptor, epitope spreading, or through activation of a clone of transfected T cells which express a native TCR with autoreactive potential.

1.3.2.5 T cell redirection

Bispecific T cell engagers consist of two single-chain antibodies fused together. One antibody is directed against a T cell antigen, usually CD3, and the other against the target cell. The most advanced in clinical studies is blinatumomab, which has specificity for both CD19 and CD3, re-directing T cells against B cells. In a dose-finding study in non-Hodgkin lymphoma, complete or partial responses were seen in all seven patients treated at the highest dose level. Two patients with CLL were treated, both at lower dose levels, one of whom had a partial response.⁴⁰⁰ Many of the clinical responses observed have proved durable, raising the possibility that this form of T cell engagement may be able to recruit endogenous tumour-specific T cells capable of long-term disease control.³²⁷

A related agent makes use of a single chain antibody fused to HLA. Expansion of CMV-reactive CD8+ T cells is observed in many patients with CLL. A streptavidin-linked antihuman CD20 single-chain variable fragment (scFv), coupled to biotinylated HLA class I loaded with CMV peptide, is used to target CMV-reactive T cells to autologous B cells. This reagent redirected CMV reactive CD8+ T cells to recognise and kill CLL cells *in vitro*.⁴⁰¹

1.3.3 Incorporating iNKT cells into immunotherapy

As discussed in section 1.2.2, iNKT cells possess a semi-invariant TCR capable of recognising glycolipid antigens in the context of the MHC Class I-like molecule CD1d. In response to TCR engagement, iNKT cells produce Th1- and Th2-type cytokines and upregulate surface co-stimulatory molecules such as CD40L. Together, these soluble and contactdependent signals can transactivate other cell types, such as NK cells and DCs.

The interactions between iNKT cells and other cells of the innate and adaptive immune system have been discussed in section 1.2.6. Specific capabilities of iNKT cells that might usefully be exploited for tumour immunotherapy include direct cytotoxicity against CD1d-positive target cells in the presence of α -GalCer, ³¹⁰ induction of DC maturation and consequent enhancement of CD4+ and CD8+ T cell responses to DC-presented antigens, ²⁵⁸ transactivation of NK cells and enhancement of NK cytotoxicity, ⁴⁰² provision of T cell help for the generation of B cell immunity, ²⁶⁶ and depletion of MDSCs and TAMs. ^{282,283}

Other lymphocyte populations, such as conventional CD4+ and CD8+ T cells or NK cells, can exhibit many of these effects, but iNKT cells possess a unique combination of features.

- Universal agonist: T cells bearing the specific iNKT cell TCR can be detected most individuals, if at a low frequency.²²⁰ Thus, a population of iNKT cells capable of responding to α -GalCer in the context of CD1d is usually present, irrespective of tissue type or of the constitution of rest of the T cell repertoire.
- Universal presenting molecule: Unlike HLA, the CD1d locus is minimally polymorphic.¹⁹³ This allows the specific activation of iNKT cells by α -GalCer on CD1d to be exploited in all individuals, without tailoring the therapy to the tissue type.
- **Rapid activation upon first stimulation:** iNKT cells are potent cytokine producers, and like effector memory cells, do not require priming—cytokine release occurs upon the first stimulation with α-GalCer.

Together with the observations of increased cancer susceptibility in some iNKT celldeficient mouse models (discussed in section 1.2.7.3), and experimental evidence that iNKT cell activation can protect against inoculated tumours, these factors have prompted extensive investigation of the potential for iNKT cells in tumour immunotherapy.

Broadly speaking, three principal tumour immunotherapy strategies exploiting iNKT cells have been employed:

- 1. Simple activation of iNKT cells, resulting in cytokine release and consequent transactivation of other immune system cells, such as NK cells. Methods of achieving this include systemic administration of α -GalCer, or administration of unloaded APCs that have been pulsed with α -GalCer.
- 2. *Ex vivo* expansion of iNKT cells for use as effector cells, which can lyse CD1d-positive target tumour cells in the presence of α -GalCer.
- 3. Administration of α -GalCer in conjunction with tumour antigen, as an adjuvant. Methods of achieving this include α -GalCer pulsing of tumour-loaded DCs, α -GalCer

Chapter 1. Introduction

pulsing of whole tumour cell vaccines, or co-administration of α -GalCer with a protein or peptide vaccine.

In the following sections, I shall review preclinical and clinical studies of iNKT celldirected tumour immunotherapy approaches for the treatment of malignancy.

1.3.3.1 Activation of iNKT cells

Systemic α -GalCer administration An improvement in the survival of mice inoculated with subcutaneous B16 melanoma tumours following intravenous α -GalCer was one of earliest observations of the biological activity of α -GalCer, ⁴⁰³ and preceded the discovery of iNKT cells. Numerous studies have since confirmed the protective effect of intravenous or intraperitoneal α -GalCer.

Intravenous α -GalCer has been administered to patients in clinical trials. In a phase 1 dose-finding study, α -GalCer was delivered intravenously once weekly for three weeks at each of seven escalating dose levels to patients with a variety of solid cancers. Doses of 50 – 4800 μ g/m2 were administered, and were generally well tolerated. Intravenous α -GalCer led to a rapid and profound reduction in circulating iNKT cell numbers within 24 hours, which persisted for at least a week. Serum cytokine responses were restricted to patients with the highest numbers of circulating iNKT cells pre-treatment. No clinical responses were observed.³⁰³

A phase I/II trial of intravenous α -GalCer for chronic hepatitis B confirmed its safety and capacity to produce a rapid decrease in circulating iNKT cells in humans. Significant reductions in viral yield were not observed.⁴⁰⁴

Thus, despite initial promise in preclinical models, intravenous administration of α -GalCer has not yet proved effective. Several factors may account for this failure. The systemic administration of α -GalCer to mice results in an initial expansion of iNKT cells followed by anergy. These anergic iNKT cells are refractory to subsequent α -GalCer administration, failing to produce IFN- γ . Moreover, in mice which have received a single dose of α -GalCer one month previously, the administration of α -GalCer alongside tumour no longer protects against melanoma metastasis.⁴⁰⁵ Repeated systemic α -GalCer administration leads to a Th2 bais in whole restimulated splenocytes.⁴⁰⁶ Finally, circulating iNKT cell numbers are typically much lower in humans than in mice.

The impressive effects of systemic α -GalCer in early preclinical tumour models may be reconciled with the disappointing results in clinical trials: A single dose of systemic α -GalCer administered to a mouse with high numbers of circulating iNKT cells leads to a dramatic but short-lived increase in pro-inflammatory cytokines, including IFN- γ . This transiently activates effector cells, such including NK cells. Provided α -GalCer is given at, or near, the time of tumour inoculation, transactivated NK cells may eradicate the tumour. After initial activation, however, iNKT cells become refractory to further stimulation, and lose their cytokine-producing and cytotoxic functions: α -GalCer is no longer protective against tumour challenges.

α-GalCer-pulsed unloaded dendritic cells Unlike intravenous α-GalCer, which results in a rapid but brief rise in serum cytokines, α-GalCer loaded onto DCs results in a delayed and more sustained rise in serum IL-12, IL-4 and IFN-γ. Moreover, compared to free intravenous α-GalCer, α-GalCer-loaded DCs provide superior protection against lung metastases of B16 melanoma in mice.^{407,408}

A number of phase I clinical studies of α -GalCer pulsed DCs for the treatment of cancer have been undertaken, and are summarised in table 1.17. It is notable that none of these studies used DCs that had also been pulsed with a tumour-specific antigen—the DCs were employed as a means of activating iNKT cells rather than to present tumour antigens. It is also important to note that only two of the studies used-mature moDC; the remainder used either immature moDC or GM-CSF treated PBMCs as APCs. Apart from the study by Kunii et al,⁴⁰⁹ in which iNKT cells and APCs were co-administered, the most significant immunologic and clinical responses appeared to occur following treatment with 'professional' DCs (i.e. moDC).

1.3.3.2 Use of iNKT cells as effectors

As the frequency of circulating iNKT cells is reduced in many patients with cancer, some authors have proposed the *ex vivo* expansion of iNKT cells for use as effectors or as adjuvants in subsequent cellular immunotherapy. Preclinical studies lend support to this approach: The adoptive transfer of iNKT cells to iNKT-deficient mice provides relative protection against methylcholanthrene-induced sarcoma.³⁰¹

A mouse model demonstrates the capacity of *ex vivo* expanded iNKT cells to eradicate a CD1d-transfected B cell malignancy *in vivo*.³¹⁰ The effect was CD1d-specific, as tumour lines transfected with CD1c were not killed. This is consistent with a similar report in a T cell malignancy,³⁰⁹ and suggests that target cell CD1d expression is a prerequisite for iNKT cell cytotoxicity.

iNKT cells could prove useful effector cells against leukaemias: Human myelomonocytic leukemia and T cell acute lymphoblastic leukaemia cells express CD1d and are sensitive to killing by iNKT cells.^{229,416} Importantly CD1d is expressed on CLL cells, and α -GalCer pulsed human CLL cells can be lysed by iNKT cells *in vitro*.⁴¹⁷

The feasibility and safety of adoptive iNKT cell transfer was demonstrated in phase I clinical study of *in vitro* expanded iNKT cells in six patients with advanced and recurrent non-small cell lung cancer.⁴¹⁸ However, despite transient increases in circulating iNKT cells, no clinical responses were seen in this setting.

Transfer of iNKT cells has also been used in a hybrid approach: In a phase I clinical study in head and neck cancer, α -GalCer loaded APCs were administered submucosally, followed by intra-arterial injection of IL-2 and α -GalCer-treated PBMCs, which were rich in iNKT cells. This resulted in a substantial increase in circulating iNKT cell numbers in seven out of eight patients, and tumour regression in three.⁴⁰⁹

APC	Route	Patients (n)	Tumours	Immunological responses	Clinical responses	References
GM-CSF/IL-2 PBMCs	Intravenous	11	Non-small cell lung cancer	1/11 patients showed marked increase in iNKT cells and IFN-γ production	None	410
GM-CSF/IL-2 PBMCs	Intravenous	17	Non-small cell lung cancer	Transient increase in iNKT nos in 6/17. Increased α-GalCer induced IFN-γ ELISPOTs in 10/17	None	411
Immature moDCs	Intravenous	12	Various	Decline, then increase in iNKT cells in 12/12. Increased IFN-y, in 12/12. NK activation.	'Biochemical response' in 5/12	412
Mature moDC*	Intravenous	ы	Various (including 3 myeloma)	Increased iNKT cells in 5/5. Increased IL12p40, IP-10 + MIP-1β. Increased CMV-specific CD8+ T cells	Two partial responses	413
GM-CSF/IL-2 PBMCs	Nasal submucosa	9	Head and neck cancer	Increased iNKT cells in $4/9$, and IFN- γ , ELISPots in $8/9$	One partial response	414
GM-CSF/IL-2 PBMCs plus 'iNKT' [†]	Nasal submucosa (iNKTs intra- arterial)	Ø	Head and neck cancer	Increased iNKT cells in 7/8. Increased IFN-γ ELISPots in 7/8	Three partial responses	409
Mature moDC [*] plus lenalidomide	Intravenous	6	Myeloma	Reduced circulating iNKT cells. Increased NK cells at day 7 [‡] Increase in CD16+ monocytes. [‡]	Paraprotein fell in 3/4 assessable [‡]	415
* Maturation coc [†] IL-2 and α-Gal([‡] May represent	ktail: IL-1β, IL-6, Cer-activated PBM a response to cone	TNF-α, PGE ₂ [Cs romitant lena	lidomide			

Table 1.17 Clinical trials of α -GalCer-pulsed DCs for cancer immunotherapy

1.3.3.3 iNKT cells as adjuvants

α-GalCer-pulsed loaded dendritic cells The clinical trial results presented in table 1.17 imply that simply activating patient iNKT cells with α-GalCer pulsed DCs is not sufficient to eradicate tumour. As iNKT cells are able to induce DC maturation, a rational step is to recruit iNKT cells to DCs that are presenting tumour peptides. That is, to present α-GalCer and tumour peptide on the same DC.

In mice, optimal enhancement of peptide-specific CD4+ and CD8+ responses by α -GalCer requires presentation of antigen and α -GalCer on the same DC.^{258,419} This co-operation extends to DCs loaded with whole tumour: addition of α -GalCer to tumour-loaded as opposed to unloaded DCs results in enhanced tumour protection.⁴²⁰

Studies performed on human cells *in vitro* also support this concept: moDCs transfected to over-express IL-12 and pulsed with α -GalCer and a TAA peptide led to better expansion of TAA-specific CD8+ T cells than did non- α -GalCer pulsed ones.²⁷⁶

Interestingly, α -GalCer can even enhance the efficacy of non-professional APCs in some settings. B cells pulsed with α -GalCer and peptide upregulate CD86 after administration to mice, and produce enhanced CD8+ T cell responses to the peptide compared to B cells without α -GalCer, including tumour protection. The responses observed in response to α -GalCer-pulsed B cells were nearly as good as those seen with α -GalCer-pulsed DCs.⁴²¹ This is particularly pertinent in the case of CLL, a B cell malignancy, as it implies that α -GalCer-pulsed tumour cells may be an effective immunogen.

 α -GalCer-pulsed tumour cells Generation of clinical-grade moDCs is a costly exercise, typically requiring several cell purification steps and a period of *in vitro* culture.³⁸⁰ An alternative strategy is to use whole tumour cells as a vaccine, using these to carry both α -GalCer and tumour antigen to host resident DCs.

Preclinical studies demonstrate that α -GalCer-pulsed irradiated tumour cells delivered as a whole tumour vaccine can produce durable antitumour immunity. In this model, resident APCs take up the α -GalCer-loaded tumour cells and present TAAs alongside α -GalCer. Presentation of α -GalCer to iNKT cells results in DC maturation and the local production of cytokines, which together lead to enhanced CD4+ and CD8+ T cell-dependent antitumour immunity.⁴²² This effect was enhanced when the target tumour cells (B16 melanoma cells) were transfected to express high levels of CD1d. Importantly, vaccination with CD1d-transfected B16 cells protected against subsequent challenge with untransfected B16 cells. The efficacy of this vaccination approach has also been demonstrated in the therapeutic setting: α -GalCer-pulsed B16 melanoma cells administered three days after a challenge with unmodified B16 melanoma cells protected against development of metastases.⁴²³

Preclinical studies have explored the use of α -GalCer pulsed tumour cells as vaccines for haematologic malignancies, many of which express CD1d.

• The non-immunogenic murine B cell lymphoma cell line A20 expresses CD1d at a low level. Lymphoma cells pulsed with α -GalCer led to iNKT cell activation and expansion, and proved a potent prophylactic and therapeutic tumour vaccine *in*

vitro. The tumour protection elicited by $\alpha\mbox{-}Gal\mbox{Cer-pulsed}$ lymphoma cells was CD4+ T cell-dependent. 424

- Plasmacytoma cells were irradiated and pulsed with α -GalCer before intravenous administration. The dying plasmacytoma cells were taken up by splenic DCs, leading to maturation of the DCs, and generation of protective immunity against subsequent tumour challenge. This protection was tumour-specific, and required both CD4+ and CD8+ T cells.⁴²⁵
- Pulsing of the WEHI-3B myelomonocytic leukaemia cell line with α -GalCer before intravenous inoculation protected against development of tumour upon rechallenge. In contrast, a T cell leukaemia cell line (EL-4) had to be transfected with CD1d to observe protection by α -GalCer pulsing.⁴²²

Administration of α -GalCer-pulsed tumour cells thus has the potential to offer a simple and effective immunotherapeutic strategy, at least for tumour cells which express CD1d.

There may be a barriers to the effective clinical administration of autologous α -GalCerpulsed tumour cells, including lack of availability of tumour tissue, concerns about the induction of metastasis, and for some malignancies, lack of CD1d expression on the tumour. One novel solution has been to transfect allogeneic fibroblasts with mRNA encoding TAAs. These fibroblasts, which express CD1d, are then pulsed with α -GalCer and administered as a tumour vaccine.⁴²⁶

Co-administration of antigen with α -GalCer The role of α -GalCer as an adjuvant is not restricted to whole tumour cell vaccines: Responses to peptide, protein and DNA vaccination can also be enhanced with α -GalCer.

As has already been discussed, the co-administration of the ovalbumin (OVA) with α -GalCer results in enhanced OVA-specific T cell responses, an effect mediated by α -GalCer matured DCs. As with intravenous α -GalCer administration, the α -GalCer needs to be delivered at the same time as OVA to yield an optimal effect, at least in the intravenous setting.²⁵⁸

In the infectious disease field, preclinical models have demonstrated the efficacy of α -GalCer as an adjuvant when co-administered with vaccines of diverse types. Responses were augmented when α -GalCer was co-administered with vaccines comprised of irradiated malaria sporozoites,⁴²⁷ herpes simplex glycoprotein,⁴²⁸ inactivated influenza A virus,⁴²⁹ and HIV DNA.⁴³⁰

1.3.3.4 Other approaches

Other approaches that utilise iNKT cells for cancer immunotherapy have been proposed or investigated.

One novel approach has been to generate a fusion protein comprising an α -GalCer-loaded soluble CD1d molecule linked to a scFv with anti-tumour specificity. This simultaneously

targets iNKT cells to the tumour and activates them. In a preclinical study, this fusion protein inhibited the growth and metastasis of a B16 melanoma. 431

A role for iNKT cells after allo-SCT has been proposed. Studies in mice suggest that iNKT cells can co-operate with Tregs to suppress GVHD, apparently without abrogating the graft versus tumour effect.⁴³² Delayed iNKT cell reconstitution after haploidentical allo-SCT for children with haematologic malignancies has been associated with increased relapse risk, leading to the suggestion that donor iNKT cells could be expanded *ex vivo*, and returned to the recipient.⁴³³

Finally, responses to iNKT cell-based immunotherapies can be enhanced by combining them with other adjuvants or with strategies to suppress negative regulation of immune responses. As well as causing upregulation of costimulatory markers, Toll-like receptor agonists enhance glycolipid presentation by human APCs and result in increased iNKT cell IFN- γ production.⁴³⁴ Abrogation of the Programmed Cell Death-1 (PD-1) pathway results in enhanced tumour responses.⁴³⁵ Finally, depetion of T_{reg} populations enhances antigen-specific CD8+ T cell responses to tumour-loaded, α -GalCer pulsed DCs.⁴²⁰

1.4 Conclusions

CLL is a common malignancy, caused by the proliferation of clonal mature B lymphocytes in blood, bone marrow and the lymphoreticular system. The malignant cells are antigenexperienced, chronically-stimulated memory B cells, typically carrying an autoreactive IgM B cell receptor (BCR). Constant stimulation of the BCR by autoantigens or ubiquitous microbial antigens gives rise to the characteristic CD5+ CLL phenotype. Although CLL cells undergo spontaneous apoptosis *in vitro*, interactions between the malignant cells and T, myeloid and stromal cells, sustain the malignant clone *in vivo*. Over time, additional genomic changes occur within the clone, resulting in a more clinically-aggressive phenotype. Effective treatments exist, the mainstay being chemoimmunotherapy with a purine analogue, alkylating agent and the anti-CD20 antibody rituximab. However, the only curative treatment, allo-SCT is only available to a minority of patients because of advanced age or co-morbidities.

CLL appears to be a promising candidate for adoptive immunotherapy, because prolonged remissions after allo-SCT and after T cell redirection therapies indicate that CLL is susceptible to T cell killing, because a proportion of patients possess circulating T cells with specificity against their leukaemic clone, the number of these T cells increasing after therapeutic tumour vaccination, and because spontaneous remissions, although rare, are well documented.

Invariant natural killer T cells, a rare subset of T cells, possess a semi-invariant T cell receptor which confers reactivity to glycolipid antigens such as α -GalCer presented by the MHC Class I-like molecule, CD1d. Functionally, iNKT cells combine properties of T cells, such as the capacity to induce DC maturation and to provide B cell help, with characteristics reminiscent of NK cells such as cytotoxicity, rapid cytokine release and even NK receptor-mediated alloreactivity. Patients with some malignancies have reduced iNKT cell numbers or exhibit impaired iNKT cell function, suggesting that iNKT cells may

play a role in tumour surveillance or control. Consistent with this, the activation of iNKT cells promotes anti-tumour immunity in a number of mouse models.

Numerous adoptive immunotherapy strategies have been proposed or tested in CLL, including peptide vaccination, whole tumour vaccination, DC vaccination, adoptive T cell transfer and T cell redirection.

Since CD1d is minimally polymorphic, and because iNKT cells are detectable in most individuals, the CD1d/iNKT cell axis can be readily exploited for immunotherapies without paying heed to tissue type or NK receptor profile. Potential roles for iNKT cells in tumour immunotherapy include:

- Non-specific activation of immune system cells through α-GalCer administration and consequent production of proinflammatory cytokines by patient iNKT cells
- An adjuvant effect elicited by co-administration of α -GalCer with free tumour antigen, with a CD1d-expressing whole tumour cell, or with a DC which is bearing tumour antigen
- Direct cytotoxicity against CD1d-positive target tumour cells in the presence of $\alpha\text{-}$ GalCer

As CLL cells express CD1d, and appear susceptible to immunotherapeutic approaches, the investigation of iNKT cell numbers and function in patients with CLL is warranted, with a view to exploiting the iNKT cell/CD1d axis in future immunotherapies for CLL.

1.5 Aims

Chronic lymphocytic leukaemia is an attractive candidate for adoptive immunotherapy, and iNKT cells have the potential to act as useful adjuvants in tumour vaccination or as effector cells targeting CD1d expressing tumour cells such as CLL. However, numerous defects of the immune system have been described in patients with CLL, including functional and immunophenotypic defects of DCs, abnormal T cell numbers and function, and a humoral immunodeficiency. Investigation of the iNKT/CD1d axis in CLL is therefore warranted to inform future trials of iNKT cell-directed immunotherapy for this and related diseases.

The aims of the experimental programme outlined in this thesis are as follows:

- 1. To determine the frequency and immunophenotype of iNKT cells from patients with CLL,
- 2. To investigate CD1d expression patterns on APCs in patients with CLL,
- 3. To analyse the function of iNKT cells from patients with CLL, and
- 4. To explore potential roles of iNKT cells in adoptive immunotherapy of CLL

Chapter 2

Methods

2.1 General Materials and Methods

2.1.1 Culture media and buffers

Dulbecco's phosphate-buffered saline (DPBS) Calcium- and magnesium-free DPBS was purchased from Invitrogen (Auckland, New Zealand), comprising 200 mg/L KCl, 200 mg/L KH₂PO₄, 8 g/L NaCl and 2160 mg/L Na₂HPO₄ – $7H_2O$.

Flow cytometry buffer 10 mM ethylenediaminetetraacetic acid (EDTA) (Sigma-Aldrich, St Louis, MO), 0.01% sodium azide (Sigma-Aldrich) and 2% foetal calf serum (FCS; SAFC Biosciences, Lenexa, KS) were added to 1 L DPBS.

Immunomagnetic bead buffer 0.1% bovine serum albumin (ICP Bio, Auckland, New Zealand) and 2 mM EDTA (Sigma-Aldrich) were added to DPBS, and the pH adjusted to 7.4. The buffer was filtered through 0.2 μ m AcrodiscTM syringe filters (Pall Life Sciences, Hamilton, New Zealand) before use.

Iscove's Modified Dulbecco's Medium (IMDM) IMDM supplemented with GlutaMAX^m, 25 mM 4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid (HEPES) buffer and 3.024 g/L NaHCO₃ was purchased from Invitrogen.

Complete IMDM (cIMDM) IMDM was supplemented with 100 U/mL penicillin, 100 mg/mL streptomycin, 55 μ M 2-mercaptoethanol (2-ME) and 5% human group AB serum (Invitrogen).

Roswell Park Memorial Institute (RPMI-1640) Phenol red- and HEPES-free RPMI containing 2 g/L NaHCO₃ was purchased from Sigma-Aldrich.

2.1.2 C1R cell lines

The C1R cell line is a human B lymphoblastoid cell line which lacks expression of HLA-A and HLA-B antigens.⁴³⁶ The untransfected C1R cell line and a C1R cell line stably transfected to express CD1d were a gift from the laboratory of Professor Vincenzo Cerundolo (Oxford, UK). Both cell lines were maintained in cIMDM. CD1d expression status was verified by flow cytometry before cell use.

2.1.3 Cell counting

Cell numbers were determined by staining an aliquot with 0.4% trypan blue (Invitrogen) and counting unstained (live) cells using a haemocytometer (Weber Scientific, Middlesex, England).

2.1.4 Culture conditions

All cells were cultured at 37 °C with 5% CO_2 and 95% humidity in a HERAcell[™] tissue culture incubator (Thermo Scientific, North Shore, New Zealand).

2.2 Patients and ethical considerations

Study participants were recruited by invitation letter and by poster advertisement (appendix A). All participants were provided with written information regarding the study (appendix B). Exclusion criteria included any other active malignancy or previous chemotherapy for a disease other than CLL, as determined by completion of a participant questionnaire (appendix C), and by review of hospital records where relevant. All donors provided written informed consent (appendix D).

All patients met morphologic and flow cytometric criteria for a diagnosis of CLL⁴³⁷. The Rai¹²⁶ and Binet¹²⁷ clinical stages of patients with CLL were determined by physical examination and by analysis of peripheral blood counts performed on the day of PBMC cryopreservation.

Ethical approval for this study was obtained from the Central Regional Ethics Committee of New Zealand (application reference CEN/08/01/002). This study was conducted in accordance with the University of Otago Code of Ethical Conduct.

2.3 Venepuncture and PBMC cryopreservation

Peripheral blood was drawn into heparinized tubes before dilution 1:1 in DPBS. Diluted blood was layered over a sodium diatrizoate and polysaccharide solution (Lymphoprep[™]; Axis-Shield, Oslo, Norway) before centrifugation at 800 g for 25 minutes at room temperature. The PBMC fraction was aspirated, washed and resuspended in 90% FCS with

10% dimethyl sulfoxide (DMSO) (Sigma-Aldrich) and frozen in a controlled-rate freezing container. Samples were stored at -180 $^\circ C$ in gaseous phase liquid nitrogen until use.

Additional blood samples were drawn into EDTA and serum-separating vacuum tubes (BD Vacutainer[™], BD, North Ryde, Australia) for determination of absolute differential blood count using a Sysmex XE 2100 analyser (Roche Diagnostics, Auckland, New Zealand), and of serum immunoglobulin G (IgG) levels using an immunoturbidimetric assay (Tina-Quant[™]IgG Gen2; Roche Diagnostics), respectively.

2.4 B cell depletion and enrichment

B cells were depleted and positively selected using anti-CD19-coated immunomagnetic beads (Dynabeads[™] CD19 pan-B; Invitrogen Dynal AS, Oslo, Norway). Immunomagnetic beads were washed as per manufacturer's instructions, and PBMCs were counted and resuspended at 1×10^6 /mL to 2.5×10^7 /mL in immunomagnetic bead buffer in 5 mL polystyrene tubes. Between 0.5 and 4 beads per target cell were added. The cell suspension was incubated for 30 minutes at 4 °C with tilting and rotation. Sample tubes were placed in a DynaMag[™] magnet (Invitrogen) for two minutes, and the supernatant containing unbound cells aspirated (B cell-depleted PBMCs). Between one and four serial rounds of B cell depletion were performed to achieve a residual level of CD5+CD19+ CLL cells of < 5% in B cell-depleted PBMCs, as confirmed by flow cytometric analysis.

For some experiments, positively-selected CD19+ cells were retained and immunomagnetic beads were detached using anti-antigen-binding fragment (Fab) antibodies specific for the bead-bound anti-CD19 antibody (DETACHaBEAD^M CD19; Invitrogen Dynal AS), as per manufacturer's instructions. The purity of CD19+ CD5+ CLL cells within the positively-selected B cell population was \geq 95%.

For some experiments, CLL cells were sorted by fluorescence-activated cell sorting (FACS): PBMCs were stained for surface markers as outlined in section 2.6, then resuspended at 1 x10⁶/mL in sterile azide-free flow cytometry buffer. CLL cells were sorted on a BD FACSVantage DiVa flow cytometer (Becton Dickinson, San Jose, CA) based on co-expression of CD19 and CD5. Purity of CLL cells was verified after FACS sorting, and was \geq 98.5%.

2.5 Loading and tetramerization of CD1d monomer

Biotinylated human CD1d monomer at 2 mg/mL was obtained from the National Institutes of Health (NIH) Tetramer Core Facility. The concentration was adjusted to 1 mg/mL by addition of an equal volume of phosphate-buffered saline (PBS)/PI, comprising DPBS with 2 mM EDTA and a protease inhibitor cocktail to give a final concentration of 1 μ M pepstatin A, 2 μ M leupeptin, 5 μ M bestatin, 80 nM aprotinin, 100 μ M 4-(2-aminoethyl) benzenesulfonyl fluoride hydrochloride and 1.5 μ M E-64 (Halt Protease Inhibitor Cocktail, Thermo Scientific). 20 μ L α -GalCer 500 μ g/mL stock (a gift from Industrial Research Laboratories, Lower Hutt, New Zealand) was sonicated at 37 °C for fifteen minutes (Ultrasonic cleaner, Unisonics, Sydney, Australia), then added carefully to 100 μ L of 1 mg/mL CD1d monomer solution. 100 μ L of DPBS containing 0.5% Tween-20 (Sigma) was added, and mixed by pipetting. The volume was made up to 1 mL by addition of PBS/PI. The CD1d monomer and α -GalCer solution was incubated in a 37 °C water bath for three hours. The loaded CD1d monomer was concentrated three times in a Vivaspin 500 10 kDa centrifugal ultrafiltration concentrator (Sigma-Aldrich), then resuspended in 50 μ L PBS/PI. Aliquots of unloaded CD1d tetramer were simultaneously prepared using the same method, but without addition of α -GalCer, for subsequent use as negative staining controls.

Loaded and unloaded CD1d monomer was tetramerized and labelled as follows: 50μ L aliquots of monomer were protected from light and stored on ice during the addition of 300μ L phycoerythrin (PE)-labelled streptavidin (BD Biosciences), added in 30μ L aliquots every fifteen minutes, with thorough mixing. Loaded and unloaded PE-labelled CD1d tetramer aliquots were stored at 4 °C until use.

2.6 Immunofluorescent staining of cell surface markers for flow cytometry

Fluorescent-labelled antibodies were obtained from BD Biosciences (Franklin Lakes, NJ), Beckman Coulter (Brea, CA), Biolegend (San Diego, CA), and Invitrogen. The antibody clones, isotypes and fluorophores used are listed in table 2.1.

B cell-depleted or whole PBMCs were washed twice and resuspended at 1 to 10×10^6 /mL in PBS. 200 μ L aliquots were dispensed into round-bottomed 96-well plates and centrifuged. Crystallizable fragment (Fc) receptors were blocked by incubation of cells for 15 minutes with 2 mg/mL polyclonal human IgG (Intragam PTM, CSL Limited, Broad-meadows, Australia). For surface staining, cells were incubated for 20 minutes at 4 °C with the fluorescent-labelled antibodies or tetramer indicated at optimal concentrations as determined by prior experiments, before two washes in flow cytometry buffer. Cells were stained with an amine-reactive viability dye (LIVE/DEAD FixableTM Blue; Molecular Probes, Eugene, OR) according to manufacturer's instructions then resuspended in FACS buffer, or were resuspended in flow buffer containing 0.5 μ g/mL 4',6-diamidino-2-phenylindole (DAPI; Molecular Probes), to allow exclusion of dead cells.

Table 2.1 Fluorescent-labelled antibodies List of antibodies and antibody cocktails used for flow cytometry. APC = allophycocyanin; FITC = fluorescein isothiocyanate; Lin 1 = lineage cocktail 1 (anti-CD3, -CD14, -CD16, -CD19, -CD20 and -CD56); PE = phycoerythrin; PerCP = peridininchlorophyll protein complex. All isotypes are mouse anti-human, unless otherwise stated

Antigen(s)	Clone(s)	Isotype(s)	Fluorophore(s)	Supplier
CD1c	L161	IgG ₁ к	PE	Biolegend
CD1d	CD1d42	IgG_1 к	PE	BD Biosciences
CD3	UCHT1	$IgG_1 \kappa$	APC	BD Biosciences
CD3	SK7	Ig G_1 к	APC-H7, FITC, PE	BD Biosciences
CD3, CD16 + CD56, CD45, CD19	SK7, B73.1 + NCAM16.2, 2D1, SJ25C1	IgG ₁ κ, IgG _{2b} κ	FITC, PE, PerCP, APC	BD Biosciences
CD4	OKT4	IgG _{2b} к	Pacific Blue	Biolegend
CD4	SK3	Ig G_1 к	APC	Biolegend
CD5	UCHT2	Ig G_1 к	APC	Biolegend
CD5	L17F12	IgG _{2a} к	PE	BD Biosciences
CD8	RPA-T8	IgG_1 к	Alexa 700	BD Biosciences
CD8	SK1	Ig G_1 к	APC	BD Biosciences
CD11c	Bu15	Ig G_1 к	APC	Biolegend
CD11c	S-HCL-3	IgG _{2b} к	PE	BD Biosciences
CD14	ΜφΡ9	IgG _{2b} к	APC, PE	BD Biosciences
CD16	3G8	Ig G_1 к	FITC	Biolegend
CD19	SJ25C1	Ig ${ m G}_1$ к	APC, PE	BD Biosciences
CD19	4G7	Ig G_1 к	FITC	BD Biosciences
CD23	9P25	Ig G_1 к	FITC	Beckman Coulter
CD25	2A3	Ig G_1 к	FITC	BD Biosciences
CD40	5C3	IgG_1 к	FITC	BD Biosciences
CD54	HA58	Ig G_1 к	PE	BD Biosciences
CD80	L307.4	Ig G_1 к	PE	BD Biosciences
CD86	IT2.2	IgG _{2b} к	PE	Biolegend

Continued on page 68

Antigen(s)	Clone(s)	Isotype(s)	Fluorophore(s)	Supplier
CD154	TRAP1	IgG ₁ к	APC	BD Biosciences
CD161	DX12	Ig G_1 к	FITC	BD Biosciences
FoxP3	259D/C7	$IgG_1 \kappa$	PE	BD Biosciences
Granzyme B	GB11	Ig G_1 к	FITC	BD Biosciences
HLA Class I	B9.12.1	Ig G_1 к	FITC	Beckman Coulter
HLA-DR	L243	IgG _{2a} к	FITC	BD Biosciences
HLA-DR	L243	IgG _{2a} к	PerCP	Biolegend
IFN-γ	B27	Ig G_1 к	Alexa 488	Invitrogen
IL-4	MP4-25D2	Rat Ig G_1 κ	APC	Biolegend
iNKT cell	6B11	$IgG_1 \kappa$	FITC, PE	BD Biosciences
Lin 1*	SK7, ΜφΡ9, 3G8, SJ25C1, L27, NCAM16.2 [*]	IgG ₁ к, IgG _{2b} к	FITC	BD Biosciences
ZAP-70	1E7.2	IgG_1 к	PE	Biolegend

Table 2.1, continued from page 67

^{*} Lineage cocktail 1 (Lin1) contains antibody clones directed against CD3, CD14, CD16, CD19, CD20 and CD56, respectively.

2.7 Immunofluorescent staining of intracellular cytokines, FoxP3, granzyme B and ZAP-70 for flow cytometry

PBMCs were resuspended to 1×10^6 /mL in cIMDM. For intracellular cytokine and granzyme B staining, cells were stimulated by addition of ionomycin 1 µg/mL (Merck, Manukau City, New Zealand), phorbol 12-myristate 13-acetate 10 ng/mL (PMA; Sigma-Aldrich) and protein transport was inhibited by addition of brefeldin A 2 µg/mL (eBioscience, San Diego, CA). Cells were then incubated at 37 °C for four hours. No stimulation step was used for ZAP-70 staining.

Cells were washed, blocked with human IgG, and stained for surface markers, as described in section 2.6. Cells were then fixed, permeabilized, stained for intracellular markers, and washed using a fixation/permeabilization kit (BD Cytofix/Cytoperm[™]; BD Biosciences) as per manufacturer's instructions, before flow cytometric analysis. For intracellular FoxP3 staining a specific FoxP3 staining buffer set (eBioscience) was used for the fixation and permeabilisation steps as per manufacturer's instructions.

Stained unactivated PBMCs, and isotype control-stained activated PBMCs were used as negative controls, and the latter used to set gates for positive intracellular cytokine and granzyme B staining. For ZAP-70 staining, CD5+ CD19- T cells were used as an internal positive control to set the gate for ZAP-70 positivity, as previously described.¹³³ Cases of CLL were considered ZAP-70 positive if > 20% of CD19+ CD5+ cells stained positively for ZAP-70.

2.8 Flow cytometric analysis

Flow cytometric data were acquired using an LSR II flow cytometer (Becton Dickinson)—the laser and filter configuration used is shown in table 2.2. Automated compensation was performed using the appropriate fluorophore-labelled antibodies bound to anti-mouse immunoglobulin (Ig) coated particles (BD Compbeads[™]; BD Biosciences). For non-antibody stains, automated compensation was performed using stained and unstained populations of positive control cells. Data were analysed using FlowJo[™] software (Tree Star Incorporated, Ashland, OR).

The principal immunofluorescence panels used for the flow cytometry experiments presented in this thesis are listed in table 2.3. Except where otherwise stated, cells were serially gated to exclude doublets (on forward scatter height versus forward scatter area then side scatter height versus side scatter area plots) and to exclude dead cells and debris (DAPI or LIVE/DEAD fixable blue negative), before application of an appropriate forward and side scatter gate.

Lymphocytes were identified using a low forward and side scatter gate (figure 3.5) except in functional assays or where activation markers were assessed, when the forward and

Table 2.2 Flow cytometer configuration Configuration of the LSR II flow cytometer used for the experiments in this thesis. APC = allophycocyanin; BP = band pass; CFSE = carboxyfluorescein succinimidyl ester; CTO = CellTracker^M Orange; DAPI = 4',6-diamidino-2-phenylindole; FITC = fluorescein isothiocyanate; LP = long pass; PE = phycoerythrin; PerCP = peridinin-chlorophyll protein complex.

Excitation wavelength (nm) (colour)	Filter wavelengths (nm)	Fluorophore(s) or parameter
355 (ultraviolet)	450/50 BP	DAPI, LIVE/DEAD fixable blue
405 (violet)	450/50 BP	Pacific blue
405 (violet)	560/40 BP	Pacific orange
488 (blue)	515/20 BP & 505 LP	FITC, Alexa 488, CFSE
488 (blue)	685/35 BP & 640 LP	PerCP
488 (blue)	488/10 BP	Side scatter
532 (green)	575/26 BP	PE, CTO
640 (red)	670/14 BP	APC
640 (red)	780/60 BP & 740 LP	APC-H7
640 (red)	710/50 BP & 685 LP	Alexa 700

side scatter gates were extended to include activated (blasting) cells. Within the lymphocyte gate, T cells were identified as CD19-CD3+, NK cells were identified as CD19-CD3and CD16/CD56+, NK-like T cells were identified as CD19-CD3+ and CD16/CD56+, B cells were identified as CD19+, CLL cells were identified as CD19+CD5+, and 'normal' B cells were identified as CD19+CD5-.

To identify dendritic cells an intermediate forward and side scatter gate was applied (figure 4.2a). Within this gate, myeloid DCs were identified as Lin1-, HLA-DR+ and CD11c high. Plasmacytoid DCs were identified as Lin1-, HLA-DR+ and CD123 high.

Monocytes were identified using a high forward and side scatter gate (figure 4.5a). Within this gate, CD14hi monocytes were identified as CD14 high and CD16-. CD16+ monocytes were identified as CD14 low and CD16+.

Except where otherwise indicated, relative numbers of PBMC subsets were expressed as a proportion of live PBMCs, determined by flow cytometry of cryopreserved samples.

For some experiments, absolute numbers of PBMC subsets were estimated from Coulter full blood counts of whole blood performed on the same day as PBMC cryopreservation, using the following formula: absolute number of circulating PBMCs = absolute lymphocyte count + absolute monocyte count.

For quantitative flow cytometric analyses, positively stained samples were analysed alongside identically-treated fluorescence minus one (FMO) controls, or controls stained with an isotype-matched control antibody. Net median fluorescence intensity was calculated **Table 2.3 Immunofluorescent antibody panels** Principal immunofluorescence panels used for the multicolour flow cytometry experiments presented in this thesis.

Panel purpose	Fluorophores and antigens
CLL and B cell identification and surface phenotype	DAPI or LIVE/DEAD fixable blue, FITC-CD19, APC-CD5, APC-H7-CD3
B cell activation status	DAPI or LIVE/DEAD fixable blue, FITC-CD40 or -CD25 or -HLA-DR, PE-CD80 or -CD86 or -CD54, APC-CD19, APC-H7-CD3
CLL cell ZAP-70 expression	FITC-CD19, PE-ZAP-70, APC-CD5
iNKT and T cell enumeration and surface phenotype	DAPI or LIVE/DEAD fixable blue, Pacific blue-CD4, FITC-CD25 or -CD161, PE-anti-iNKT cell (clone 6B11) or PE-CD1d tetramer, APC-CD19, Alexa 700-CD8, APC-H7-CD3
iNKT and T cell intracellular cytokines	LIVE/DEAD fixable blue, Pacific blue-CD4, Alexa 488-IFN-γ, PE-anti-iNKT cell (clone 6B11) or PE-CD1d tetramer, APC-IL-4, Alexa 700-CD8, APC-H7-CD3
iNKT and T cell proliferation	DAPI, Pacific blue-CD4, CFSE, PE-anti-iNKT cell (clone 6B11) or PE-CD1d tetramer, APC-CD19, Alexa 700-CD8, APC-H7-CD3
Treg identification	Pacific blue-CD4, FITC-CD25, PE-FoxP3, APC-H7-CD3
NK and NK-like T cell identification	DAPI or LIVE/DEAD fixable blue, FITC-CD161, PE-CD3, APC-CD19
Myeloid dendritic cell enumeration and phenotype	DAPI or LIVE/DEAD fixable blue, FITC-Lin1, PE-CD1d or CD1c or CD86 or CD80, PerCP-HLA-DR, APC-CD11c
Monocyte enumeration and phenotype	DAPI or LIVE/DEAD fixable blue, FITC-CD16, PE-CD1d or CD86 or CD80, APC-CD14

by subtracting the median fluorescence intensity of the FMO or isotype-matched control stained sample from that of the positively-stained sample.

2.9 IFN-γ ELISpot assay

IFN-γ ELISpot was performed using a human IFN-γ ELISpot kit (BD Biosciences) according to manufacturer's instructions. 5 x10⁵ B cell-depleted PBMCs were washed, resuspended in cIMDM and dispensed into pre-coated ELISpot plates before addition of α-GalCer 200 ng/mL, PBS with 0.5% Tween-20 vehicle as negative control, or phytohaemagglutinin-P 1 μ g/mL (PHA-P; Sigma-Aldrich) as positive control. For some experiments, an unlabelled purified anti-CD1d monoclonal antibody (clone CD1d42; BD Biosciences) was added at 0.5 μ g/mL to confirm the CD1d dependence of α-GalCer-induced IFN-γ ELISpots. Plates were cultured for 18 hours at 37 °C.

Following culture, the cell suspension was aspirated, and the wells washed twice with deionised water, then thrice with 200 μ L/well of the manufacturer's supplied wash buffer. 100 μ L/well detection antibody was added, and the plates incubated for two hours at 20 °C. Detection antibody was discarded, and the wells washed thrice with 200 μ L/well wash buffer before addition of 100 μ L/well streptavidin-horseradishperoxidase (HRP) solution and a further one hour incubation at 20 °C. Streptavidin-HRP was discarded, and the wells washed four times with 200 μ L/well wash buffer, then twice with 200 μ L/well PBS. 100 μ L/well 3-amino-9-ethylcarbazole (AEC) substrate was added, and the plates monitored visually for spot development. After 20 to 40 minutes, the reaction was halted by washing wells twice with deionised water, and the plate dried overnight.

ELISpots were enumerated using an automated ELISpot reader (Autoimmun Diagnostika, Strassberg, Germany). Spot-forming units (SFUs) in response to vehicle were subtracted from SFUs in response to α -GalCer to calculate α -GalCer-induced ELISpots.

2.10 iNKT cell proliferation assays

 2×10^5 to 2×10^6 PBMCs were cultured at 37 °C in cIMDM with either 100 ng/mL α -GalCer or PBS/0.5% Tween-20 vehicle. Recombinant human IL-2 100 U/mL (Chiron Corporation, Emeryville, CA) was added at 24 hours. For some experiments, recombinant human IL-7, IL-15 or IL-21 (all from eBioscience, San Diego, CA) were used instead of IL-2.

Cells were harvested after seven days. The absolute number of live cells was enumerated by Trypan blue staining (as described in section 2.1.3), and the proportion of iNKT cells was determined by flow cytometry.

To assess the effect of CLL cells on iNKT cell proliferation, for some experiments positivelyselected autologous CLL cells were added back to the B-cell depleted PBMCs before the addition of α -GalCer and IL-2, as above.

2.11 Generation and maintenance of iNKT cell lines

Following initial proliferation of iNKT cells as above, iNKT cells were sorted by FACS (as described in section 2.4) based on co-expression of CD3 and the V α 24J α 18 TCR- α chain, detected with the 6B11 antibody clone.

Sorted iNKT cells (4 x10³ to 2 x10⁵) were dispensed into 24-well plates. 2 x 10⁵ to 1 x10⁶ CD1d-transfected C1R cells (C1R-CD1d) were cultured in the presence of 200 ng/mL α -GalCer for 24 hours, γ -irradiated to 50 Gy (Gammacell 3000 Elan irradiator, Best Theratronics, Ottowa, Canada), and washed three times, before addition to sorted iNKT cells for restimulation. 100 U/mL recombinant human IL-2 was added 24 hours after the C1R-CD1d cells.

Resultant polyclonal iNKT cell lines were split once the concentration reached $\geq 2 \times 10^6$ cells per well. Cell lines were restimulated with α -GalCer-pulsed irradiated C1R-CD1d cells every two to three weeks, and maintained in media containing 50 U/mL IL-2. For some cell lines, FACS sorting was repeated to maintain the purity of iNKT cells \geq 95%.

2.12 Cytokine bead array analysis

2 x10⁴ C1R-CD1d cells were pulsed with 200 ng/mL α -GalCer or vehicle for two hours, γ -irradiated to 50 Gy, then added to 4 x10⁴ cultured iNKT cells. After four hours, supernatant was harvested for cytokine analysis. In separate experiments, 2 x10⁶ whole or B cell-depleted PBMCs were cultured in the presence of α -GalCer or vehicle. IL-2 100 U/mL was added at 17 to 24 hours. Supernatant was taken at 6 to 72 hours. Supernatants were cryopreserved at -80 °C until analysis.

Supernatant cytokine concentrations were determined by cytokine bead array (MilliplexTM; Millipore Corporation, Billerica, MA), according to bead manufacturer's instructions. Briefly, 25 μ L of each culture supernatant and cytokine standard was dispensed into pre-wetted filter plates in duplicate, and diluted with assay buffer or culture medium, respectively. Capture antibody-conjugated microbeads were sonicated, vortexed and mixed, and added to each well. The plate was sealed and incubated with agitation for one hour at 20 °C. Fluid was removed by vacuum filtration, and the wells washed twice with wash buffer. Detection antibodies were added, and the plate incubated for 30 minutes at 20 °C with agitation. PE-labelled streptavidin was added to each well before a further 30 minute inubation at 20 °C with agitation. Fluid was removed by vacuum filtration, and each well washed twice, before the contents were resuspended in 100 μ L sheath fluid (Bio-Rad Laboratories, Hercules, CA) with agitation.

Plates were analysed on a Bio-Plex[™] analyser (Bio-Rad Laboratories). A cytokine standard curve was generated using a spline-fitting curve model, and was used to determine cytokine concentrations in the experimental samples. The lower limit of detection for each cytokine was 3.2 pg/mL, determined by the lowest dilution of the cytokine standard used.

2.13 Cytotoxicity assays

2.13.1 Lactate dehydrogenase release assay

2.5 $\times 10^4$ untransfected C1R or C1R-CD1d target cells were cultured with 200 ng/mL α -GalCer or PBS/Tween vehicle for one hour in phenol-red free RPMI-1640 with 5% human AB serum before addition of iNKT effector cells at indicated effector:target ratios, in triplicate. Effector only, target only and background (medium only) controls were included, and to determine maximal lactate dehydrogenase (LDH) release, additional aliquots of target cells were incubated with 1/10 LDH assay lysis solution (Sigma-Aldrich).

After incubation for four hours at 37 °C, supernatants were harvested and LDH content determined using an LDH assay kit (Sigma-Aldrich), as per manufacturer's instructions. Briefly, 50 μ L of each supernatant was dispensed into flat-bottomed 96-well plates, and 100 μ L of LDH assay mixture (comprising LDH substrate, dye and co-factor solutions) added. The plates were incubated for 30 minutes at 20 °C in the dark, after which 15 μ L of 1N HCl was added to each well to terminate the reaction.

LDH activity was measured by spectrophotometry using a Versamax[™] microplate reader (Molecular Devices, Surrey Hills, VIC, Australia). Background absorption at 690 nm was subtracted from absorption at 490 nm. Specific lysis was calculated according to the formula (experimental LDH release – spontaneous effector LDH release – spontaneous target LDH release) ÷ (maximal target LDH release – spontaneous target LDH release), as previously described.⁴³⁸

2.13.2 VITAL assay

The *in vivo/in vitro* technique for assessing lysis (VITAL) assay was performed on *in vitro* cultured PBMCs as previously described.⁴³⁹ Briefly, C1R-CD1d target cells were cultured for two hours with 100 ng/mL α -GalCer or with vehicle, then resuspended at 2 x10⁶/mL in PBS.

The α -GalCer-pulsed C1R-CD1d target cells were labelled with cell tracker orange (CTO; Molecular Probes) by addition of 10 μ M CTO followed by incubation at 37 °C for fifteen minutes. The cells were then centrifuged and resuspended in warmed cIMDM, incubated at 37 °C for a further twenty minutes, and washed twice in PBS.

Unpulsed C1R-CD1d target cells were labelled with carboxyfluorescein succinimidyl ester (CFSE) as follows: CFSE was added to the cell suspension to a final concentration of 0.2μ M. The cell suspension was mixed thoroughly, then incubated for eight minutes at 20 °C, mixing twice during the incubation. An equal volume of FCS was added to quench the reaction. Cells were washed twice, then resuspended in cIMDM.

Each population of labelled target cells was resuspended at 2×10^5 /mL in cIMDM. 25 μ L of this suspension was dispensed to wells of a 96-well round-bottomed plate, in duplicate (5 $\times 10^3$ of each target cell population/well). Additional wells containing each target population alone were used for compensation controls. Whole healthy donor PBMCs were added as effectors, at the effector:target ratios indicated.

Following a six hour incubation at 37 °C, cells were harvested, washed, resuspended in buffer containing propidium iodide (PI; BD Biosciences) for dead cell exclusion, and analysed by flow cytometry to determine the relative proportions of CTO-labelled α -GalCerpulsed and CFSE-labelled unpulsed target cells.

2.14 IFN- γ , α -GalCer and resiquimod effects on CLL cell surface markers

Leukaemic CLL cells were immunomagnetically selected, and immunomagnetic beads removed, as detailed in section 2.4. 1 x10⁶ CD19+ cells (\geq 95% CD19 and CD5 copositive) were cultured in cIMDM for 24 to 48 hours in the presence of vehicle alone or with 1000 U/mL recombinant human IFN- γ (Peprotech, Rocky Hill, NJ), 100 ng/mL α -GalCer, or resiquimod at the indicated concentrations (Invivogen, San Diego, CA). Cells were harvested and stained for MHC Class I, Class II, CD1d and CD86 expression in parallel with FMO or isotype controls, as detailed in section 2.6.

2.15 Vaccine recall responses

For the assessment of vaccine recall responses, the following vaccines were used:

- VAXIGRIP[™](Sanofi Pasteur, Lyon, France), a purified, inactivated adjuvant-free vaccine containing influenza virus strains A/Brisbane/59/2007IVR-148, A/Uruguay/-716/2007 NYMC X-175C and B/Florida/4/2006
- ADT Booster™(CSL Biotherapies, Parkville, Australia), containing diphtheria toxoid 4 IU/mL, tetanus toxoid 40 IU/mL, and aluminium hydroxide adjuvant.

2.15.1 Thymidine incorporation

Thawed PBMCs from vaccinated donors were cultured at 2×10^5 /well in 96-well plates in the presence of various concentrations of influenza or tetanus toxoid vaccines in cIMDM for five days, with the addition of 100 ng/mL α -GalCer or vehicle, in triplicate.

Twelve hours before the end of the culture period, ³H thymidine (Amersham Biosciences, Little Chalfont, United Kingdom) was added at 1 μ Ci [0.037 MBq]/well. At the end of the culture period, plate contents were aspirated through glass fibre filter mats (PerkinElmer, Waltham, MA) using an automated plate harvester (Tomtec, Hamden, CT). Filter mats were dried, and placed into plastic sample bags before addition of scillation liquid cocktail (Beta ScintTM; Wallac, Turku, Finland). ³H thymidine incorporation was measured using an automated beta scintillation counter (1450 Microbeta Plus; Wallac).

2.15.2 CFSE dilution

For some experiments, CFSE dilution was used to assess the proliferation of T cell subsets. Thawed PBMCs were labelled with 0.2 μ M CFSE, as described in section 2.13.2. 1 x10⁶/well CFSE-labelled PBMCs were dispensed into 24-well plates before addition of vaccine and/or α -GalCer. Cells were cultured for five days at 37 °C, then harvested and stained for surface markers. Flow cytometry was used to determine CFSE dilution within T cell subsets.

2.16 In vitro alloresponse to CLL cells

CLL cells were purified by flow cytometric cell sorting or positive immunomagnetic selection and incubated for 18 hours with vehicle, 200 ng/mL α -GalCer, 1 μ g/mL resiquimod (Invivogen, San Diego, CA), or both, then γ -irradiated (50 Gy) and washed four times. Treated, irradiated CLL cells were added at 4 x10⁴/well to 2 x10⁶/well healthy donor PBMCs in 96-well plates, and cultured in cIMDM for five days. ³H thymidine was added during the final twelve hours of culture, and thymidine incorporation measured by scintillation counter as described in section 2.15.1.

For some experiments, responding PBMCs were labelled with 0.2 μ M CFSE, as described in section 2.13.2, before addition of 4 x10⁴/well irradiated CLL cells as stimulators to 2 x10⁵ responders, in 96-well plates. After incubation at 37 °C for five days, cultures were harvested, stained for surface markers and analysed by flow cytometry to determine the number of divided (CFSE-low) cells within the iNKT and CD4+ and CD8+ T cell subsets.

2.17 Autologous responses to CLL cells

CLL cells were purified by FACS or positive immunomagnetic selection as described in section 2.4, then incubated for 18 hours with vehicle or 200 ng/mL α -GalCer, before γ -irradiation (50 Gy), four washes, and resuspension in cIMDM. For some experiments, additional aliquots of unpulsed or α -GalCer-pulsed CLL cells were cryopreserved at -180 °C in 90% FCS, 10% DMSO for future stimulations.

Between 1×10^5 and 1×10^6 CLL cells (the same number of treated and untreated cells were used for each donor) were added to 1×10^6 autologous B cell-depleted PBMCs in cIMDM in 48-well plates. 1 ng/mL IL-7 (eBioscience) was added at 24 hours. At one week intervals, further aliquots of CLL cells were thawed, washed, resuspended in cIMDM, and added to the responding cell cultures, with a further 1 ng/mL IL-7 added 24 hours later, to a total of three stimulations. One week after the third stimulation, cultures were harvested, absolute live cell counts determined by trypan blue exclusion, and flow cytometry performed to determine iNKT cell numbers.

For some experiments, 1 x10⁶ responding PBMCs were labelled with 0.2 μ M CFSE and resuspended in cIMDM, before addition of 5 x10⁵ autologous untreated or α -GalCer-

treated CLL cells in the absence of exogenous cytokines. Proliferation of PBMC subsets was assessed after five days of culture by flow cytometry.

2.18 Statistical analyses

Data were analysed using Prism^m 5.0 (GraphPad Software, La Jolla, CA). Non-parametric tests were used for all analyses except where otherwise stated: The Mann Whitney test was used for unpaired data, the Wilcoxon matched pairs test for paired data, and Kruskal Wallis analysis of variance (ANOVA) with Dunn's post-test for multiple comparisons. Bars and error bars represent medians and interquartile ranges except where otherwise stated. Categorical data were compared using Fisher's exact test. Survival curves were compared using the log-rank (Mantel-Cox) test. Linear and nonlinear regressions were performed using the least squares fitting method, without weighting. For some analyses, Pearson's correlation coefficient was calculated. For all analyses, two-tailed P values were calculated, and a P value < 0.05 was considered significant.

Chapter 3

Frequency and immunophenotype of iNKT cells in patients with CLL

3.1 Introduction

As discussed in section 1.3.3, preclinical studies suggest that iNKT cells have a number of potential roles in the immunotherapy of cancer, including:

- Production of cytokines such as IFN- γ with the potential to enhance NK cytotoxicity or modulate the number and function of immunosuppressive cells.⁴⁰²
- Activation of DCs via CD40L signals to enhance peptide-specific CD4+ and CD8+ T cell responses against a co-presented tumour antigen.²⁵⁸
- Direct cytotoxicity against CD1d positive tumour cells in the presence of strong ligands (such as α -GalCer) and/or activating NK receptor signals.³¹⁰

The number of circulating iNKT cells may dictate responses to iNKT cell-directed immunotherapy: In a phase I clinical study, cytokine responses to systemically administered α -GalCer were stronger in patients with higher numbers of circulating iNKT cells.³⁰³ However, reduced numbers and impaired function of circulating iNKT cells have been reported in patients with a range of malignancies, including advanced solid cancers, prostate cancer and multiple myeloma.^{210,303,304,307}

Subsets of iNKT cells with differing function have been described, as detailed in section 1.2.2.3. In humans, CD4+ iNKT cells produce Th2-type cytokines, whereas their CD4- counterparts do not.^{218,219} This may be of functional significance, as in mice, only the CD4- iNKT cell subset was capable of tumour rejection in a sarcoma model.²²²

As discussed in section 1.1.3.5, numerous abnormalities of innate and adaptive immunity have been described in patients with untreated CLL. These include a significant increase in the number of circulating T cells with an inversion of the CD4/CD8 T cell ratio, ^{83,84} the presence of oligoclonal T cell expansions in both CD4+ and CD8+ T cell subsets, ^{87,88} and impaired T cell proliferative responses to common mitogens.⁴⁴⁰ The treatment of CLL also affects T cell numbers: the purine analogue fludarabine, which is widely used as a

component of first-line chemotherapy, induces a profound and prolonged depletion of T cells, particularly of the CD4+ subset. $^{\rm 175}$

The numbers of iNKT cells have not previously been reported in patients with CLL. Therefore, as a first step to exploring the feasibility of iNKT cell-mediated immunotherapy of CLL, I investigated the number and immunophenotype of iNKT cells in patients with untreated and fludarabine-treated CLL.

3.2 Aims

I hypothesised that, as in other malignancies, the frequency of circulating iNKT cells would be reduced in patients with CLL, and that iNKT cells would display an abnormal immunophenotype in patients, compared to healthy controls.

Specifically, these experiments aimed to:

- Enumerate circulating iNKT cells from the peripheral blood of patients with untreated and fludarabine-treated CLL and of healthy age-matched controls.
- Determine the immunophenotype of circulating iNKT cells in patients and controls, focussing on the expression of CD4, CD8 and activation markers.

3.3 Results

3.3.1 Patient and control characteristics

Patients with untreated (n = 30) and treated (n = 10) CLL were recruited. Because the frequency of circulating iNKT cells declines with age, 210,223,441 healthy age-matched controls (n = 31) were also recruited. As circulating iNKT cell numbers are known to be reduced in patients with of solid cancers, 303,304 a history of any active malignancy other than CLL was used as an exclusion criterion in both patient and control groups.

The number, sex and age distribution of controls and untreated and treated patients is given in table 3.1. The age distribution of the three cohorts was closely matched (figure 3.1, table 3.1). The trend towards a male preponderance among patients who had received treatment for CLL (difference non-significant, Fisher's exact test) is consistent with the literature.⁸

Clinical stage was determined on the day of PBMC cryopreservation by clinical examination and peripheral blood count. The majority of patients included in this study were at an early clinical stage: 17% of untreated and 30% of treated patients had features of more advanced Binet stage B or C disease.

Treated patients had previously received the following lines of chemotherapy: chlorambucil (40%); fludarabine monotherapy (40%); fludarabine with cyclophosphamide (60%); fludarabine, cyclophosphamide and rituximab (20%); cyclophosphamide, doxorubicin, vincristine and prednisone (10%). The median time since the last cycle of chemotherapy was 22 months (range 3 – 87 months). All but one of the chemotherapy-treated patients had received at least one cycle of a fludarabine-containing regimen.

The distributions of haemoglobin concentration, platelet and lymphocyte counts, taken on the day of phlebotomy for the study, are given in figure 3.1. As would be expected, absolute lymphocyte count was significantly higher in untreated patients than in controls or patients who had received chemotherapy. Platelet count was significantly lower among chemotherapy-treated patients than in healthy controls or in untreated patients, likely to represent a consequence of chemotherapy-induced myelosuppression.

Figure 3.1 Age and peripheral blood counts of study participants Absolute peripheral blood counts and haemoglobin were determined by Coulter counter analysis of whole EDTA-anticoagulated blood taken on the same day as PBMC isolation. (a) Age, (b) haemoglobin, (c) lymphocyte count, (d) platelet count distribution of study participants (* p < 0.05; *** p < 0.0001; Kruskall Wallis test with Dunn's multiple comparison).



* One patient in the untreated This patient's lymphocyte or x10 ⁹ /L again since taking page	Median time since last chemotherapy (range), days	Median platelet count (range), x10 ⁹ /L	Median haemoglobin (range), g/L	Median lymphocyte count (range), x 10 ⁹ /L	Binet stage B or C (%)	Median age (range), years	Male (%)	n	Characteristic
d CLL cohort with a CLL score ount was persistently $> 5 \times 10^{-10}$ s this patient has been in	1	245 (131 – 329)	137 (117 – 171)	1.9 (0.6 – 2.8)	I	64 (43 – 86)	40	31	Controls
of 5 had a lymphocyte count < 5) ⁹ /L before donating blood for cluded in the CLL cohort.	I	188.5 (82 – 392)	136 (102 – 164)	29.6 (4.7 – 175.1) *	17	63 (47 – 86)	47	30	Untreated CLL
$x10^9/L$ on the day of phlebotomy this study, and has risen to > 10	713 (103 – 2649)	121 (87 – 200)	134.5 (97 – 166)	4.2 (1.0 - 102.3)	30	62 (47 - 79)	70	10	Treated CLL

Table 3.1 Control and patient characteristics.

	CLL cells (% of lymphocytes)				
		Post-depletion			
Lymphocyte count (x10 ⁹ /L)	Pre-depletion	Single round	Three rounds		
16.9	59.9	1.6	0.4		
19.0	86.3	6.3	1.7		
26.9	82.9	13.7	2.8		
38.0	94.0	10.8	0.3		
39.9	84.6	3.1	0.3		
61.9	94.6	26.0	0.8		
84.0	92.5	5.1	1.3		
129.5	95.5	31.2	0.3		
175.1	95.9	13.4	0.7		

Table 3.2 Immunomagnetic depletion of leukemic cells from CLL PBMC samples

3.3.2 Immunomagnetic depletion of B cells

Invariant natural killer T cells constitute a small proportion of circulating T cells in humans, particularly in older individuals,^{210,442} posing challenges to their analysis.

Flow cytometry allows the analysis of multiple parameters on individual cells at high speed, and was therefore selected as the most appropriate technique to enumerate and phenotyping of iNKT cells. However, the overwhelming proportion of malignant B cells (characterised by the immunophenotype CD5+CD19+ or CD5+CD23+) observed in the peripheral blood of some patients with CLL (table 3.2; figure 3.2) complicated flow cytometric analysis by necessitating the acquisition of extremely large numbers of events to detect any iNKT cell events.

Detailed analysis of this rare T cell subset therefore demanded depletion of malignant B cells from the stored PBMC samples. I chose to deplete B cells using anti-CD19 immunomagnetic beads. The reasons for using a negative isolation technique, rather than a positive isolation (selecting T cells), was that the negative isolation method leaves T and iNKT cells untouched and unactivated for subsequent analysis, and also allows characterisation of other PBMC populations such as DCs, monocytes and NK cells.

In some patients with CLL, particularly those with a very high peripheral blood lymphocyte count, a single round of CD19 depletion was insufficient to fully remove residual CLL cells. Three sequential rounds of immunomagnetic depletion successfully depleted CLL cells to <5% of residual lymphocytes in all cases (table 3.2; figure 3.2).



Each round of immunomagnetic depletion involves a 30-minute incubation at 4 °C with tilting and rotation. To exclude the possibility that this led to selective depletion or activation of PBMC subsets, I compared the numbers and activation status of PBMC populations after one and four rounds of immunomagnetic depletion.

I observed no major differences in the proportions of CD4+, CD8+, NK, NK-like T and iNKT cells, myeloid and plasmacytoid DCs, and CD14 high and CD16+ monocytes among viable CD19- cells (figure 3.3).

Figure 3.3 Effect of serial CD19 immunomagnetic depletion on major and minor PBMC subsets. Healthy donor PBMCs were analysed by flow cytometry to determine the proportions of CD4+ and CD8+ T cells, NK cells, NK-like T cells, iNKT cells, myeloid and pDCs and monocytes, after one or four sequential rounds of CD19 immunomagnetic depletion, as a percentage of viable CD19- cells. Performed in triplicate; bars and error bars represent mean and standard error of the mean of replicates, respectively.



Expression of the activation marker CD86 was determined on viable monocytes (gated as high forward and side scatter, CD14+ events) and on viable mDCs (gated as described in section 2.8). Serial depletion rounds did not result in marked monocyte or dendritic cell activation (figure 3.4a).

Chapter 3. Frequency and immunophenotype of iNKT cells in patients with CLL

Expression of the activation marker CD69 was determined on viable T cells (gated as CD3+CD19- events with low to high forward and side scatter), and did not change between one and four serial rounds of immunomagnetic B cell depletion (figure 3.4b).

Together, these experiments indicated that B cells could be effectively removed from patient and control samples using serial rounds of CD19 immunomagnetic depletion without major numerical or immunophenotypic changes in the remaining PBMC populations.

3.3.3 Flow cytometric identification of invariant natural killer T cells

A number of reagents haven been employed for the flow cytometric enumeration of iNKT cells, including antibodies against the V α 24 and V β 11 chains, CD1d tetramers loaded with α -GalCer, and the 6B11 antibody clone, which is directed against the complementarity determining region (CDR) 3 of the V α 24J α 18 TCR- α chain expressed by human iNKT cells.^{218,443} As not all V α 24 and V β 11 co-expressing T cells are either invariant or CD1d-reactive, ²¹⁸ I used the two more specific reagents to identify iNKT cells: the 6B11 antibody cD1d tetramer.

The low frequency of circulating iNKT cells in humans necessitated a stringent gating strategy including exclusion of doublet events using forward scatter and side scatter height versus area plots, exclusion of dead cells using a viability dye, and exclusion of B cells using an anti-CD19 antibody, illustrated in figure 3.5. To enhance iNKT cell discrimination, the brightest available fluorophore, PE, was used to label the α -GalCer-loaded CD1d tetramer and the 6B11 antibody clone. Together, these measures resulted in a false positive event rate, as determined by staining with unloaded CD1d tetramer and with a PE-labelled isotype control antibody, of 0.001% of T cells (figure 3.6).

Comparing the 6B11 antibody clone with an α -GalCer-loaded CD1d tetramer, I found that both could reliably identify iNKT cells, and that the numbers assessed using the two techniques correlated closely (figure 3.6), consistent with published observations.^{218,286}

3.3.4 Enumeration of circulating iNKT cells in patients and controls

For the enumeration of iNKT cells, the 6B11 antibody was selected because of lower background staining than that observed using the CD1d tetramer (figure 3.6). A median of 154914 T cells (range 48949 – 323685) were analysed per donor. False positive iNKT cell events, as determined using isotype control antibody, occurred at rate of 0.00003% of T cell events. This was lower than the lower limit of detection as determined by the median number of T cell events analysed, at 0.0006%. Thus, the median lower limit of iNKT cell detection was approximately 1 in 10^5 T cell events, or 0.001%.

Among healthy controls (n = 31), median iNKT cell frequency was 0.022% of T cells (range 0.004 - 0.302%). An age-related decline in iNKT cells was seen (95% confidence interval for slope -0.060 to -0.012; non-linear regression). The frequency of iNKT cells did
Figure 3.4 Activation status of monocytes, mDCs and T cells after serial immunomagnetic B cell depletion. Healthy donor PBMCs were analysed by flow cytometry after one or four sequential rounds of CD19 immunomagnetic depletion. (a) expression of the activation marker CD86 on monocytes and on myeloid DCs. (b) T cell activation, as determined by expression of CD69 and increase in side scatter.



Chapter 3. Frequency and immunophenotype of iNKT cells in patients with CLL

Figure 3.5 Flow cytometric gating to identify iNKT cells PBMCs from a healthy donor with a high proportion of iNKT cells stained and analysed by flow cytometry. Light gray lines illustrate gates. Gates were sequentially applied in the following order: forward scatter singlets; side scatter singlets; live cells; lymphocytes (by forward and side scatter); CD19 negative cells; CD3+6B11+ cells (iNKT cells).



Figure 3.6 Comparison of CD1d tetramer and 6B11 antibody clone for iNKT cell identification (a) Identification of iNKT cells with the 6B11 antibody clone, (b) identification of iNKT cells with α -GalCer-loaded CD1d tetramer in the same donor. Numbers represent percentage of live CD19- lymphocytes within the iNKT cell gate. (c) Comparison of iNKT cell numbers as assessed by 6B11 antibody or α -GalCer-loaded CD1d tetramer in healthy donors (n = 18; r² = 0.97, linear regression)



(a)

not differ significantly between males and females (median 0.019% of T cells in males, 0.026% in females; p = 0.35).

Compared to healthy controls, the frequency of iNKT cells in patients, expressed as a percentage of circulating T cells, was significantly reduced (median 0.02% in controls, 0.01% in patients; figure 3.7a).

To calculate absolute numbers of circulating iNKT cells and other T cell subsets, the absolute lymphocyte count, as determined by analysis of whole blood taken on the day of PBMC cryopreservation using a commercial blood count analyser, was used. The number of T cells as a proportion of all lymphocytes, had been determined by flow cytometric analysis of lymphocyte CD3 expression of cryopreserved PBMCs, before B cell depletion steps. Together, this provided a measure of the absolute number of circulating T cells, which was used to calculate absolute numbers of iNKT cells and of other T cell subsets.

Considering absolute numbers of circulating iNKT cells, the difference between patients and controls was no longer apparent (figure 3.7b). This was due to an absolute T cell lymphocytosis observed among patients with CLL (median 1.0 x10⁶/L T cells in controls, 2.5 x 10⁶/L in patients; p < 0.0001), a phenomenon that has been described previously.^{83,84}

Stratification of results by disease status did not reveal any relationship between iNKT cell frequency and the clinical stage, lymphocyte count, ZAP-70 status or history of prior fludarabine chemotherapy (figure 3.8). As discussed in section 1.1.5.7, fludarabine has a profound and prolonged effect on T cell populations.¹⁷⁵ Therefore, for this, and for future comparisons of chemotherapy-treated versus untreated patients, the single patient who had received chlorambucil chemotherapy only (and who had never received fludarabine) was not included. Similar results were obtained throughout when the chlorambucil-treated patient was included.

Together, these data suggest that absolute iNKT cell numbers are preserved in CLL, but are reduced as a proportion of T cells, because of an absolute increase in the number of T cells which do not bear the invariant $V\alpha 24J\alpha 18$ receptor.

3.3.5 Analysis of T cell subsets

In order to place the results of iNKT cell numbers into context, I analysed the numbers of other major T cell subsets: CD4+, CD8+, CD4-CD8-, NK-like T and T_{reg} cells. These are presented as percentages of all T cells in figure 3.9. The results indicate a relative reduction in CD4+ T cells and increase in CD8+ T cells in patients, a finding that was more marked among those who had received fludarabine. Reductions in the CD4/CD8 ratio are well described in CLL.^{83,84}

In terms of absolute numbers, increases in all T cell subsets were evident in untreated patients (figure 3.10). In patients who had received fludarabine, a trend towards reduced numbers of CD4+ T cells was observed. These findings are consistent with published data utilising counting beads to accurately determine absolute counts in smaller group of patients.⁴⁴⁴

Figure 3.7 Relative and absolute iNKT cell numbers in patients and controls. Invariant natural killer T cells were enumerated by flow cytometry of B-cell depleted PBMCs from 40 patients with CLL and 31 healthy age-matched controls using the 6B11 antibody clone, and expressed as a proportion of all T cells (a), or as an absolute number (b). (Bars represent medians; * p < 0.05; n.s. = not significant, Mann Whitney test)



(a)

Figure 3.8 NKT cell frequency stratified by stage, lymphocyte count, ZAP-70 status, and prior treatment. iNKT cell frequency, expressed as a proportion of T cells, were determined in PBMCs from 30 patients with CLL and 31 healthy age-matched controls. Patient results are stratified according to (a) Binet clinical stage, (b) absolute lymphocyte count, (c) ZAP-70 expression status, and (d) history of prior fludarabine chemotherapy. (Bars represent medians)



The data I present do not demonstrate a relative increase in T_{reg} numbers among patients with CLL (figure 3.9), although a trend towards an absolute increase in T_{reg} numbers was seen (figure 3.10). This is at odds with some observations, ⁹⁸ but a recent systematic analysis of T cell subset numbers in patients with B cell malignancies also reported normal T_{reg} numbers in patients with CLL.⁴⁴⁴

Figure 3.9 Relative proportions of T cell subsets in controls and patients with CLL. Flow cytometry of T cell subsets was performed, gating on CD4+, CD8+, double negative (DN; CD4-CD8-), NK-like T (CD161+CD3+) and T_{reg} (CD4+FoxP3+) subsets of CD3+ T cells. T cell subsets are expressed as a percentage of all T cells. ** p < 0.01; * p < 0.05; Kruskall Wallis test with Dunn's post tests.



Chapter 3. Frequency and immunophenotype of iNKT cells in patients with CLL

Figure 3.10 Absolute numbers of T cell subsets in controls and patients with CLL. Flow cytometry of T cell subsets was performed, gating on CD4+, CD8+, double negative (DN; CD4-CD8-), NK-like T (CD161+CD3+) and T_{reg} (CD4+FoxP3+) subsets of CD3+ T cells. T cell subsets are expressed as absolute numbers. *** p < 0.001; ** p < 0.01; * p < 0.05; Kruskall Wallis test with Dunn's post tests.



3.3.6 Immunophenotype of iNKT cells

Subsets of human iNKT cells with differing function have been identified. The CD4subset is associated with production of Th1 cytokines such as IFN- γ , whereas the CD4+ subset also produces T helper 2 (Th2) cytokines such as IL-4 and IL-13.^{218,220} The NK marker CD161 is expressed on the majority of human iNKT cells, and may provide a costimulatory function.²²⁷ Activation of T cells, including iNKT cells, results in upregulation of CD25. In one study, upregulation of CD25 on iNKT cells was documented in patients with tuberculosis.²⁸⁵

I used multiparameter flow cytometry to investigate iNKT cell expression of CD4, CD8, CD161 and the activation marker CD25 in patients with CLL. The immunophenotype of circulating iNKT cells was similar in patients and controls (figure 3.11). Although I observed a trend towards a higher proportion of CD4+ iNKT cells in patients (figure 3.11), this was not statistically significant. A statistically significant inverse correlation between the proportion of iNKT cells which were CD4+ and the number of iNKT cells (as a proportion of T cells) was observed (figure 3.12), a finding which has previously been reported. 218

3.3.7 iNKT cell numbers do not correlate with clinical outcome

Clinical outcome was assessed by review of patient records for the 29 out of 30 patients with untreated CLL for whom follow-up data were available. Events defining progression-free survival were death from any cause or progression of CLL requiring treatment (including palliative treatments or treatment for disease-related autoimmune cytopenias).

During a median follow-up of 38.7 months following phlebotomy (range 12.4 – 48.8 months), eight out of twenty-nine patients progressed: four patients received chlorambucil-based chemotherapy, one received FCR chemotherapy, one was treated with corticosteroids for autoimmune haemolytic anaemia, one patient had radiotherapy to a symptomatic lymph node and one patient died of metastatic lung cancer.

The median iNKT cell frequency was 0.009% of T cells (range 0.004 - 0.051) among patients who progressed or died and 0.011% of T cells (range 0.001 - 0.113) among patients who remained alive and treatment-free (difference not significant, Mann Witney test).

The median absolute iNKT cell number was 253/mL blood (range 97 – 2452) among patients who progressed or died and 335/mL (range 18 – 2940) among patients who remained alive and treatment-free during follow-up (difference not significant, Mann Witney test).

Survival curve analysis showed no difference in progression-free survival between those with a high and those with a low iNKT cell frequency (figure 3.13a) or between those with a high and those with a low absolute number of circulating iNKT cells (figure 3.13b).

Figure 3.11 Immunophenotype of circulating iNKT cells (a) Representative flow cytometry plots showing CD4 and CD8 status of T and iNKT cells. (b) Patient (n = 38) and control (n = 28) PBMCs were stained, and expression of CD4, CD8, CD161 and CD25 determined by flow cytometry, gated on iNKT cells (identified using 6B11 antibody clone). Bars represent medians, error bars represent interquartile ranges. Difference not significant for any comparison, Mann Whitney test. (c) CD4 status of iNKT cells presented according to fludarabine treatment status. Difference not significant for any comparison, one-way ANOVA.



Figure 3.11 (continued) Immunophenotype of circulating iNKT cells (c) CD4 status of iNKT cells presented according to fludarabine treatment status. Difference not significant for any comparison, one-way ANOVA.



(c)

Patients: prior treatment

Figure 3.12 Correlation between iNKT cell numbers and iNKT cell CD4 status Patient (n = 40) and control (n = 30) PBMCs were analysed by flow cytometry for numbers of iNKT cells (identified as 6B11+CD3+), and for iNKT cell CD4 expression. The percentage of iNKT cells which are CD4+ is plotted against the iNKT cell number (as a percentage of all T cells). $r^2 = 0.136$, p < 0.01 of zero slope, nonlinear regression, semilog fit.



Figure 3.13 Progression-free survival does not correlate with iNKT cell numbers Progression-free survival was determined for the 29 patients with untreated CLL for whom follow-up data were available. (a) Progression-free survival did not differ between those with a high iNKT cell frequency ($\geq 0.01\%$ of T cells, solid line, n = 14) and those with a low iNKT cell frequency (< 0.01% of T cells, dashed line, n = 15), whether determined from the date of phlebotomy for iNKT cell frequency assessment or from the date of diagnosis. (b) Progression-free survival did not differ between those with a high absolute iNKT cell number (\geq 330 per mL blood, solid line, n = 14) and those with a low absolute iNKT cell number (< 330 per mL blood, dashed line, n = 15), whether determined from the date of phlebotomy for iNKT cell frequency assessment or from the date of phlebotomy for iNKT cell frequency assessment or from the date of phlebotomy for iNKT cell frequency.

(a)



3.4 Summary

In this chapter, I have determined the numbers and immunophenotype of circulating iNKT cells from patients with CLL.

I recruited a cohort of patients with untreated and treated CLL, and a group of healthy controls who were closely age-matched. The sensitivity of flow cytometric detection of iNKT cells was optimised by depletion of contaminating leukaemic B cells, the selection of a highly specific fluorophore (the 6B11 antibody clone), and the use of a stringent gating strategy including exclusion of dead cells, B cells and doublet events.

My results indicate that although reduced as a proportion of T cells, absolute numbers of iNKT cells are preserved in patients with untreated CLL, with a decline among those who had received fludarabine chemotherapy. The combination of a reduced frequency but normal absolute number of iNKT cells may be explained by to a failure of iNKT cells to fully participate in the expansion of other T cell subsets observed in patients with untreated CLL. In contrast, reduced absolute numbers of all T cell subsets, including iNKT cells, were seen in those who had received fludarabine-based chemotherapies.

The immunophenotype of circulating iNKT cells is similar in patients to that in controls: no significant differences in CD4, CD8, CD161 and CD25 expression were observed, although there was a trend towards a higher proportion of CD4+ iNKT cells in patients. This trend in keeping with a general inverse correlation between the frequency of iNKT cells overall and the proportion of iNKT cells which were CD4+.

Finally, I sought, but did not find, a correlation between iNKT cell frequency and number and progression free survival among patients with untreated CLL. This suggests that a deficiency of circulating iNKT cells does not directly contribute to CLL disease progression.

In chapter 4, I shall evaluate expression of CD1d, the molecule which presents glycolipid antigens to iNKT cells, on PBMCs from patients with CLL and controls.

Chapter 4

CD1d expression on PBMC subsets

4.1 Introduction

In chapter 3, I established that absolute numbers of circulating iNKT cells were normal in untreated patients with CLL, and were reduced but detectable in patients who had previously received fludarabine-containing therapies. I found no major differences in the immunophenotype of circulating iNKT cells between patients and controls.

The activation of iNKT cells through their invariant TCR requires the presentation of glycolipid antigens on the MHC Class I-like molecule, CD1d. Many cells of haematopoietic origin express CD1d, including B cells, monocytes and mDCs.⁴⁴⁵

Numeric, phenotypic and functional defects of APCs have been described in patients with CLL, including reduced frequency of circulating DCs,⁷⁸ reduced MHC Class I expression on circulating DCs,⁸⁰ and reduced allostimulatory of moDCs generated from patients with active disease.⁸² The expression of CD1d on circulating APC subsets has not been reported in CLL.

The lysis of tumour cells by iNKT cells requires target cell CD1d expression.^{309,310} Expression of CD1d on tumour cells also augmented the efficacy of vaccination with α -GalCer-pulsed tumour cells in preclinical studies.^{422,423} A single published report indicates that CLL cells do express CD1d and, and that they can be lysed by allogeneic iNKT cell lines.⁴¹⁷

I used flow cytometry to examine the numbers of circulating APC subsets and CD1d expression on APCs and CLL cells, on PBMC samples from patients with CLL and from healthy controls.

4.2 Aims

The specific aims of the experiments outlined in this chapter were:

- to enumerate circulating APCs from patients with CLL
- to determine CD1d expression levels on circulating APCs
- to determine CD1d expression levels on CLL cells.

4.3 Results

4.3.1 Relative CD1d expression levels on PBMC subsets

Activation of iNKT cells by α -GalCer requires its presentation on the MHC Class I-like molecule, CD1d. Within PBMCs, the highest levels of CD1d expression were found on mDCs and on monocytes. Leukemic B cells expressed CD1d at a lower level, similar to that of normal B cells (figure 4.1).

Higher levels of CD1d have been reported on CLL cells in cases with unmutated IgVH genes.⁴¹⁷ I do not have data on the IgVH status of the patients in this study. Of the samples exhibiting ZAP-70 positivity (which correlates with unmutated IgVH status), there was no trend towards higher CD1d expression.

In patients with myeloma, another malignancy of mature B cells, CD1d expression on the malignant B cells declines with progression of the disease.⁴⁴⁶ I therefore compared CD1d expression levels on CLL cells between patients with early (Binet stage A) and clinically advanced (Binet stages B or C) disease (figure 4.1c). I could find no evidence of reduced CD1d expression on CLL cells in patients with advanced as opposed to early stage disease (figure 4.1).

4.3.2 Myeloid dendritic cell numbers and CD1d expression

Numeric, immunophenotypic and functional defects of circulating mDC have been reported in CLL.^{78,80} Therefore, I investigated the numbers of, and CD1d expression on, mDC from patients with CLL in more detail.

Flow cytometric gating of mDCs was performed by selecting a forward and side scatter gate intermediate between lymphocytes and monocytes, excluding cells expressing lineage markers using a Lin-1 antibody cocktail, then selecting cells which expressed high levels of HLA-DR and CD11c. Cells identified in this manner were overwhelmingly positive for CD1c, also known as Blood Dendritic Cell Antigen 1 (BDCA-1), and considered a specific marker for mDCs (median 98.5% CD1c positive, range 94.4% – 100%, n = 16; figure 4.2a).⁴⁴⁷

As expected, mDC numbers were reduced as a proportion of PBMCs in most patients due to large numbers of circulating leukaemic cells. In contrast, the absolute numbers of circulating mDCs were similar to controls in patients with untreated CLL, with a non-significant trend towards reduction among patients who had received a fludarabine-based chemotherapy (median 8.4 x 10^3 /mL in controls, 7.9 x 10^3 /mL in untreated patients, 6.2 x 10^3 /mL in fludarabine-treated patients, figure 4.2b).

Figure 4.1 Expression of CD1d on PBMC from patients with CLL. CD1d expression on PB-MCs from patients with CLL was determined by flow cytometry, gating on T cells (CD19-, CD5+ lymphocytes), normal B cells (CD19+, CD5- lymphocytes), CLL cells (CD19+, CD5+ lymphocytes), myeloid DCs (intermediate forward and side scatter, Lin1-, HLA-DR high, CD11c high) and monocytes (high forward and side scatter, CD14+). (a) CD1d expression on PBMC subsets from an individual patient with CLL. (b) CD1d expression levels in patients with untreated CLL (n = 10). Bars represent median, error bars represent range, *** p < 0.001, ** p < 0.01, Friedman test with Dunn's multiple comparison. (c) CD1d expression on CLL cells in patients with early (Binet stage A; n = 10) or clinically advanced (Binet stages B or C; n = 9) disease. Bars represent medians; n.s. = not significant, Mann Whitney test.



(c)



CLL Binet stage



Assessed by flow cytometry using an anti-CD1d antibody, the surface expression of CD1d on mDC was the same in patients with CLL as in healthy donors (figures 4.3a and 4.3b).

Hypogammaglobulinemia is a common feature of CLL. MoDCs generated in the absence of human IgG have reduced CD1d expression, a finding ascribed to IgG binding to moDC via Fc γ RIIa.⁴⁴⁸ I hypothesized that mDC CD1d expression may correlate positively with serum IgG level. However, I found no relationship between CD1d expression and serum IgG levels in patients or in healthy donors (r² = 0.00), with normal mDC CD1d expression even in patients with marked hypogammaglobulinaemia (figure 4.3c).

4.3.3 Monocyte numbers and CD1d expression

As already noted, monocytes express CD1d at a level intermediate between that of B cells and mDCs (figure 4.1b). As monocytes constitute a high proportion of PBMCs, they may be important for the presentation of α -GalCer to iNKT cells. Abnormalities of the monocyte population have been described in patients with CLL, including a monocytosis and increased MHC class II expression.⁴⁴⁹ I therefore investigated monocyte numbers and CD1d expression in more detail.

Two principal monocytes subpopulations have been identified in human PBMCs: a large population of monocytes with high expression of CD14 and low CD16 expression (CD14hi monocytes), and a smaller population of monocytes that co-express CD16 and CD14 (CD16+ monocytes). CD16+ monocytes are also known as 'inflammatory monocytes', as they possess phenotypic features of tissue macrophages, ⁴⁵⁰ are potent producers of TNF- α in vitro, ⁴⁵¹ and are increased in number in proinflammatory states such as sepsis. ⁴⁵² These proinflammatory characteristics do not appear to translate to tumour protection, however, as CD16+ monocytes are reported to be increased in numbers in patients with breast cancer and cholangiocarcinoma, and to carry negative prognostic value. ^{453,454} In vitro, CD16+ monocytes differentiate into moDCs with Th2-skewing properties. ^{455,456}

I investigated the numbers of these two monocyte subsets in patients with CLL and controls, and examined CD1d expression levels on each subset.

The absolute number of CD14hi monocytes was similar in patients and controls. A trend towards a higher frequency of CD16+ monocytes in fludarabine-treated patients with CLL was not statistically significant (median 5.9% of monocytes CD16+ in controls, 6.7% in untreated patients, 11.1% in fludarabine-treated patients, p = 0.06, Kruskal-Wallis test; figure 4.4).

Flow cytometric analysis indicated that CD14hi monocytes consistently expressed CD1d at a significantly higher level than their CD16+ counterparts in both patients and controls (p < 0.0001, Wilcoxon matched pairs test; figure 4.5a and 4.5b). Comparing patients with controls, expression of CD1d on the two monocyte subsets was normal in patients with CLL (figure 4.5c).

Figure 4.3 Expression of CD1d on patient and control mDC. (a) Representative flow cytometry plots demonstrating expression of CD1d by patient and control mDC. (b) CD1d expression on patient and control mDCs; box represents median +/- interquartile range; whiskers represent range. (c) Myeloid DC CD1d expression in controls (n = 20) and patients with CLL (n = 20), expressed according to serum IgG level. Dotted line represents linear regression ($r^2 = 0.00$).







Figure 4.4 Absolute numbers of monocyte subsets. The CD14hi and CD16+ monocyte subsets were identified by flow cytometry of PBMCs, as indicated in figure 4.5a, and absolute numbers of each subset calculated using full blood counts performed on whole blood on the same day as PBMC storage. Differences not significant for any comparison, Kruskal-Wallis test with Dunn's post-tests.



Figure 4.5 CD1d expression on monocyte subsets. (a) Example of gating from a patient with CLL, illustrating reduced CD1d expression on the CD16+ subset of monocytes, compared to CD14hi monocytes. Gates are indicated by gray lines. For CD1d expression histograms, the black line represents anti-CD1d PE, and solid gray represents isotype-matched control antibody. (b) comparison of CD1d expression on CD14hi and CD16+ monocyte subsets; *** p < 0.001, Wilcoxon matched pairs test. (c) Expression of CD1d on monocyte subsets was studied in controls (n = 12) and patients (n = 11) (bars represent medians, error bars represent range; n.s. = not significant, Mann Whitney test).



4.4 Summary

In this chapter, I have investigated the expression of CD1d on PBMC subsets in patients and controls.

CLL cells expressed CD1d at a similar level to that of normal B cells, with no evidence of a reduction in CD1d expression in those with more clinically advanced disease.

The highest levels of CD1d expression were found on mDCs. I found normal absolute numbers of circulating mDCs in untreated CLL patients, with a trend towards a reduction in those who had previously received fludarabine-based chemotherapy. The expression of CD1d on mDC was normal in patients, even among those with marked hypogamma-globulinaemia.

The numbers of the major monocyte subsets were similar in patients with CLL to healthy controls, although there was a non-significant trend towards higher numbers of 'inflammatory' CD16+ monocytes in patients with CLL who had received a fludarabine-based chemotherapy. If confirmed with larger numbers, this would be in keeping with the finding of increased numbers of CD16+ monocytes in patients with other malignancies, such as breast cancer and cholangiocarcinoma. 453,454,457

Although overall monocyte CD1d expression levels were similar in patients with CLL compared to controls, the CD14hi and CD16+ subsets of monocytes exhibited differential expression of CD1d. The CD14hi subset expressed CD1d at significantly a higher level than the minor population of CD16+ monocytes in both healthy donors and patients.

Taken together, these results suggest that patients with fludarabine-treated CLL exhibit changes in myeloid APCs populations, with reduced mDC numbers and a trend towards higher CD16+ monocyte numbers. However, the overall expression of CD1d on both mDCs and monocytes is largely intact in patients with both treated and untreated CLL. These results confirm that CLL cells themselves express CD1d, at a level similar to that of normal B cells.

Having established that phenotypically-normal iNKT cells can be detected in the peripheral blood of patients with CLL in chapter 3, and that CD1d expression on circulating APCs is intact in this chapter, I proceed to investigate the function of iNKT cells from patients with CLL in chapter 5.

Chapter 5

Function of iNKT cells from patients with CLL

5.1 Introduction

In chapters 3 and 4, I established that the iNKT cell and CD1d axis is essentially numerically and immunophenotypically intact in patients with untreated CLL, although absolute numbers of iNKT cells and mDCs were reduced in patients who had previously received a fludarabine-containing therapy. In this chapter, I aim to evaluate the function of iNKT cells from patients with CLL.

As discussed in section 1.3.3, properties of iNKT cells which might be exploited for use in immunotherapeutic strategies include the production of cytokines (resulting in transactivation of NK cells), the maturation of dendritic cells (resulting in enhanced T cell responses to peptide antigens), and the lysis of CD1d-expressing target cells.

Section 1.1.3.5 details the functional defects of T cells described in patients with CLL, which include increased IL-4 production,⁹⁵ defective immunological synapse formation,⁹⁷ and impaired *in vitro* expansion in response to mitogenic stimuli.⁹⁴

Abnormalities of iNKT cell function have been described in patients with other malignancies. The production of IFN- γ by iNKT cells is reduced in patients with advanced myeloma and prostate cancer.^{304,307} In patients with advanced prostate cancer, *ex vivo* iNKT cell proliferation was impaired,³⁰⁴ while in contrast, iNKT cell expansion from patients with myeloma was maintained.³⁰⁷

The function of iNKT cells from patients with CLL has not previously been described. In view of the wide-ranging defects of T cell function described in patients with CLL, ³³⁴ and the abnormal iNKT cell function observed in patients with other malignancies, I hypothesised that, despite possessing a normal immunophenotype, iNKT cells from patients with CLL would exhibit abnormalities in cytokine production, proliferation or cytotoxicity.

5.2 Aims

To assess function of iNKT cells from patients with CLL, I performed a series of experiments to assess iNKT cell cytokine production and *in vitro* proliferation in whole PBMCs cultures. To enable analysis of iNKT cell cytotoxicity, I generated and characterised the function of iNKT cell lines from patients and controls.

The specific aims of the experiments detailed in this chapter were:

- To evaluate cytokine production by iNKT cells from patients with CLL
- To determine the capacity of iNKT cells from patients with CLL to proliferate *in vitro*
- To generate and characterise iNKT cell lines from patients with CLL

5.3 Results

5.3.1 Cytokine production by iNKT cells

Invariant natural killer T cells are potent producers of cytokines. Among others, iNKT cells are capable of producing both Th1-type cytokines (such as IFN- γ and IL-2) and Th2-type cytokines (such as IL-4, IL-5, IL-10 and IL-13). Defects of iNKT cell cytokine production have been described in patients with advanced myeloma and prostate cancer.^{304,307}

Assessment of cytokine production is challenging because iNKT cells constitute a very small proportion of circulating T cells in humans (typically 0.01% in patients with CLL, as determined in chapter 3). I therefore employed two sensitive techniques: an ELISpot assay, and intracellular cytokine staining (ICS).

Preliminary experiments demonstrated that the addition of α -GalCer to PBMCs enabled specific detection of iNKT cells: the numbers of α -GalCer-induced IFN- γ ELISpots exceeded those produced in the presence of vehicle alone, and were inhibited by the addition of anti-CD1d antibody (figure 5.1).

I performed IFN- γ ELISpot assays on B-cell depleted PBMCs from patients with CLL and controls (figure 5.2). A significant increase in ELISpots over background was observed with the addition of α -GalCer in both groups. In the presence of α -GalCer, at least 50 IFN- γ ELISpots were observed, and the number of IFN- γ ELISpots was at least double that of control, in six out of twelve healthy donors and in six out of eleven patients. The difference between the number of α -GalCer induced ELISpots observed in patient and healthy donor samples was not statistically significant.

Although this indicates that α -GalCer-induced IFN- γ production is intact in CLL, the ELISpot assay does not determine which cell type is releasing IFN- γ . The cytokine could be produced by iNKT cells themselves, or by neighbouring cell types (such as NK or T cells). To directly assess cytokine production by iNKT cells, I used flow cytometry with ICS. This technique allows simultaneous evaluation of multiple surface and intracellular markers. The results of ICS of T cells have previously been reported in CLL: reported findings include enhanced T cell production of IFN- γ and IL-2 in patients with early CLL,

Figure 5.1 α**-GalCer-induced IFN-**γ **ELISpot assay.** PBMCs from individuals with high or low numbers of circulating iNKT cells were incubated with α-GalCer, α-GalCer with anti-CD1d antibody, vehicle (negative control) or PMA and ionomycin (positive control) at 5 x10⁵ cells per well. The number of IFN-γ ELISpots is given below each image.



Figure 5.2 IFN- γ **ELISpot production by patient and control PBMCs in response to** α -GalCer. 1 x10⁶ B-cell depleted PBMCs were incubated for 24 hours in the presence of α -GalCer or vehicle control. ELISpots were developed and enumerated by automated counter. (a) Representative α -GalCer-induced IFN- γ ELISpots from two controls and two patients with CLL. (b) ELISpots from patients with CLL (n = 11) and age-matched healthy controls (n = 11). Dashed line represents a threshold value of 50 ELISpots per well. ** p < 0.01, Wilcoxon matched-pairs test.



and in one study, a Th2 bias with increased IL-4 production in patients with progressive CLL. $^{458-460}$

Expression of intracellular IFN- γ and IL-4, the archetypal Th1 and Th2 cytokines respectively, was assessed in PMA and ionomycin-stimulated PBMCs from healthy donors and controls, gating separately on iNKT and T cell populations. In comparison to normal T cells, iNKT cells produced significantly more IFN- γ (figure 5.3). Moreover, compared to 'conventional' T cells, a significantly higher proportion of iNKT cells were dual cytokine producers, with some individual iNKT cells producing both IFN- γ and IL-4, as previously reported.⁴⁶¹ Separate analysis of CD4+ and CD4- iNKT cell subsets in healthy donors confirmed the previously reported finding that the CD4+ subset of iNKT cells produces more IL-4 than the CD4- subset (figure 5.4).²¹⁸

Significantly higher T cell IFN- γ production was observed in patients compared to controls, consistent with previous reports (figure 5.5a).^{458,460} Although there was a trend towards higher IFN- γ production from patient iNKT cells, this was not statistically significant, and iNKT cell IL-4 expression was the same in patient and control groups (figure 5.5b).

Taken together, the ELISpot and ICS assays performed indicate that circulating iNKT cells in patients with CLL have an intact potential to produce IFN- γ , and do not exhibit a Th2 bias compared with iNKT cells from healthy controls.

5.3.2 Granzyme B expression by iNKT cells

The serine protease granzyme B is expressed by cytotoxic CD8+ T cells and NK cells, and contributes to target cell lysis. Granzyme B appears to play a role in iNKT cell cytotoxicity: in one study, α -GalCer-induced cell lysis of primary human myelomonocytic leukemia cells by iNKT cells was perforin- and granzyme B-dependent.²²⁹

The expression of granzyme B is modulated by cytokines, particularly by IL-2 and its related cytokines IL-15 and IL-21, which enhance granzyme B expression.⁴⁶² CLL cells express the IL-2 receptor alpha (CD25) and can take up IL-2 directly, depriving T cells of this signal.¹⁰⁶ Moreover, supernatant from cultures of CLL cells actively inhibits T cell IL-2 production.⁹¹ I therefore wished to determine granzyme B expression of iNKT cells from patients with CLL as a first step towards establishing their cytotoxic function.

I used intracellular staining to determine granzyme B expression levels on NK, T and iNKT cells in patients and controls. Granzyme B expression of iNKT cells was lower than on NK cells, and at a similar level to T cells overall (figure 5.6a and 5.6b). Comparing patients with controls, I found no major differences between granzyme B expression levels on NK, T or iNKT subsets (figure 5.6c). As with the ICS data above, iNKT cell granzyme B expression data should be interpreted with caution, as fewer iNKT cell events than NK or T cell events were available for analysis.

Figure 5.3 Intracellular cytokine staining of iNKT cells. Patient and control PBMCs were incubated with PMA and ionomycin for 6 hours then permeabilised and stained for intracellular IFN- γ and IL-4. (a) Representative intracellular cytokine staining showing cytokine profiles of T cells and iNKT cells. (b) Comparison of cytokine profiles of T and iNKT cells (n = 22). 'Th1' refers to cells producing IFN- γ alone; 'Th2' refers cells producing IL-4 alone; 'Dual' refers to cells expressing both IFN- γ and IL-4 producing. Bars within boxes represent medians, boxes represent interquartile range, and whiskers represent range. *** p < 0.0001, * p<0.05, Wilcoxon signed rank test.







Figure 5.4 Differential cytokine production by CD4+ and CD4- subsets of iNKT cells. ICS was performed on PBMCs from healthy donors (n = 6), gating on the CD4- and CD4+ subpopulations of iNKT cells. * p < 0.05, Mann Whitney test..



5.3.3 Proliferative capacity of iNKT cells

Next, I wished to establish whether iNKT cells from patients with CLL were able to proliferate when PBMCs were stimulated with α -GalCer. I reasoned that proliferation of iNKT cells in response to α -GalCer within whole PBMC cultures would be a useful marker of iNKT cell function, as it would require α -GalCer presentation by patient APCs, recognition of the α -GalCer/CD1d complex by patient iNKT cells, and the presence of functional iNKT cell signalling pathways downstream of the TCR.

Preliminary experiments using healthy donor samples indicated that proliferation of iNKT cells could be induced by culturing whole PBMCs with α -GalCer in the presence of 100 U/mL IL-2 (figure 5.7a). An α -GalCer concentration of 100 ng/mL was chosen for subsequent experiments. Although IL-2, IL-7 and IL-15 were able to support iNKT cell proliferation, IL-21 was not (figure 5.7b). Used at 100 U/mL, IL-2 appeared at least as effective as IL-7 or IL-15 in promoting iNKT cell proliferation in the presence of α -GalCer (figure 5.7c). A further experiment indicated that delaying the addition of IL-2 until 24 hours after the start of the cultures significantly improved iNKT cell proliferation (figure 5.7d). On the basis of these data, I chose to study iNKT cell proliferation in the presence of 100 ng/mL α -GalCer, with addition of 100 U/mL IL-2 24 hours after start of culture.

To allow comparison between patient and control samples, PBMCs were depleted of Bcells using anti-CD19 immunomagnetic beads before incubation with α -GalCer and then IL-2. An α -GalCer-dependent increase in iNKT cell numbers (figure 5.8a) was observed in all 20 controls and in 22 out of 25 patients studied (figure 5.8b). The median fold increase in iNKT cells over the seven-day culture period was 37.9 in controls, 26.5 in untreated patients, and 74.1 in fludarabine-treated patients (difference not significant; **Figure 5.5 Cytokine production by T and iNKT cells in patients and controls.** Intracellular cytokine staining was performed on PBMCs from healthy donors (n = 11) and patients with CLL (= 11). (a) Cytokine profile of T cells. (b) Cytokine profile of iNKT cells. * p < 0.05, Mann Whitney test.



Figure 5.6 Intracellular granzyme B expression of iNKT cells. PBMCs from healthy donors (n = 4) and patients with CLL (n = 4) were stained for intracellular granzyme B alongside surface markers. (a) Illustrative flow cytometry plot showing granzyme B expression on B cells, T cells and iNKT cells. (b) Comparison of granzyme B expression levels in NK, T and iNKT cells. ** p < 0.01; * p < 0.05; Friedman test with Dunn's post-tests. (c) Graph summarising the proportions NK, T and iNKT cells expressing granzyme B in patients and controls.



Figure 5.7 Optimising cytokine conditions for iNKT cell proliferation. (a) 1 x 10⁶ healthy donor PBMCs were incubated with the indicated concentration of α-GalCer and 100 U/mL IL-2. Cells were harvested after 7 days, and iNKT cells enumerated by flow cytometry. (b) 1 x 10⁶ PBMCs from two healthy donors were incubated with 100 ng/mL α-GalCer and the indicated cytokines at 100 U/mL for IL-2 and 10 ng/mL for the other cytokines. Cells were harvested after 7 days of culture, and live iNKT cell numbers determined by flow cytometry. *** p < 0.001; ** p < 0.01; two way ANOVA with Bonferroni post-tests. (c) 2 x 10⁵ PBMCs from a healthy donor were cultured in triplicate in the presence of α-GalCer and the indicated concentration of cytokine for 7 days, then harvested, counted, and analysed by flow cytometry to determine the numbers of iNKT cells. Bars represent means, error bars represent standard error of the mean (SEM). *** p < 0.001, ** p < 0.01, * p < 0.05, one-way ANOVA with Tukey post-test. (d) Effect of delayed addition of IL-2 on iNKT cell expansion. PBMCs from three healthy donors were cultured with 100 ng/mL α-GalCer in triplicate. 100 U/mL IL-2 was added immediately or 24 hours after start of culture. Cells were harvested after 7 days, and iNKT cells enumerated by flow cytometry. p < 0.01 for effect of IL-2 timing, two-way ANOVA.



figure 5.8c).

Figure 5.8 iNKT cell proliferation from patients and controls. (a) Representative flow cytometry plots demonstrating expansion of iNKT cells after culture of 5×10^5 patient B-cell depleted PBMCs with α -GalCer and IL-2 for seven days. (b) Expansion of iNKT cells from patients (n = 25) and healthy controls (n = 20). *** p < 0.001, Wilcoxon matched-pairs. (c) The same data as in (b) presented as fold expansion of iNKT cells over seven days in untreated (n = 20) and fludarabine-treated (n = 5) patients compared to controls (n = 20). Difference not significant for any comparison, Kruskal Wallis test with Dunn's post-tests.



5.3.4 Immunophenotype of iNKT cells after expansion

The CD4- subpopulation of human iNKT cells is known to exhibit a Th1-type cytokine profile, while the CD4+ subpopulation produces both IFN- γ and IL-4 upon stimulation.^{218,219}

In mice, it is the CD4- subset of iNKT cells which mediates anti-tumour immunity.²²² I assessed the CD4 status of iNKT cells following *in vitro* culture, to determine whether proliferating iNKT cells acquire a CD4+ bias.

The proportion of iNKT cells that were CD4+ increased following the culture of PBMCs with α -GalCer and IL-2 (figure 5.9a and 5.9b). This was observed in both patient and control groups, was statistically significant, and was accompanied by a reciprocal decrease in the proportion of CD8+ and double negative iNKT cells (figure 5.9c and 5.9d). The difference in the proportion of iNKT cells CD4+ between patients and controls was not statistically significant (Mann Witney test).

The culture of human iNKT cells in the presence of IL-7 or IL-15, rather than IL-2, has been reported to favour proliferation of the CD4- subset.⁴⁶³ Therefore, I compared IL-2 with two other gamma chain cytokines, IL-7 and IL-15 again, assessing the proportion of iNKT cells that were CD4+ following a seven day culture of healthy control PBMCs with α -GalCer and each cytokine (figure 5.10). I found no clear evidence that IL-7 or IL-15 favoured the proliferation of CD4- iNKT cells in the four donors studied. Rather, the final proportion of CD4+ iNKT cells appeared to be related to the starting proportion with the exception of one of the four donors, in whom the addition of low doses of IL-2 led to a decline in CD4+ iNKT cells and an increase in the CD4-CD8- iNKT cell population.

5.3.5 Effect of leukaemic cells on iNKT cell proliferation

In the experiments above, I had depleted patient PBMCs of B cells before culture. In patients with CLL, a marked excess of leukaemic cells over T cells in the blood, bone marrow and lymph nodes is typical. Leukaemic CLL cells can impair T cell function *in vitro*,⁹¹ and in a mouse model, the adoptive transfer of CD1d-expressing B cells impairs iNKT cell proliferation.²⁴⁶ Therefore, I wished to determine whether the presence of CLL cells impaired human iNKT cell proliferation *in vitro*. To determine this, I used autologous CLL cells, as allogeneic CLL cells would be likely to provide NK and T cell signals in these mixed PBMC cultures, which might complicate interpretation of the data.

First, whole PBMCs from eight patients with CLL (constituting a median of 55.2% malignant B cells, range 20.6% – 87.7%) were cultured with α -GalCer and IL-2 for 7 days (figure 5.11). An increase in iNKT cell numbers was seen in all cases, indicating that leukaemic cells do not completely prevent iNKT cell proliferation *in vitro*.

To quantify the effect of CLL cells on iNKT cell proliferation, I immunomagnetically depleted patient PBMCs of B cells before adding back positively-selected autologous CLL cells in defined numbers. The addition of CLL cells did not impair *in vitro* iNKT cell proliferation. In contrast, the addition of autologous CLL cells to B cell-depleted PBMCs at a ratio of 2:1 resulted in a significant increase in iNKT cell proliferation (p < 0.05 compared to no CLL cells added, Wilcoxon matched pairs test; figure 5.12a). The addition of CLL cells did not affect the final proportion of iNKT cells which were CD4+ (figure 5.12b).

These experiments indicate that iNKT cells from patients with CLL retain their ability to proliferate, and that under the *in vitro* conditions studied, leukaemic cells do not inhibit, and may promote, iNKT cell proliferation.
Figure 5.9 Immunophenotype of iNKT cells after expansion. Illustrative flow cytometry plots showing iNKT cell numbers and CD4 and CD8 status before (a) and after (b) 7 days of culture of healthy donor PBMCs with α -GalCer and IL-2. Numbers within gates indicate percentages.



Figure 5.9 (continued) Immunophenotype of iNKT cells after expansion. (c) Summary graph showing changes in CD4 and CD8 status with iNKT cell expansion in patients (n = 21) and controls (n = 20) (bars represent medians). (d) Differences in iNKT cell phenotype before and after expansion in patients and controls. ** p < 0.01, * p < 0.05, Wilcoxon signed rank test.



(d)



Figure 5.10 Effect of IL2, IL7 and IL15 on iNKT subsets. PBMCs from four healthy donors were cultured with 100 ng/mL α -GalCer in the presence of the indicated concentration of each cytokine for seven days. CD4 status of the proliferating iNKT cells was assessed by flow cytometry. IL-2 concentration is indicated on lower x axis; IL-7 and IL-15 concentrations are indicated on the upper x axis. The starting proportion of iNKT cells which were CD4+ is given in the box in each chart. Points represent means of triplicate analyses.



Figure 5.11 Proliferation of iNKT cells in the presence of leukaemic cells. Whole PBMCs (without prior B cell depletion) from 8 patients with CLL were cultured at 2 x10⁶/well with α -GalCer 100 ng/mL and 100 U/mL IL-2 added at 24 hours. After 7 days, wells were harvested, and iNKT cells enumerated. ** p < 0.01, Wilcoxon signed rank test.



5.3.6 Generation of iNKT cell lines

Having established that iNKT cells from patients with CLL retain the capacity to proliferate *in vitro*, I attempted to generate iNKT cell lines from patients. The purpose of this was twofold:

- 1. To provide a pure population of iNKT cells for further analyses, including for cytotoxicity assays, and
- 2. To investigate the feasibility of *ex vivo* iNKT cell expansion from patients with CLL, for potential therapeutic purposes

In order to generate iNKT cell lines, PBMCs were initially cultured in the presence of α -GalCer and IL-2 as for the proliferation experiments above. FACS was used to select cells staining positively with both the anti-iNKT cell antibody (clone 6B11) and with anti-CD3.

In a preliminary experiment, I tested various irradiated cells as 'feeder cells' to support ongoing proliferation of sorted iNKT cells in the presence of IL-2 (figure 5.13). The HLA Class I-low human lymphoblastoid cell line, C1R, was at least as effective as autologous PBMCs in leading to iNKT cell expansion. Surprisingly, an untransfected C1R cell line was as effective as one transfected to express CD1d. Nonetheless, I used the CD1d-transfected C1R (C1R-CD1d) cell line for restimulation of iNKT cells.

Four patient and four control samples were selected for generation of iNKT cell lines, based on the presence of sufficient iNKT cells after seven day culture to enable cell sorting (median 2.73% iNKT cells after culture; range 0.73% – 5.74%). Using flow cytometric cell sorting, iNKT cells were enriched to a median of 89.0% of live cells (figure 5.14). How-

Figure 5.12 Effect of addition of autologous CLL cells on iNKT cell expansion. 2×10^5 B cell depleted PBMCs from six patients with CLL were cultured in the presence of 100 ng/mL α -GalCer and 100 U/mL IL-2, with the addition of purified B cells (> 95% with the CLL phenotype CD19+CD5+) at the indicated numbers. After 7 days, iNKT cell numbers were evaluated by flow cytometry. (a) Absolute numbers of iNKT cells at day seven, according to number of CLL cells added per well. Bars represent medians, boxes represent interquartile range, and whiskers represent range. * p < 0.05, Wilcoxon matched pairs test. (b) Proportion of iNKT cells expressing CD4+, according to number of CLL cells added. Bars represent medians, boxes represent interquartile range, and whiskers represent range.



50

25

100

200

0

Ò

400

(a)

Figure 5.13 Effect of feeder cells on iNKT cell expansion. Feeder cells were irradiated to 50 Gy and washed, then added to 4×10^4 FACS-sorted iNKT cells at a 5:1 feeder:iNKT cell ratio, in the presence of α -GalCer, in duplicate. 100 U/mL IL-2 was added at 24 hours. After one week, wells were harvested, and iNKT cells enumerated by flow cytometry. None = no feeder cells added; auto-PBMC = whole autologous PBMCs; allo-PBMC = mixed PBMCs from three different healthy donors; C1R = untransfected C1R cell line; C1R-CD1d = CD1d-transfected C1R cell line.



ever, iNKT cell purity declined after two restimulations with CD1d-C1R cells. A second sort was performed at this stage, to yield iNKT cell lines of median 99.3% purity (range 95.2% - 99.8%) (figure 5.15a). After two restimulations, the most frequent contaminating cell types were CD4+ 'conventional' T cells (median 7.2% of live cells), followed by CD8+ 'conventional' T cells (median 1.1%) and NK cells (median 0.5%).

The absolute numbers of iNKT cells rose rapidly, with a median 25.8-fold increase during the seven day culture of PBMCs with IL-2 and α -GalCer and a further 44.1-fold increase over two further 2-weekly restimulations before approaching a plateau. Overall, there was a median 4186-fold increase in iNKT cell numbers by the end of the fourth restimulation, beyond which iNKT cell numbers fell, in some instances precipitously (figure 5.15b).

The immunophenotype of the iNKT cell lines generated was assessed by flow cytometry: the iNKT cell populations were overwhelmingly CD4 positive (median 99.8% of cells CD4+), and a minority expressed CD161 (figure 5.16). This contrasts with the phenotype of circulating iNKT cells, around half of which were CD4+, and the majority of which expressed CD161 (figure 3.11).

Having generated iNKT cell lines from patients with CLL and healthy controls, I proceeded to characterise their function.

Figure 5.14 Flow cytometric cell sorting of iNKT cells. Representative flow cytometry plots showing enrichment of iNKT cells using FACS, positively selecting for CD3 and 6B11 co-positive cells. Numbers represent percentage of events within iNKT cell gate (CD3+ 6B11+).



5.3.7 Cytokine profile of iNKT cell lines

To determine the cytokine profile of the iNKT cell lines generated, I performed intracellular cytokine staining for IFN- γ and IL-4 after stimulation with PMA and ionomycin (figure 5.17a). Patient and control iNKT cell lines were dual producers of both cytokines, consistent with their overwhelmingly CD4+ status (figure 5.17b).

To further assess the cytokine profile of the iNKT cell lines generated, I co-cultured iNKT cell lines with C1R-CD1d cells and either vehicle or α -GalCer (figure 5.18). Background levels of cytokine production were observed in the presence of vehicle and CD1d-transfected C1R cells, but cytokine production was markedly enhanced in the presence of α -GalCer, indicating at least a component of α -GalCer specificity.

The cytokine profiles of control- and patient-derived iNKT cell lines were similar. Both produced both Th1-type cytokines such as IFN- γ and TNF- α in combination with Th2-type cytokines such as IL-4 and IL-13. iNKT cell lines produced GM-CSF, a finding which has previously been reported.²²⁰ α -GalCer-induced IL-10 production was observed at a low level in a patient-derived cell line. The iNKT cell lines tested did not produce detectable quantities of IL-17.

5.3.8 Cytotoxicity of iNKT cell lines

Intracellular staining confirmed that iNKT cell lines expressed granzyme B (figure 5.19). An LDH release cytotoxicity assay confirmed that both a patient and a control iNKT cell line could lyse a C1R target cell, in a CD1d- and α -GalCer dependent fashion (figure 5.20).



Figure 5.16 Phenotype of iNKT cell lines. Surface immunophenotype of patient (n = 4) and control (n = 4) iNKT cell lines. Bars represent medians, error bars represent range. Difference non significant for any comparison, Mann Whitney test.



Together, these data indicate that iNKT lines derived from patients with CLL have similar functional properties to those from healthy donors: they produce both Th1 and Th2-type cytokines, exhibit intracellular granzyme B expression, and have the capacity to lyse a CD1d-expressing cell line in the presence of α -GalCer.

Figure 5.17 Intracellular staining of iNKT cell lines. The iNKT cell lines generated from three patients and three controls were studied by intracellular cytokine staining. (a) Flow cytometry plots showing Intracellular cytokine staining. (b) Cytokine expression represented as a graph. Bars represent medians, error bars represent range.



Anti-IL-4 APC



Figure 5.18 Cytokine production by iNKT cell lines. (a) α -GalCer induced cytokine production by a representative control and patient iNKT cell line. iNKT cell lines were cultured with CD1dexpressing C1R cells and either α -GalCer or vehicle. Supernatants were taken at 24 hours and cytokines assayed by bead array. nd = not detectable. (b) In a separate experiment, supernatant cytokine levels were measured after culturing three control and two patient iNKT cell lines with CD1d-expressing C1R cells in the presence or absence of α -GalCer. Dashed line represents lower limit of detection for GM-CSF. nd = not detectable.



(a)

Figure 5.19 Granzyme B expression of iNKT cell lines. Intracellular granzyme B expression of iNKT cell lines from two patients and two controls, determined by flow cytometry. Black line represents anti-granzyme B FITC; shaded histogram represents isotype control.



Figure 5.20 Cytotoxicity of iNKT cell lines. Cytotoxicity of an iNKT cell line from a patient and control against untransfected and CD1d-transfected C1R cells, in the presence and absence of a-GalCer, as assessed by LDH release assay.



C1R and vehicle

-— — — C1R and α-GalCer

5.4 Summary

In this chapter, I have assessed the function of iNKT cells from in patients with CLL.

Cytokine production by iNKT cells from patients with CLL is intact, as assessed by α -GalCer-induced IFN- γ ELISpots and by intracellular cytokine staining for IFN- γ and IL-4. Both patient and control iNKT cells express granzyme B, at a level similar to that of T cells.

Proliferation of iNKT cells from mixed PBMC cultures was achieved using α -GalCer and delayed addition of IL-2. The degree of iNKT cell expansion seen in PBMCs from patients was similar to that observed in healthy, age-matched controls. Moreover, the addition of autologous CLL cells to the cultures did not impair iNKT cell proliferation. The culture conditions employed appeared to favour expansion of the CD4+ subset of iNKT cells in both patient and control groups.

I have demonstrated a method of generating iNKT cell lines from patients with CLL, and have determined the immunophenotype and cytokine profile of these lines. The iNKT cell lines produced were overwhelmingly CD4+, and consistent with this surface phenotype, expressed both IFN- γ and IL-4 intracellularly. Cell lines from both patients and controls produced Th1 and Th2-type cytokines and GM-CSF upon exposure to α -GalCer and a CD1d-expressing cell line. IL-17 was not produced.

Finally, I have shown that both patient- and control-derived iNKT cell lines express granzyme B, and have demonstrated the capacity of a CLL patient-derived iNKT cell line to lyse a target cell line in a CD1d- and α -GalCer-dependent manner. The degree of cytotoxicity was modest in this experiment, a finding which will be discussed in section 7.2.4.5.

Together, these findings indicate that iNKT cell function is fundamentally intact in CLL. In chapter 6, I shall explore potential immunotherapeutic strategies using iNKT cells.

Chapter 6

Application of iNKT cells to CLL immunotherapy

6.1 Introduction

In chapters 3 to 5, I established that the iNKT cell and CD1d axis is essentially numerically, immunophenotypically and functionally intact in patients with CLL. In this chapter, I aim to explore potential immunotherapeutic strategies for CLL, which exploit the iNKT/CD1d axis.

There are three principal mechanisms by which iNKT cells might contribute to the immunological control of a malignancy: the transactivation of NK or other effector cells, direct cytotoxicity against tumour cells, or the provision of adjuvant activity to enhance T cell responses to peptide antigens.

Transactivation of natural killer effector cells The transactivation of NK cells by activated iNKT cells is a critical antitumour effector mechanism of α -GalCer in some murine models of cancer.^{271,272} Moreover, iNKT cell activation of NK cells has also been observed in human cell cultures *in vitro*.²⁷⁶ In human clinical trials, the administration of intravenous α -GalCer resulted in increases in circulating cytokines and NK cell activation in some individuals.^{303,404}

Direct cytotoxicity against CD1d-expressing target cells As outlined in section 1.2.5, iNKT cells can exhibit direct cytotoxicity against CD1d-expressing target cells via a number of pathways including perforin/granzyme B and Fas/FasL. Human iNKT cells have been shown to lyse allogeneic CD1d-expressing haematopoietic malignancies *in vitro*, including chronic lymphocytic leukaemia, myeloid leukaemias and T cell acute lymphoblastic leukaemia.^{229,416,417} In chapter 5 I demonstrated the feasibility of generation of iNKT cell lines from patients with CLL, raising the possibility that these might be used as effector cells. Although human iNKT cell lines can lyse allogeneic human CLL cell lines,⁴¹⁷ this has not been demonstrated in the autologous setting, where NK receptor mismatches are absent.

Provision of adjuvant activity The co-administration of α -GalCer with a protein or peptide antigen results in enhanced 'conventional' CD4+ and CD8+ T cell responses to the protein, ^{258–260} as discussed in section 1.2.6. This interaction requires CD40/CD40L, and not IFN- γ , and can result in protection against subsequent challenge with a tumour expressing the protein. ²⁵⁸ The adjuvant activity of α -GalCer has also been exploited using α -GalCer-pulsed whole tumour cells as a vaccine—this approach has proved effective against CD1d-expressing haematopoietic malignancies in preclinical studies ^{422,424,425}

In this chapter, I present a series of *in vitro* experiments using PBMCs from patients and healthy controls, aimed at exploring the potential applicability of each of these mechanisms to the immunotherapy of CLL.

6.2 Aims

The aims of the experiments presented in this chapter are to identify promising strategies for the incorporation of iNKT cells in the immunotherapy of CLL. Specifically, these experiments aim to:

- Assess the effect of α -GalCer addition to whole PBMC cultures on supernatant cytokine levels and APC activation status
- Determine the capacity of patient-derived iNKT cell lines to lyse autologous CLL cells
- Investigate the effect of $\alpha\mbox{-}Gal\mbox{Cer}$ on T cell responses to a co-administered protein antigen
- Explore the capacity of $\alpha\mbox{-}Gal\mbox{Cer-pulsed}$ CLL cells to elicit expansion of tumour-reactive T cells

6.3 Results

6.3.1 α-GalCer-induced cytokine release in whole PBMC cultures

In mice, the systemic administration of α -GalCer results in a rapid increase in circulating IFN- γ levels, which is at least partly responsible for the transactivation of NK cells and for enhancement of NK cytotoxicity.²⁷³ However, in clinical trials of intravenous α -GalCer, systemic cytokine responses and NK cell activation were restricted to patients with the highest numbers of circulating iNKT cells.^{303,404}

Having previously established that iNKT cell cytokine production is intact in patients with CLL using the highly-sensitive techniques of α -GalCer-induced IFN- γ ELISpot enumeration and intracellular cytokine staining (figures 5.3 and 5.5), I wished to determine whether the addition of α -GalCer to patient PBMCs would induce appreciable levels of cytokines in PBMC culture supernatants. I reasoned that supernatant cytokine concentrations would

depend upon the degree of α -GalCer-induced transactivation of other cell types (such as NK and T cells) as well as cytokines released by iNKT cells alone.

Cryopreserved supernatants of PBMC cultures used in the iNKT cell proliferation experiments outlined in section 5.3.3 were analysed for cytokines. In a preliminary experiment, PBMCs from three healthy donors aged 26 to 34 years old were cultured with α -GalCer or vehicle. IL-2 100U/mL was added at 24 hours. Supernatants were analysed at various timepoints (figure 6.1). In the two donors with higher numbers of iNKT cells, the addition of α -GalCer resulted in an increase in supernatant concentrations of IFN- γ and IL-13 beyond 24 hours.

The effect of α -GalCer addition on the cytokine levels of supernatants from cultured patient and control PBMCs was examined (figure 6.2). The addition of α -GalCer led to a significant increase in supernatant IFN- γ in controls only: an α -GalCer-dependent induction of IFN- γ was not observed in patient samples. The addition of α -GalCer did not lead to significant induction of GM-CSF, TNF- α , IL-4, IL-10 or IL-13 in either patient or control samples.

Several explanations for this lack of a cytokine response in patients could be proposed, including the lower relative numbers of iNKT cells (median 0.01% of T cells) seen in patients than in age-matched healthy controls (median 0.02% of T cells), the possibility of inhibitory effects of residual CLL cells in the B-cell depleted patient samples, or defects in other cell types, such as NK cells, within the CLL patient PBMCs. Nonetheless, in this *in vitro* setting, the undirected administration of α -GalCer did not elicit a major cytokine response in mixed patient PBMCs.

6.3.2 Effect of α-GalCer on APC co-stimulatory markers

In mice, the systemic administration of α -GalCer results in upregulation of co-stimulatory markers on splenic DC.²⁵⁸ I wished to determine whether the addition of α -GalCer to whole human PBMC cultures would result in a similar effect. To do so, I examined the expression of activation markers on monocytes and B cells, which are more frequent and therefore more readily analysed than DCs.

The incubation of human PBMCs with α -GalCer led to significantly increased upregulation of monocyte MHC Class II only in a donor with a high percentage of iNKT cells (figure 6.3a). In donors with smaller proportions of iNKT cells, upregulation of the activation markers CD80, CD86 and class II was inconsistent (figure 6.3b).

In mice, the systemic administration of α -GalCer leads to upregulation of CD86 on circulating B cells. I investigated this using human PBMCs. Examining PBMCs from a healthy donor with a high number of iNKT cells, the culture of PBMCs with α -GalCer led to slight upregulation of CD86 and CD54 on B cells at 60 hours. CD80, CD25, CD40 and HLA-DR did not appear to be upregulated under these conditions (figure 6.4a to 6.4f).

These data suggest that, although α -GalCer might be capable of inducing human monocyte and B cell activation when iNKT cells are present in sufficiently high proportions, this phenomenon is unlikely to be observed in mixed PBMC cultures from most humans in whom the frequency of iNKT cells is low. Given the lack of effect of α -GalCer on monoFigure 6.1 Detection of cytokines in supernatants of cultured PBMCs. 2×10^6 PBMCs from each of three healthy donors, were cultured with α -GalCer or vehicle and 100 U/mL IL-2. Supernatants were taken at the indicated times, and cytokine concentrations evaluated by cytokine bead array. Percentages in the charts indicate the number of iNKT cells (as a percentage of all T cells) before culture. Dashed lines represent lower limit of detection of assay.





Figure 6.3 Effect of α **-GalCer on monocyte activation in PBMC cultures.** 2 x 10⁶ PBMCs from each of four healthy donors were cultured with 100 ng/mL α -GalCer or vehicle. After 18 hours, cells were harvested and APC maturation assessed by flow cytometry for monocyte CD80, CD86 and HLA-DR (MHC Class II) expression. (a) Histograms showing monocyte expression levels after culture with vehicle (shaded area) or α -GalCer (black line) from a healthy donor with a high number of iNKT cells. (b) Graphs showing monocyte expression levels from four different healthy donors. Percentages beneath graph represent iNKT cell numbers as a % of all iNKT cells.

(a)



Figure 6.4 Effect of addition of α **-GalCer on B cells.** PBMCs from a healthy donor with a high number of iNKT cells (1.3% of T cells) were cultured in the presence of vehicle or α -GalCer, and expression of (a) CD25, (b) CD40, (c) CD54, (d) CD80, (e) CD86, and (f) HLA-DR on B cells assessed by flow cytometry at the indicated time points. Numbers in histograms indicate net median fluorescence intensity of the indicated marker on B cells.



cyte HLA-DR expression in healthy donors with an iNKT cell frequency of 0.03% of T cells, I considered it unlikely that I would be able to demonstrate this effect in cultures of PBMCs from CLL patients, where iNKT cells constitute a median of just 0.01% of T cells (figure 3.7).

6.3.3 Activation of CLL cells by patient-derived iNKT cell lines is not α -GalCer-dependent

Having shown that α -GalCer-induced activation of B cells was weak in mixed PBMC cultures, even in a donor with a relatively high proportion of iNKT cells (figure 6.4), I wished to establish whether purified iNKT cells could interact directly with CLL cells to result in CLL cell activation.

Patient CLL cells were co-cultured with autologous patient-derived iNKT cell lines at various ratios in the presence or absence of α -GalCer. Activation status of the CLL cells was determined by flow cytometry. The results indicate that co-culture of CLL cells with autologous iNKT cell lines results in the upregulation of CD86 and CD80 on the leukaemic cells, but that this effect was not dependent upon the addition of α -GalCer (figure 6.5). Contamination with the IL-2 used during iNKT cell line generation is unlikely to explain this, as the iNKT cells had been washed four times during the iNKT cell cryopreservation, thawing and viable cell counting steps.

One explanation for the α -GalCer-independent activation of CLL cells by autologous iNKT cell lines may be via a CD40/CD40L interaction. B cells express CD40, while CD40L (CD154) is upregulated by T cells upon activation and exposure to IL-2.⁴⁶⁴ Flow cytometric analysis confirmed that leukaemic CLL cells expressed CD40 (figure 6.6a), and that a patient-derived iNKT cell line expressed surface CD40L (CD154) (figure 6.6b).

6.3.4 Lysis of autologous CLL cells by patient-derived iNKT cell lines

Healthy donor iNKT cell lines can lyse allogeneic CLL cell lines.⁴¹⁷ Having demonstrated that iNKT cell lines can be derived from patients with CLL (figure 5.15), and that a patient-derived iNKT cell line can lyse a CD1d-positive target cell (figure 5.20), I wished to determine whether patient-derived iNKT cells can lyse autologous CLL cells. Unlike allogeneic CLL cells or the MHC Class I-low C1R cell line, autologous leukaemic cells would not provide activating NK signals, and may provide inhibitory NK signals, to iNKT cells.

First, I examined killing of α -GalCer pulsed target cells in whole PBMC cultures. I used the VITAL assay, which employs of differential fluorescent labelling to determine the relative proportions of α -GalCer and vehicle-pulsed CD1d-expressing target cells remaining following co-culture with whole PBMCs as effector cells.⁴³⁹

The VITAL assay enabled discrimination between α -GalCer-pulsed and unpulsed target cells (figure 6.7a). When assessed using two healthy donors, up to 16% specific lysis of α -GalCer-pulsed C1R cells was observed at an effector:target ratio of 20:1 (figure 6.7b).

Figure 6.5 Activation of CLL cells by iNKT cell lines is α -GalCer-independent. 2 x 10⁴ immunomagnetically-selected patient CLL cells (> 95% CD5+ CD19+) were cultured alone, or in the presence of the indicated numbers of cryopreserved, washed, viable autologous iNKT cells (> 98% CD3+ 6B11+), in the presence of α -GalCer or vehicle. After 20 hours, cells were harvested, and CLL cell expression of CD80, CD86 and MHC Class II determined by flow cytometry, gating on CD5+ CD19+ cells.



Figure 6.6 Expression of CD40 on CLL cells and CD40L on iNKT cells. (a) Immunomagnetically-selected patient CLL cells were stained with anti-CD40 antibody or isotype control, gated on CD19+ CD5+ cells. (b) A patient-derived iNKT cell line was stained with anti-CD154 (anti-CD40L) antibody or isotype control, gated on CD3+ 6B11+ cells. Solid black lines represent anti-CD40 or anti-CD154 antibody stains; filled gray histograms represent matched isotype control stains.



The low maximal specific lysis percentages observed may be because iNKT cells account for a small proportion of nucleated cells within whole PBMCs. Thus the 'effective' effector:target ratios were much lower than the 'nominal' effector:target ratios of up to 20:1.

I therefore focussed on using iNKT cell lines rather than whole PBMCs as effector cells, having already demonstrated that iNKT cell lines could lyse a CD1d-transfected target cell at effector:target ratios of 3:1 *in vitro* (figure 5.20).

A patient-derived iNKT cell line was incubated with autologous α -GalCer-pulsed or unpulsed CLL cells at various effector:target ratios, and target cell lysis determined by LDH release. This indicated that a patient-derived iNKT cell line was capable of lysis of autologous CLL cells in an α -GalCer-dependent manner (figure 6.8a). A further experiment showed inconsistent results, with α -GalCer-dependent lysis of autologous and allogeneic CLL cells by patient-derived iNKT cells in three out of six target and effector cell pairs (figure 6.8b).

The degree of specific lysis was low, and varied between target and effector cell pairs (figure 6.8b), lysis of autologous CLL cells being demonstrated with one donor (P026) but not another (P060). This may be due to differences in susceptibility of the three patients' CLL cells to lysis, differences in function of the two patient-derived iNKT cell lines, and to differences in NK receptor mismatches in each effector:target pair. Availability of the patient-derived iNKT cell lines limited my ability to explore these factors further: compared with the lymphoblastoid C1R cell line, CLL cells exhibited a lower maximal LDH release. This necessitated relatively large numbers of CLL cell target cells, and therefore of iNKT cell effector cells, to obtain detectable LDH release signals. **Figure 6.7 VITAL assay to assess iNKT cell target killing in whole PBMCs.** CD1d-transfected C1R target cells were pulsed with α -GalCer and labelled with CTO dye, or were unpulsed and labelled with CSFE. 5 x10³ of each of the α -GalCer-pulsed and unpulsed target cells were added to whole PBMCs from two healthy donors at the indicated effector:target cell ratios, in duplicate. After a six hour incubation, the ratio of α -GalCer-pulsed to vehicle-pulsed target cells was determined by flow cytometry, and was used to determine the percentage specific lysis of α -GalCer-pulsed targets. (a) Example of flow cytometry plots, demonstrating changing ratios of α -GalCer pulsed and vehicle-pulsed target cells. (b) Specific lysis of α -GalCer pulsed target cells by PBMCs from two healthy donors. The percentage above each graph represents the frequency of iNKT cells (as a proportion of T cells) in that donor's PBMCs



(b)



Figure 6.8 Lysis of CLL cells by patient-derived iNKT cell lines. Patient-derived iNKT cell lines were incubated for six hours at 37 °C with 3 x10⁴ vehicle- or α -GalCer-pulsed CLL cells at the indicated effector:target ratios. Lysis was assessed by LDH release assay. (a) Lysis of autologous CLL cells by a patient-derived iNKT cell line. Error bars represent standard error of means of triplicate repeats. (b) Separate experiment showing lysis of autologous and allogeneic CLL cells by two patient-derived iNKT cell lines, all at a 5:1 effector:target ratio. Bars and error bars represent means and standard error of means of triplicate repeats. p < 0.05 for α -GalCer treatment effect, 2-way ANOVA.



(b)



(a)

These data suggest that while patient-derived iNKT cell lines may be able to lyse CLL cells in the presence of α -GalCer *in vitro*, the degree of lysis is variable, and is not efficient, requiring high effector:target cell ratios. Moreover, an earlier experiment had indicated that following a twenty-hour co-culture of CLL cells with iNKT cells at effector:target ratios of up to 4:1, viable target CLL cells remained readily detectable by flow cytometry (figure 6.5).

6.3.5 Expression of MHC Class I, but not CD1d, is upregulated by CLL cell exposure to IFN- γ

Killing of CD1d-expressing C1R cells by an iNKT cell line was detectable at a lower effector:target ratios than killing of CLL cells (figure 5.20, figure 6.8a). Expression levels of CD1d are higher on the CD1d-transfected C1R cell line than on CLL cells (figure 6.9a). As the capacity of IFN- γ to induce upregulation of MHC Class I has long been recognised, ⁴⁶⁵ I hypothesised that expression of the MHC Class I-like molecule CD1d would also be increased upon culture of CLL cells in the presence of IFN- γ .

To test this hypothesis, CLL cells were cultured in the presence of IFN- γ or vehicle for 48 hours. Significant upregulation of MHC Class I was observed in the presence of IFN- γ (figure 6.9b). In contrast, CLL cells did not upregulate CD1d upon exposure to IFN- γ (figure 6.9c).

6.3.6 Effect of α-GalCer on vaccine recall responses

In mice, α -GalCer can act as a powerful adjuvant to protein vaccination.²⁵⁸ I wished to explore similar potential adjuvant effects in human PBMC samples. To do this, I examined the effect of α -GalCer on vaccine recall responses using influenza and tetanus/diphtheria toxoid vaccines, to which many healthy donors have previously been exposed. Proliferation of PBMCs, determined by ³H thymidine incorporation, was used to assess the vaccine recall response.

The addition of α -GalCer significantly enhanced PBMC proliferation in response to influenza vaccine in an influenza-vaccinated donor with a high number of iNKT cells, but not in an influenza-vaccinated donor with a lower proportion of iNKT cells (figure 6.10a). Similar findings were observed when the tetanus/diphtheria toxoid vaccine recall response was assessed (figure 6.10b).

This ³H thymidine incorporation assay has two important limitations. First, it does not indicate which cells are proliferating: α -GalCer could be acting as an adjuvant to enhance proliferation of conventional CD4+ and CD8+ T cells in response to the vaccine, or the vaccine could act as an adjuvant to enhance proliferation of iNKT cells in response to α -GalCer. Second, the assessment of proliferation by ³H thymidine incorporation is dependent upon the timing of thymidine addition: if thymidine is added before or after the peak of the proliferative response, the response may be underestimated or missed.

To overcome both of these limitations, a flow cytometric assay incorporating CFSE dilution was used to measure proliferation. This allows immunophenotypic identification **Figure 6.9 Effect of IFN-** γ **on CLL MHC Class I and CD1d expression.** (a) CD1d expression levels on CLL cells and CD1d-transfected and on untransfected C1R cell lines. (b), (c) Immunomagnetically purified CLL cells (\geq 95% CD19+) were cultured for 48 hours with 1000 U/mL IFN- γ or vehicle control. Cells were harvested, and expression of MHC Class I and CD1d was determined by flow cytometry. p < 0.0001 for IFN- γ effect on CLL cell MHC Class I expression (two-way ANOVA); difference not significant for IFN- γ effect on CLL cell CD1d expression.



Figure 6.10 Effect of *α***-GalCer on vaccine recall responses.** The PBMCs of vaccinated healthy donors were cultured for five days with human influenza vaccine (a) or diphtheria and tetanus toxoid vaccine (b) at the indicated concentrations, in the presence of α-GalCer or vehicle. ³H thymidine was added for the final 12 hours of culture, and proliferation was assessed by ³H thymidine incorporation. Numbers above each graph represent donor iNKT cell numbers (as a percentage of all T cells). * p < 0.05, ** p < 0.01, unpaired T test.







(a)

of the divided cells. Moreover, once a cell has divided, the CFSE content of its progeny remains low even if the cells are no longer actively proliferating at the time of analysis.

This assay indicated that the addition of α -GalCer resulted in an increase in the proliferation of both 'conventional' T cells and iNKT cells (figure 6.11a). The experiment was repeated using an α -GalCer-loaded CD1d tetramer instead of the 6B11 antibody to identify iNKT cells, to exclude the possibility that non-invariant but CD1d-reactive iNKT cells were responsible for the apparent increase in divided 'conventional' T cells in the presence of α -GalCer. Application of the α -GalCer-loaded CD1d tetramer confirmed the enhancement of 'conventional' T cell proliferation in the presence of α -GalCer (figure 6.11b).

Interestingly, these experiments indicated that the presence of the influenza vaccine enhanced the proliferation of iNKT cells at least as much as the presence of α -GalCer enhanced the proliferation of 'conventional' T cells, suggesting that co-administration of peptide and glycolipid antigens may result in a reciprocal adjuvant effect.

Having demonstrated the adjuvant effect of α -GalCer on an influenza vaccine recall response in a healthy donor with a high proportion of iNKT cells, I assessed this in healthy donors and patients with lower proportions of iNKT cells. Using a titrated concentration of the influenza vaccine, I assessed the effect of α -GalCer addition on recall responses in vaccinated healthy controls and patients with CLL (figure 6.12). This did not yield consistent results, with a statistically significant increase in the recall response after addition of α -GalCer in one of six controls, and a decrease in one of four patients.

These data suggest that although α -GalCer can act as an adjuvant to support vaccine recall responses when iNKT cells are present at a high frequency, this effect is not observed in whole PBMCs from many human donors. The very low frequency of iNKT cells typically present in mixed PBMC cultures may account for this finding.

6.3.7 Pulsing CLL cells with α -GalCer induces allogeneic iNKT cell expansion

Given the low proportion of iNKT cells in patient PBMCs, a strategy to both expand iNKT cells and to deliver tumour antigen would appear attractive. Since CLL cells are readily obtained and express CD1d, I explored the possibility of loading CLL cells with α -GalCer directly, utilising the tumour cells themselves to expand iNKT cell numbers and deliver tumour antigen.

CLL cell lines can present α -GalCer to allogeneic iNKT cell lines,⁴¹⁷ but the capacity of CLL cells to induce iNKT cell proliferation in mixed PBMC cultures, where iNKT cells are present at very low frequencies, has not been established.

Immunomagnetically purified CLL cells were pulsed with α -GalCer or vehicle, washed, then added to healthy donor PBMCs. Expansion of iNKT cells was observed in eight of nine responder/stimulator pairs tested (figure 6.13a and 6.13b), indicating that CLL cells are able to retain α -GalCer and to elicit proliferation of allogeneic iNKT cells within mixed PBMC cultures.

Figure 6.11 Immunophenotype of divided cells after α -GalCer-enhanced vaccine recall response. PBMCs from an influenza-vaccinated donor with a high percentage of iNKT cells (1.3% of T cells) were pulsed with CFSE then cultured for five days in the presence of influenza vaccine, α -GalCer, both, or vehicle alone. After harvest, surface stains were applied, and flow cytometry performed, gating on live CD3+ lymphocyte events. CFSE low, 6B11+ or tetramer+ cells represent divided iNKT cells. CFSE low, 6B11- or tetramer- cells represent divided 'conventional' T cells. (a) iNKT cells identified using anti-iNKT cell antibody (clone 6B11). (b) Separate experiment, identifying iNKT cells using α -GalCer-loaded CD1d tetramer. Numbers on flow cytometry plots indicate percentage of events within each quadrant.



CFSE

Figure 6.12 Influenza vaccine recall responses with and without α -GalCer. 2 x10⁵ B-cell depleted PBMCs from vaccinated patients and controls were cultured for five days in the presence of 2009 influenza vaccine (1:1000 dilution) with 100 ng/mL α -GalCer or vehicle. ** p < 0.01, * p < 0.05, two-tailed T test of triplicate repeats.



Figure 6.13 Proliferation of allogeneic iNKT cells in response to α -GalCer pulsed CLL cells. Immunomagnetically-selected B cells from patients with CLL ($\geq 95\%$ CD5+CD19+) were pulsed overnight with vehicle or α -GalCer, then washed four times before addition at 4 x 10⁴/well to 2 x10⁵ allogeneic healthy donor PBMCs. After five days, iNKT cell proportions were assessed by flow cytometry. (a) Representative flow cytometry plots demonstrating increased iNKT cell numbers after addition of α -GalCer-pulsed CLL cells. (b) Before/after graph showing results for nine patient and healthy donor pairs. ** p < 0.01, Wilcoxon matched pairs test.



155

6.3.8 Autologous α -GalCer-pulsed CLL cells induce proliferation of iNKT cells

Functional NK receptors are expressed by iNKT cells, and NK receptor mismatches can lead to greater iNKT cell proliferation in the allogeneic compared to the autologous setting.²²⁸ Therefore, I wished to establish whether α -GalCer-pulsed CLL cells can support the proliferation of *autologous* iNKT cells proliferation, without the potential effect of NK receptor mismatches.

CLL cells were pulsed with α -GalCer or vehicle and used to stimulate autologous B-cell depleted PBMCs once weekly, in the presence of 1 ng/mL IL-7. After three restimulations, iNKT cells were enumerated. Pulsing CLL cells with α -GalCer resulted in iNKT expansion in six out of eight patients tested (figure 6.14a and 6.14b). This indicates that α -GalCer-pulsed CLL cells can support iNKT cell proliferation in the autologous, as well as in the allogeneic, setting.

6.3.9 α-GalCer treatment of CLL cells leads to increased proliferation of allogeneic PBMCs

Unmodified CLL cells are poor stimulators in autologous and allogeneic mixed lymphocyte reactions^{466,467} As concomitant activation of iNKT cells can enhance the human *in vitro* alloresponse,⁴⁶⁸ I explored whether α -GalCer treatment could enhance the *in vitro* proliferation of PBMCs induced by allogeneic CLL cells.

CLL cells were purified by FACS to high purity, to reduce the possibility that residual monocytes or myeloid DCs may be mediating the α -GalCer effect. After overnight treatment with vehicle or α -GalCer, CLL cells were irradiated and added to healthy donor PBMCs, and the proliferative response was measured by thymidine incorporation at day five. In five of six stimulator/responder pairs tested, α -GalCer treatment of leukemic cells resulted in significantly increased proliferation of responding PBMCs (figure 6.15; p < 0.001 for α -GalCer treated versus untreated comparison, two-way ANOVA).

6.3.10 Effect of α -GalCer treatment is additional to that of TLR 7/8 ligation

TLR agonists can co-operate with α -GalCer to further enhance antigen-specific T cell responses *in vivo*.³¹¹ CLL cells express TLR 7,⁴⁶⁹ and a TLR 7 agonist leads to co-stimulatory marker up-regulation on leukemic cells and enhancement of their allostimulatory capacity.⁴⁷⁰ I confirmed that treatment of CLL cells with the TLR 7/8 agonist resiquimod leads to upregulation of CD86 (figure 6.16a and 6.16b). Furthermore, culture of CLL cells with resiquimod, but not with α -GalCer, resulted in upregulation of MHC class I and II on leukemic cells (figure 6.17a and 6.17b).

The proliferative responses of healthy donor PBMCs against CLL cells treated with vehicle, α -GalCer, resiquimod or both compounds were assessed. Results from two responders with a high frequency of iNKT cells indicated that the α -GalCer-induced increase in proliferation was additional to that of resiquimod (figure 6.18a). In contrast, results from

Figure 6.14 α -GalCer-pulsed CLL cells support proliferation of autologous iNKT cells. Immunomagnetically selected B cells from patients with CLL (\geq 95% CD5+CD19+) were pulsed overnight with α -GalCer or vehicle, irradiated, washed four times, then cryopreserved. Aliquots of 1 x10⁵ to 1 x10⁶ CLL cells were thawed, washed, and added back to 1 x10⁶ autologous B cell-depleted PBMCs (< 5% residual CD19+ cells), with the addition of 1 ng/mL IL-7. After two further weekly stimulations, flow cytometry was performed to assess iNKT cell numbers. (a) representative flow cytometry plots from three patients, demonstrating proliferation of iNKT cells in response to autologous α -GalCer-pulsed CLL cells. (b) iNKT cell numbers following stimulation with autologous vehicle- or α -GalCer-pulsed CLL cells in eight patients. * p < 0.05, Wilcoxon matched pairs test.



Figure 6.15 α -GalCer treatment of CLL cells leads to increased proliferation of allogeneic PBMCs. FACS-purified CLL cells (> 98% CD19+CD5+) were pulsed overnight with α -GalCer or vehicle, irradiated to 50 Gy, washed four times, then added to healthy donor PBMCs for five days. ³H thymidine was added during the final 12 hours of culture and ³H thymidine incorporation determined by scintillation counter. p < 0.001 for α -GalCer treatment effect, 2-way ANOVA.


Figure 6.16 Upregulation of CD86 on CLL cells by resiquimod. Immunomagnetically selected CLL cells (\geq 95% CD19+CD5+) were cultured for 24 hours in the presence of the indicated concentration of resiquimod before flow cytometric analysis of CD86 expression. (a) Representative flow cytometry histogram showing CD86 expression on CLL cells from a single patient. (b) CD86 expression on CLL cells from four patients.



Figure 6.17 Resiquimod, but not α -GalCer, leads to upregulation of MHC Class I and II on CLL cells. Immunomagnetically-selected B cells from six patients with CLL ($\geq 95\%$ CD5+CD19+) were cultured for 48 hours with 100 ng/mL α -GalCer, 1 μ g/mL resiquimod or vehicle in duplicate, before assessment of MHC Class I and Class II expression by flow cytometry. (a) MHC Class I expression. p < 0.001 for vehicle versus resiquimod and for α -GalCer versus resiquimod comparisons, not significant for vehicle versus α -GalCer comparison, two-way ANOVA with Bonferroni post-tests. (b) MHC Class II expression. p < 0.001 for vehicle versus resiquimod and for α -GalCer comparison, two-way ANOVA with Bonferroni Post-tests.





two responders with low iNKT cell frequencies showed no effect of α -GalCer treatment of CLL cells.

Together, these results suggest that α -GalCer enhancement of allogeneic PBMC proliferation in response to CLL cells is independent of co-stimulatory marker upregulation, and requires sufficient iNKT cells. This is consistent with a published finding that iNKT cells can augment the human *in vitro* alloresponse.⁴⁶⁸

6.3.11 Increased allogeneic PBMC proliferation in response to α-GalCerpulsed CLL cells is primarily due to iNKT cell proliferation

Given the observation that in vaccine recall responses, α -GalCer and a protein antigen could co-operate to enhance the proliferation of both iNKT and 'conventional' T cells (figure 6.11), I aimed to separately assess the proliferation of 'conventional' T and iNKT cells in response to resiquimod and α -GalCer-treated CLL cells.

CLL cells from three patients were treated with α -GalCer, resiquimod, both or vehicle, were washed, then added to CFSE-labelled healthy donor PBMCs as responder cells. The proliferation of T cell subsets was assessed by CFSE dilution after five days. This experiment indicated that, as previously demonstrated (figure 6.13), the addition of α -GalCer resulted in enhanced proliferation of iNKT cells. The addition of resiquimod led to enhanced proliferation of 'conventional' CD4+ and CD8+ T cells. However, α -GalCer treatment of CLL cells did not consistently result in enhanced proliferation of 'conventional' T cells in this short term culture (figure 6.19).

6.3.12 Autologous α -GalCer-pulsed CLL cells induce proliferation of conventional T cells

Leukaemic cells treated with α -GalCer can induce proliferation of autologous iNKT cells (figure 6.14). To determine whether proliferation of autologous 'conventional' T cells was also enhanced by α -GalCer treatment of leukaemic cells, vehicle- or α -GalCer-treated irradiated CLL cells were added back to B cell-depleted, CFSE-labelled PBMCs and cultured for 5 days in the absence of exogenous cytokines.

As expected, divided iNKT cells (CD3+, CD1d tetramer+, CFSE low) were detected only where CLL cells had been treated with α -GalCer (figure 6.20a). However, as well as inducing iNKT cell proliferation, α -GalCer-treatment of CLL cells resulted in a modestly increased proportion of divided conventional T cells (CD3+, CD1d tetramer negative, CFSE low) in seven out of ten patients tested (p < 0.05; figure 6.20b). These divided non-CD1d restricted T cells comprised both CD4+ and CD8+ cells (constituting a median of 43% and 24% of divided T cells in the α -GalCer-treated samples, respectively).

These data confirm that α -GalCer-treated CLL cells induce proliferation of autologous iNKT cells. Moreover, they suggest that α -GalCer treatment of autologous leukaemic cells leads to enhanced proliferation of 'conventional' T cells.

Figure 6.18 Effect of α -GalCer on allogeneic PBMC proliferation in response to CLL cells is additional to that of resiquimod. FACS-purified CLL cells (> 98% CD19+CD5+) were pulsed overnight with α -GalCer, resiquimod, both or vehicle, γ -irradiated to 50Gy, washed four times, then added at 4 x10⁴/well to 2 x10⁶/well healthy donor PBMCs as responders for five days, in triplicate. ³H thymidine was added during the final 12 hours of culture and ³H thymidine incorporation determined by scintillation counter. (a) Results from two responders with a high frequency of iNKT cells (median 0.69% of T cells). (b) Results from two responders with a low frequency of iNKT cells (median 0.03% of T cells). Solid white bars represent untreated CLL cells, solid gray bars represent α -GalCer-treated CLL cells; hatched bars represent resiquimod-treated CLL cells; solid black bars represent CLL cells treated with both α -GalCer and resiquimod. *** p < 0.001, ** p < 0.05, two-way ANOVA.



Stimulator/responder pair

(a)

vehicle, irradiated to 50 Gy, washed four times, then added at 4×10^4 /well as stimulators to 2×10^5 /well CFSE-labelled healthy donor PBMCs as responders for five days, in duplicate. Flow cytometry was used to determine the numbers of divided iNKT and 'conventional' CD4+ and CD8+ T Figure 6.19 Increased allogeneic PBMC proliferation in response to α -GalCer treatment of CLL cells is due to iNKT cell, not 'conventional' T cell, proliferation. Immunomagnetically selected CLL cells ($\geq 95\%$ CD19+CD5+) were pulsed overnight with α -GalCer, resignimod, both or cells for each stimulator/responder pair.



Figure 6.20 α-GalCer-pulsed CLL cells may support proliferation of 'conventional' T cells. Immunomagnetically selected B cells from patients with CLL (≥ 95% CD5+CD19+) were pulsed overnight with α-GalCer or vehicle, irradiated, washed four times, and added at 1 x10⁶/well to 5 x10⁵ CFSE-pulsed autologous B cell-depleted PBMCs (< 5% residual CD19+ cells). After five days, flow cytometry was performed to identify divided (CFSE-low) iNKT (CD19- CD3+ CD1d tetramer+) and 'conventional' T cells (CD19- CD3+ CD1d tetramer-). (a) representative flow cytometry plots, demonstrating enhanced proliferation of non-CD1d restricted cells in response to autologous α-GalCer-pulsed CLL cells. (b) numbers of divided 'conventional' T cells following stimulation with autologous vehicle- or α-GalCer-pulsed CLL cells in ten patients. * p < 0.05, Wilcoxon matched pairs test.



6.4 Summary

In this chapter, I have explored the potential utility of the iNKT cell/CD1d axis in the immunotherapy of CLL using a series of *in vitro* assays. I have investigated several potential methods of exploiting this axis:

Administration of α -GalCer to induce a cytokine response In contrast with findings using a healthy donor with a high frequency of iNKT cells, I was unable to demonstrate significant α -GalCer-induced cytokine production within the supernatants of PBMCs from patients with CLL. This may be because due to low frequency of iNKT cells in most patients. In a study using PBMCs from healthy human donors, the addition of α -GalCer did not result in marked IFN- γ production or NK cell activation unless previously-cultured activated iNKT cells were added to the cultures to a level of 0.1% of all cells.⁴⁰²

Administration of α -GalCer to induce APC maturation While the administration of α -GalCer led to upregulation of B cell and APC co-stimulatory molecules in a donor with a high frequency of iNKT cells, this effect was not consistently observed in donors with frequencies of iNKT cells closer to those of patients with CLL *in vitro*.

Direct cytotoxicity against CLL cells by iNKT cells Killing of allogeneic CLL cells by iNKT cell lines has previously been demonstrated. ⁴¹⁷ I endeavoured to demonstrate this in the autologous setting, where potential activating NK signals are not present. While I was able to demonstrate lysis of autologous CLL cells by a patient-derived iNKT cell line, these experiments were limited by the number of effector iNKT cells available — large numbers of target cells were needed to detect lysis in the assay used, and high effector:target ratios were required. The lysis of CLL cells by iNKT cell lines appeared less efficient than lysis of the CD1d-transfected C1R cell line, which expresses CD1d at a much higher level. I therefore assessed the effects of IFN- γ on CLL cell CD1d expression. I found that, although IFN- γ resulted in increased CLL cell MHC class I expression, it did not result in upregulation of CD1d.

 α -GalCer as an adjuvant to protein vaccines To investigate the potential role of α -GalCer as an adjuvant to a protein vaccine, I assessed the impact of α -GalCer on vaccine recall responses. These experiments demonstrated that α -GalCer can enhance the vaccine recall response within whole PBMCs when iNKT cells are present at a high frequency, leading to enhanced proliferation of non-CD1d restricted T cells as well as of iNKT cells.

The demonstration of an adjuvant effect using human PBMCs is important, as the efficacy of α -GalCer-treated tumour cells or α -GalCer-treated tumour-loaded DCs as an immuno-therapy relies upon the enhancement of peptide-specific CD4+ and CD8+ T cell responses via DC activation (see section 1.3.3.3).^{258,419,424,425}

The restriction of the adjuvant effect to a donor with a high iNKT cell percentage is not surprising, given the low frequency of iNKT cells in the majority of individuals, and the limited cell numbers in each *in vitro* culture well. Nonetheless, it is possible that the low

frequency of iNKT cells in the majority of healthy donors, and particularly in older adults with CLL, may limit any adjuvant effect in clinical practice.

Effect of α -GalCer pulsed CLL cells in allogeneic setting As CLL cells themselves express CD1d, pulsing CLL cells with α -GalCer offers a simple method for simultaneously delivering tumour antigen and the glycolipid adjuvant. I demonstrated that washed, irradiated, α -GalCer-pulsed CLL cells support the expansion of iNKT cells in both allogeneic and autologous healthy donor PBMCs.

Combination of \alpha-GalCer with resiquimod The TLR 7/8 ligand resiquimod results in enhanced CD86 and MHC class I and II expression on CLL cells. CLL cells treated with both α -GalCer and resiquimod, elicited an enhanced alloresponse compared to the cells treated with each alone. The effect of α -GalCer treatment of CLL cells was restricted to responders with a higher frequency of iNKT cells. When the proliferation of iNKT and T cells was examined by CFSE dilution, the effects of α -GalCer (enhancing iNKT cell proliferation) and resiquimod (enhancing proliferation of non CD1d-restricted CD4+ and CD8+ T cells) appeared to be independent, without evidence of synergy.

Again, whether or not the restriction of the α -GalCer treatment effect to donors with a high iNKT cell frequency *in vitro* would limit the applicability of iNKT cell-directed immunotherapies in clinical practice is not clear.

Effect of α -GalCer pulsed CLL cells in autologous setting Using CFSE dilution, I was able to confirm that α -GalCer-pulsed CLL cells induced expansion of autologous iNKT cells within PBMC cultures. Moreover, compared to vehicle-treated CLL cells, α -GalCer-treated CLL cells appeared to elicit enhanced proliferation of 'conventional' non CD1d-restricted T cells, although effect was small.

Together, these findings offer some insight into the uses and limitations of the iNKT cell and CD1d axis in the immunotherapy of CLL. Many of the effects of *in vitro* iNKT cell stimulation were limited by the low frequency of iNKT cells present in most healthy donor and patient PBMC samples. However, when iNKT cells were present at sufficiently high frequencies, α -GalCer could induce cytokine release, induce upregulation of co-stimulatory markers on APCs, and augment T cell proliferative responses to a protein vaccine. Importantly, α -GalCer-treated CLL cells were able to induce proliferation of *autologous* iNKT cells, and α -GalCer treatment led enhanced proliferation of 'conventional' CD4+ and CD8+ T cells.

The findings presented in this chapter, their relation to the published literature, and their implications for the exploitation of iNKT cells in the immunotherapy of CLL, will be discussed in chapter 7.

Chapter 7

Discussion

7.1 Introduction

In the course of this experimental programme, I have described the iNKT cell and CD1d axis in patients with CLL in numeric, immunophenotypic and functional terms, and have performed a series of *in vitro* experiments aimed at identifying ways to exploit this axis for the immunotherapy of CLL.

In this chapter, I shall summarise these results and discuss their significance and limitations, with reference to the published literature. I shall then place the findings in the broader context of immunotherapy for cancer, outlining the ideal characteristics of, and factors associated with, a successful immunotherapy. The potential benefits and drawbacks of exploiting the iNKT cell/CD1d axis for the cellular immunotherapy of cancer will be discussed. Finally, I shall outline a potential immunotherapeutic strategy for the treatment of CLL that makes use of the iNKT cell and CD1d axis.

7.2 Summary and discussion of results

7.2.1 Numbers and immunophenotype of iNKT cells

In section 3.3.4, I enumerated, and determined the immunophenotype of, iNKT cells in patients with CLL and in healthy age-matched controls. Multiparameter flow cytometry was chosen for this purpose, as it allows the simultaneous analysis of multiple surface and intracellular markers on a 'per cell' basis. However, the very small number of iNKT cells present in most donors (a median of 0.01% of T cells in patients) posed technical challenges to their flow cytometric identification. Optimising detection of these rare events required a number of measures, including the immunomagnetic depletion of malignant B cells before analysis, the inclusion of channels to exclude non-viable and leukaemic B cells, the use of a bright fluorophore and a highly specific antibody for iNKT cell identification, the application of a stringent gating strategy during data analysis which includes exclusion of doublet events, and the acquisition of a large number of T cell events (median of > 10^5 events per donor).

Together, application of these measures enabled a lower limit of detection of iNKT cells of approximately 1 in 10^5 T cells, which approaches the lower limit of detection for flow cytometric events.⁴⁷¹

The specificity of staining for iNKT cells using the 6B11 antibody clone was confirmed by application of an isotype control antibody (figure 3.6a), concordance with results using an α -GalCer-loaded CD1d tetramer (figure 3.6c), and by the finding that 6B11-positive iNKT cell lines exhibited α -GalCer dependent cytokine release (figure 5.18) and cytotoxicity (figure 5.20).

7.2.1.1 Numbers of iNKT cells in healthy controls

The frequency of iNKT cells observed in healthy controls is towards the lower end of the ranges published in the literature (see table 7.1). This may be due to the older age of controls in this study compared to the majority of published studies.

An age-related decline in iNKT cell numbers is well-described, ^{210,223} and contrasts with age-related increases in NK cell numbers, and with stable overall T cell numbers with aging. ²¹⁰ The iNKT cell frequency reported here is similar to that observed in a study of healthy donors with a similar median age. ³⁰⁸

The effect of gender on iNKT cell frequencies is not well established. In one study of 90 healthy donors, iNKT cell frequency was not significantly different between males and females, ²¹⁸ while in another, iNKT cell frequency was lower in males. ²¹⁰ The data presented in this thesis show no significant effect of gender on iNKT cell number or frequency among healthy controls (section 3.3.4). A study examining *in vitro* iNKT cell proliferation found no effect of gender. ⁴⁷²

7.2.1.2 Numbers of iNKT cells in patients with CLL

Relative iNKT cell numbers The frequency of iNKT cells I observed in patients with CLL are presented alongside other reports of iNKT cells in patients with cancer in table 7.2. The two studies including larger numbers of patients with cancer reported similar iNKT cell frequencies to those reported here in patients with CLL. ^{303,308} In contrast, studies examining smaller numbers of patients found higher iNKT cell frequencies. ^{418,475,476} Possible reasons for this discrepancy include differing disease biology, differences in donor age distribution, methodological factors, or sampling error in the latter studies due to small patient numbers.

Considering the two larger series, compared to healthy controls, median iNKT cell numbers were \sim 2-fold reduced in patients with head and neck squamous cell carcinoma, ³⁰⁸ and median iNKT cell frequency was \sim 3-fold reduced in patients with a range of advanced cancers. ³⁰³ The 2-fold reduction in iNKT cell frequency that I observed in this study of patients with CLL is of similar magnitude.

I sought, but did not find, any trend towards lower iNKT cell frequencies in patients with more clinically advanced disease (figure 3.8), although given the relatively small numbers

Table 7.1 Circulating iNKT cell numbers reported in healthy donors

				Ahsoliite niimher	1
Donor characteristics	u	iNKT cell identification	% of T cells	per mL blood	Reference
Median age 64y	31	CD3+6B11+	0.022	195	This stud
Age range 21 – 52y	37	$V\alpha 24 + V\beta 11 +$	I	1013	303
Median age 54y	31	$CD3 + V\alpha 24 + V\beta 11 +$	0.053	I	473
Mean age 44y	49	$CD3 + V\alpha 24 + V\beta 11 +$	I	857	474
Median age 65y	33	$CD3 + V\alpha 24 + V\beta 11 +$	0.021	373	308
Median age 29y	50	CD3+6B11+	I	2910	285
Mean age 30y	49	$CD3 + V\alpha 24 + V\beta 11 +$	0.044 *	I	223
Mean age 78y	73	$CD3 + V\alpha 24 + V\beta 11 +$	0.004 *	Ι	223
Age range 15 – 52y	06	CD3+6B11+	0.17 ^{*,†}	Ι	218
Ages not stated	20	CD3+Tetramer+ ‡	0.034		220
Mean age 24y	31	CD3+Tetramer+ *	0.086 [§]	I	286

of patients with advanced CLL in this study, a true difference could have been missed (a type II statistical error).

The preclinical and clinical data suggesting a role for iNKT cells in cancer immunosurveillance have already been discussed in section 1.2.7.3. However, whether the reductions in circulating iNKT cell frequencies observed in some patients with cancer is a cause or effect of the malignancy, or simply an epiphenomenon, remains unclear. Although this study adds to the body of data, it does not help to answer this question. Since lower iNKT cell numbers predict poor prognosis in patients with head and neck squamous cell carcinoma,³⁰⁸ a longitudinal study examining the impact of iNKT cell numbers on CLL prognosis may be of interest. Given the indolent nature of CLL, long follow-up would be required, so such a study may be best performed using previously cryopreserved PB-MCs.

Absolute iNKT cell numbers In contrast to the reduction in the relative frequency of iNKT cells, I observed a trend towards increased *absolute* numbers of iNKT cells in patients with untreated CLL. This has to be interpreted in the context of the long-recognised phenomenon of increased T cell numbers in patients with untreated CLL.^{83,84} The data presented in figure 3.10 confirm the presence of a statistically significant increase in the major T cell subsets in patients with untreated CLL in this study. The pathologic significance of this increase in T cell numbers has been discussed in section 1.1.3.5.

In this study, absolute numbers of PBMC subsets were calculated from absolute lymphocyte counts determined on a commercial blood count analyser, before PBMC separation. As this involves several steps, this could lead to a wider coefficient of variation for the determination of absolute cell counts, and could introduce sources of error, compared to the direct analysis of blood counts. Absolute counts could also be determined in a single step by flow cytometric analysis of whole blood spiked with a known number of 'counting beads', polystyrene beads with distinct forward and side scatter properties on flow cytometric analysis. This type of analysis was not performed in the current study, and cannot be applied retrospectively.

The discordance between reduced relative iNKT cell numbers but normal or increased absolute numbers suggests that iNKT cells do not fully participate in this T cell expansion. Whether this is because iNKT cells are resistant to the CLL cell / T cell interactions which support increased numbers of 'conventional' T cells, or because iNKT cell frequencies are lower as part of the aetiology or pathophysiology of CLL is not clear.

Effect of fludarabine chemotherapy Fludarabine-based chemotherapies are known to results in a prolonged depletion of all T cell subsets, particularly of CD4+ T cells, with full T cell reconstitution remaining incomplete at three years after treatment. In contrast, NK cell numbers are relatively preserved.¹⁷⁵ As the first candidates for any investigational therapy of CLL are likely to be patients who have relapsed following, or who are refractory to, fludarabine-based treatments, I recruited a cohort of patients who had received fludarabine treatment, and evaluated iNKT cell numbers in this group.

Among those who had received fludarabine treatment, the numbers of all T cell subsets were reduced compared to untreated patients (figure 3.10). This included an absolute

Table 7.2 Circulating iNKT cell numbers reported in patients with cancer

			Median iN	KT cell numbers	
Donor characteristics	u	iNKT cell identification	% of T cells	Absolute number per mL blood	Reference
Chronic lympocytic leukaemia (median age 63y)	40	CD3+6B11+	0.010	244	This study
Advanced cancer (median age 54y)	24	$V\alpha 24 + V\beta 11 +$	I	333	303
Head and neck squamous cell carcinoma (mean age 65y)	47	$CD3 + V\alpha 24 + V\beta 11 +$	0.009	103	308
Advanced non small cell lung cancer cancer (median age 66y)	9	$CD3 + V\alpha 24 + V\beta 11 +$	0.03	Ι	418
Myeloma (ages not given)	~	$CD3 + V\alpha 24 + V\beta 11 +$	0.01	Ι	477
Renal cell carcinoma (ages not given)	11	$CD3 + V\alpha 24 + V\beta 11 +$	0.05	Ι	475
Glioma (median age 40y)	6	$CD3 + V\alpha 24 + V\beta 11 +$	0.07 *,†	Ι	476
 * Number of iNKT cells expressed as i † iNKT cells expressed as a percentag. 	a mean, e of lym	not median phocytes, not as a percen	tage of T cells		

(but not relative, see figure 3.8) reduction in the numbers of iNKT cells. This suggests that the magnitude of iNKT cell depletion after fludarabine is similar to that of depletion of other T cell subsets.

This was a cross-sectional study, so confounding factors such as differing disease or immunological characteristics among patients who required treatment could be responsible for the reduction in absolute iNKT cell numbers observed. As discussed in section 7.2.1.1, a gender difference in iNKT cell numbers has been reported, ²¹⁰ although another study did not confirm this, ²¹⁸ and I did not find significant differences in iNKT cell number or frequency between healthy male and female donors in my own data. Nonetheless, as there was an excess of males among fludarabine-treated patients, I cannot exclude the possibility that gender contributed to the lower iNKT cell numbers observed in this group.

Fully defining the effect of fludarabine-based chemotherapy on the iNKT cell population would be more reliably achieved in a longitudinal study, examining iNKT cell numbers before and after fludarabine treatment in the same individuals.

7.2.1.3 Immunophenotype of iNKT cells

Healthy donor iNKT cell immunophenotype I examined the CD4, CD8, CD161 and CD25 expression of circulating iNKT cells in patients and controls (figure 3.11). The results in this study are compared with those reported elsewhere in table 7.3.

The proportions of CD4+ and CD161+ iNKT cells I observed in healthy donors were similar to those previously published. Expression levels of CD25 on T cells lie in a continuum rather than forming discrete positive and negative populations, so are sensitive to the positioning of gates during analysis. The CD25 expression levels I recorded are not therefore readily comparable with those of the one other study that reported them. However, I did observe a consistently lower proportion of CD8+ iNKT cells than that reported in other studies.

Of note, the *level* of CD8 expression on CD8+ iNKT cells was lower than that of 'conventional' CD8+ T cells in this study (figure 3.11a). This was a consistent finding across all experiments, and was not due to inconsistent staining, because the difference between the two cell subsets was observed within the same sample. The CD8 expressed by iNKT cells is reported to be in the form of the CD8aa homodimer, rather than the functional CD8aβ heterodimer more commonly expressed by 'conventional' T cells.²²⁰ The antibody clone I employed (RPA-T8) is said to recognise CD8a, so should have detected the CD8aa homodimer on iNKT cells. Nonetheless, antibody clones could differ in their relative detection of the homodimeric and heterodimeric forms of CD8. Thus, possible reasons for the lower iNKT cell CD8 levels I observed include a genuine reduction in surface expression, or an effect of the selected antibody clone. This could be explored further using alternative anti-CD8 antibody clones. However, since the CD8aα homodimer is not believed to be functional, ⁴⁷⁸ and because, unlike CD4 expression, no functional significance of CD8 expression on iNKT cells has been reported, I did not pursue this observation further.

Patient iNKT cell immunophenotype Comparing patients with controls, I did not find statistically significant differences in iNKT cell immunophenotype. However, a non-significant trend towards a higher proportion of CD4+ iNKT cells in patients was observed. This may relate to the lower relative frequency of iNKT cells in patients, as I confirmed that there was a negative correlation between the proportion of iNKT cells which were CD4+ and overall iNKT cell frequency (figure 3.12), as previously reported.²¹⁸

I found no significant difference in the CD4 status of iNKT cells between patients who had and who had not received fludarabine chemotherapy (figure 3.11c). This could imply that fludarabine does not disproportionately deplete CD4+ iNKT cells, but as this is a cross-sectional study, the results could be confounded by other factors.

7.2.2 Relationship between iNKT cell numbers and progression-free survival

In section 3.3.7, I examined the association between circulating iNKT cell numbers and progression-free survival among patients with untreated CLL. The low rate of disease progression seen (28% over a three year follow-up period) highlights the fact that this is a cohort of patients with indolent CLL. Only one patient died during follow-up.

I found no correlation between iNKT cell frequency or absolute numbers and CLL disease progression. My findings contrast with studies reporting that iNKT cell frequency correlates with overall survival in patients with head and neck squamous cell carcinoma,³⁰⁸ and that absolute numbers of iNKT cells correlate with overall survival in patients with acute myeloid leukaemia.³⁰⁵

The findings I present should be interpreted with some caution. First, I do not have data on the most established prognostic markers in early CLL, FISH cytogenetic findings (particularly 17p and 11q deletion status) or IgVH rearrangement status. Thus, it is not possible to perform a multivariate analysis that includes these parameters. Second, my cohort is relatively small (29 patients) and fewer than half progressed or died during follow-up, so there is a chance that an association could have been found had a larger cohort, or longer follow-up period, been available (although I do note that the two studies above each found an association between iNKT cell numbers and survival using a similar-sized cohort). Third, as I have sampled the peripheral blood only, I cannot exclude the possibility that changes in iNKT cell tissue distribution take place in high-risk or progressive CLL, and that peripheral blood iNKT cell frequency does not reflect total iNKT cell numbers.

Despite these limitations, in conjunction with the finding that iNKT cell frequency does not correlate with CLL disease stage, lymphocyte count or ZAP-70 status (figure 3.8), the lack of an association between iNKT cell frequency or number and progression-free survival suggests that iNKT cell deficiency does not play a central role in the progression of CLL.

			iNKT ce	lls of give	en phenoty	/pe (%)	
Donor characteristics	n	iNKT cell identification	CD4+	CD8+	CD161+	CD25+	Reference
Median age 64y	28	CD3+6B11+	42	14	63	26	This study
Ages not given	20	CD3+Tetramer+ *	51 [†]	51 [†]	72 †		220
Mean age 24y	31	CD3+Tetramer+ *	56 [†]	31 [†]	I	I	286
Age range 15 – 52y	30	CD3+Tetramer+ *	41 [†]	25 [†]	[†]	3.5 [†]	218
Young, mean age 29y	19	$CD3 + V\alpha 24 + V\beta 11 +$	17 *	29 [†]		I	223
Elderly, mean age 74y	30	$CD3 + V\alpha 24 + V\beta 11 +$	44 *	28 *			223
$\frac{1}{2}$ "Tetramer' refers to α -Ga	Cer-load	ed CD1d tetramer					

Table 7.3 Reported immunophenotype of circulating iNKT cells in healthy donors

 $^{\dagger}~$ Expressed as mean, not median

7.2.3 CD1d expression

In section 4.3.1, I investigated CD1d expression on subsets of PBMCs from patients with CLL. I quantified CD1d expression using flow cytometry. Compared to real-time reverse transcription quantitative polymerase chain reaction (RT-PCR), flow cytometry has the advantage of assessing expression at the protein level, and unlike both real-time RT-PCR and Western blotting, allows simultaneous analysis of expression on multiple cell types within mixed PBMCs, without need for cell purification. Flow cytometric quantification of surface and intracellular markers is, however, susceptible to systematic errors. I have outlined common sources of error and their possible solutions in table 7.4.

Having observed the precautions suggested in table 7.4, I have confidence in the findings presented, but would still consider expression quantification by flow cytometry to be a semi-quantitative rather than fully quantitative technique.

7.2.3.1 Expression of CD1d by CLL cells

Downregulation of MHC Class I has long been recognised as a mechanism of immune evasion in cancer, ^{479,480} and has been reported on leukaemic cells in CLL.⁴⁸¹ CD1d is a MHC Class I-like, β_2 M-associated molecule, and may also be downregulated in B cell malignancies: patients with unmutated IgVH CLL exhibited lower CD1d expression than those with mutated IgVH, ⁴¹⁷ while in myeloma, progressive disease is associated with loss of CD1d expression on the malignant B cells.⁴⁴⁶ I therefore chose to analyse CLL cell CD1d expression.

My observation that CD19+CD5+ leukaemic cells expressed CD1d at a similar level to that of normal B cells (figure 4.1b) was consistent with reported findings.⁴¹⁷ Although I did not observe a progressive decline in CD1d progression among patients with advanced CLL (figure 4.1c), it is possible that a difference would emerge if a larger group of patients with more advanced disease were studied.

7.2.3.2 Number and CD1d expression of mDC

Numbers of mDC The reported defects of DC from patients with CLL have been discussed in section 1.1.3.5. One report suggested that mDC are reduced in patients with CLL, but notably, this study expressed mDC as a proportion of total PBMCs, the denominator including the very large numbers of leukaemic cells—absolute numbers of circulating mDC were not given.⁷⁸

I used absolute full blood counts performed on the same day as PBMC cryopreservation to derive figures for the absolute numbers of circulating mDC (figure 4.2b). This showed normal absolute numbers of mDC in patients with untreated CLL, although I found a significant reduction among those who had received fludarabine chemotherapy.

The significance of the reduction among fludarabine-treated patients is unclear: fludarabine has not been previously reported to cause a durable decline in mDC numbers. Indeed, in one longitudinal study of patients with the indolent B cell malignancy hairy cell leukaemia, mDC numbers were low pre-treatment and *recovered* after chemotherapy with cladribine, a purine analogue similar to fludarabine.⁴⁸² As this was a cross-sectional rather than longitudinal study, the observed reduction in mDC among fludarabine-treated patients could be due to disease or immunological characteristics rather than an effect of the treatment *per se*.

CD1d expression on mDC The expression of CD1d on *in vitro* generated human moDC is affected by culture conditions, the presence of TGF- β^{483} and the absence of IgG⁴⁴⁸ being associated with reduced CD1d levels. Raised serum TGF- β^{102} and reduced serum IgG levels⁶⁵ have both been described in patients with CLL. In view of this, I had hypothesised that CD1d expression would be reduced on circulating mDC from patients with CLL.

Contrary to my hypothesis, I found normal CD1d expression levels on patient mDC, with no evidence of a relationship between mDC CD1d expression and serum IgG levels (figure 4.3). Thus, at least in patients with untreated CLL, mDC are normal in number and express CD1d at normal levels, suggesting that they may retain the capacity to present α -GalCer to iNKT cells.

7.2.3.3 Number and CD1d expression of monocytes

Monocytes may be important for the maintenance of the leukaemic clone in CLL: the addition of monocytes supports the survival of CLL cells *in vitro*, an effect which may be mediated by soluble CD14,³³¹ while monocytes can inhibit rituximab-induced antibody-dependent cellular cytotoxicity (ADCC) against CLL cells.⁴⁸⁴

Data on the numbers of circulating monocytes in CLL are conflicting, one study indicating higher monocyte numbers among patients with CLL and high circulating leukocyte counts, ⁴⁴⁹ while another describes normal monocyte numbers. ⁴⁸⁵

My data indicate that CD14hi monocytes are normal in number in CLL, but there was a non-significant trend towards an increase in the number of 'inflammatory' CD16+ monocytes in patients, especially in those who had received a fludarabine-based chemotherapy (figure 4.4). If confirmed with a larger cohort, this would be in keeping with reports of increased CD16+ monocytes in patients with other malignancies, such as gastric cancer, breast cancer and cholangiocarcinoma.^{453,454,457}

Examining CD1d expression levels, the CD16+ monocytes expressed CD1d at a significantly lower level than their CD14hi counterparts (figure 4.5). However, even among patients with untreated CLL, the majority of monocytes were CD14hi, so the overall level of monocyte CD1d expression was similar to that in controls. The finding of differential CD1d expression of these monocyte subsets has not previously been reported, and it may be of interest to determine whether the capacity of 'inflammatory' CD16+ monocytes to activate iNKT cells is altered relative to that of the major CD14hi subset.

The CD16+ monocyte subset constitutes around 10% of monocytes in healthy individuals, and is expanded in infections and in patients with cancer.⁴⁸⁶ The functional role of CD16+ monocytes has not been well defined, in part because mice lack a CD16 (Fc γ RIII) expressing monocyte population. However, human CD14loCD16+ monocytes

express high levels of the fractalkine receptor CX3CR1, in contrast to CD14hiCD16- monocytes, which have the opposite phenotype.^{487,488} In mice, CX3CR1+ monocytes infiltrate the peritoneum during ovarian carcinoma progression, and suppress naive CD8+ and CD4+ T cell responses.⁴⁸⁹ It is not clear whether or not the CX3CR1-expressing population of circulating monocytes (CD16+ in humans) generates myeloid-derived suppressor cells.

These results have some potential limitations. The flow cytometric identification of CD16+ monocytes can be complicated by contamination with NK cells or granulocytes. Addition of an anti-MHC Class II antibody to the monocyte identification panel can overcome this (class II being highly expressed by monocytes, but not by NK cells).⁴⁹⁰ In my study, granulocytes are excluded by the density centrifugation step prior to PBMC storage, and should additionally be excluded by the forward and side scatter gate. My preliminary experiments indicated that monocytes identified by forward and side scatter were > 95% MHC Class II positive. I excluded CD14 low events from the CD16+ gate (figure 4.5a), which should exclude any NK cells which have entered the monocyte gate. Together, these measures should have limited NK cell or granulocyte contamination. However, MHC Class II was not routinely included in my monocyte identification panel, so I cannot exclude a small number of contaminating events. Finally, as I have reported CD1d expression levels as median, rather than mean, fluorescence intensities, the presence of a minor population of contaminating cells should not affect reported expression levels.

The effect of fludarabine on monocyte subpopulations has not been reported. However, total monocytes are modestly reduced after fludarabine-containing chemotherapies, as might be expected for a myelosuppressive chemotherapy.¹⁷⁵ The monocytic cell line U937 undergoes apoptosis upon exposure to fludarabine, but in response to lower fludarabine concentrations, upregulates ICAM-1 and produces IL-8.⁴⁹¹

Other factors influence CD16+ monocyte numbers, including exercise (which temporarily raises CD16+ monocyte numbers), exogenous corticosteroids (which reduce CD16+ monocyte numbers) and gender (women exhibiting lower CD16+ monocyte numbers).⁴⁹⁰ In this study, all subjects were at rest immediately before phlebotomy and donors taking systemic corticosteroids were excluded. However, there was a greater proportion of men in the fludarabine-treated CLL patient group (table 3.1), which could have contributed to the finding of higher CD16+ monocyte numbers in this group.

Finally, it is not possible to attribute any of the observed changes in monocyte subsets in fludarabine-treated patients to immunological or disease characteristics or to the treatment itself; this would require a longitudinal study.

These limitations notwithstanding, I have demonstrated that monocyte numbers are retained in CLL, that overall monocyte CD1d expression is intact, and have made the interesting observation that CD1d expression differs between CD14hi and CD16+ monocyte subsets, a finding which has not previously been reported.

7.2.4 Function of iNKT cells

In chapter 5, I used a number of assays to evaluate the function of iNKT cells from patients with CLL. The cytokine production and proliferative capacity of iNKT cells within whole

re flow cytometry
Solution(s)
Avoid fluorophores which emit in autofluorescence-prone channels (e.g. FITC) Measure FMO fluorescence in identically treated and gated cells, and subtract from result (i.e. report net fluorescence)
Rigorous use of FMO controls Measure FMO fluorescence in identically treated and gated cells, and subtract from result (i.e. report net fluorescence)
Handle all samples identically Stain and analyse all samples within the same session
Count cells and stain identical cell numbers for each sample Use non-limiting dilution of antibody
Use blocking reagents (e.g. polyclonal IgG) to minimise non-specific binding
Report fluorescence intensities as median rather than mean fluorescent intensity Analyse pure cell populations, or gate carefully on cell of interest

PBMCs were assessed. I went on to generate iNKT cell lines from patients and controls, and characterised their immunophenotype, cytokine profile and cytotoxicity.

7.2.4.1 Cytokine production by iNKT cells

A reduction in IFN- γ production by iNKT cells has been documented in patients with advanced prostate cancer and myeloma.^{304,307} I hypothesised that a similar defect may exist in patients with CLL.

α-GalCer induced IFN-γ **ELISpot assay** Figure 5.2 demonstrates intact IFN-γ-induced ELISpot formation in patients with CLL. In patients with myeloma, α-GalCer-induced IFNγ ELISpots were reduced in patients with progressive disease.³⁰⁷ Few of the patients I studied by ELISpot had advanced or treated CLL, although ELISpots remained detectable in three patients who had clinically advanced CLL (Binet clinical stage B or C) at the time of analysis, and in three separate patients who had previously received a fludarabine-based chemotherapy.

Preliminary experiments had indicated that rigorous titration, and the use of a 'resting period' after B-cell depletion of PBMCs, were necessary to obtain ELISpot results within a measurable range and to reduce background spot formation. Compared with the measurement of intracellular cytokines, the ELISpot assay has a relatively narrow 'dynamic range', with the capacity to reliably count between around 20 and 400 ELISpots per well. This may have the effect of 'clipping' the results: some donors with negative results might have had detectable numbers of ELISpots had higher cell numbers of cells per well been used, and some donors may have generated more ELISpots than the automated counter could reliably detect. In these cases, titrations with lower numbers of cells per well would have resolved further differences. I did not have sufficient PBMCs to perform ELISpot assays at numerous titration levels for each donor, so selected a single number of cells per well for all donors. As such, ELISpot could be considered a semi-quantitative rather than fully quantitative assay.

The ELISpot results indicated that α -GalCer-induced IFN- γ production is intact in the PB-MCs of patients with CLL. This result could be considered at odds with the data presented in figure 3.7, which indicate that iNKT cell numbers are reduced as a proportion of T cells in patients. However, as discussed above, ELISpot can be considered semi- rather than fully quantitative. Moreover, the identity of the IFN- γ -producing cell types is not determined by ELISpot. Thus, iNKT cells may be activated by α -GalCer, and induce neighbouring cells (such as 'conventional' T cells or NK cells) to produce IFN- γ . As will be discussed in the next section, IFN- γ expression of T cells is *increased* in CLL, thus the higher number of ELISpots than expected may in part be due to enhanced secondary production of IFN- γ by admixed T cells in this group.

Despite its limitations, I consider the demonstration of α -GalCer-induced IFN- γ ELISpot production important, because it implies that the processes of α -GalCer presentation by APCs, recognition by iNKT cells and downstream production (or induction) of cytokine release are all intact in patients with CLL. Moreover, ELISpot provides an indication of

 α -GalCer-induced cytokine production in mixed PBMCs which is independent of the flow cytometric identification of iNKT cells.

Intracellular cytokine staining Unlike ELISpot, intracellular cytokine staining (ICS) allows immunophenotypic characterisation of individual cytokine-producing cells. I used a seven colour flow cytometry panel to simultaneously characterise IFN- γ and IL-4 production of iNKT and T cell subsets following stimulation with PMA and ionomycin.

Whereas cytokine-producing T cells predominantly produced either IFN- γ or IL-4, but not both, the majority of IL-4 producing iNKT cells also produced IFN- γ (figure 5.3a). Dual production of cytokines by iNKT cells has been demonstrated in both mice^{492,493} and humans,^{461,494} and contrasts with the Th1 and Th2 paradigm of 'conventional' T helper cells, in which Th1 and Th2 differentiation are regarded as mutually exclusive pathways. Unlike T helper cells, murine NKT cells activate the transcription of IL-4 and IFN- γ early during thymic development, and harbour high levels of mRNA transcripts for both cytokines.⁴⁹⁵ The molecular mechanisms underlying this have been discussed elsewhere.^{221,240}

Overall, the proportion of iNKT cells which produced IFN- γ was higher than the proportion of T cells producing this cytokine (figure 5.3b). This is in keeping with published reports.²²⁰ Considering the CD4+ and CD4- subsets of iNKT cells separately, my observation of increased IL-4 production in CD4+ iNKT cells was also in keeping with published findings.^{218,220}

The observation that in patients with CLL a larger proportion of T cells expressed IFN- γ on ICS than in controls (figure 5.5a), is consistent with prior reports.^{458,460} In contrast, I found no difference between patient and control T cell expression of IL-4. This appears to be at odds with some reports which describe increased T cell IL-4 production in CLL.^{95,496} Possible explanations for this include the degree of leukaemic cell depletion, which was more thorough in my study than in those showing increased T cell IL-4 production: IL-6 produced by residual CLL cells may induce T cell IL-4 production.³³⁴ Another factor may be disease stage: one study suggested that IL-4 production by T cells is greater in patients with more advanced disease,⁴⁵⁸ and the majority of patients in this study had early stage CLL (table 3.1).

In contrast with the differences observed in T cell cytokine production, the expression of both IFN- γ and IL-4 by patient iNKT cells was similar to that of controls (figure 5.5b).

One disadvantage of evaluating cytokine profile by ICS on PMA and ionomycin-stimulated PBMCs is that the T cell stimulation is far from physiological—the cytokine profile elicited may not reflect that observed on stimulation with a peptide (or glycolipid) ligand. Non-etheless, my findings by ICS are consistent with those observed by ELISpot, and together these two techniques provide evidence that the capacity of iNKT cells to produce IFN- γ and IL-4 is grossly intact in the peripheral blood of patients with CLL.

7.2.4.2 Proliferation of iNKT cells

In section 5.3.3, I evaluated the proliferative capacity of iNKT cells from patients with CLL. The *in vitro* proliferation of iNKT cells is reduced in healthy elderly individuals⁴⁴¹ and in patients with advanced prostate cancer,³⁰⁴ while it appears to be intact in patients with melanoma⁴⁹⁷ and myeloma.³⁰⁷

To assess iNKT cell proliferation, I added α -GalCer followed by IL-2 to B-cell depleted PBMCs. The rationale of using PBMCs rather than purified T cells was to simultaneously assess the integrity of α -GalCer presentation by patient APCs, recognition by iNKT cells, and the downstream proliferative response. The choice of added cytokine, a high concentration of IL-2, was made because preliminary experiments had indicated that this resulted in optimal early iNKT cell proliferation (figure 5.7).

Proliferation of iNKT cells was similar in patients and controls, with nearly all patients, and all controls, exhibiting some degree of iNKT cell expansion (figure 5.8). Fold proliferation of iNKT cells was similar in the two groups. The *in vitro* culture conditions employed favoured proliferation of the CD4+ subset of iNKT cells in both patients and controls (figure 5.9), a finding which has previously been observed in healthy donor PB-MCs.⁴⁶³

The presence of CLL cells in *in vitro* culture systems is typically associated with inhibition of T cell proliferation,⁹⁴ an effect which may be mediated by CLL cell expression of CD200,⁴⁹⁸ production of IL-6,⁹¹ or secretion of soluble IL-2 receptor.⁴⁹⁹ I hypothesised that the addition of CLL leukaemic cells would inhibit iNKT cell proliferation.

In contrast, my findings showed that patient iNKT cells were able to proliferate in whole PBMC cultures, without prior B cell depletion (figure 5.11). Moreover, the addition of CLL cells appeared to enhance rather than reduce iNKT cell proliferation in these *in vitro* conditions (figure 5.12a). This is in keeping with a finding that depletion of CD19+ B cells from human PBMCs resulted in impaired *in vitro* iNKT cell proliferation, and that α -GalCer-loaded B cells are able to induce iNKT cell proliferation.²⁶⁹

The presentation of α -GalCer by B cells rather than by mDCs ('professional' APCs) has been associated with increased production of IL-4 by iNKT cells.²⁴⁶ Given that CD4+ iNKT cells are associated with IL-4 production (figure 5.4), I analysed the CD4 status of iNKT cells after proliferation in the presence of various numbers of added CLL cells, and did not find any difference (figure 5.12b).

Together, these findings demonstrate that iNKT cells within the PBMCs of patients with CLL retain their capacity to proliferate *in vitro*, and indicate that iNKT cell proliferation is not impaired by the presence of CLL cells.

7.2.4.3 Generation of iNKT cell lines

Having demonstrated the capacity of iNKT cells from patients with CLL to proliferate, I wished to generate iNKT cell lines to facilitate further investigation of iNKT cell function.

Table 7.5 lists published methods for the generation of polyclonal iNKT cell lines. Some protocols begin with initial iNKT cell expansion within PBMCs, using α -GalCer together with one or more gamma chain cytokines. This is followed by iNKT cell selection using immunomagnetic bead or FACS, followed by a series of restimulations to expand iNKT cell numbers further. Other reported methods omit the initial expansion of iNKT cells within whole PBMCs: iNKT cells are instead enriched from PBMCs directly using immunomagnetic selection or FACS, 276,500 or are specifically stimulated by the addition of autologous or artificial APCs to PBMCs or anti-CD3 mAb. 477,500,501

I chose to employ initial iNKT cell expansion within whole PBMC, because the starting proportion of iNKT cells was very low in both patient and control groups (figure 3.7), and because I had already demonstrated that iNKT cell proliferation could be successfully achieved in PBMCs in this cohort (figure 5.8). Following iNKT cell expansion, FACS was used to purify the iNKT cell population to a relatively high purity (figure 5.14).

I evaluated several potential feeder cells for their capacity to support iNKT cell proliferation, and found that the C1R cell line was as effective as autologous PBMC (figure 5.13). I chose to use the C1R cell line because the availability of patient autologous PBMCs was limited. Interestingly, CD1d expression by the C1R cell line was not required to support iNKT cell expansion, a finding which might be accounted for by the presence of alternative iNKT cell stimuli, such as KIR mismatches between the C1R cell line and the donor iNKT cells, or by low levels of CD1d expression on the untransfected C1R cell line. Nonetheless, I chose to use the CD1d-transfected C1R cell line as a feeder cell for iNKT cell expansion.

Maintenance of the polyclonal iNKT cell lines was hampered by declining iNKT cell purity after two restimulations (figure 5.15). The contaminating cells were a mixture of 'conventional' CD4+ and CD8+ T cells and NK cells, suggesting that the feeder cell line used, CD1d-transfected C1R, was stimulating these subsets. Although characterised as a HLA class I-low cell line, the C1R cell line does express the HLA-Cw4 protein and low levels of HLA-B35,⁵⁰² as well as expressing MHC class II. It is not surprising, therefore, that the C1R cell line can stimulate 'conventional' CD4+ and CD8+ T cells, as well as stimulating NK alloreactivity.

Purity of the polyclonal iNKT cell lines was readily re-established by repeat FACS. However, the number of iNKT cells in culture began to decline after a fourth restimulation (figure 5.15). From the start of PBMC culture to the end of the fourth restimulation, iNKT cells expanded a median of 9352 times, corresponding to just over 13 iNKT cell population doublings: $log_2(9352) \approx 13$. Under optimal *in vitro* conditions, human T cells would be expected to achieve a much higher number of population doublings. The older age of the donors in this study is not likely to be the explanation for this, as T cells from elderly individuals have been shown to be capable of achieving > 30 population doublings *in vitro*.⁵⁰³ Rather, the failure of iNKT cells to maintain expansion in my cultures may have been due IL-2 induced apoptosis, a phenomenon which has long been described in 'conventional' T cells,⁵⁰⁴ and which may be due to upregulation of Fas ligand on antigenand IL-2-stimulated T cells.⁵⁰⁵

Improved viability, and therefore larger iNKT cell numbers, may have been obtained by selecting alternative cytokines: The viability of cultured T cells has been reported to be higher in the presence of IL-7 rather than IL-2. 506 As outlined in table 7.5, and as my

data confirm (figure 5.7), both IL-7 and IL-15 can support iNKT cell proliferation *in vitro*. Compared with IL-2, IL-7 and IL-15 may also favour the expansion of CD4- iNKT cells, a subset associated with anti-tumour activity.⁴³³

The experiments presented demonstrate the feasibility of generating iNKT cell lines from the peripheral blood of patients with CLL. Although the optimisation of purification, restimulation and expansion methods could lead to higher iNKT cell line purity and numbers, and to cell lines with a different cytokine profile, such comparisons were beyond the immediate scope of this project. I proceeded to characterise the iNKT cell lines I had generated with the method outlined above.

7.2.4.4 Phenotype and cytokine profile of iNKT cell lines

The iNKT cell lines generated were overwhelmingly CD4+ (figure 5.16), a finding which has previously been reported.^{463,472} The selective expansion of CD4+ iNKT cells in the presence of IL-2 may be explained by published observations that expression of the IL-2 receptor α chain, CD25, is higher on the CD4+ than the CD4- subset of iNKT cells.^{219,220}

In contrast with the cytokine profile of circulating iNKT cells (figure 5.3), the iNKT cell lines generated from both patients and controls exhibited dual expression of both IFN- γ and IL-4 (figure 5.17) on ICS. This property is consistent with the CD4+ immunopheno-type of the iNKT cell lines, and has previously been described.^{463,516}

In order to demonstrate α -GalCer reactivity of the iNKT cell lines generated, and to further assess their cytokine profile, I added the CD1d-transfected C1R cell line to iNKT cell lines in the presence or abscence of α -GalCer, and determined cytokine levels in the supernatants (figure 5.18).

Although the treatment of C1R-CD1d cells with α -GalCer led to a marked enhancement of cytokine production in co-cultures with iNKT cells, a baseline level of cytokine production was seen in the presence of vehicle-treated C1R-CD1d cells. Several explanations could account for this. First, vehicle-treated C1R-CD1d cells could have presented weak endogenous glycolipids to iNKT cells, resulting in low-level iNKT cell activation. Second, vehicle-treated C1R-CD1d cells could have partially activated iNKT cells via KIR mismatches, as iNKT cells are known to express functional NK receptors.²²⁸ Third, the polyclonal iNKT cell lines could be contaminated with 'conventional' T cells able to recognize peptide antigens presented by MHC on the C1R cells (as discussed in section 7.2.4.3, C1R cells express MHC Class II and some Class I antigens). Fourth, the polyclonal iNKT cell lines could be contaminated with NK cells able to recognize mismatching KIR ligands on the C1R cells. Finally, contaminants within the cultures such as bacterial endotoxins could have induced spontaneous cytokine production. As the increase in cytokine production in the presence of α -GalCer was unequivocal, I did not explore the background levels of production further. However, if investigation were critical, repetition of the experiment with the following additional controls could be informative: iNKT cells alone (in the absence of C1R cells), untransfected C1R cells and α -GalCer-treated C1R-CD1d cells in the presence of a blocking anti-CD1d antibody.

The finding of the Th2-type cytokines IL-4 and IL-13 in the supernatants of the α -GalCer-

Table 7.5 Methods of poly	yclonal iNKT cell line generation		
Step	Method	Advantages and disadvantages	Reference(s)
Initial stimulation	α-GalCer addition to whole PBMCs	Enriches iNKT cells prior to purification step, below	507-510
	IL-2	Favours CD4+ iNKT cells	463
Cytokine addition	IL-7	Favours CD4- iNKT cells	463,511
	IL-15	Enhanced cytotoxicity, favours CD4- iNKT cells	463,511
	FACS	High purity of sorting, requires specialised equipment, infection control issues	509,510,512,513
iNKT purification	Immunomagnetic separation	Little specialised equipment required, can be performed in sterile hood, purity lower than FACS	509
	Selective restimulation only	Requires no purification steps, significant contamination with non-iNKT cells possible	514
	Autologous APC	No stimulation of NK or T cells, availability of cells may be limited	307,476,477,509,515
	CD1d-transfected cell line	Readily available, may also stimulate NK and T cells	276,509
Restimulation	Anti-CD3 mAb	Readily available, non-selective (will also stimulate T cells)	509,512
	Plate-bound anti-iNKT cell mAb (6B11 clone)	Readily available, no requirement for donor APCs	516
	CD1d-Ig coated artificial APCs	Readily available, no requirement for donor APCs	501

treated co-cultures alongside the Th1-type cytokines IFN- γ and TNF- α was consistent with the intracellular cytokine findings.

In contrast with the high levels of IL-4 and IL-13 production, production of the immunosuppressive cytokine IL-10 by the iNKT cell lines was not observed. Although murine iNKT cells have been shown to produce IL-10 directly,²⁴¹ IL-10 production by human iNKT cells has not been demonstrated. However, human iNKT cells are able to induce the secondary production of IL-10 by APCs.²⁶²

Production of GM-CSF by human iNKT cells has been demonstrated, and may be of importance in normal haematopoiesis: a proportion of haematopoietic progenitor cells express CD1d, and the addition of human iNKT cells results in a dose-dependent increase in myeloid colony formation *in vitro*.⁵¹³ My data indicate that both patient- and control-derived iNKT cell lines produced similar amounts of GM-CSF in an α -GalCer-dependent manner (figure 5.18).

Although a subpopulation of murine iNKT cells produce IL-17,²⁴¹ production of IL-17 by human iNKT cells has never been conclusively demonstrated. Rather, human IL-17 producing cells appear to be non-invariant CD161+ NK-like T cells.⁵¹⁷ My data showing the absence of IL-17 production by both patient and control-derived CLL cell lines are consistent with this.

In conclusion, the surface immunophenotypic and intracellular cytokine properties of *in vitro* generated iNKT cell lines appear to differ from those of iNKT cells analysed directly *ex vivo*. Nonetheless, the finding that iNKT lines can be generated from patients with CLL, and have similar characteristics to those generated from healthy controls, suggests that critical impairments of iNKT cell proliferative capacity and cytokine production are not present in patients with CLL.

7.2.4.5 Cytotoxicity of iNKT cells

The expression of granzyme B by *in vitro* generated iNKT cell lines (figure 5.19) is consistent with the known effects of IL-2, which leads to upregulation of granzyme expression by NK and by T cells.⁵¹⁸

Using a LDH release assay, I was able to demonstrate α -GalCer and CD1d-dependent lysis of a C1R cell line (figure 5.20). This demonstrates that the iNKT cell lines generated exhibit antigen specificity, and that their cytotoxic function is intact.

Although specific, degree of cytotoxicity seen was relatively modest in this experiment. A number of reasons may account for this. First, the effector:target ratios I used were modest, in part because of restrictions on the number of effector cells available from preceding iNKT cell line expansion. Second, the LDH release assay may not be the most sensitive assay available—alternative cytotoxicity assays will be discussed in section 7.2.5.1. Finally, the iNKT cell lines, which I derived in the presence of IL-2, were overwhelmingly CD4+, and may not be the optimal cytotoxic effectors—CD4- iNKT cells are reported to exhibit greater cytotoxicity than their CD4+ counterparts.

Several molecular mechanisms have been implicated in cytotoxicity by iNKT cells, as previously discussed in section 1.2.5. The LDH release assay measures completed lysis of target cells,⁵¹⁹ and does not therefore distinguish between apoptosis or necrosis. Therefore, any or all of the described mechanisms of iNKT cell cytotoxicity may be operating.

The calculation method I employed takes account of spontaneous LDH release from both effector and target cells, and as such should provide a robust measure of cytotoxicity.⁴³⁸ However, the LDH release assay carries the disadvantage that relatively high numbers of viable target (and therefore effector) cells were required to yield detectable supernatant LDH concentrations. In conjunction with the fall in iNKT cell line numbers after repeated restimulation (figure 5.15b), this prevented the assessment of cytotoxicity by all cell lines generated. Alternative methods of detecting target cell killing may have overcome this limitation, and will be discussed in section 7.2.5.1.

As a consequence of their prolonged culture in the presence of IL-2, the cytokine production and cytotoxicity of iNKT cell lines may not reflect those of *ex vivo* iNKT cells. The scarcity of iNKT cells in patient and control PBMCs prohibited direct assessment of *ex vivo* cytotoxicity. However, the expression of IFN- γ , IL-4 and granzyme B by *ex vivo* patient iNKT cells appeared to be intact (figures 5.5b and 5.6), and my data do offer a proof of principle that the generation of iNKT cell lines of high purity and with specific cytotoxicity against CD1d positive targets is possible from CLL patient PBMCs.

7.2.5 Potential roles of iNKT cells in immunotherapy

In chapter 6, I used a series of *in vitro* assays to assess various methods of exploiting iNKT cells for the adoptive immunotherapy of CLL. In this section, I shall discuss my experimental findings, and relate them to findings in preclinical and clinical studies.

7.2.5.1 iNKT cells as effectors

Cytokine production One proposed method of utilising iNKT cells for immunotherapy is to administer α -GalCer systemically, and to rely on iNKT cell-derived cytokines to transactivate other effectors, such as NK cells, as discussed in section 1.2.6.4.

In a preliminary experiment, α -GalCer induced the production of IFN- γ and IL-13 when added to the PBMCs of donors with high frequencies of circulating iNKT cells (figure 6.1).

In a larger group of controls, but not in patients, α -GalCer addition to cultured PBMCs led to a significant increase in IFN- γ production. There was a non-significant trend towards increased IL-10 and IL-4 levels in the supernatant of α -GalCer-treated PBMCs from both controls and patients (figure 6.2). This experiment showed no evidence of α -GalCer-induced production of GM-CSF, TNF- α or IL-13.

The failure to demonstrate α -GalCer-induced IFN- γ production in patients may be explained by the observation that iNKT cells constitute a smaller percentage of T cells in patients than controls (figure 3.7). Indeed, in a study using PBMCs from healthy human donors, the addition of previously cultured, activated iNKT cells to a level of at least 0.1% of PBMCs was necessary to elicit a significant effect of α -GalCer on supernatant IFN- γ levels or NK cell activity.⁴⁰²

This experimental finding may appear at odds with the α -GalCer-induced ELISpot data presented in figure 5.2, in which significant α -GalCer-induced IFN- γ production could be observed in both patients and controls. The difference between the findings in the two assays may be statistical chance. However, the two assays do differ in an important manner: ELISpot aims to quantify the number of IFN- γ -producing (or -inducing) iNKT cells, whereas the measurement of IFN- γ in culture supernatants is also dependent on the amount of IFN- γ produced by each cell. Thus, transactivated T or NK cells, for example, might produce higher levels of IFN- γ in healthy donor than in patient PBMCs.

Despite the lack of an IFN- γ response, a trend towards increased concentrations of the potentially immunosuppressive cytokines IL-4 and IL-10 was seen when α -GalCer was added to both patient and control PBMCs. The iNKT cell lines I generated did not produce IL-10 upon stimulation, but α -GalCer-activated iNKT cells have been shown to induce IL-10 production by APCs.²⁶²

Together, these data suggest that the addition of α -GalCer to whole PBMCs from patients with CLL is not likely to result in a cytokine milieu favouring activation of innate cells and cytotoxicity, and may instead result in the production of immunosuppressive cytokines.

Clinical data do not lend support to the concept of systemic α -GalCer administration for the immunotherapy of cancer: In a phase 1 clinical trial of intravenous α -GalCer administration to patients with solid tumours, only one of 24 patients had a persistent rise in serum IFN- γ and IL-12 levels. The one responding patient had a relatively high number of circulating iNKT cells before α -GalCer administration, at around 1000 iNKT cells/mL blood.³⁰³

Maturation of APCs Next, I looked for evidence that α -GalCer could induce APC activation in cultures of whole human PBMCs. A preliminary experiment demonstrated α -GalCer-induced upregulation of CD86 and HLA-DR on monocytes in PBMCs from a donor with a high proportion of iNKT cells. However, this effect was not consistently observed in healthy donors with more typical proportions of iNKT cells (figure 6.3). I concluded that I would be unlikely to be able to demonstrate α -GalCer-induced APC activation in PBMCs from patients with CLL.

B cells are APCs in their own right, and the administration of systemic α -GalCer to mice results in early B cell activation.⁵²⁰ I briefly explored this, by examining B cell activation markers after α -GalCer treatment of PBMCs from a donor with a high proportion of iNKT cells. This experiment (figure 6.4) demonstrated slight upregulation of B cell CD54 and CD86 expression 60 hours after α -GalCer administration. The finding that significant B cell upregulation of activation markers was not observed at the earlier eight and 24 hour timepoints, and that class II expression was not upregulated, suggested that any direct effect of iNKT cells on B cell activation within human PBMCs was weak, even when studied using PBMCs from a donor with a high frequency of iNKT cells. I therefore chose not to explore this experimental line further in patients, in whom iNKT cells are typically found at a much lower frequency in PBMCs.

In contrast to the weak effects on α -GalCer on B cell activation in mixed PBMC cultures, the co-culture of purified iNKT cells with autologous CLL cells at a 1:1 to 4:1 ratio led

to upregulation of CD80 and CD86 on the leukemic cells (figure 6.5). Interestingly, this effect was not α -GalCer-dependent.

The purified iNKT cell lines used had been generated in the presence of IL-2 and using repeated TCR stimulation, conditions which have been associated with surface CD40 ligand (also referred to as CD40L or CD154) expression by CD4+ T cells.⁴⁶⁴ CD154 expression by T cells can mediate CLL cell activation via CD40.⁵²¹ Flow cytometry confirmed that the purified patient iNKT cell lines expressed surface CD154 and that CLL cells expressed CD40 (figure 6.6). The degree of CD154 expression was dim, although I did not assess intracellular CD154 in the iNKT cells, which may have been present at a higher level than surface CD154. My data do not prove that a CD40/CD154 interaction is occurring, but this is one possible explanation for the observed α -GalCer-independent CLL cell activation by iNKT cells. CD40L (CD154) expression on iNKT cells, and α -GalCer-independent activation of B cells by iNKT cells have been reported elsewhere—my findings are consistent with this.⁵²²

Cytotoxicity As discussed in section 1.3.3.2, the α -GalCer-dependent lysis of human CD1d-expressing target cells, including myelomonocytic leukaemia and CLL cells, by iNKT cells has been demonstrated *in vitro*.^{229,417} Moreover, pre-clinical studies demonstrate the *in vivo* potential of iNKT cells to eradicate CD1d-expressing lymphoid malignancies.^{309,310}

Having demonstrated that a patient-derived iNKT cell line could lyse a CD1d-transfected target cell line (figure 5.20), I performed experiments aimed at demonstrating lysis of leukaemic CLL cells by patient-derived iNKT cells. Using an LDH release assay, I found some evidence of lysis of CLL cells by an autologous iNKT cell line (figure 6.8a). However, lysis was inconsistent when tested in a larger series of effector:target pairs (figure 6.8b). This contrasts with the effective lysis of CD1d-transfected C1R cells by iNKT cell lines (figure 5.20).

A number of reasons may account for the apparent resistance of CLL cells to killing by iNKT cells. Compared to CD1d-transfected C1R target cells, CLL cells express CD1d at a very low level (figure 4.1b and data not shown). As discussed in section 1.2.2.3, iNKT cells express NK inhibitory and activating receptors, and CLL cells are characteristically resistant to NK cell lysis.⁵²³ Unlike C1R cells, CLL cells would be expected to express the inhibitory NK receptor ligand HLA-C. CLL cells have also been reported to express other ligands for inhibitory NK receptors such as HLA-E⁵²³ and HLA-G.³³⁷

Methodological difficulties may have contributed to my inability to conclusively demonstrate killing of CLL cells by iNKT cells. The cytotoxicity experiments were limited by the inefficiency of lysis of CLL cells by iNKT cells, the limited numbers of iNKT effector cells available from preceding cultures, and the requirement for large numbers of targets (and therefore effectors) using the LDH release assay. I therefore considered ways to enhance the susceptibility of CLL cells to lysis by iNKT cells.

Expression of CD1d on target cells is necessary for *in vivo* lysis by iNKT cells in the presence of α -GalCer,³¹⁰ the level of CD1d expression determining the susceptibility of the target cell to lysis.³⁰⁹ As CLL cells express CD1d at a relatively low level, I wished to determine whether CD1d could be upregulated, to enhance their susceptibility to lysis.

Upregulation of MHC Class I by IFN- γ is well-described, an effect which I was able to confirm in primary CLL cells (figure 6.9b). I hypothesised that IFN- γ would also result in upregulation of CD1d on CLL cells, potentially enhancing their susceptibility to lysis by iNKT cells. However, treatment of CLL cells with IFN- γ did not result in upregulation of CD1d (figure 6.9c).

A number of other methods for assessing cytotoxicity against CLL cells have been reported, some of which are listed in table 7.6. I performed preliminary experiments using dihexyloxacarbocyanine iodide (DiOC6) and Annexin V staining to detect CLL cell killing by iNKT cells, but these killing assays proved difficult to interpret, the results being highly dependent upon the precise flow cytometric gating strategy used. Flow cytometric detection of apoptotic cells demands careful titration of incubation times, and highly consistent flow cytometric thresholding and gating, as cells which have been reduced to debris would no longer be detectable within a typical lymphocyte forward and side scatter gate by flow cytometry. This could lead to under- or overestimation of cell lysis, depending on the flow cytometric gates or thresholds used.

A more sensitive release assay, such as ⁵¹Cr or calcein AM release, may be more suitable assays for the detection of killing of CLL cells. Like LDH release, these techniques detect completed target cell lysis (irrespective of the mechanism of target cell death), but unlike LDH release, require relatively small numbers of target (and therefore effector) cells. Neither assay relies on flow cytometric gating.

Although I might have been able to demonstrate killing of CLL cells by autologous iNKT cells had I used an alternative killing assay, I do not believe that iNKT cells would prove useful effector cells against CLL cells in clinical practice for several reasons. First, iNKT cells do not specifically lyse leukaemic cells—CLL cells express CD1d at a lower level than normal monocytes and mDCs, and at the same level as normal B cells, suggesting that these non-malignant cells would be at least as susceptible to killing by iNKT cells. Second, lysis of CLL cells by iNKT cells is inefficient, requiring high effector:target ratios to achieve modest degrees of specific lysis. Third, iNKT cells are present at very low frequencies in patient PBMCs, suggesting that their effective use as effector cells would require their expansion, potentially demanding prolonged *ex vivo* culture.

For these reasons, I opted to assess other potential roles of iNKT cells in the cellular immunotherapy of CLL.

7.2.5.2 iNKT cells as adjuvants

As outlined in section 1.2.6 and illustrated in figure 1.6, α -GalCer-activated iNKT cells can act as adjuvants to peptide antigens. The co-administration of a peptide antigen with α -GalCer results in presentation of both antigens on the same DC. Recognition of α -GalCer on DC-expressed CD1d by iNKT cells leads to a reciprocal interaction between the DC and iNKT cell that leads to upregulation of costimulatory markers on the DC and production of IL-12.^{242,258} This DC activation (or 'licensing') enhances 'conventional' T cell responses against the peptide antigen co-presented on MHC.^{258–260} This adjuvant activity is critical in models of α -GalCer-based immunotherapy for B cell malignancies, where the administration of α -GalCer-treated whole tumour cells leads to enhanced CD4+ and

Table 7.6 Methods of assessing cytotoxicity against CLL cells

Method	Comments	Reference(s)
⁵¹ Cr release	Gold standard assay Few target and effector cells required Detects completed lysis of targets Utilises toxic radioisotope	337,360,361,524
LDH release	Many target and effector cells required Requires careful controls, as effectors as well as targets can release LDH Detects completed lysis of targets	372
Calcein AM release	Few target and effector cells required Detects completed lysis of targets	337,477
Annexin V staining	Flow cytometric method Few target and effector cells required	525
DiOC6 staining	Flow cytometric method Few target and effector cells required	417,526
Fluorescent caspase substrates	Flow cytometric method Few target and effector cells required Specific for apoptosis induction	337
CD107a upregulation	Measures effector cell degranulation, not killing of target cells	337,527
Granzyme B ELISpot	Measures effector cell granzyme B release, not killing of target cells Sensitive assay	325,366

CD8+ protection against the malignancy. 424,425

I therefore sought to assess the potential for $\alpha\mbox{-}Gal\mbox{Cer}$ as an adjuvant in patients with CLL.

Vaccine recall responses present a simple and potentially clinically-relevant method of assessing responses to a peptide antigen. Initially I used the incorporation of radiolabelled thymidine to assess antigen-specific T cell proliferation.⁵²⁸ This assesses proliferation in response to antigen, but does not identify the dividing cells within whole PBMCs. An initial experiment suggested that the addition of α -GalCer can augment recall responses to both influenza and diphtheria/tetanus vaccines, at least in a donor with a high proportion of iNKT cells (figure 6.10).

The diphtheria/tetanus vaccine used (ADTTM Booster) contains an aluminium hydroxide adjuvant. To avoid the possibility of enhancement of iNKT cell proliferation by aluminium hydroxide, potentially confounding the analysis, I chose to perform further assays using the influenza vaccine (VaxigripTM), which comprises formaldehyde-inactivated influenza virus particles, without an added adjuvant. Donors who had been previously vaccinated with the relevant year's influenza vaccine were selected for this analysis.

The use of CFSE to assess lymphocyte proliferation allows simultaneous assessment of the immunophenotype of dividing cells. 529,530 CFSE dilution indicated that the proliferative response of conventional T cells to influenza vaccine was indeed increased by α -GalCer, but that a reciprocal interaction also occurred: influenza vaccine increased the proliferative response of iNKT cells to α -GalCer (figure 6.11).

When a larger number of controls and patients were studied, the effect of α -GalCer on the influenza vaccine recall response was much more mixed, with a statistically significant reduction in ³H thymidine incorporation after α -GalCer addition in one patient (figure 6.12). Methodological factors may have affected this result, however: it is notable that the two donors with the most marked proliferative responses in response to vaccine alone showed a trend towards reduced proliferation when α -GalCer was added. It is possible that the addition of α -GalCer augmented early T and/or iNKT cell proliferation sufficiently to result in exhaustion of the culture media, and that by the time ³H thymidine was added to the cultures at day five, the cells were already dying. Further experiments using limiting dilutions of the vaccine, or shorter culture periods, would have to be performed to investigate this possibility.

Possible mechanisms for the enhancement of 'conventional' T cell proliferation in response to α -GalCer addition include cytokine production in response to α -GalCer or α -GalCer-induced maturation of APCs presenting vaccine-derived peptides. The experiments presented here cannot distinguish these two mechanisms. As discussed in section 1.2.4, iNKT cells can produce cytokines such as IL-2 and IL-21, which may directly support the proliferation of 'conventional' T or NK cells, cytokines such as GM-CSF and TNF- α that may activate DCs, and cytokines such as IFN- γ and IL-4 that may polarize responding 'conventional' T cells. iNKT cells can also interact with DCs directly, via CD40/CD40L, to activate DCs and enhance immunological responses to co-presented peptide agonists (see section 1.2.6.1). Had the adjuvant effect of α -GalCer been more consistent, I could have designed Transwell studies to determine the relative contribution of cytokine-mediated and contact-dependent pathways. Together, the experiments I performed suggest that although α -GalCer has the potential to augment vaccine recall responses in whole human PBMCs, this effect is small or insignificant in the majority of the older individuals in this study, in whom iNKT cells constitute a very small proportion of T cells.

7.2.5.3 CLL cells as a whole tumour vaccine

Proliferation of iNKT cells in response to α **-GalCer-pulsed CLL cells** Potential sources of TAAs include whole tumour cells, tumour lysates, extracted or synthetic proteins or peptides, and tumour RNA. I chose to employ whole tumour (leukaemic) cells, as these were readily available, incorporate numerous potential antigens, and, as demonstrated in section 4.3.1, express CD1d.

First, I sought to assess whether CLL cells pulsed with α -GalCer, could induce iNKT cell proliferation. These experiments confirmed that α -GalCer-pulsed CLL cells led to proliferation of iNKT cells within whole PBMC cultures, both in the autologous and allogeneic settings (figures 6.13a and 6.14). The use of the autologous setting is important, as iNKT cells can express activating KIRs, which contribute to alloreactivity:²²⁸ The finding that autologous α -GalCer-pulsed CLL cells induce iNKT cell proliferation demonstrates that such KIR signals are not essential for CLL-induced iNKT cell proliferation.

Several potential mechanisms might account for the proliferation of iNKT cells induced by irradiated, α -GalCer-pulsed CLL cells. First, irradiated CLL cells, which express CD1d, may present α -GalCer directly to iNKT cells. Second, the irradiated CLL cells may be phagocytosed by other APCs within the PBMC cultures, resulting in re-presentation of the CLL-cell associated α -GalCer by CD1d-expressing APCs such as mDCs or monocytes. Third, the α -GalCer may dissociate from the irradiated CLL cells—free α -GalCer may then be taken up and presented by other APCs within the PBMC cultures.

The data presented in this thesis do not allow differentiation of these mechanisms. Experiments using defined cell populations, for instance assessing proliferation after adding α -GalCer-pulsed CLL cells to purified autologous iNKT cells, would demonstrate whether irradiated CLL cells are capable of direct α -GalCer presentation to iNKT cells. The dissociation of α -GalCer from the CLL cells in culture could be excluded using transwell experiments.

Effect of α -GalCer on proliferation of PBMCs in response to allogeneic CLL cells Having demonstrated that iNKT cell proliferation occurs in response to α -GalCer-pulsed CLL cells, it was not surprising that α -GalCer treatment led to increased proliferation of allogeneic PBMCs in response to CLL cells, as assessed by ³H thymidine incorporation in responding PBMCs (figure 6.15).

In a preclinical model, the combination of α -GalCer and a TLR ligand results in a substantial potentiation of the immune response.³¹¹ In human PBMCs, TLR ligands have been shown to enhance presentation of α -GalCer by DCs, and enhance iNKT cell IFN- γ production.⁴³⁴

CLL cells express a range of TLRs, including TLR7.³⁷⁰ TLR7 agonists such as loxoribine and the imidazoquinoline S28690 result in upregulation of activation markers on CLL cells.⁴⁷⁰ To explore the effect of combining a TLR ligand and α -GalCer, I used the TLR7/8 agonist, resiquimod. Flow cytometric analyses confirmed that resiquimod leads to CLL cell upregulation of CD80 and CD86, and demonstrated that, while resiquimod led to a significant and consistent increase in MHC Class I and II expression on CLL cells, α -GalCer had no effect on expression of these markers (figures 6.16 and 6.17).

Resiquimod- and α -GalCer-treatment of CLL cells appeared to augment proliferation of allogeneic PBMCs in an independent fashion (figure 6.18). Analysing the numbers of T cells in CLL-stimulated PBMCs, α -GalCer treatment resulted in iNKT cell proliferation (as would be expected), while resiquimod treatment resulted in increased numbers of 'conventional' CD4 and CD8 T cells. There was no consistent effect of α -GalCer treatment on the number of CD4 and CD8 'conventional' T cell numbers in this experiment (figure 6.19).

 α -GalCer treatment of CLL cells leads to increased proliferation of autologous 'conventional' T cells In a subsequent experiment, I examined the effect of α -GalCer treatment of CLL cells on the *autologous* T cell response using CFSE dilution (figure 6.20). This confirmed that α -GalCer-treated CLL cells led to proliferation of autologous iNKT cells.

A modest increase in proliferation of 'conventional' T cells was also seen in the α -GalCertreated group. One explanation for this observation is that the dividing 'conventional' T cells represent leukemia-reactive T cells, and that their proliferation is enhanced by α -GalCer treatment of the leukemic cells. This would be consistent with the findings of preclinical studies of α -GalCer-treated tumour cells for the immunotherapy of B cell malignancies.^{424,425} However, the enhancement of proliferation observed in my experiment was modest, and I was not able to demonstrate anti-tumour activity of the divided T cells due to the very small number of cells available.

An alternative explanation would be that iNKT and CLL cells both enhance background T cell proliferation in the cultures. A modest degree of spontaneous T cell proliferation is seen in the medium-only control (figure 6.20a), and could represent T cell proliferation in response to culture contaminants or to contents of the media, such as human serum. Irradiated CLL cells might enhance this background proliferation through release of stimulatory cytokines, and this effect could be enhanced further by iNKT cell-produced cytokines, such as IL-2, in the α -GalCer-treated CLL cell group.

The increased proliferation of 'conventional' T cells in response to α -GalCer treatment of CLL cells in the autologous but not the allogeneic setting was unexpected. The impact of NK alloreactivity may provide a possible explanation for the failure to demonstrate an effect of α -GalCer in the allogeneic setting: In the allogeneic setting, NK alloreactivity would be expected to provide strong early cytokine signals even in the absence of α -GalCer—iNKT cells, which are far less abundant in PBMCs than NK cells, are redundant. In the autologous setting, however, no NK alloreactivity occurs, so the impact of iNKT cell recruitment by α -GalCer becomes significant.

7.2.6 Strengths and limitations of this study

This is the first study, of which I am aware, to characterise iNKT cells from patients with CLL. The previous section reviewed the findings of this study, discussing the methodology, and placing the findings in the context of the published literature. In this section, I shall consider the design and methodological strengths and weaknesses of this study.

Participant characteristics and numbers The frequency of circulating iNKT cells declines with age, ^{210,223,441} hence the importance of the close age-matching between patient and controls in this study. By excluding patients and controls with active malignancy or who are taking systemic immunosuppressive medications, these potential confounders were avoided. Nonetheless, other confounders may exist: both patients and controls who had both the capacity and motivation to attend a research institute to provide blood samples.

There is a wide variation in iNKT cell numbers in healthy humans. This requires the analysis of large numbers of individuals to detect statistically significant differences in numbers of iNKT cells between different groups. I recruited sufficient participants in this study to demonstrate statistical significance in the difference in the proportion of iNKT cells (as a percentage of T cells), but did not find significant differences in absolute iNKT cell numbers or in iNKT cell immunophenotype. It is possible that additional differences would have emerged had a larger number of patients and/or controls been studied.

The numbers of patients within the smaller subgroups who had clinically-advanced or fludarabine-treated CLL were small. As such, differences in iNKT cell numbers or function in these groups may have been missed. Nonetheless, this study may be considered to provide a 'proof of principle' that iNKT cells are present and can proliferate in such patients.

Limitations on the numbers of cryopreserved PBMCs, availability of reagents, and time precluded performance of the full suite of functional iNKT cell analyses in all patient and control samples. Had I performed functional analyses on all samples, it is possible that differences in iNKT cell function would have emerged. For example, although I observed no statistically-significant differences in *ex vivo* iNKT cell proliferation between patients and controls, a trend towards reduced proliferation in patients was observed. Had more samples been included in this analysis, it is possible that a statistically-significant differences. Nonetheless, although I cannot exclude small differences, the studies of iNKT cell function performed effectively rule out severe defects in iNKT cell function in patients with CLL.

Cross-sectional design CLL is an indolent malignancy, the progression from early CLL to advanced disease requiring therapy typically taking many years. As such, time limitations demanded a cross-sectional rather than longitudinal design for this study, taking individual samples from patients at a fixed point in time, rather than serial samples from patients before and after disease progression and treatment. Performing longitudinal studies in such an indolent malignancy would likely require a tissue bank containing large numbers of cryopreserved PBMCs, which was not available.
A longitudinal study would be better suited to the exploration of relationships between CLL progression and iNKT cell numbers, phenotype and function, by allowing the use of more powerful paired or repeated-measures statistical tests to compare iNKT cell parameters before and after disease progression or treatment.

The cross-sectional design of this study does carry an advantage: as a single individual was responsible for the collection of all clinical data, and for the collection, separation and cryopreservation of both patient and control samples, variability in sample handling was minimised.

Peripheral blood analysis This study has examined iNKT cells within the peripheral blood. In common with most other translational studies examining T cell responses, I chose to study peripheral blood because of its ready availability, and because it simultaneously provided a source of both viable immune system cells and leukaemic cells suitable for phenotypic and functional analyses.

As discussed in section 1.2.2.1, iNKT cells can be detected within organs, and are particularly enriched in the liver.²¹² It is plausible that differential migration of iNKT cells may occur in disease states such as CLL: an apparent reduction of iNKT cells observed in the peripheral blood may actually represent migration of the cells into solid tissues. Published reports offer potential mechanisms for such iNKT cell migration: in mice, iNKT cells express CCR4, which is required for their migration into the lung,⁵³¹ while human CLL cells can express CCL22, a ligand for CCR4.⁵³² Thus, a limitation of this study is that the peripheral blood iNKT cells analysed may not be numerically, immunophenotypically or functionally representative of tissue iNKT cells.

The results of clinical trials fail to clarify the relevance of peripheral blood iNKT cell numbers to the efficacy of iNKT cell-directed immunotherapy: in a phase I study of intravenous α -GalCer in patients with cancer, cytokine responses to α -GalCer were restricted to those with the highest pre-treatment numbers of circulating iNKT cells.³⁰³ In contrast, in a phase I/II clinical study of vaccination with α -GalCer-pulsed APCs, the number of circulating iNKT cells before treatment bore little relation to the number of α -GalCer-induced IFN- γ ELISpots after vaccination.⁴¹¹

Thus, although peripheral blood provided a convenient source of iNKT cells for analysis, the relevance of the circulating iNKT cells I characterised to the efficacy of any iNKT cell-based immunotherapy remains unclear.

Applicability of findings from iNKT cell lines The methodological challenges experienced in the generation of iNKT cell lines have been discussed in section 7.2.4.3. In comparison to circulating iNKT cells, the cell lines generated were biased towards CD4 expression and IL-4 production, suggesting that they are not fully representative of *ex vivo* iNKT cells. However, iNKT cell numbers were too small to allow direct *ex vivo* isolation and functional assessment, and the data on iNKT cell lines does provide a proof of principle that functional lines can be derived from patients with CLL.

As discussed in section 7.2.4.3, is possible that iNKT lines generated in the presence of cytokines other than IL-2, such as IL-7 or IL-15, may have exhibited a different cytokine

Chapter 7. Discussion

profile, or may have exhibited better long-term viability during prolonged culture. However, time and the limited numbers of patient PBMCs available prohibited an exhaustive exploration of iNKT cell line generation.

Use of PBMCs for functional assays Many of the functional assays performed used mixed PBMCs, the rationale of which was to model the impact of iNKT cell engagement *in vivo*, where iNKT cells represent a very small proportion of lymphocytes, and where the effects of iNKT cell activation may be modulated by interactions with other cell types.

These assays failed to demonstrate significant cytokine production, APC maturation or vaccine recall response augmentation in response to α -GalCer. It seems likely that had iNKT cells been present at a sufficiently high frequency, these effects would have been demonstrable, as they were observed in a donor with a high proportion of iNKT cells. However, the aim of these experiments was to help select the most promising means to exploit the iNKT cell/CD1d axis in patients, in whom circulating iNKT cells are rare.

Although mixed PBMCs include a mixed population of lymphocytes and APCs, they exclude a number of cells which may interact with iNKT cells *in vivo*: interactions between iNKT cells and granulocytes, ^{277,279} hepatocytes, ⁵³³ and even atherosclerotic plaques ⁵³⁴ have been described. In addition, the microenvironment of a cell culture well differs substantially from any condition found *in vivo*, in terms of cytokine, hormone and metabolite concentrations, cell motility and concentration, and physical shear. The *in vitro* effect of iNKT cell activation may not, therefore, be representative of that which would be observed *in vivo*. Where possible, I have sought to correlate my *in vitro* findings in human PBMCs with reported *in vivo* findings in mice, and with findings in human clinical studies.

Effects of α -GalCer-pulsed CLL cells I have demonstrated that CLL cells are able to retain α -GalCer and to induce the proliferation of iNKT cells. It is possible that monocytes or mDCs contaminating the purified CLL cells were responsible for the carriage of α -GalCer in these experiments. However, the effect was still readily detectable when highly-purified (FACS-selected) CLL cells were used.

That α -GalCer-pulsed CLL cells can result in iNKT cell proliferation in the autologous as well as allogeneic setting is important, as it suggests a potential immunotherapeutic strategy. Moreover, this effect could be detected after after irradiation, washing and cryopreservation of the CLL cells. The rationale of irradiating and cryopreserving the CLL cells in this experiment was to mimic the preparation of autologous tumour cells as a whole tumour vaccine in a putative future phase I study.

Although the enhanced proliferation of 'conventional' T cells in response to α -GalCertreated leukaemic cells shown in figure 6.20 could indicate enhanced antitumour responses, it is possible that the proliferating T cells are not tumour-specific, or even that they are tumour-specific but are immunosuppressive rather than cytotoxic. I was unable to functionally characterise the T cells that had proliferated in response to α -GalCertreated CLL cells because of the small numbers of cells available, and methodological difficulties with the chosen cytotoxicity assay (LDH release). The limited sensitivity of the LDH release assay for the assessment of cytotoxicity against CLL cells has already been discussed, and alternative assays discussed in section 7.2.5.1. Nonetheless, the efficacy of α -GalCer-pulsed tumour cells as a vaccine against B cell and myeloid malignancies has been demonstrated in pre-clinical studies (section 1.3.3.3).

Despite the limitations of this study, this is the first study, of which the author is aware, to assess the number and function of iNKT cells from patients with CLL. It establishes that iNKT cells are detectable in nearly all patients with CLL, including most of those who have previously received fludarabine-containing chemotherapies, that iNKT cell immunophenotype and cytokine profile is intact, and that patient iNKT cells are capable of proliferation. Patient APCs express CD1d at normal levels. This study provides 'proof of principle' that functional iNKT cell lines can be derived from patients with CLL. Through a series of *in vitro* experiments, this study suggests that the low frequency of iNKT cells in patients may limit responses to systemic α -GalCer, but that iNKT proliferation can be induced by α -GalCer-treated autologous leukaemic cells. Finally, this study suggests that α -GalCer treatment of autologous leukaemic cells may enhance the proliferation of non-CD1d restricted T cells in response to tumour.

7.3 Cellular immunotherapy of CLL

7.3.1 Factors associated with successful cellular immunotherapy

In this section, I shall outline factors which have been associated with successful immunotherapy of cancer in humans.

Clinical trials of therapeutic cancer vaccination and adoptive immunotherapies can offer useful insights. For example, a review of patients who received adoptive transfer of *in vitro*-expanded tumour-infiltrating lymphocytes for the treatment of metastatic melanoma offers some insight into the role of prior lymphodepleting chemotherapy.⁵³⁵ Adoptive transfer of autologous T cells bearing a chimeric antigen receptor (CAR) directed against CD19 has been used for the treatment of B cell malignancies in phase 1 trials, and although clinical experience to date is limited, a systematic review has identified factors associated with successful treatment.⁵³⁶

Allogeneic stem cell transplantation (allo-SCT) has been in routine clinical use for the treatment of haematological malignancies since the 1970s, and been progressively refined as factors associated with long-term disease-free survival have been identified. The large numbers of patients included in registry studies, clinical trials and case series may provide useful indications of the factors associated with successful immunotherapy.

Susceptible malignancy Early experience with *in vivo* and *in vitro* use of IL-2 suggested that melanoma and renal carcinoma were the most immunogenic solid malignancies, ⁵³⁷ and the impressive response rates following adoptive immunotherapy of melanoma would seem to validate this. ⁵³⁵ However, some of the most convincing outcome data for therapeutic cancer vaccination are for prostate cancer, ⁵³⁸ and transgenic T cells have shown

promising activity against tumour types that were not traditionally regarded as highly immunogenic, such as ovarian and synovial cell carcinoma.^{385,539}

Clinical experience of allo-SCT suggests that some haematological malignancies are more readily eradicated by this procedure than others. Chronic myeloid leukaemia has a particularly high disease-free survival rates after allo-SCT.⁵⁴⁰ In contrast, the benefit of allo-SCT for myeloma is more controversial.⁵⁴¹ CLL, like most other indolent B cell malignancies (with the possible exception of myeloma) appears to be susceptible to allo-SCT.⁵⁴²

The experience of allo-SCT for the treatment of non-haematologic malignancies, in contrast, has proved disappointing, with response rates to reduced intensity conditioning allogeneic stem cell transplantion (RIC-allo) generally lower than for haematologic malignancies, ⁵⁴³ with the possible exception of metastatic renal cell carcinoma. ⁵⁴⁴ Possible reasons for the failure of most non-haematologic malignancies to respond to allo-SCT include:

- Lack of disease remission: The outcome of allo-SCT for haematopoietic malignancies is superior if the patient is transplanted when their disease is in remission.⁵⁴⁵ Whether because of inherent chemoresistance, the inability of chemotherapies to penetrate solid tumours, or other microenvironmental factors, the complete remission rates after chemotherapy for most metastatic solid carcinomas are lower than for most haematological malignancies.
- Unfavourable cytogenetic features: the outcome of patients receiving allo-SCT for refractory leukaemia not in remission is particularly poor in those with poor-risk cytogenetic changes.⁵⁴⁶ Unfavourable cytogenetic changes may be more frequent in solid tumours: whole genome sequencing suggests that the numbers of mutations in protein-coding regions are greater in solid cancers than in acute myeloid leukaemia.⁵⁴⁷
- Lack of MHC class II expression: Most haematopoietic tumours express MHC class II: B cells and myeloid APCs express MHC class II constitutively, while T cells express class II upon activation.⁵⁴⁸ In contrast, most non-haematopoietic cells do not express MHC class II (although class II expression has been demonstrated on astrocytes, ⁵⁴⁹ endothelial cells, ⁵⁵⁰ activated hepatocytes, ⁵⁵¹ melanomas ⁵⁵² and adipocytes ⁵⁵³). It is tempting to speculate that following allo-SCT, malignant tumours of haematopoietic origin present tumour-associated antigens on MHC class II as well as on MHC class I, and that this contributes to tumour eradication by donor T cells. Consistent with this notion, the transfection of human carcinoma cells with MHC class II and CD80 allows them to present MHC class II-restricted endogenous antigens to tumour-specific CD4+ T lymphocytes. ⁵⁵⁴

Whatever the reasons, some tumours appear more susceptible to eradication by certain immunotherapies than others. It is likely that, as for chemotherapeutic agents, different tumours will respond best to different immunotherapeutic strategies.

Disease remission status Many tumour immunotherapy studies are undertaken in patients with active, often bulky, malignancies. However, experience of allo-SCT suggests that disease remission at the time of this adoptive immunotherapy is associated with

markedly improved survival allo-SCT.^{545,555} There are numerous potential explanations for this observation:

- Chemosensitivity of the malignancy: malignancies which are resistant to pretransplant chemotherapies are likely to be resistant to transplant conditioning regimens
- Inherent aggressive behaviour of the malignancy: remission status may be a proxy for poor-risk features, such as adverse cytogenetic changes within the tumour
- Kinetics of tumour growth: rapidly-growing malignancies may be over-represented among cases not in remission, and may progress before donor T cells engraft
- Patient frailty: frail patients, those with co-morbidities or with a history of severe infections may be over-represented among those with refractory disease as they receive less intensive chemotherapy, and may also be at higher risk of transplant-related mortality
- Resistance to immunological eradication: refractory malignancies may be those which lack susceptibility to immunological control, due either to tumour factors (such as MHC Class I downregulation or HLA-G expression) or to features of the tumour microenvironment (such as the presence of intratumoural myeloid-derived suppressor cells)

Whether achieving remission truly makes a subsequent immunotherapy more effective, or whether it is simply a proxy for a treatment-sensitive tumour or a 'fit' patient, is moot. Nonetheless, experience from allo-SCT would suggest that cellular immunotherapies might be most effective in patients who are in remission prior to treatment. The fact that early clinical trials tend to be performed in patients with active malignancy may contribute to the disappointing outcomes reported in clinical trials of therapeutic cancer vaccination to date.⁵³⁸

Sufficient T cell repertoire The T cell-mediated eradication of malignant cells presumably requires a T cell repertoire capable of recognising at least one TAA. However, with advancing age, human CD4+ and CD8+ T cell populations become skewed and develop oligoclonal expansions.⁵⁵⁶ In aging mice, similar changes occur, and result in demonstrable gaps in the T cell repertoire.⁵⁵⁷

The presence of T cell specific for a single tumour-associated antigen is not sufficient for disease control: among patients vaccinated against a melanoma-associated antigen, disease progression was documented even in patients with antigen-specific CD8+ T cells accounting for > 10% of the CD8+ T cell repertoire.⁵⁵⁸

The presence of 'helper' CD4+ T cells as well as 'cytotoxic' CD8+ T cells recognising TAAs may contribute to successful immunotherapy: clinical responses were higher among patients vaccinated with DCs bearing MHC Class I and II epitopes than among patients vaccinated with DCs bearing Class I epitopes alone.⁵⁵⁹ One clinical case of dramatic regression of metastatic melanoma following adoptive transfer of MHC Class II-restricted tumour-infiltrating lymphocytes provides proof of principle that CD4+ T cells can have a role in tumour eradication.⁵⁶⁰

Prolonged disease-free survival after allo-SCT requires T cells: in registry studies of allo-SCT, depletion of T cells from the haematopoietic stem cell graft results in significantly higher leukaemia relapse rates after allo-SCT.⁵⁶¹

Consistent with the notion that a broad T cell repertoire is required for optimal tumour control, the age of human haematopoietic stem cell donors has a significant impact on disease-free survival after allo-SCT: in registry studies, the use of older donors is associated with higher relapse rates in the recipient.⁵⁶² One explanation for this would be that haematopoietic stem cell grafts from older donors have relatively skewed and deficient T cell repertoires, and as such are incapable of mounting broad T cell responses against recipient tumours.

One potential pitfall of any *autologous* cellular immunotherapy is the possibility that patients with cancer have deficient T cell repertoires which are incapable of mounting effective antitumour T cell responses. This might be overcome by using allogeneic (including umbilical cord or haploidentical) rather than autologous T cells, by transfecting patient T cells with modified or chimeric T cell receptors, or by redirecting T cells towards the tumour using bispecific antibodies. The adoptive transfer of T cells transfected with either a TCR specific for a TAA, or with a CAR directed against a TAA, can result in clinical remissions, even among heavily pretreated patients with cancer.^{385,536}

Serial clinical trials of adoptive transfer of *in vitro*-expanded tumour infiltrating lymphocytes, without a modified TCR, have shown clinical response rates of up to 50% in patients with metastatic melanoma.^{535,563} This suggests that some patients with advanced cancer *do* possess a T cell repertoire capable of eradicating their own malignancy following the reversal of an immunosuppressive tumour environment using lymphodepleting chemotherapy and *in vitro* expansion and activation of tumour-infiltrating T cells.

Lymphodepletion Lymphodepleting chemotherapy, such as cyclophosphamide and/or fludarabine, may contribute to the efficacy of the subsequent immune response against the malignancy.

In preclinical models, lymphodepletion before adoptive transfer of T cells or vaccination results in enhanced T cell proliferation. Lymphodepletion is associated with homeostatic increases in serum IL-7 and IL-15 levels, which drive proliferation and persistence of the adoptively transferred or antigen-stimulated T cells.⁵⁶⁴ As in mice, lymphodepletion of human patients results in raised serum levels of IL-7 and IL-15.⁵⁶⁵

Clinical trials of adoptive T cell therapy for melanoma suggest that response rates are higher if the patient receives lymphodepleting chemo- and radiotherapy before adoptive T cell transfer.^{535,536,565} Mechanisms for this effect may include the depletion of immunosuppressive T_{reg} cells as well as the induction of homeostatic T cell proliferation. A systematic review of clinical trials of adoptive transfer of autologous T cells transfected with an anti-CD19 CAR for the treatment of B cell malignancies found better clinical responses in studies that used pre-transfer lymphodepletion.⁵³⁶

Host lymphodepletion before haematopoietic stem cell administration is a critical component of allo-SCT, being required to prevent host rejection of the allogeneic donor stem cells, so whether this is an important contributor to the subsequent immune response against the recipient malignancy is uncertain.

A theoretical concern of lymphodepletion prior to therapeutic cancer vaccination is that it may deplete elements of the patient's T cell repertoire necessary for subsequent tumour eradication. Alternatives to lymphodepletion could include the administration of exogenous cytokines to mimic IL-7 and IL-15-driven 'homeostatic' T cell proliferation.

Disruption of immunological tolerance Immunological tolerance of malignancies can be mediated by T_{reg} cells, MDSCs, by metabolic changes within the tumour microenvironment, or by upregulation of T cell inhibitory molecules, such as CTLA-4 or PD-1.

Several of these mechanisms of tolerance may be amenable to specific disruption in the context of cellular immunotherapy. Depletion of $T_{reg}s$ with the anti-CD25 mAb daclizumab enhanced T cell responses to a subsequently-administered TAA in patients with metastatic breast cancer.⁵⁶⁶ The chemotherapy drug 5-fluorouracil, routinely used for the treatment of colorectal carcinoma, results a in specific depletion of MDSCs in mice.⁵⁶⁷ Finally, anti-CTLA-4 mAbs have demonstrated single-agent activity against melanoma, prostate and renal cell carcinoma in clinical trials.⁵⁶⁸

Allo-SCT may disrupt this tolerance through lymphodepletion, chemo- or radiotherapyinduced modification of the tumour microenvironment, and by reconstitution of the recipient immune system with naïve lymphocytes.

Engagement of innate lymphocytes T cell signals alone may be insufficient to eradicate tumour: the engagement of innate lymphocyte populations such as NK cells (or perhaps iNKT cells) may be important for effective tumour eradication.

The potential efficacy of NK cells as effectors in tumour immunotherapy has been demonstrated: in a phase I clinical trial, activated alloreactive NK cells administered after host lymphodepletion expanded *in vivo*, and in some cases eradicated refractory acute myeloid leukaemia.⁵⁶⁹

Registry data comparing outcomes of allo-SCT from identical twin (syngeneic) with HLAidentical sibling donors shows that the use of a syngeneic donor is associated with a substantially increased relapse risk.⁵⁷⁰ As both types of donor are HLA identical, the implication is that mismatches in other genes, such as minor histocompatibility antigens and/or NK receptors are responsible for the differences in relapse risk. Similarly, in CLL, relapse risk following allo-SCT is lower if an HLA-identical unrelated donor rather than a sibling donor is used.¹⁵⁷ Again, this suggests that factors other than the HLA complex affect the immunological control of haematological malignancies.

The choice of conditioning chemotherapy before allo-SCT may affect subsequent NK cell function: the anti-CD52 mAb, alemtuzumab targets NK cells, whereas polyclonal anti-thymocyte globulin (ATG) does not. Alemtuzumab results in more profound early depletion of NK cells following allo-SCT than does ATG.⁵⁷¹ In myeloma, the use of alemtuzumab is associated with higher subsequent disease relapse rates than ATG).⁵⁷²

Together, these empirical observations suggest that the efficacy of an immunotherapy might be enhanced if innate lymphocytes, such as NK cells, are activated.

Activation of antigen-presenting cells Activated or mature antigen-presenting cells (APCs) are required for the optimal generation of T cell responses. The use of adjuvants that can activate and mature dendritic cells, such as TLR ligands, is common practice in vaccination against infectious diseases.⁵⁷³ Ensuring activation of APCs may also assist the generation of antitumour immunity—in a meta-analysis of clinical trials of DC vaccination for the treatment of prostate cancer, the use of mature DCs was associated with a significantly higher response rate than immature DCs.⁵⁷⁴

Clinical reports suggest that TLR signals can augment anti-tumour immunity. Clinical regression of tumours after administration of bacteria or bacterial toxins has been reported since the nineteenth century.⁵⁷⁵ The intralesional administration of mycobacterial cell wall products is associated with local regression of lesions in patients with sarcoma and melanoma,⁵⁷⁶ while intravesical administration of BCG remains in routine use as an effective therapy for bladder cancer.⁵⁷⁷ Topical imiquimod (a TLR7/8 agonist) induces local remissions in cutaneous carcinomas.⁵⁷⁷ In CLL, lymphomatous skin deposits have also been reported to regress after topical imiquimod administration.³⁵⁷ Finally, rare spontaneous remissions of acute myeloid leukaemia have been associated with episodes of sepsis, an effect which might be TLR-mediated.⁵⁷⁸

TLR ligands appear to be of value when used in conjunction with therapeutic cancer vaccines. The addition of a synthetic deoxycytidyl-deoxyguanosin oligodeoxynucleotide (CpG ODN) to a tumour-associated peptide antigen resulted in improved immunological responses in patients with melanoma.⁵⁷⁹ A change in formulation of incomplete Freund's adjuvant was associated with a decline in immunological responses to a therapeutic peptide vaccine against melanoma.⁵⁸⁰

Dendritic cells treated with the cytokine GM-CSF develop an inflammatory phenotype and induce stronger T cell responses than those generated in the presence of fms-related tyrosine kinase 3 ligand (Flt3L).⁵⁸¹ In men with castration-resistant metastatic prostate cancer, the infusion of PBMCs treated with a fusion protein combining prostatic acid phosphatase (a TAA) with GM-CSF *ex-vivo* led to prolongation of overall survival, compared with infusion of untreated PBMCs.⁵⁸²

Incorporation of the transgenes B7.1, ICAM-1 and LFA-3, which encode three APC-expressed costimulatory molecules, led to enhanced *in vitro* responses to a poxviral gene vaccine for prostate cancer. The combined vaccine improves overall survival in men with hormone-refractory metastatic prostate cancer. ⁵⁸³

DCs express CD40, and can be matured through provision of CD40 ligand or agonistic CD40 antibodies. In a phase I study of patients with myeloma, intranodal administration of DCs pulsed with idiotype protein and matured with CD40 ligand led to immunologic response in all nine patients and stable disease at five years in four patients.⁵⁸⁴

In allo-SCT, the combination of conditioning chemotherapy and/or radiotherapy, and the consequent neutropenia with associated infective episodes, are likely to provide 'danger' signals, resulting in DC activation. As well as potentially stimulating anti-tumour immunity, this may also provoke GVHD: in a xenograft model, depletion of activated DCs by targeting the DC surface activation marker CD83 prevented GVHD.⁵⁸⁵

Together, these clinical observations suggest that the generation of mature activated (or 'licensed') DCs, for example through the co-administation of a vaccine with a TLR ligand,

with GM-CSF or with a CD40-stimulating agent, may support the successful generation of antitumour immunity.

Summary In this section, I have considered how clinical experience of cellular immunotherapies, including allo-SCT, can provide insights into the factors associated with successful tumour immunotherapy. Although many of these clinical studies are uncontrolled, and most deal with malignancies other than CLL, I would tentatively suggest that in broad terms, an immunotherapeutic strategy is more likely to be successful if it:

- Targets a malignancy which is shown to be susceptible to the immunotherapy
- Is employed while the malignancy is in remission
- Is used in recipients who retain a broad T cell repertoire, employs healthy allogeneic T cells, or utilises T cell redirection
- Is preceded by host lymphodepletion
- Is preceded by therapies to disrupt immunological tolerance
- Involves engagement of innate lymphocytes (such as NK cells) as well as T cells
- Activates antigen presenting cells (for example by providing a TLR signal)

Table 7.7 (page 204) summaries these suggestions, and proposes means by which each of these factors might be addressed in practice.

In the next section I shall consider how iNKT cell engagement might help (or hinder) the development of a cellular immunotherapy which meets these requirements.

7.3.2 Potential role of iNKT cell engagement in CLL immunotherapy

In section 7.3.1, I proposed a series of factors associated with successful cellular immunotherapy in clinical practice. In this section, I shall consider how the engagement of iNKT cells might help to address these factors.

Experimental evidence that the engagement of iNKT cells can augment antitumour immunity has been discussed in section 1.3.3 (page 55). Briefly, preclinical models suggest that iNKT cells can enhance antitumour responses through three mechanisms: the transactivation of NK cells leading to enhanced NK cell cytotoxicity against tumour cells, the direct lysis of CD1d-expressing target tumour cells by iNKT cells in the presence of α -GalCer, and the activation (or 'licensing') of α -GalCer-presenting DCs leading to enhanced 'conventional' CD4+ and CD8+ T cell responses against co-presented TAAs.

The data presented in section 6.3.1 indicate that α -GalCer-induced cytokine production in whole PBMC cultures is limited. Moreover, even if administration of α -GalCer to patients could elicit sufficient cytokine production to activate NK cells, NK cytotoxicity against autologous CLL cells is likely to be constrained by CLL cell expression of inhibitory MHC Class I molecules, and by HLA-G expression on CLL cells.³³⁷

In section 5.3.6, I show that iNKT cell lines can be generated from the peripheral blood of patients with CLL. However, these cell lines had a CD4+ phenotype, produced large

Table 7.7 Factors associated with successful cellular immunotherapy

Requirement	Potential solution(s)
Susceptible malignancy	Select appropriate immunotherapy strategy for the disease
Disease in remission	Chemotherapy, radiotherapy or surgery before immunotherapy
T cell repertoire sufficient to recognise tumour	Verify presence of tumour-reactive autologous T cells Use allogeneic T cell source Use transgenic T cells Use redirected T cells
Expansion of tumour-reactive T cells	Lymphodeplete before immunotherapy Administer exogenous cytokines to patient <i>Ex vivo</i> T cell expansion
Disruption of tolerance	Chemotherapy or radiotherapy Anti-CD25, anti-CTLA4, or anti-PD1 mAb
Engagement of innate lymphocytes	Opsonise tumour with mAb Elicit IFN-γ release Inhibit HLA-G
Activation of APCs	Chemotherapy or radiotherapy TLR ligand GM-CSF Irradiated tumour as vaccine

amounts of Th2-type cytokines such as IL-13, and exhibited only modest cytotoxicity against CD1d-transfected C1R cells, and only in the presence of α -GalCer (section 5.3.8). Cytotoxicity against allogeneic and autologous CLL cells was even more limited (section 6.3.4). Thus, I suggest that the notion of using iNKT cells to directly lyse CLL cells is unlikely to prove effective, due to the relatively insensitivity of CLL cells to iNKT cell lysis (perhaps related to the relatively low level of CD1d expression by CLL cells), the low frequency of circulating effector iNKT cells in patients with CLL, and the requirement for CD1d-presented α -GalCer for cell lysis.

I suggest that the capacity of iNKT cells to enhance proliferation of antigen-specific CD4+ and CD8+ T cells appears the most promising role for these cells in cellular immuno-therapy. As discussed in section 1.3.3 (page 55), preclinical models have shown that α -GalCer-pulsed tumour cells can induce 'conventional' CD4+ and CD8+ T cell responses against CD1d-expressing haematologic malignancies. ^{422,424,425}

The data presented in this thesis lend some support to the latter approach in CLL. The data presented in chapter 3 confirm that iNKT cells can be detected in the peripheral blood of patients with CLL, and are of normal phenotype. In chapter 4, I confirm that CLL cells express CD1d, and that CD1d expression on mDCs is normal in patients with CLL. The data presented in chapter 5 show that iNKT cells from patients with CLL exhibit normal cytokine production, and retain proliferative capacity. Finally, in chapter 6, I demonstrate that α -GalCer-pulsed CLL cells are able to recruit iNKT cells in both the allogeneic and autologous settings, and present data suggesting that α -GalCer pulsing of tumour cells may enhance proliferation of non-CD1d restricted 'conventional' T cells.

7.3.3 Limitations of iNKT cell-directed immunotherapies

Two major factors may limit the application of iNKT cells to tumour immunotherapy against CLL: the very low frequency of circulating iNKT cells in humans, and the potential for iNKT cells to exhibit immunosuppressive activity.

Numbers of iNKT cells Both in healthy controls and in patients with CLL, the number of iNKT cells varies widely between individuals (figure 3.7). However, in general, iNKT cells constitute a small proportion of T cells in patients with cancer (table 7.2).

During the experiments presented in this thesis, I found that the low frequency of iNKT cells in PBMCs from patients with CLL hindered *in vitro* analysis of their function. It is possible that the low frequency of iNKT cells in older patients, and in those with cancer in particular, may also severely limit the efficacy of any iNKT cell-based immunotherapy *in vivo*. On the other hand, it is by no means clear that the low frequency of circulating iNKT cells is a barrier to their exploitation.

In vivo, trafficking of T cells through lymphoid organs means that even T cells at a very low frequency are able to come into contact with their cognate antigen, to expand and to promote antibody production—the *in vitro* setting cannot emulate this. Moreover, as discussed in section 1.2.2.1, iNKT cells are enriched in certain organs, including the liver, spleen and thymus, which I was not able to sample. In keeping with this, some clinical data suggest that a low frequency of circulating iNKT cells does not necessarily

preclude a response to an iNKT cell-directed therapy: In a phase I/II clinical study of vaccination with α -GalCer-pulsed APCs, the number of circulating iNKT cells prior to treatment bore little relation to the number of α -GalCer-induced IFN- γ ELISpots following vaccination.⁴¹¹

If iNKT cell numbers do prove limiting, immunotherapy strategies might be designed to induce iNKT cell proliferation. Although the *in vitro* expansion protocol I used in sections 5.3.6 to 5.3.8 was suboptimal, generating CD4+ Th2-type cytokine-producing iNKT cells with limited cytotoxicity, it provides proof of principle that functional iNKT cell lines can be expanded from the peripheral blood of patients with CLL. As discussed in section 7.2.4.3, alternative *in vitro* iNKT cell expansion strategies could be explored. There is also evidence that *in vivo* expansion of human iNKT cells is feasible: the administration of α -GalCer-pulsed moDCs resulted in increased circulating iNKT cell numbers in patients with cancer.^{412,413}

Thus, low iNKT cell frequencies in humans may hinder their successful exploitation for cancer immunotherapy, although this will not be certain until clinical trials are undertaken. *In vitro* or *in vivo* expansion of iNKT cells might help to overcome this limitation.

Immunosuppression by iNKT cells As well as showing the capacity to enhance antitumour responses in some preclinical models, iNKT cells appear to suppress anti-tumour immunity in certain scenarios. Wild-type mice were more susceptible to a CD1dtransfected T cell lymphoma than iNKT cell-deficient strains. In the wild-type (iNKT cell-replete) mice, inoculation with the CD1d-transfected lymphoma was associated with production of the immunosuppressive cytokine, IL-13.³¹³ In another model, mice treated with radiotherapy for a 4T1 mammary carcinoma had a better response in the absence of iNKT cells.³¹²

A number of effector mechanisms may account for the immunosuppressive activity of iNKT cells observed in some settings:

- The CD4+ subpopulation of iNKT cells can produce large quantities of the immunosuppressive Th2 cytokines IL-4, IL-10 and IL-13. Unlike their CD4- counterparts, CD4+ iNKT cells fail to reject a sarcoma in mice.²²² Presentation of α -GalCer by non-professional APCs such as B cells also results in Th2-skewed iNKT cell cytokine production.²⁴⁶
- A population of Foxp3 + iNKT cells can be induced in the presence of TGF-β, and can directly suppress T cell proliferation through a contact-dependent mechanism.²⁶⁴ This may be of relevance in CLL, as the leukaemic cells are known to produce TGF-β in culture, and patients with active CLL have raised plasma TGF-β levels.¹⁰²
- Human iNKT cells can induce monocytes to differentiate into IL-10-producing immunosuppressive APCs.²⁶²
- Type II NKT cells, which are CD1d restricted, but which unlike iNKT cells do not recognise α -GalCer, can suppress antitumour immunity via production of IL-13.³¹⁴

The factors influencing the relative production of Th1 and Th2 cytokines by iNKT cells

have already been discussed in section 1.2.4. Briefly, the activation of iNKT cells in the presence of weak or endogenous ligands, non-professional APCs, high doses of IL-2, or inhibitory NK receptors, tends to favour the production of Th2 cytokines and toler-ance.

Consideration of the mode of iNKT cell activation, and of the cytokine and cellular milieu in which it occurs, may help to avoid the unintended induction of immunosuppressive iNKT cell activity. Selectivity for type I NKT cells (iNKT cells) over immunosuppressive type II NKT cells can be achieved by using α -GalCer for NKT cell activation (table 1.9, figure 1.3). Targeting iNKT cells to professional APCs might be achieved by administering α -GalCer on *in vitro*-generated mature moDCs, or on dying tumour cells (which are then taken up by native DCs). The production of immunosuppressive TGF- β by leukaemic cells may be minimised by the application of iNKT cell-based immunotherapy after induction of disease remission. Finally, as discussed in section 7.2.4.3, if iNKT cells are to be expanded *in vitro*, using cytokines other than IL-2 might avoid the selection of CD4+ Th2 cytokineproducing iNKT cells.⁴³³

Thus, iNKT cells clearly have the potential to be immunosuppressive. Any tumour immunotherapy strategy using iNKT cells should be designed to avoid the induction of immunosuppressive iNKT cells, and clinical trials of iNKT cell-based immunotherapies should include secondary endpoints that detect the induction of tumour tolerance.

Summary Preclinical data suggest that recruitment of iNKT cells with α -GalCer can lead to enhancement of CD4+ and CD8+ 'conventional' T cell responses against tumours, including CD1d-expressing haematologic malignancies. I have established that the iNKT cell axis is functional in patients with CLL, and that α -GalCer-pulsed CLL cells can elicit iNKT cell proliferation and may enhance 'conventional' T cell responses.

Although preclinical data suggest that iNKT cell recruitment may enhance antitumour responses, two factors may limit their use in clinical practice: the very low iNKT cell frequency in human PBMCs, and the immunosuppressive potential of iNKT cells.

The low frequency of iNKT cells is not necessarily a barrier to their effective exploitation in clinical immunotherapy, as the iNKT cell frequency within tissues may be higher, and some immunotherapeutic strategies incorporate *in vitro* or *in vivo* iNKT cell expansion. Immunosuppression by iNKT cells might be minimised by measures to disrupt any immunosuppressive cytokine or cellular environments, and by avoiding iNKT cell expansion strategies which favour Th2 cytokine production.

In the next section, I shall attempt to incorporate these ideas into a proposal for a potential cellular immunotherapy of CLL.

7.3.4 Proposal for a cellular immunotherapy of CLL

As discussed in section 1.3.1, clinical experience indicates that CLL is amenable to immunologic control and eradication. In this section, I propose a strategy for cellular immunotherapy of CLL, which aims to incorporate some of the factors associated with successful immunotherapy suggested in section 7.3.1. Experience from clinical trials of immunotherapy for melanoma, and from allo-SCT, suggests that application of a tumour immunotherapy while the disease is in remission enhances the chance of long-term disease control. The current 'gold standard' of treatment for CLL, at least in patients without the 17p cytogenetic deletion, is chemotherapy with fludarabine, cyclophosphamide and rituximab (FCR), both as initial treatment, ^{5,140} and at the time of relapse. ^{586,587}

Following FCR chemotherapy, T cell depletion lasts at least a year, although NK cell numbers and function are relatively preserved.¹⁷⁵ The combination of fludarabine and cyclophosphamide is commonly used for conditioning prior to allogeneic stem cell transplantation, and has also been used for lymphodepletion prior to T cell transfer in clinical trials of adoptive immunotherapy for cancer.⁵⁸⁸ Fludarabine treatment of patients with CLL leads to a depletion of T_{reg} cell numbers and function,¹⁸² while low-dose cyclophosphamide has also been associated with T_{reg} depletion.

Standard chemoimmunotherapy of CLL with FCR, then, results in a high rate of disease remission, induces lymphodepletion, and may disrupt T_{reg} -mediated immune tolerance. The period following FCR chemotherapy may represent a good opportunity to introduce a cellular immunotherapy.

As discussed in section 1.3.3.3, preclinical data indicate that α -GalCer-pulsed tumour cells can act as a powerful whole-tumour vaccine against experimental B cell malignancies.^{424,425} The translational experiments presented in this thesis indicate that α -GalCer-pulsed CLL cells can lead to proliferation of patient iNKT cells and may enhance the proliferation of autologous non-CD1d restricted T cells.

Leukaemic cells are abundant in the peripheral blood of untreated patients with CLL, and can be readily purified and stored. I propose that treated irradiated CLL cells may provide a useful whole tumour vaccine. An intravenous route of vaccination may be favoured, as preclinical studies indicate that intravenous, but not subcutaneous, injection of α -GalCer-loaded plasmacytoma cells, resulted in CD4+ and CD8+ T cell responses and tumour protection.⁴²⁵ Unlike solid tumours, CLL is always a systemic malignancy, and FCR chemoimmunotherapy is not curative, so intravenous administration does not carry the risk of metastasis.

A variety of strategies for the treatment of CLL cells prior to irradiation and infusion as a whole tumour vaccine could be used:

• **a-GalCer treatment:** Treatment of leukaemic cells with α -GalCer can recruit autologous iNKT cells, which may lead to activation of endogenous DCs presenting CLL tumour-associated antigens, and may enhance subsequent 'conventional' T cell responses against CLL. The experiments in this thesis lend some support to this concept, by demonstrating that patients with CLL have detectable iNKT cells, that CLL cells express CD1d, that patient iNKT cells are able to proliferate and produce cytokines, and that autologous α -GalCer-treated CLL cells can lead to iNKT cell proliferation. Expression of CD1d on CLL cells is low compared to mDCs (figure 4.1), and higher levels of CD1d expression may support the strategy of vaccination with whole α -GalCer-treated tumour cells.⁴²² Although the treatment of CLL cells with IFN- γ did not increase their CD1d expression (figure 6.9), other strategies such as CD1d transfection might enhance the ability of CLL cells to carry and to present

Figure 7.1 A whole tumour cell vaccination strategy for CLL. Leukaemic cells are isolated from the peripheral blood of patients with CLL requiring treatment, are treated to enhance immunogenicity, irradiated and cryopreserved as a whole tumour cell vaccine. The patient is treated with standard fludarabine, cyclophosphamide and rituximab chemoimmunotherapy. After treatment, repeated doses of the tumour vaccine are administered during T cell reconstitution.



 α -GalCer. Finally, I have not been able to conclusively demonstrate that α -GalCer treatment of leukaemic B cells enhances proliferation of tumour-reactive 'conventional' T cells, although this has been shown in pre-clinical models of B cell malignancies. ^{424,425}

- TLR ligand treatment: As discussed in section 1.3.2.2, CLL cells express TLR ligands, and treatment of CLL cells with TLR7 and TLR9 ligands leads to upregulation of costimulatory markers and enhancement of immunologic responses against leukaemic cells.³⁶⁹ Experiments presented in this thesis confirm this (figures 6.16 and 6.18). In pre-clinical models, TLR ligands and α-GalCer can be used together to enhance immunological responses to malignancies,³¹¹ an effect which has also been shown using human cells *in vitro*.⁴³⁴
- **CD40 ligation:** As discussed in section 1.3.2.2, CLL cells express CD40, and treatment with CD40 ligand leads to CLL cell upregulation of costimulatory molecules and enhancement of allogeneic CD4 and CD8 T cell (and autologous CD4 T cell) responses to tumour.^{359,360}

Figure 7.1 illustrates a proposed strategy, in which leukaemic cells would be isolated from the peripheral blood of patients with CLL, possibly including immunomagnetic selection. Isolated leukaemic cells would be treated using one or more of the above strategies, irradiated, and cryopreserved in batches for later use as a whole tumour vaccine. Patients would then receive standard chemoimmunotherapy (FCR) to a maximal response. Following completion of chemotherapy, a series of intravenous vaccinations with treated irradiated tumour cells would be administered.

Chapter 7. Discussion

Figure 7.2 Rationale of tumour vaccination strategy. Patients are immunised with α -GalCerbearing, irradiated leukaemic cells. These are taken up by resident DCs, which present both TAAs on MHC class I and II, and α -GalCer on CD1d. Recognition of α -GalCer results in iNKT cell proliferation and cytokine production, which in turn leads to DC activation. This enhances T cell responses to the co-presented TAAs, resulting in augmented cytotoxicity against residual, unmodified CLL cells.



The rationale for treatment of CLL cells with α -GalCer before whole tumour vaccination is outlined in figure 7.2. α -GalCer-pulsed whole irradiated tumour cells would be taken up by patients' resident DCs. The DCs would present TAAs alongside α -GalCer. Recruitment of iNKT cells by α -GalCer on the DCs would augment DC expression of costimulatory molecules, and enhance proliferation of non-CD1d restricted T cells in response to the co-presented peptide TAAs. These conventional T cells then provide ongoing tumour immunity. The use of repeated doses may be important: Following early vaccinations, iNKT cell numbers might expand, enhancing their adjuvant activity for subsequent vaccine doses. Alternative strategies for enhancing immunogenicity of CLL cells, such as treatment with TLR or CD40 ligands could be used alone or in addition to this α -GalCer treatment.

In additional to clinical outcome measures such as minimal residual disease eradication, laboratory assessments of response to immunotherapy might include analysis of serum cytokines, CLL cell-induced T cell proliferation, CLL cell-induced IFN- γ ELISpot formation, and cytotoxicity against CLL cells, all before and after vaccination.

This tumour vaccination approach has several potential limitations. First, the restricted T cell repertoire of fludarabine-treated patients with CLL may not include sufficient potentially tumour-reactive T cells to eradicate the malignancy. Second, production of potentially-immunosuppressive cytokines in response to treated tumour cells might result in suppression rather than enhancement of the antitumour response. Third, in relation to the use of α -GalCer, an adjuvant effect of iNKT cells has not been reported in humans, and I have not been able to conclusively demonstrate it in this thesis. Fourth, because median overall and progression-free survival are long after first treatment with FCR alone, a very long follow-up period, or the use of a surrogate endpoint, would be required for any study. I shall address each of these limitations in turn.

First, despite their restricted T cell repertoire, translational studies have suggested that tumour-reactive T cells are present in the peripheral blood of patients with CLL, including some of those who have been previously treated with fludarabine-based chemotherapies.^{323,325} Although fludarabine treatment of patients with CLL results in relative preservation of memory T cells compared to naïve T cells, ⁵⁸⁹ T cell receptor gene rearrangement excision circles (TRECs) could be detected in ten of thirteen older adults (median age 58 years) after fludarabine chemotherapy, ⁵⁹⁰ indicating that thymic production of naïve T cells does occur after fludarabine. Together, these findings suggest that, while the numbers of naïve T cells are reduced after fludarabine-containing chemotherapies, mature T cells, which may include CLL-reactive T cells, are retained, and thymic reconstitution of the naïve T cell pool can occur. Finally, indirect evidence suggesting immunological control of CLL by autologous T cells after fludarabine comes from long-term follow-up data for FCR-treated patients with CLL, which show a plateau in progression-free survival curves beyond eight years.^{591,592}

Second, the application of a tumour vaccination strategy after FCR chemotherapy should avoid promoting immunosuppression, as the frequency of TGF- β -producing leukaemic B cells and Tregs would be reduced by pre-vaccination chemotherapy. Although my experiments show that *in vitro* cultured iNKT cells are capable of producing immunosuppressive cytokines, this may be related to the use of high doses of IL-2 during cell culture, as it was not as evident in iNKT cells tested *ex vivo*. By avoiding prolonged *in vitro* iNKT cell culture and high doses of IL-2, a study using a design similar to this should avoid induction of an immunosuppressive phenotype.

Third, the experiments in this thesis show that although at a low frequency, iNKT cells can be detected in the peripheral blood of nearly all patients with CLL, and that they are capable of proliferating *in vitro*, including in response to autologous α -GalCer-pulsed irradiated leukaemic cells. Clinical trials have demonstrated the capacity of immuno-therapeutic strategies to induce *in vivo* iNKT cell expansion in patients with malignancies (table 1.17). Conclusive evidence for an adjuvant effect of iNKT cell agonists in humans is lacking, but this has been shown in preclinical models of immunotherapy for B cell malignancies.

Fourth, phase II or phase III trials examining the efficacy of an immunotherapeutic approach such as that suggested here could use eradication of MRD as a surrogate endpoint to overcome the very long follow-up periods required to detect differences in progression free survival after first line chemotherapy. Very low levels of residual CLL cells can be detected in the blood or bone marrow of most FCR-treated patients using polychromatic flow cytometry or molecular techniques, ^{593,594} and levels of MRD correlate with clinical outcome after FCR chemotherapy. ⁵⁹⁵ Alternatively, higher-risk groups of patients could be selected, such as those with the 11q or 17p deletions, or unmutated IGVH regions, as

earlier relapses would be expected in these groups after standard first-line chemoimmunotherapy.

To summarize, the proposed immunotherapeutic strategy combines standard chemoimmunotherapy, to induce both disease remission and lymphodepletion, with subsequent tumour vaccination incorporating iNKT cell engagement and/or upregulation of CLL cell costimulatory markers with TLR or CD40 ligands. Such an immunotherapy strategy addresses many of the factors associated with successful immunotherapy proposed in table 7.7.

7.4 Conclusions and future directions

The experiments presented in this thesis provide the first data characterising iNKT cell numbers and function in patients with CLL. The findings indicate that although iNKT cells constitute a reduced proportion of T cells, they are present in normal absolute numbers in the peripheral blood of patients with CLL. The immunophenotype of circulating iNKT cells in patients is normal in terms of CD4, CD8 and CD25 expression. The frequency of circulating iNKT cells is not associated with CLL disease stage or ZAP-70 status, and does not predict disease progression, suggesting that iNKT cells are not central to the pathogenesis of CLL progression.

The expression of the glycolipid-presenting molecule CD1d on circulating APCs is intact in CLL, although patients have a higher proportion of CD16+ monocytes, which express CD1d at lower levels than their CD14high counterparts.

In terms of function, intracellular cytokine profile and granzyme B expression are normal in iNKT cells of patients with CLL, and functional iNKT cell lines can be derived from patients with CLL. These cell lines produce both Th1- and Th2-type cytokines, and can lyse CD1d-transfected target cells in an α -GalCer-dependent manner.

Despite these findings, the addition of α -GalCer to whole PBMCs from patients with CLL does not result in substantial cytokine production, a finding attributable to the low iNKT cell frequencies in most individuals. The lysis of CLL cells by patient-derived iNKT cells is inefficient, requiring high effector to target ratios. However, the addition of α -GalCer-pulsed CLL cells to B cell-depleted PBMCs induces proliferation of iNKT cells in both allogeneic and autologous settings, and α -GalCer-treatment enhances the proliferation of non-CD1d restricted T cells in response to irradiated autologous leukaemic cells.

The results presented in chapters 3 to 5 suggest that despite a relative reduction in iNKT cell numbers, absolute numbers are normal and iNKT cell cytokine production and proliferative capacity is retained. This suggests that the iNKT cell/CD1d axis is essentially intact in patients with CLL. However, the results presented in chapter 6 highlight two key challenges to any iNKT cell-based tumour immunotherapy.

First, the scarcity of circulating iNKT cells may limit the efficacy of an α -GalCer-based immunotherapy—immunologic effects of α -GalCer which are readily detectable in mice, or in human donors with high numbers of iNKT cells, are difficult to demonstrate in those with iNKT cell numbers more typical of older adults, including CLL patients

Second, iNKT cells, particularly those which have undergone prolonged *in vitro* culture in the presence of IL-2, are capable of producing high concentrations of Th2-type cytokines, which could hinder rather than help the generation of antitumour responses.

To overcome the small numbers of iNKT cells, patients with higher numbers of circulating cells could be selected for inclusion of any clinical studies. However, this would limit the applicability of any immunotherapy, and does not take account of the possibility that iNKT cells may be more numerous in tissues other than peripheral blood. A more attractive solution may be to design an immunotherapeutic strategy which incorporates *ex vivo* or *in vivo* expansion of iNKT cells.

Although I was able to demonstrate *ex vivo* generation of iNKT cell lines from patients with CLL, the process was laborious, requiring several purification steps and prolonged culture periods, and yielding limited numbers of iNKT cells. Moreover, the resultant iNKT cells produced high levels of IL-4 and IL-13, which may not be desirable characteristics for application in antitumour immunotherapy. Refining iNKT cell expansion methods may overcome a number of these limitations. However, if expansion of iNKT cells could be achieved *in vivo*, prolonged *in vitro* culture might be avoided altogether.

Experience in phase I clinical trials suggests that α -GalCer-pulsed APCs can lead to *in vivo* iNKT cell proliferation in humans (table 1.17). My *in vitro* data indicate that α -GalCer-pulsed CLL cells can induce iNKT cell proliferation. Compared to mDCs, CLL cells are very readily obtained and purified, and would intrinsically express the full range of patient-specific TAAs.

The use of α -GalCer-pulsed tumour cells represents an attractive and simple means of delivering TAAs alongside α -GalCer adjuvant. As discussed in section 1.3.3.3, preclinical data lends support to the approach of using whole tumour cells pulsed with α -GalCer as a vaccine in B cell malignancies.

Further experiments could be performed to elucidate the mechanisms by which α -GalCerpulsed CLL cells result in iNKT cell proliferation (either by directly presentating α -GalCer to iNKT cells, or through re-presentation by other APCs). *In vitro* experiments could explore the co-operation between various TLR ligands and α -GalCer in the enhancement of T cell responses against autologous CLL cells. Demonstrating that the responding T cells have cytotoxic potential against autologous CLL cells would also be important, and might be achieved with granzyme B ELISpot assays, or by using an alternative cytotoxicity assay.

Application of the immunotherapeutic strategy of α -GalCer-pulsed tumour cells in one or more of the murine models of CLL discussed in section 1.1.3.6 could provide additional support for future clinical trials of α -GalCer-pulsed CLL cells as a tumour vaccine.

Should translational or early phase clinical studies fail to demonstrate an effect of α -GalCer in the enhancement of anti-tumour responses in humans, the impact of engaging other unconventional T cells could be explored in future. Humans express group 1 CD1 molecules, including CD1a, CD1b and CD1c. Group 1 CD1-restricted T cells can be isolated from human peripheral blood, respond to distinct lipid agonists (table 1.8), and appear to have a role in the immunological response to tuberculosis.¹⁹⁵ They are capable of Th1-type cytokine production, and their frequency in human peripheral blood is at least order of magnitude greater than that of iNKT cells.⁵⁹⁶ The availability of CD1

tetramers and the development of a transgenic mouse expressing human group 1 CD1 molecules will facilitate future study of this field.^{196,596} As described in section 1.2.2.2, phosphoantigen-reactive V γ 9V δ 2 T cells and MR1-restricted MAIT cells represent other human invariant T cell populations that share functional characteristics with, but are more prevalent than, iNKT cells. It is not yet known whether tumour vaccination strategies incorporating ligands for these cells are effective.

In conclusion, the experiments presented in this thesis provide the first evidence that the iNKT cell and CD1d axis is largely phenotypically and functionally intact in patients with CLL, that functional iNKT cell lines can be generated from the peripheral blood of patients with CLL, and that autologous α -GalCer-treated irradiated leukaemic cells can induce the proliferation of patient iNKT cells. On the basis of *in vitro* functional assays of patient PBMCs, I suggest that the most promising means of employing iNKT cells for the cellular immunotherapy of CLL may be to exploit their adjuvant function to enhance 'conventional' CD4+ and CD8+ T cell responses against tumour-associated peptide antigens. Finally, I propose a feasible whole tumour vaccination strategy using α -GalCer-treated irradiated leukaemic cells for the treatment of CLL.

References

- 1. Celsus, A. C. (1935) *De Medicina*, On Medicine. (Harvard University Press, Boston) Vol. 1, 5th edition.
- Muller-Hermelink, H. K, Montserrat, E, Catovsky, D, Campo, E, Harris, N. L, & Stein, H. (2008) WHO Classification of Tumours of Haematopoietic and Lymphoid Tissues, eds. Swerdlow, S. H, Campo, E. Harris, N. L, Jaffe, E. S, Pieri, S. A, Stein, H, Thiele, J, & Vardiman, J. W. (IARC, Lyon, France), 4th edition, pp. 180–182.
- 3. Giannopoulos, K, Dmoszyńska, A, Kowal, M, Roliński, J, Gostick, E, Price, D. A, Greiner, J, Rojewski, M, Stilgenbauer, S, Döhner, H, & Schmitt, M. (2010) Peptide vaccination elicits leukemia-associated antigen-specific cytotoxic CD8+ T-cell responses in patients with chronic lymphocytic leukemia. *Leukemia* 24, 798–805.
- 4. Hallek, M, Cheson, B. D, Catovsky, D, Caligaris-Cappio, F, Dighiero, G, Döhner, H, Hillmen, P, Keating, M. J, Montserrat, E, Rai, K. R, Kipps, T. J, & on Chronic Lymphocytic Leukemia, I. W. (2008) Guidelines for the diagnosis and treatment of chronic lymphocytic leukemia: a report from the International Workshop on Chronic Lymphocytic Leukemia updating the National Cancer Institute-Working Group 1996 guidelines. *Blood* **111**, 5446–5456.
- 5. Gribben, J. G. (2010) How I treat CLL up front. Blood 115, 187–197.
- Rawstron, A. C, Bennett, F. L, O'Connor, S. J. M, Kwok, M, Fenton, J. A. L, Plummer, M, de Tute, R, Owen, R. G, Richards, S. J, Jack, A. S, & Hillmen, P. (2008) Monoclonal B-cell lymphocytosis and chronic lymphocytic leukemia. *New England Journal of Medicine* 359, 575–583.
- Rawstron, A. C, Green, M. J, Kuzmicki, A, Kennedy, B, Fenton, J. A. L, Evans, P. A. S, O'Connor, S. J. M, Richards, S. J, Morgan, G. J, Jack, A. S, & Hillmen, P. (2002) Monoclonal B lymphocytes with the characteristics of "indolent" chronic lymphocytic leukemia are present in 3.5% of adults with normal blood counts. *Blood* 100, 635– 639.
- 8. Diehl, L. F, Karnell, L. H, & Menck, H. R. (1999) The American College of Surgeons Commission on Cancer and the American Cancer Society. The National Cancer Data Base report on age, gender, treatment, and outcomes of patients with chronic lymphocytic leukemia. *Cancer* **86**, 2684–2692.
- 9. Tracey, M. C & Carter, J. M. (2005) Ethnicity variables in the incidence rates of

leukemias in New Zealand populations: Implications for stem-cell transplantation. *American Journal of Hematology* **79**, 114–118.

- Dores, G. M, Anderson, W. F, Curtis, R. E, Landgren, O, Ostroumova, E, Bluhm, E. C, Rabkin, C. S, Devesa, S. S, & Linet, M. S. (2007) Chronic lymphocytic leukaemia and small lymphocytic lymphoma: overview of the descriptive epidemiology. *British Journal of Haematology* 139, 809–819.
- 11. Wu, S.-J, Huang, S.-Y, Lin, C.-T, Lin, Y.-J, Chang, C.-J, & Tien, H.-F. (2010) The incidence of chronic lymphocytic leukemia in Taiwan, 1986-2005: a distinct increasing trend with birth-cohort effect. *Blood* **116**, 4430–4435.
- 12. Pan, J. W. Y, Cook, L. S, Schwartz, S. M, & Weis, N. S. (2002) Incidence of leukemia in Asian migrants to the United States and their descendants. *Cancer Causes & Control* **13**, 791–795.
- Goldin, L. R, Bjorkholm, M, Kristinsson, S. Y, Turesson, I, & Landgren, O. (2009) Elevated risk of chronic lymphocytic leukemia and other indolent non-Hodgkin's lymphomas among relatives of patients with chronic lymphocytic leukemia. *Haematologica* 94, 647–653.
- Rawstron, A. C, Yuille, M. R, Fuller, J, Cullen, M, Kennedy, B, Richards, S. J, Jack, A. S, Matutes, E, Catovsky, D, Hillmen, P, & Houlston, R. S. (2002) Inherited predisposition to CLL is detectable as subclinical monoclonal B-lymphocyte expansion. *Blood* 100, 2289–2290.
- Enjuanes, A, Benavente, Y, Bosch, F, Martín-Guerrero, I, Colomer, D, Pérez-Alvarez, S, Reina, O, Ardanaz, M. T, Jares, P, García-Orad, A, Pujana, M. A, Montserrat, E, de Sanjosé, S, & Campo, E. (2008) Genetic variants in apoptosis and immunoregulation-related genes are associated with risk of chronic lymphocytic leukemia. *Cancer Research* 68, 10178–10186.
- 16. Khalade, A, Jaakkola, M. S, Pukkala, E, & Jaakkola, J. J. K. (2010) Exposure to benzene at work and the risk of leukemia: a systematic review and meta-analysis. *Environmental Health* **9**, 31.
- McLean, D, Mannetje, A. T, Dryson, E, Walls, C, McKenzie, F, Maule, M, Cheng, S, Cunningham, C, Kromhout, H, Boffetta, P, Blair, A, & Pearce, N. (2009) Leukaemia and occupation: a New Zealand Cancer Registry-based case-control Study. *International Journal of Epidemiology* 38, 594–606.
- 18. Zheng, T, Blair, A, Zhang, Y, Weisenburger, D. D, & Zahm, S. H. (2002) Occupation and risk of non-Hodgkin's lymphoma and chronic lymphocytic leukemia. *Journal of Occupational and Environmental Medicine* **44**, 469–474.
- Nowakowski, G. S, Hoyer, J. D, Shanafelt, T. D, Zent, C. S, Call, T. G, Bone, N. D, Laplant, B, Dewald, G. W, Tschumper, R. C, Jelinek, D. F, Witzig, T. E, & Kay, N. E. (2009) Percentage of Smudge Cells on Routine Blood Smear Predicts Survival in Chronic Lymphocytic Leukemia. *Journal of Clinical Oncology* 27, 1844–1849.
- 20. Moreau, E. J, Matutes, E, A'Hern, R. P, Morilla, A. M, Morilla, R. M, Owusu-Ankomah, K. A, Seon, B. K, & Catovsky, D. (1997) Improvement of the chronic

lymphocytic leukemia scoring system with the monoclonal antibody SN8 (CD79b). *American journal of Clinical Pathology* **108**, 378–382.

- 21. Bain, B. J, Clark, D. M, & Wilkins, B. (2010) *Bone marrow pathology*. (Wiley-Blackwell, Oxford), 4th edition.
- 22. Ioachim, H. L & Medeiros, L. J. (2009) *Ioachim's lymph node pathology*. (Wolters Kluwer/Lippincott Williams & Wilkins, Philadelphia), 4th edition.
- 23. Ternynck, T, Dighiero, G, Follezou, J, & Binet, J. L. (1974) Comparison of normal and CLL lymphocyte surface Ig determinants using peroxidase-labeled antibodies. I. Detection and quantitation of light chain determinants. *Blood* **43**, 789–795.
- 24. Landgren, O, Albitar, M, Ma, W, Abbasi, F, Hayes, R. B, Ghia, P, Marti, G. E, & Caporaso, N. E. (2009) B-cell clones as early markers for chronic lymphocytic leukemia. *New England Journal of Medicine* **360**, 659–667.
- 25. Freedman, A. S, Boyd, A. W, Bieber, F. R, Daley, J, Rosen, K, Horowitz, J. C, Levy, D. N, & Nadler, L. M. (1987) Normal cellular counterparts of B cell chronic lymphocytic leukemia. *Blood* **70**, 418–427.
- 26. Klein, U, Tu, Y, Stolovitzky, G. A, Mattioli, M, Cattoretti, G, Husson, H, Freedman, A, Inghirami, G, Cro, L, Baldini, L, Neri, A, Califano, A, & Dalla-Favera, R. (2001) Gene expression profiling of B cell chronic lymphocytic leukemia reveals a homogeneous phenotype related to memory B cells. *Journal of Experimental Medicine* **194**, 1625–1638.
- 27. Malisan, F, Fluckiger, A. C, Ho, S, Guret, C, Banchereau, J, & Martinez-Valdez, H. (1996) B-chronic lymphocytic leukemias can undergo isotype switching in vivo and can be induced to differentiate and switch in vitro. *Blood* **87**, 717–724.
- 28. Krysov, S, Potter, K. N, Mockridge, C. I, Coelho, V, Wheatley, I, Packham, G, & Stevenson, F. K. (2010) Surface IgM of CLL cells displays unusual glycans indicative of engagement of antigen in vivo. *Blood* **115**, 4198–4205.
- 29. Fais, F, Ghiotto, F, Hashimoto, S, Sellars, B, Valetto, A, Allen, S. L, Schulman, P, Vinciguerra, V. P, Rai, K, Rassenti, L. Z, Kipps, T. J, Dighiero, G, Schroeder, H. W. J, Ferrarini, M, & Chiorazzi, N. (1998) Chronic lymphocytic leukemia B cells express restricted sets of mutated and unmutated antigen receptors. *Journal of Clinical Investigation* **102**, 1515–1525.
- 30. Chen, L, Zhang, Y, Zheng, W, Wu, Y, Qiao, C, Fan, L, Xu, W, & Li, J. (2008) Distinctive IgVH gene segments usage and mutation status in Chinese patients with chronic lymphocytic leukemia. *Leukemia Research* **32**, 1491–1498.
- Johnson, T. A, Rassenti, L. Z, & Kipps, T. J. (1997) Ig VH1 genes expressed in B cell chronic lymphocytic leukemia exhibit distinctive molecular features. *Journal of Immunology* 158, 235–246.
- 32. Messmer, B. T, Albesiano, E, Efremov, D. G, Ghiotto, F, Allen, S. L, Kolitz, J, Foa, R, Damle, R. N, Fais, F, Messmer, D, Rai, K. R, Ferrarini, M, & Chiorazzi, N. (2004) Multiple distinct sets of stereotyped antigen receptors indicate a role for antigen

in promoting chronic lymphocytic leukemia. *Journal of Experimental Medicine* **200**, 519–525.

- 33. Murray, F, Darzentas, N, Hadzidimitriou, A, Tobin, G, Boudjogra, M, Scielzo, C, Laoutaris, N, Karlsson, K, Baran-Marzsak, F, Tsaftaris, A, Moreno, C, Anagnostopoulos, A, Caligaris-Cappio, F, Vaur, D, Ouzounis, C, Belessi, C, Ghia, P, Davi, F, Rosenquist, R, & Stamatopoulos, K. (2008) Stereotyped patterns of somatic hypermutation in subsets of patients with chronic lymphocytic leukemia: implications for the role of antigen selection in leukemogenesis. *Blood* 111, 1524–1533.
- 34. Stamatopoulos, K, Belessi, C, Moreno, C, Boudjograh, M, Guida, G, Smilevska, T, Belhoul, L, Stella, S, Stavroyianni, N, Crespo, M, Hadzidimitriou, A, Sutton, L, Bosch, F, Laoutaris, N, Anagnostopoulos, A, Montserrat, E, Fassas, A, Dighiero, G, Caligaris-Cappio, F, Merle-Beral, H, Ghia, P, & Davi, F. (2007) Over 20% of patients with chronic lymphocytic leukemia carry stereotyped receptors: Pathogenetic implications and clinical correlations. *Blood* 109, 259–270.
- 35. Cong, Y. Z, Rabin, E, & Wortis, H. H. (1991) Treatment of murine CD5- B cells with anti-Ig, but not LPS, induces surface CD5: two B-cell activation pathways. *International Immunology* **3**, 467–476.
- Hayakawa, K, Asano, M, Shinton, S. A, Gui, M, Allman, D, Stewart, C. L, Silver, J, & Hardy, R. R. (1999) Positive selection of natural autoreactive B cells. *Science* 285, 113–116.
- Damle, R. N, Ghiotto, F, Valetto, A, Albesiano, E, Fais, F, Yan, X.-J, Sison, C. P, Allen, S. L, Kolitz, J, Schulman, P, Vinciguerra, V. P, Budde, P, Frey, J, Rai, K. R, Ferrarini, M, & Chiorazzi, N. (2002) B-cell chronic lymphocytic leukemia cells express a surface membrane phenotype of activated, antigen-experienced B lymphocytes. *Blood* 99, 4087–4093.
- Mockridge, C. I, Potter, K. N, Wheatley, I, Neville, L. A, Packham, G, & Stevenson, F. K. (2007) Reversible anergy of sIgM-mediated signaling in the two subsets of CLL defined by VH-gene mutational status. *Blood* 109, 4424–4431.
- Gary-Gouy, H, Sainz-Perez, A, Marteau, J, Marfaing-Koka, A, Delic, J, Merle-Beral, H, Galanaud, P, & Dalloul, A. (2007) Natural phosphorylation of CD5 in chronic lymphocytic leukemia B cells and analysis of CD5-regulated genes in a B cell line suggest a role for CD5 in malignant phenotype. *Journal of Immunology* 179, 4335– 4344.
- Zenz, T, Mertens, D, Küppers, R, Döhner, H, & Stilgenbauer, S. (2010) From pathogenesis to treatment of chronic lymphocytic leukaemia. *Nature Reviews Cancer* 10, 37–50.
- 41. Borche, L, Lim, A, Binet, J. L, & Dighiero, G. (1990) Evidence that chronic lymphocytic leukemia B lymphocytes are frequently committed to production of natural autoantibodies. *Blood* **76**, 562–569.
- 42. Sthoeger, Z. M, Wakai, M, Tse, D. B, Vinciguerra, V. P, Allen, S. L, Budman, D. R, Lichtman, S. M, Schulman, P, Weiselberg, L. R, & Chiorazzi, N. (1989) Produc-

tion of autoantibodies by CD5-expressing B lymphocytes from patients with chronic lymphocytic leukemia. *Journal of Experimental Medicine* **169**, 255–268.

- 43. Duan, B & Morel, L. (2006) Role of B-1a cells in autoimmunity. *Autoimmunity Reviews* **5**, 403–408.
- Hervé, M, Xu, K, Ng, Y.-S, Wardemann, H, Albesiano, E, Messmer, B. T, Chiorazzi, N, & Meffre, E. (2005) Unmutated and mutated chronic lymphocytic leukemias derive from self-reactive B cell precursors despite expressing different antibody reactivity. *Journal of Clinical Investigation* 115, 1636–1643.
- 45. Chu, C. C, Catera, R, Hatzi, K, Yan, X. J, Zhang, L, Wang, X. B, Fales, H. M, Allen, S. L, Kolitz, J. E, Rai, K. R, & Chiorazzi, N. (2008) Chronic lymphocytic leukemia antibodies with a common stereotypic rearrangement recognize nonmuscle myosin heavy chain IIA. *Blood* **112**, 5122–5129.
- 46. Lanemo Myhrinder, A, Hellqvist, E, Sidorova, E, Soderberg, A, Baxendale, H, Dahle, C, Willander, K, Tobin, G, Backman, E, Soderberg, O, Rosenquist, R, Horkko, S, & Rosen, A. (2008) A new perspective: molecular motifs on oxidized LDL, apoptotic cells, and bacteria are targets for chronic lymphocytic leukemia antibodies. *Blood* 111, 3838–3848.
- 47. Catera, R, Silverman, G. J, Hatzi, K, Seiler, T, Didier, S, Zhang, L, Herve, M, Meffre, E, Oscier, D. G, Vlassara, H, Scofield, R. H, Chen, Y, Allen, S. L, Kolitz, J, Rai, K. R, Chu, C. C, & Chiorazzi, N. (2008) Chronic lymphocytic leukemia cells recognize conserved epitopes associated with apoptosis and oxidation. *Molecular Medicine* 14, 665–674.
- 48. Klein, U & Dalla-Favera, R. (2010) New insights into the pathogenesis of chronic lymphocytic leukemia. *Seminars in Cancer Biology* **20**, 377–383.
- 49. Schena, M, Larsson, L. G, Gottardi, D, Gaidano, G, Carlsson, M, Nilsson, K, & Caligaris-Cappio, F. (1992) Growth- and differentiation-associated expression of bcl-2 in B-chronic lymphocytic leukemia cells. *Blood* **79**, 2981–2989.
- 50. Nakagawa, Y, Yamaguchi, S, Hasegawa, M, Nemoto, T, Inoue, M, Suzuki, K, Hirokawa, K, & Kitagawa, M. (2004) Differential expression of survivin in bone marrow cells from patients with acute lymphocytic leukemia and chronic lymphocytic leukemia. *Leukemia Research* **28**, 487–494.
- Ding, W, Nowakowski, G. S, Knox, T. R, Boysen, J. C, Maas, M. L, Schwager, S. M, Wu, W, Wellik, L. E, Dietz, A. B, Ghosh, A. K, Secreto, C. R, Medina, K. L, Shanafelt, T. D, Zent, C. S, Call, T. G, & Kay, N. E. (2009) Bi-directional activation between mesenchymal stem cells and CLL B-cells: implication for CLL disease progression. *British Journal of Haematology* 147, 471–483.
- 52. Tsukada, N, Burger, J. A, Zvaifler, N. J, & Kipps, T. J. (2002) Distinctive features of "nurselike" cells that differentiate in the context of chronic lymphocytic leukemia. *Blood* **99**, 1030–1037.
- 53. Willimott, S, Baou, M, Huf, S, & Wagner, S. D. (2007) Separate cell culture conditions to promote proliferation or quiescent cell survival in chronic lymphocytic leukemia. *Leukemia & Lymphoma* **48**, 1647–1650.

- 54. Petlickovski, A, Laurenti, L, Li, X, Marietti, S, Chiusolo, P, Sica, S, Leone, G, & Efremov, D. G. (2005) Sustained signaling through the B-cell receptor induces Mcl-1 and promotes survival of chronic lymphocytic leukemia B cells. *Blood* **105**, 4820–4827.
- 55. Messmer, B. T, Messmer, D, Allen, S. L, Kolitz, J. E, Kudalkar, P, Cesar, D, Murphy, E. J, Koduru, P, Ferrarini, M, Zupo, S, Cutrona, G, Damle, R. N, Wasil, T, Rai, K. R, Hellerstein, M. K, & Chiorazzi, N. (2005) In vivo measurements document the dynamic cellular kinetics of chronic lymphocytic leukemia B cells. *Journal of Clinical Investigation* **115**, 755–764.
- 56. Andreeff, M, Darzynkiewicz, Z, Sharpless, T. K, Clarkson, B. D, & Melamed, M. R. (1980) Discrimination of human leukemia subtypes by flow cytometric analysis of cellular DNA and RNA. *Blood* **55**, 282–293.
- 57. Granziero, L, Ghia, P, Circosta, P, Gottardi, D, Strola, G, Geuna, M, Montagna, L, Piccoli, P, Chilosi, M, & Caligaris-Cappio, F. (2001) Survivin is expressed on CD40 stimulation and interfaces proliferation and apoptosis in B-cell chronic lymphocytic leukemia. *Blood* 97, 2777–2783.
- Patten, P. E. M, Buggins, A. G. S, Richards, J, Wotherspoon, A, Salisbury, J, Mufti, G. J, Hamblin, T. J, & Devereux, S. (2008) CD38 expression in chronic lymphocytic leukemia is regulated by the tumor microenvironment. *Blood* 111, 5173–5181.
- 59. Calissano, C, Damle, R. N, Hayes, G, Murphy, E. J, Hellerstein, M. K, Moreno, C, Sison, C, Kaufman, M. S, Kolitz, J. E, Allen, S. L, Rai, K. R, & Chiorazzi, N. (2009) In vivo intraclonal and interclonal kinetic heterogeneity in B-cell chronic lymphocytic leukemia. *Blood* **114**, 4832–4842.
- 60. Chiorazzi, N & Ferrarini, M. (2006) Evolving view of the in-vivo kinetics of chronic lymphocytic leukemia B cells. *Hematology. The American Society of Hematology Education Program Book* pp. 273–278.
- 61. Berkova, A, Zemanova, Z, Trneny, M, Schwarz, J, Karban, J, Cmunt, E, Pavlistova, L, Brezinova, J, & Michalova, K. (2009) Clonal evolution in chronic lymphocytic leukemia studied by interphase fluorescence in-situ hybridization. *Neoplasma* **56**, 455–458.
- Shanafelt, T. D, Witzig, T. E, Fink, S. R, Jenkins, R. B, Paternoster, S. F, Smoley, S. A, Stockero, K. J, Nast, D. M, Flynn, H. C, Tschumper, R. C, Geyer, S, Zent, C. S, Call, T. G, Jelinek, D. F, Kay, N. E, & Dewald, G. W. (2006) Prospective evaluation of clonal evolution during long-term follow-up of patients with untreated early-stage chronic lymphocytic leukemia. *Journal of Clinical Oncology* 24, 4634–4641.
- 63. Grubor, V, Krasnitz, A, Troge, J. E, Meth, J. L, Lakshmi, B, Kendall, J. T, Yamrom, B, Alex, G, Pai, D, Navin, N, Hufnagel, L. A, Lee, Y.-H, Cook, K, Allen, S. L, Rai, K. R, Damle, R. N, Calissano, C, Chiorazzi, N, Wigler, M, & Esposito, D. (2009) Novel genomic alterations and clonal evolution in chronic lymphocytic leukemia revealed by representational oligonucleotide microarray analysis (ROMA). *Blood* **113**, 1294–1303.

- 64. Rossi, D & Gaidano, G. (2009) Richter syndrome: molecular insights and clinical perspectives. *Hematological Oncology* **27**, 1–10.
- 65. Miller, D. G & Karnofsky, D. A. (1961) Immunologic factors and resistance to infection in chronic lymphatic leukemia. *American Journal of Medicine* **31**, 748–757.
- 66. Dearden, C. (2008) Disease-specific complications of chronic lymphocytic leukemia. *Hematology. The American Society of Hematology Education Program Book* pp. 450–456.
- 67. Ghia, P, Scielzo, C, Frenquelli, M, Muzio, M, & Caligaris-Cappio, F. (2007) From normal to clonal B cells: Chronic lymphocytic leukemia (CLL) at the crossroad between neoplasia and autoimmunity. *Autoimmunity Reviews* 7, 127–131.
- 68. Lewis, F. B, Schwartz, R. S, & Dameshek, W. (1966) X-radiation and alkylating agents as possible "trigger" mechanisms in the autoimmune complications of malignant lymphophroliferative disease. *Clinical and Experimental Immunology* **1**, 3–11.
- 69. Dearden, C, Wade, R, Else, M, Richards, S, Milligan, D, Hamblin, T, Catovsky, D, UK National Cancer Research Institute NCRI Haematological Oncology Clinical Studies Group, & NCRI CLL Working Group. (2008) The prognostic significance of a positive direct antiglobulin test in chronic lymphocytic leukemia: a beneficial effect of the combination of fludarabine and cyclophosphamide on the incidence of hemolytic anemia. *Blood* **111**, 1820–1826.
- 70. Hall, A. M, Vickers, M. A, McLeod, E, & Barker, R. N. (2005) Rh autoantigen presentation to helper T cells in chronic lymphocytic leukemia by malignant B cells. *Blood* **105**, 2007–2015.
- 71. Molica, S. (1994) Infections in chronic lymphocytic leukemia: risk factors, and impact on survival, and treatment. *Leukemia & Lymphoma* **13**, 203–214.
- 72. Ziegler, H. W, Kay, N. E, & Zarling, J. M. (1981) Deficiency of natural killer cell activity in patients with chronic lymphocytic leukemia. *International Journal of Cancer* **27**, 321–327.
- 73. Katrinakis, G, Kyriakou, D, Papadaki, H, Kalokyri, I, Markidou, F, & Eliopoulos, G. D. (1996) Defective natural killer cell activity in B-cell chronic lymphocytic leukaemia is associated with impaired release of natural killer cytotoxic factor(s) but not of tumour necrosis factor-alpha. *Acta Haematologica* **96**, 16–23.
- 74. Itälä, M, Vainio, O, & Remes, K. (1996) Functional abnormalities in granulocytes predict susceptibility to bacterial infections in chronic lymphocytic leukaemia. *European Journal of Haematology* **57**, 46–53.
- 75. Schlesinger, M, Broman, I, & Lugassy, G. (1996) The complement system is defective in chronic lymphatic leukemia patients and in their healthy relatives. *Leukemia* **10**, 1509–1513.
- 76. Rozman, C & Montserrat, E. (1995) Chronic lymphocytic leukemia. *New England Journal of Medicine* **333**, 1052–1057.
- 77. Cerutti, A, Kim, E. C, Shah, S, Schattner, E. J, Zan, H, Schaffer, A, & Casali, P. (2001)

Dysregulation of CD30+ T cells by leukemia impairs isotype switching in normal B cells. *Nature Immunology* 2, 150–156.

- Mami, N. B, Mohty, M, Aurran-Schleinitz, T, Olive, D, & Gaugler, B. (2008) Blood dendritic cells in patients with chronic lymphocytic leukaemia. *Immunobiology* 213, 493–498.
- Rezvany, M.-R, Jeddi-Tehrani, M, Biberfeld, P, Söderlund, J, Mellstedt, H, Osterborg, A, & Rabbani, H. (2001) Dendritic cells in patients with non-progressive B-chronic lymphocytic leukaemia have a normal functional capability but abnormal cytokine pattern. *British Journal of Haematology* **115**, 263–271.
- 80. Orsini, E, Guarini, A, Chiaretti, S, Mauro, F. R, & Foa, R. (2003) The circulating dendritic cell compartment in patients with chronic lymphocytic leukemia is severely defective and unable to stimulate an effective T-cell response. *Cancer Research* **63**, 4497–4506.
- Lee, J.-J, Foon, K. A, Mailliard, R. B, Muthuswamy, R, & Kalinski, P. (2008) Type 1-polarized dendritic cells loaded with autologous tumor are a potent immunogen against chronic lymphocytic leukemia. *Journal of Leukocyte Biology* 84, 319–325.
- 82. Orsini, E, Pasquale, A, Maggio, R, Calabrese, E, Mauro, F. R, Giammartini, E, Guarini, A, & Foa, R. (2004) Phenotypic and functional characterization of monocytederived dendritic cells in chronic lymphocytic leukaemia patients: influence of neoplastic CD19 cells in vivo and in vitro. *British Journal of Haematology* **125**, 720–728.
- Platsoucas, C. D, Galinski, M, Kempin, S, Reich, L, Clarkson, B, & Good, R. A. (1982) Abnormal T lymphocyte subpopulations in patients with B cell chronic lymphocytic leukemia: an analysis by monoclonal antibodies. *Journal of Immunology* 129, 2305– 2312.
- 84. Tötterman, T. H, Carlsson, M, Simonsson, B, Bengtsson, M, & Nilsson, K. (1989) T-cell activation and subset patterns are altered in B-CLL and correlate with the stage of the disease. *Blood* **74**, 786–792.
- Haiat, S, Billard, C, Quiney, C, Ajchenbaum-Cymbalista, F, & Kolb, J.-P. (2006) Role of BAFF and APRIL in human B-cell chronic lymphocytic leukaemia. *Immunology* 118, 281–292.
- 86. Bagnara, D, Kaufman, M. S, Calissano, C, Marsilio, S, Patten, P. E. M, Simone, R, Chum, P, Yan, X.-J, Allen, S. L, Kolitz, J. E, Baskar, S, Rader, C, Mellstedt, H, Rabbani, H, Lee, A, Gregersen, P. K, Rai, K. R, & Chiorazzi, N. (2011) A novel adoptive transfer model of chronic lymphocytic leukemia suggests a key role for T lymphocytes in the disease. *Blood* **117**, 5463–5472.
- Rezvany, M.-R, Jeddi-Tehrani, M, Osterborg, A, Kimby, E, Wigzell, H, & Mellstedt, H. (1999) Oligoclonal TCRBV gene usage in B-cell chronic lymphocytic leukemia: major perturbations are preferentially seen within the CD4 T-cell subset. *Blood* 94, 1063–1069.
- Goolsby, C. L, Kuchnio, M, Finn, W. G, & Peterson, L. (2000) Expansions of clonal and oligoclonal T cells in B-cell chronic lymphocytic leukemia are primarily restricted to the CD3(+)CD8(+) T-cell population. *Cytometry* 42, 188–195.

- Farace, F, Orlanducci, F, Dietrich, P. Y, Gaudin, C, Angevin, E, Courtier, M. H, Bayle, C, Hercend, T, & Triebel, F. (1994) T cell repertoire in patients with B chronic lymphocytic leukemia. Evidence for multiple in vivo T cell clonal expansions. *Journal of Immunology* 153, 4281–4290.
- 90. Cantwell, M, Hua, T, Pappas, J, & Kipps, T. J. (1997) Acquired CD40-ligand deficiency in chronic lymphocytic leukemia. *Nature Medicine* **3**, 984–989.
- Buggins, A. G. S, Patten, P. E. M, Richards, J, Thomas, N. S. B, Mufti, G. J, & Devereux, S. (2008) Tumor-derived IL-6 may contribute to the immunological defect in CLL. *Leukemia* 22, 1084–1087.
- Rossi, E, Matutes, E, Morilla, R, Owusu-Ankomah, K, Heffernan, A. M, & Catovsky, D. (1996) Zeta chain and CD28 are poorly expressed on T lymphocytes from chronic lymphocytic leukemia. *Leukemia* 10, 494–497.
- 93. Motta, M, Rassenti, L, Shelvin, B. J, Lerner, S, Kipps, T. J, Keating, M. J, & Wierda, W. G. (2005) Increased expression of CD152 (CTLA-4) by normal T lymphocytes in untreated patients with B-cell chronic lymphocytic leukemia. *Leukemia* 19, 1788–1793.
- 94. Prieto, A, Garcia-Suarez, J, Reyes, E, Lapena, P, Hernandez, M, & Alvarez-Mon, M. (1993) Diminished DNA synthesis in T cells from B chronic lymphocytic leukemia after phytohemagglutinin, anti-CD3, and phorbol myristate acetate mitogenic signals. *Experimental Hematology* 21, 1563–1569.
- 95. Mainou-Fowler, T, Proctor, S. J, Miller, S, & Dickinson, A. M. (2001) Expression and production of interleukin 4 in B-cell chronic lymphocytic leukaemia. *Leukemia & Lymphoma* **42**, 689–698.
- 96. Kay, N. E, Han, L, Bone, N, & Williams, G. (2001) Interleukin 4 content in chronic lymphocytic leukaemia (CLL) B cells and blood CD8+ T cells from B-CLL patients: impact on clonal B-cell apoptosis. *British Journal of Haematology* **112**, 760–767.
- 97. Ramsay, A. G, Johnson, A. J, Lee, A. M, Gorgün, G, Le Dieu, R, Blum, W, Byrd, J. C, & Gribben, J. G. (2008) Chronic lymphocytic leukemia T cells show impaired immunological synapse formation that can be reversed with an immunomodulating drug. *Journal of Clinical Investigation* **118**, 2427–2437.
- 98. Weiss, L, Melchardt, T, Egle, A, Grabmer, C, Greil, R, & Tinhofer, I. (2011) Regulatory T cells predict the time to initial treatment in early stage chronic lymphocytic leukemia. *Cancer* **117**, 2163–2169.
- 99. Giannopoulos, K, Schmitt, M, Kowal, M, Wlasiuk, P, Bojarska-Junak, A, Chen, J, Rolinski, J, & Dmoszynska, A. (2008) Characterization of regulatory T cells in patients with B-cell chronic lymphocytic leukemia. *Oncology Reports* **20**, 677–682.
- 100. Gorczynski, R. M, Lee, L, & Boudakov, I. (2005) Augmented Induction of CD4+CD25+ Treg using monoclonal antibodies to CD200R. *Transplantation* **79**, 1180–1183.
- 101. Görgün, G, Holderried, T. A. W, Zahrieh, D, Neuberg, D, & Gribben, J. G. (2005)

Chronic lymphocytic leukemia cells induce changes in gene expression of CD4 and CD8 T cells. *Journal of Clinical Investigation* **115**, 1797–1805.

- 102. Lotz, M, Ranheim, E, & Kipps, T. J. (1994) Transforming growth factor beta as endogenous growth inhibitor of chronic lymphocytic leukemia B cells. *Journal of Experimental Medicine* **179**, 999–1004.
- 103. Kobie, J. J, Wu, R. S, Kurt, R. A, Lou, S, Adelman, M. K, Whitesell, L. J, Ramanathapuram, L. V, Arteaga, C. L, & Akporiaye, E. T. (2003) Transforming growth factor beta inhibits the antigen-presenting functions and antitumor activity of dendritic cell vaccines. *Cancer Research* **63**, 1860–1864.
- Fayad, L, Keating, M. J, Reuben, J. M, O'Brien, S, Lee, B. N, Lerner, S, & Kurzrock, R. (2001) Interleukin-6 and interleukin-10 levels in chronic lymphocytic leukemia: correlation with phenotypic characteristics and outcome. *Blood* 97, 256–263.
- 105. de Waal Malefyt, R, Haanen, J, Spits, H, Roncarolo, M. G, te Velde, A, Figdor, C, Johnson, K, Kastelein, R, Yssel, H, & de Vries, J. E. (1991) Interleukin 10 (IL-10) and viral IL-10 strongly reduce antigen-specific human T cell proliferation by diminishing the antigen-presenting capacity of monocytes via downregulation of class II major histocompatibility complex expression. *Journal of Experimental Medicine* 174, 915–924.
- 106. Foa, R, Giovarelli, M, Jemma, C, Fierro, M. T, Lusso, P, Ferrando, M. L, Lauria, F, & Forni, G. (1985) Interleukin 2 (IL 2) and interferon-gamma production by T lymphocytes from patients with B-chronic lymphocytic leukemia: evidence that normally released IL 2 is absorbed by the neoplastic B cell population. *Blood* 66, 614–619.
- 107. Knauf, W. U, Langenmayer, I, Ehlers, B, Mohr, B, Adorf, D, Nerl, C. H, Hallek, M, Zwingers, T. H, Emmerich, B, & Thiel, E. (1997) Serum levels of soluble CD23, but not soluble CD25, predict disease progression in early stage B-cell chronic lymphocytic leukemia. *Leukemia & Lymphoma* 27, 523–532.
- 108. Hock, B. D, Starling, G. C, Patton, W. N, Salm, N, Bond, K, McArthur, L. T, & McKenzie, J. L. (2004) Identification of a circulating soluble form of CD80: levels in patients with hematological malignancies. *Leukemia & Lymphoma* 45, 2111–2118.
- 109. Hock, B. D, McKenzie, J. L, Patton, N. W, Drayson, M, Taylor, K, Wakeman, C, Kantarjian, H, Giles, F, & Albitar, M. (2006) Circulating levels and clinical significance of soluble CD40 in patients with hematologic malignancies. *Cancer* 106, 2148–2157.
- Hock, B. D, Fernyhough, L. J, Gough, S. M, Steinkasserer, A, Cox, A. G, & McKenzie, J. L. (2009) Release and clinical significance of soluble CD83 in chronic lymphocytic leukemia. *Leukemia Research* 33, 1089–1095.
- 111. Phillips, J. A, Mehta, K, Fernandez, C, & Raveché, E. S. (1992) The NZB mouse as a model for chronic lymphocytic leukemia. *Cancer Research* **52**, 437–443.
- 112. Salerno, E, Yuan, Y, Scaglione, B. J, Marti, G, Jankovic, A, Mazzella, F, Laurindo, M. F, Despres, D, Baskar, S, Rader, C, & Raveche, E. (2010) The New Zealand

black mouse as a model for the development and progression of chronic lymphocytic leukemia. *Cytometry Part B, Clinical Cytometry* **78 Suppl 1**, S98–109.

- 113. Raveché, E. S, Salerno, E, Scaglione, B. J, Manohar, V, Abbasi, F, Lin, Y. C, Fredrickson, T, Landgraf, P, Ramachandra, S, Huppi, K, Toro, J. R, Zenger, V. E, Metcalf, R. A, & Marti, G. E. (2007) Abnormal microRNA-16 locus with syntemy to human 13q14 linked to CLL in NZB mice. *Blood* 109, 5079–5086.
- 114. Klein, U, Lia, M, Crespo, M, Siegel, R, Shen, Q, Mo, T, Ambesi-Impiombato, A, Califano, A, Migliazza, A, Bhagat, G, & Dalla-Favera, R. (2010) The DLEU2/miR-15a/16-1 cluster controls B cell proliferation and its deletion leads to chronic lymphocytic leukemia. *Cancer Cell* **17**, 28–40.
- 115. Santanam, U, Zanesi, N, Efanov, A, Costinean, S, Palamarchuk, A, Hagan, J. P, Volinia, S, Alder, H, Rassenti, L, Kipps, T, Croce, C. M, & Pekarsky, Y. (2010) Chronic lymphocytic leukemia modeled in mouse by targeted miR-29 expression. *Proceedings of the National Academy of Sciences of the United States of America* 107, 12210–12215.
- 116. Bichi, R, Shinton, S. A, Martin, E. S, Koval, A, Calin, G. A, Cesari, R, Russo, G, Hardy, R. R, & Croce, C. M. (2002) Human chronic lymphocytic leukemia modeled in mouse by targeted TCL1 expression. *Proceedings of the National Academy of Sciences of the United States of America* **99**, 6955–6960.
- 117. Hamblin, T. J. (2010) The TCL1 mouse as a model for chronic lymphocytic leukemia. *Leukemia Research* **34**, 135–136.
- 118. Herling, M, Patel, K. A, Khalili, J, Schlette, E, Kobayashi, R, Medeiros, L. J, & Jones, D. (2006) TCL1 shows a regulated expression pattern in chronic lymphocytic leukemia that correlates with molecular subtypes and proliferative state. *Leukemia* **20**, 280–285.
- 119. ter Brugge, P. J, Ta, V. B. T, de Bruijn, M. J. W, Keijzers, G, Maas, A, van Gent, D. C, & Hendriks, R. W. (2009) A mouse model for chronic lymphocytic leukemia based on expression of the SV40 large T antigen. *Blood* **114**, 119–127.
- 120. Bertilaccio, M. T. S, Scielzo, C, Simonetti, G, Ponzoni, M, Apollonio, B, Fazi, C, Scarfò, L, Rocchi, M, Muzio, M, Caligaris-Cappio, F, & Ghia, P. (2010) A novel Rag2-/-gammac-/-xenograft model of human CLL. *Blood* **115**, 1605–1609.
- 121. Scaglione, B. J, Salerno, E, Balan, M, Coffman, F, Landgraf, P, Abbasi, F, Kotenko, S, Marti, G. E, & Raveche, E. S. (2007) Murine models of chronic lymphocytic leukaemia: role of microRNA-16 in the New Zealand Black mouse model. *British Journal of Haematology* **139**, 645–657.
- 122. Zapata, J. M, Krajewska, M, Morse, H. C, Choi, Y, & Reed, J. C. (2004) TNF receptorassociated factor (TRAF) domain and Bcl-2 cooperate to induce small B cell lymphoma/chronic lymphocytic leukemia in transgenic mice. *Proceedings of the National Academy of Sciences of the United States of America* **101**, 16600–16605.
- 123. Planelles, L, Carvalho-Pinto, C. E, Hardenberg, G, Smaniotto, S, Savino, W, Gomez-Caro, R, Alvarez-Mon, M, de Jong, J, Eldering, E, Martinez, A. C, Medema, J. P,

& Hahne, M. (2004) APRIL promotes B-1 cell-associated neoplasm. *Cancer Cell* **6**, 399–408.

- 124. Wierda, W. G, O'brien, S, Wang, X, Faderl, S, Ferrajoli, A, Do, K.-A, Cortes, J, Thomas, D, Garcia-Manero, G, Koller, C, Beran, M, Giles, F, Ravandi, F, Lerner, S, Kantarjian, H, & Keating, M. (2007) Prognostic nomogram and index for overall survival in previously untreated patients with chronic lymphocytic leukemia. *Blood* 109, 4679–4685.
- 125. Molica, S & Alberti, A. (1987) Prognostic value of the lymphocyte doubling time in chronic lymphocytic leukemia. *Cancer* **60**, 2712–2716.
- 126. Rai, K. R, Sawitsky, A, Cronkite, E. P, Chanana, A. D, Levy, R. N, & Pasternack, B. S. (1975) Clinical staging of chronic lymphocytic leukemia. *Blood* **46**, 219–234.
- 127. Binet, J. L, Auquier, A, Dighiero, G, Chastang, C, Piguet, H, Goasguen, J, Vaugier, G, Potron, G, Colona, P, Oberling, F, Thomas, M, Tchernia, G, Jacquillat, C, Boivin, P, Lesty, C, Duault, M. T, Monconduit, M, Belabbes, S, & Gremy, F. (1981) A new prognostic classification of chronic lymphocytic leukemia derived from a multivariate survival analysis. *Cancer* 48, 198–206.
- Zenz, T, Döhner, H, & Stilgenbauer, S. (2007) Genetics and risk-stratified approach to therapy in chronic lymphocytic leukemia. *Best Practice & Research Clinical Haematology* 20, 439–453.
- 129. Calin, G. A, Dumitru, C. D, Shimizu, M, Bichi, R, Zupo, S, Noch, E, Aldler, H, Rattan, S, Keating, M, Rai, K, Rassenti, L, Kipps, T, Negrini, M, Bullrich, F, & Croce, C. M. (2002) Frequent deletions and down-regulation of micro- RNA genes miR15 and miR16 at 13q14 in chronic lymphocytic leukemia. *Proceedings of the National Academy of Sciences of the United States of America* **99**, 15524–15529.
- 130. Krober, A, Seiler, T, Benner, A, Bullinger, L, Bruckle, E, Lichter, P, Döhner, H, & Stilgenbauer, S. (2002) V(H) mutation status, CD38 expression level, genomic aberrations, and survival in chronic lymphocytic leukemia. *Blood* **100**, 1410–1416.
- 131. Damle, R. N, Wasil, T, Fais, F, Ghiotto, F, Valetto, A, Allen, S. L, Buchbinder, A, Budman, D, Dittmar, K, Kolitz, J, Lichtman, S. M, Schulman, P, Vinciguerra, V. P, Rai, K. R, Ferrarini, M, & Chiorazzi, N. (1999) Ig V gene mutation status and CD38 expression as novel prognostic indicators in chronic lymphocytic leukemia. *Blood* 94, 1840–1847.
- 132. Oscier, D. G, Gardiner, A. C, Mould, S. J, Glide, S, Davis, Z. A, Ibbotson, R. E, Corcoran, M. M, Chapman, R. M, Thomas, P. W, Copplestone, J. A, Orchard, J. A, & Hamblin, T. J. (2002) Multivariate analysis of prognostic factors in CLL: clinical stage, IGVH gene mutational status, and loss or mutation of the p53 gene are independent prognostic factors. *Blood* 100, 1177–1184.
- 133. Crespo, M, Bosch, F, Villamor, N, Bellosillo, B, Colomer, D, Rozman, M, Marce, S, Lopez-Guillermo, A, Campo, E, & Montserrat, E. (2003) ZAP-70 expression as a surrogate for immunoglobulin-variable-region mutations in chronic lymphocytic leukemia. *New England Journal of Medicine* 348, 1764–1775.

- 134. Marti, G, Orfao, A, & Goolsby, C. (2006) ZAP-70 in CLL: towards standardization of a biomarker for patient management: history of clinical cytometry special issue. *Cytometry Part B, Clinical Cytometry* **70**, 197–200.
- 135. Wiestner, A, Rosenwald, A, Barry, T. S, Wright, G, Davis, R. E, Henrickson, S. E, Zhao, H, Ibbotson, R. E, Orchard, J. A, Davis, Z, Stetler-Stevenson, M, Raffeld, M, Arthur, D. C, Marti, G. E, Wilson, W. H, Hamblin, T. J, Oscier, D. G, & Staudt, L. M. (2003) ZAP-70 expression identifies a chronic lymphocytic leukemia subtype with unmutated immunoglobulin genes, inferior clinical outcome, and distinct gene expression profile. *Blood* **101**, 4944–4951.
- 136. Hamblin, T. J, Orchard, J. A, Ibbotson, R. E, Davis, Z, Thomas, P. W, Stevenson, F. K, & Oscier, D. G. (2002) CD38 expression and immunoglobulin variable region mutations are independent prognostic variables in chronic lymphocytic leukemia, but CD38 expression may vary during the course of the disease. *Blood* 99, 1023–1029.
- 137. Van Bockstaele, F, Verhasselt, B, & Philippé, J. (2009) Prognostic markers in chronic lymphocytic leukemia: a comprehensive review. *Blood Reviews* **23**, 25–47.
- 138. Group, C. T. C. (1999) Chemotherapeutic options in chronic lymphocytic leukemia: a meta-analysis of the randomized trials. CLL Trialists' Collaborative Group. *Journal of the National Cancer Institute* **91**, 861–868.
- 139. Catovsky, D, Richards, S, Matutes, E, Oscier, D, Dyer, M. J. S, Bezares, R. F, Pettitt, A. R, Hamblin, T, Milligan, D. W, Child, J. A, Hamilton, M. S, Dearden, C. E, Smith, A. G, Bosanquet, A. G, Davis, Z, Brito-Babapulle, V, Else, M, Wade, R, Hillmen, P, UK National Cancer Research Institute NCRI Haematological Oncology Clinical Studies Group, & NCRI Chronic Lymphocytic Leukaemia Working Group. (2007) Assessment of fludarabine plus cyclophosphamide for patients with chronic lymphocytic leukaemia (the LRF CLL4 Trial): a randomised controlled trial. *Lancet* **370**, 230–239.
- 140. Hallek, M, Fischer, K, Fingerle-Rowson, G, Fink, A. M, Busch, R, Mayer, J, Hensel, M, Hopfinger, G, Hess, G, von Grünhagen, U, Bergmann, M, Catalano, J, Zinzani, P. L, Caligaris-Cappio, F, Seymour, J. F, Berrebi, A, Jäger, U, Cazin, B, Trneny, M, Westermann, A, Wendtner, C. M, Eichhorst, B. F, Staib, P, Bühler, A, Winkler, D, Zenz, T, Böttcher, S, Ritgen, M, Mendila, M, Kneba, M, Döhner, H, Stilgenbauer, S, International Group of Investigators, & German Chronic Lymphocytic Leukaemia Study Group. (2010) Addition of rituximab to fludarabine and cyclophosphamide in patients with chronic lymphocytic leukaemia: a randomised, open-label, phase 3 trial. *Lancet* 376, 1164–1174.
- 141. Tsimberidou, A.-M, Tam, C, Abruzzo, L. V, O'brien, S, Wierda, W. G, Lerner, S, Kantarjian, H. M, & Keating, M. J. (2009) Chemoimmunotherapy may overcome the adverse prognostic significance of 11q deletion in previously untreated patients with chronic lymphocytic leukemia. *Cancer* **115**, 373–380.
- 142. Hui, D, Lam, W, Toze, C, Delorme, M, Noble, M, Klimo, P, Sutherland, J, Gill, K, Connors, J, & Sehn, L. (2008) Alemtuzumab in clinical practice: A British Columbia experience. *Leukemia & Lymphoma* **49**, 218–226.

- 143. Bowen, A. L, Zomas, A, Emmett, E, Matutes, E, Dyer, M. J, & Catovsky, D. (1997) Subcutaneous CAMPATH-1H in fludarabine-resistant/relapsed chronic lymphocytic and B-prolymphocytic leukaemia. *British Journal of Haematology* **96**, 617–619.
- 144. Laurenti, L, Piccioni, P, Cattani, P, Cingolani, A, Efremov, D, Chiusolo, P, Tarnani, M, Fadda, G, Sica, S, & Leone, G. (2004) Cytomegalovirus reactivation during alemtuzumab therapy for chronic lymphocytic leukemia: incidence and treatment with oral ganciclovir. *Haematologica* **89**, 1248–1252.
- 145. Osterborg, A, Foa, R, Bezares, R. F, Dearden, C, Dyer, M. J. S, Geisler, C, Lin, T. S, Montillo, M, van Oers, M. H. J, Wendtner, C. M, & Rai, K. R. (2009) Management guidelines for the use of alemtuzumab in chronic lymphocytic leukemia. *Leukemia* 23, 1980–1988.
- 146. Cortelezzi, A, Pasquini, M. C, Gardellini, A, Gianelli, U, Bossi, A, Reda, G, Sarina, B, Musto, P, Barcellini, W, Neri, A, & Deliliers, G. L. (2009) Low-dose subcutaneous alemtuzumab in refractory chronic lymphocytic leukaemia (CLL): results of a prospective, single-arm multicentre study. *Leukemia* 23, 2027–2033.
- 147. Teeling, J. L, French, R. R, Cragg, M. S, van den Brakel, J, Pluyter, M, Huang, H, Chan, C, Parren, P. W. H. I, Hack, C. E, Dechant, M, Valerius, T, van de Winkel, J. G. J, & Glennie, M. J. (2004) Characterization of new human CD20 monoclonal antibodies with potent cytolytic activity against non-Hodgkin lymphomas. *Blood* 104, 1793–1800.
- 148. Wierda, W. G, Kipps, T. J, Dürig, J, Griskevicius, L, Stilgenbauer, S, Mayer, J, Smolej, L, Hess, G, Griniute, R, Hernandez-Ilizaliturri, F. J, Padmanabhan, S, Gorczyca, M, Chang, C. N, Chan, G, Gupta, I, Nielsen, T. G, & Russell, C. A. (2011) Chemoimmunotherapy with ofatumumab, fludarabine, and cyclophosphamide (O-FC) in previously untreated patients with chronic lymphocytic leukemia. *Blood* **117**, 6450– 6458.
- 149. Wierda, W. G, Kipps, T. J, Mayer, J, Stilgenbauer, S, Williams, C. D, Hellmann, A, Robak, T, Furman, R. R, Hillmen, P, Trneny, M, Dyer, M. J. S, Padmanabhan, S, Piotrowska, M, Kozak, T, Chan, G, Davis, R, Losic, N, Wilms, J, Russell, C. A, Österborg, A, & Hx-CD20-406 Study Investigators. (2010) Ofatumumab as single-agent CD20 immunotherapy in fludarabine-refractory chronic lymphocytic leukemia. *Journal of Clinical Oncology* 28, 1749–1755.
- 150. Byrd, J. C, Kipps, T. J, Flinn, I. W, Castro, J, Lin, T. S, Wierda, W, Heerema, N, Woodworth, J, Hughes, S, Tangri, S, Harris, S, Wynne, D, Molina, A, Leigh, B, & O'brien, S. (2010) Phase 1/2 study of lumiliximab combined with fludarabine, cyclophosphamide, and rituximab in patients with relapsed or refractory chronic lymphocytic leukemia. *Blood* **115**, 489–495.
- Thornton, P. D, Hamblin, M, Treleaven, J. G, Matutes, E, Lakhani, A. K, & Catovsky, D. (1999) High dose methyl prednisolone in refractory chronic lymphocytic leukaemia. *Leukemia & Lymphoma* 34, 167–170.
- 152. Castro, J. E, Sandoval-Sus, J. D, Bole, J, Rassenti, L, & Kipps, T. J. (2008) Rituximab in combination with high-dose methylprednisolone for the treatment of fludarabine refractory high-risk chronic lymphocytic leukemia. *Leukemia* **22**, 2048–2053.

- 153. Gribben, J. G. (2007) Stem-cell transplantation in chronic lymphocytic leukaemia. *Best Practice & Research Clinical Haematology* **20**, 513–527.
- 154. Gribben, J. G, Zahrieh, D, Stephans, K, Bartlett-Pandite, L, Alyea, E. P, Fisher, D. C, Freedman, A. S, Mauch, P, Schlossman, R, Sequist, L. V, Soiffer, R. J, Marshall, B, Neuberg, D, Ritz, J, & Nadler, L. M. (2005) Autologous and allogeneic stem cell transplantations for poor-risk chronic lymphocytic leukemia. *Blood* **106**, 4389– 4396.
- 155. Ritgen, M, Stilgenbauer, S, von Neuhoff, N, Humpe, A, Bruggemann, M, Pott, C, Raff, T, Krober, A, Bunjes, D, Schlenk, R, Schmitz, N, Döhner, H, Kneba, M, & Dreger, P. (2004) Graft-versus-leukemia activity may overcome therapeutic resistance of chronic lymphocytic leukemia with unmutated immunoglobulin variable heavy-chain gene status: implications of minimal residual disease measurement with quantitative PCR. *Blood* **104**, 2600–2602.
- 156. Nishida, T, Hudecek, M, Kostic, A, Bleakley, M, Warren, E. H, Maloney, D, Storb, R, & Riddell, S. R. (2009) Development of tumor-reactive T cells after nonmyeloablative allogeneic hematopoietic stem cell transplant for chronic lymphocytic leukemia. *Clinical Cancer Research* **15**, 4759–4768.
- 157. Sorror, M. L, Storer, B. E, Sandmaier, B. M, Maris, M, Shizuru, J, Maziarz, R, Agura, E, Chauncey, T. R, Pulsipher, M. A, McSweeney, P. A, Wade, J. C, Bruno, B, Langston, A, Radich, J, Niederwieser, D, Blume, K. G, Storb, R, & Maloney, D. G. (2008) Five-year follow-up of patients with advanced chronic lymphocytic leukemia treated with allogeneic hematopoietic cell transplantation after nonmyeloablative conditioning. *Journal of Clinical Oncology* 26, 4912–4920.
- 158. Dreger, P, Corradini, P, Kimby, E, Michallet, M, Milligan, D, Schetelig, J, Wiktor-Jedrzejczak, W, Niederwieser, D, Hallek, M, & Montserrat, E. (2007) Indications for allogeneic stem cell transplantation in chronic lymphocytic leukemia: the EBMT transplant consensus. *Leukemia* **21**, 12–17.
- 159. Weide, R, Mergenthaler, U, Friesenhahn, V, Kleboth, K, Heymanns, J, Thomalla, J, & Koppler, H. (2009) Bendamustine/mitoxantrone/rituximab: a short remission induction chemoimmunotherapy for elderly patients with relapsed or refractory chronic lymphocytic leukemia. *Leukemia & Lymphoma* **50**, 1468–1474.
- 160. Ferrajoli, A, Lee, B. N, Schlette, E. J, O'Brien, S. M, Gao, H, Wen, S, Wierda, W. G, Estrov, Z, Faderl, S. H, Cohen, E. N, Li, C, Reuben, J. M, & Keating, M. J. (2008) Lenalidomide induces complete and partial remissions in patients with relapsed and refractory chronic lymphocytic leukemia. *Blood* **111**, 5291–5297.
- 161. Andritsos, L. A, Johnson, A. J, Lozanski, G, Blum, W, Kefauver, C, Awan, F, Smith, L. L, Lapalombella, R, May, S. E, Raymond, C. A, Wang, D. S, Knight, R. D, Ruppert, A. S, Lehman, A, Jarjoura, D, Chen, C. S, & Byrd, J. C. (2008) Higher doses of lenalidomide are associated with unacceptable toxicity including life-threatening tumor flare in patients with chronic lymphocytic leukemia. *Journal of Clinical Oncology* **26**, 2519–2525.
- 162. Byrd, J. C, Shinn, C, Waselenko, J. K, Fuchs, E. J, Lehman, T. A, Nguyen, P. L, Flinn, I. W, Diehl, L. F, Sausville, E, & Grever, M. R. (1998) Flavopiridol induces apoptosis

in chronic lymphocytic leukemia cells via activation of caspase-3 without evidence of bcl-2 modulation or dependence on functional p53. *Blood* **92**, 3804–3816.

- 163. Byrd, J. C, Lin, T. S, Dalton, J. T, Wu, D, Phelps, M. A, Fischer, B, Moran, M, Blum, K. A, Rovin, B, Brooker-McEldowney, M, Broering, S, Schaaf, L. J, Johnson, A. J, Lucas, D. M, Heerema, N. A, Lozanski, G, Young, D. C, Suarez, J.-R, Colevas, A. D, & Grever, M. R. (2007) Flavopiridol administered using a pharmacologically derived schedule is associated with marked clinical efficacy in refractory, genetically high-risk chronic lymphocytic leukemia. *Blood* **109**, 399–404.
- 164. Bates, D. J, Salerni, B. L, Lowrey, C. H, & Eastman, A. (2011) Vinblastine sensitizes leukemia cells to cyclin-dependent kinase inhibitors, inducing acute cell cycle phaseindependent apoptosis. *Cancer Biology & Therapy* 12, 314–25.
- 165. O'brien, S, Moore, J. O, Boyd, T. E, Larratt, L. M, Skotnicki, A, Koziner, B, Chanan-Khan, A. A, Seymour, J. F, Bociek, R. G, Pavletic, S, & Rai, K. R. (2007) Randomized phase III trial of fludarabine plus cyclophosphamide with or without oblimersen sodium (Bcl-2 antisense) in patients with relapsed or refractory chronic lymphocytic leukemia. *Journal of Clinical Oncology* 25, 1114–1120.
- 166. Oltersdorf, T, Elmore, S. W, Shoemaker, A. R, Armstrong, R. C, Augeri, D. J, Belli, B. A, Bruncko, M, Deckwerth, T. L, Dinges, J, Hajduk, P. J, Joseph, M. K, Kitada, S, Korsmeyer, S. J, Kunzer, A. R, Letai, A, Li, C, Mitten, M. J, Nettesheim, D. G, Ng, S, Nimmer, P. M, O'Connor, J. M, Oleksijew, A, Petros, A. M, Reed, J. C, Shen, W, Tahir, S. K, Thompson, C. B, Tomaselli, K. J, Wang, B, Wendt, M. D, Zhang, H, Fesik, S. W, & Rosenberg, S. H. (2005) An inhibitor of Bcl-2 family proteins induces regression of solid tumours. *Nature* 435, 677–81.
- 167. Mason, K. D, Khaw, S. L, Rayeroux, K. C, Chew, E, Lee, E. F, Fairlie, W. D, Grigg, A. P, Seymour, J. F, Szer, J, Huang, D. C, & Roberts, A. W. (2009) The BH3 mimetic compound, ABT-737, synergizes with a range of cytotoxic chemotherapy agents in chronic lymphocytic leukemia. *Leukemia* 23, 2034–41.
- 168. Roberts, A. W, Seymour, J. F, Brown, J. R, Wierda, W. G, Kipps, T. J, Khaw, S. L, Carney, D. A, He, S. Z, Huang, D. C, Xiong, H, Cui, Y, Busman, T. A, McKeegan, E. M, Krivoshik, A. P, Enschede, S. H, & Humerickhouse, R. (2012) Substantial susceptibility of chronic lymphocytic leukemia to BCL2 inhibition: results of a phase I study of navitoclax in patients with relapsed or refractory disease. *Journal of Clinical Oncology* **30**, 488–96.
- 169. Honigberg, L. A, Smith, A. M, Sirisawad, M, Verner, E, Loury, D, Chang, B, Li, S, Pan, Z, Thamm, D. H, Miller, R. A, & Buggy, J. J. (2010) The Bruton tyrosine kinase inhibitor PCI-32765 blocks B-cell activation and is efficacious in models of autoimmune disease and B-cell malignancy. *Proceedings of the National Academy of Sciences of the United States of America* 107, 13075–80.
- 170. Herman, S. E, Gordon, A. L, Hertlein, E, Ramanunni, A, Zhang, X, Jaglowski, S, Flynn, J, Jones, J, Blum, K. A, Buggy, J. J, Hamdy, A, Johnson, A. J, & Byrd, J. C. (2011) Bruton tyrosine kinase represents a promising therapeutic target for treatment of chronic lymphocytic leukemia and is effectively targeted by pci-32765. *Blood* **117**, 6287–96.
- 171. Byrd, J. C, Furman, R. R, Coutre, S. E, Burger, J. A, Blum, K. A, Sharman, J. P, Flinn, I. W, Grant, B. W, Heerema, N. A, Johnson, A. J, Navarro, T, James, D. F, Hedrick, E, & O'Brien, S. M. (2012) *The Bruton's tyrosine kinase (BTK) inhibitor PCI-32765 (P) in treatment-naive (TN) chronic lymphocytic leukemia (CLL) patients (pts): Interim results of a phase Ib/II study.* (American Society of Oncology Annual Meeting).
- 172. Lannutti, B. J, Meadows, S. A, Herman, S. E, Kashishian, A, Steiner, B, Johnson, A. J, Byrd, J. C, Tyner, J. W, Loriaux, M. M, Deininger, M, Druker, B. J, Puri, K. D, Ulrich, R. G, & Giese, N. A. (2011) CAL-101, a p110delta selective phosphatidylinositol-3kinase inhibitor for the treatment of B-cell malignancies, inhibits PI3K signaling and cellular viability. *Blood* **117**, 591–4.
- 173. Furman, R. R, Barrientos, J. C, Sharman, J. P, De Vos, S, Leonard, J, Coutre, S. E, Schreeder, M. T, Wagner-Johnston, N. D, Boyd, T. E, Fowler, N. H, Flinn, I. W, Boccia, R. V, Holes, L, Lannutti, B. J, Johnson, D, Jahn, T. M, & Miller, L. L. (2012) A phase I/II study of the selective phosphatidylinositol 3-kinase-delta (PI3K8) inhibitor, GS-1101 (CAL-101), with ofatumumab in patients with previously treated chronic lymphocytic leukemia (CLL). (American Society of Oncology Annual Meeting).
- 174. Wiestner, A. (2012) Emerging role of kinase targeted strategies in chronic lymphocytic leukemia. *Blood* [Epub ahead of print].
- 175. Ysebaert, L, Gross, E, Kuhlein, E, Blanc, A, Corre, J, Fournie, J. J, Laurent, G, & Quillet-Mary, A. (2010) Immune recovery after fludarabine-cyclophosphamiderituximab treatment in B-chronic lymphocytic leukemia: implication for maintenance immunotherapy. *Leukemia* **24**, 1310–1316.
- 176. Robinson, K. S, Williams, M. E, van der Jagt, R. H, Cohen, P, Herst, J. A, Tulpule, A, Schwartzberg, L. S, Lemieux, B, & Cheson, B. D. (2008) Phase ii multicenter study of bendamustine plus rituximab in patients with relapsed indolent B-cell and mantle cell non-Hodgkin's lymphoma. *Journal of Clinical Oncology* **26**, 4473–9.
- 177. Pettitt, A. R, Jackson, R, Carruthers, S, Dodd, J, Dodd, S, Oates, M, Johnson, G. G, Schuh, A, Matutes, E, Dearden, C. E, Catovsky, D, Radford, J. A, Bloor, A, Follows, G. A, Devereux, S, Kruger, A, Blundell, J, Agrawal, S, Allsup, D, Proctor, S, Heartin, E, Oscier, D, Hamblin, T. J, Rawstron, A, & Hillmen, P. (2012) Alemtuzumab in combination with methylprednisolone is a highly effective induction regimen for patients with chronic lymphocytic leukemia and deletion of tp53: final results of the national cancer research institute cll206 trial. *Journal of Clinical Oncology* 30, 1647–55.
- 178. Lepretre, S, Aurran, T, Mahe, B, Cazin, B, Tournilhac, O, Maisonneuve, H, Casasnovas, O, Delmer, A, Leblond, V, Royer, B, Corront, B, Chevret, S, Delepine, R, Vaudaux, S, Van Den Neste, E, Bene, M. C, Letestu, R, Cymbalista, F, & Feugier, P. (2012) Excess mortality after treatment with fludarabine and cyclophosphamide in combination with alemtuzumab in previously untreated patients with chronic lymphocytic leukemia in a randomized phase 3 trial. *Blood* **119**, 5104–10.
- 179. Rao, S. P, Sancho, J, Campos-Rivera, J, Boutin, P. M, Severy, P. B, Weeden, T, Shankara, S, Roberts, B. L, & Kaplan, J. M. (2012) Human peripheral blood mononuclear

cells exhibit heterogeneous CD52 expression levels and show differential sensitivity to alemtuzumab mediated cytolysis. *PLoS ONE* **7**, e39416.

- 180. Auffermann-Gretzinger, S, Eger, L, Schetelig, J, Bornhauser, M, Heidenreich, F, & Ehninger, G. (2007) Alemtuzumab depletes dendritic cells more effectively in blood than in skin: a pilot study in patients with chronic lymphocytic leukemia. *Transplantation* **83**, 1268–72.
- 181. Bedognetti, D, Zoppoli, G, Massucco, C, Zanardi, E, Zupo, S, Bruzzone, A, Sertoli, M. R, Balleari, E, Racchi, O, Messina, M, Caltabiano, G, Icardi, G, Durando, P, Marincola, F. M, Boccardo, F, Ferrarini, M, Ansaldi, F, & De Maria, A. (2011) Impaired response to influenza vaccine associated with persistent memory B cell depletion in non-Hodgkin's lymphoma patients treated with rituximab-containing regimens. *Journal of Immunology* **186**, 6044–6055.
- 182. Beyer, M, Kochanek, M, Darabi, K, Popov, A, Jensen, M, Endl, E, Knolle, P. A, Thomas, R. K, von Bergwelt-Baildon, M, Debey, S, Hallek, M, & Schultze, J. L. (2005) Reduced frequencies and suppressive function of CD4+CD25hi regulatory T cells in patients with chronic lymphocytic leukemia after therapy with fludarabine. *Blood* 106, 2018–2025.
- 183. Hosoda, T, Yokoyama, A, Yoneda, M, Yamamoto, R, Ohashi, K, Kagoo, T, Ueno, H, Boku, S, & Yano, T. (2012) Bendamustine can severely impair T-cell immunity against cytomegalovirus. *Leukemia & Lymphoma* [Epub ahead of print].
- 184. Lundin, J, Porwit-MacDonald, A, Rossmann, E. D, Karlsson, C, Edman, P, Rezvany, M. R, Kimby, E, Osterborg, A, & Mellstedt, H. (2004) Cellular immune reconstitution after subcutaneous alemtuzumab (anti-CD52 monoclonal antibody, CAMPATH-1H) treatment as first-line therapy for B-cell chronic lymphocytic leukaemia. *Leukemia* 18, 484–90.
- 185. Sageshima, J, Ciancio, G, Guerra, G, Gaynor, J. J, Cova, D, Zarak, A, Chen, L, Mattiazzi, A, Kupin, W, Roth, D, Hanson, L, Livingstone, A. S, Ruiz, P, & Burke, G. W. (2011) Prolonged lymphocyte depletion by single-dose rabbit anti-thymocyte globulin and alemtuzumab in kidney transplantation. *Transplant Immunology* 25, 104–11.
- 186. Aue, G, Njuguna, N, Tian, X, Soto, S, Hughes, T, Vire, B, Keyvanfar, K, Gibellini, F, Valdez, J, Boss, C, Samsel, L, McCoy, J. P., J, Wilson, W. H, Pittaluga, S, & Wiestner, A. (2009) Lenalidomide-induced upregulation of CD80 on tumor cells correlates with t-cell activation, the rapid onset of a cytokine release syndrome and leukemic cell clearance in chronic lymphocytic leukemia. *Haematologica* **94**, 1266–73.
- 187. Maloney, D. G, Grillo-Lopez, A. J, White, C. A, Bodkin, D, Schilder, R. J, Neidhart, J. A, Janakiraman, N, Foon, K. A, Liles, T. M, Dallaire, B. K, Wey, K, Royston, I, Davis, T, & Levy, R. (1997) IDEC-C2B8 (rituximab) anti-CD20 monoclonal antibody therapy in patients with relapsed low-grade non-Hodgkin's lymphoma. *Blood* 90, 2188–95.
- 188. Yri, O. E, Torfoss, D, Hungnes, O, Tierens, A, Waalen, K, Nordoy, T, Dudman, S, Kilander, A, Wader, K. F, Ostenstad, B, Ekanger, R, Meyer, P, & Kolstad, A. (2011)

Rituximab blocks protective serologic response to influenza A (H1N1) 2009 vaccination in lymphoma patients during or within 6 months after treatment. *Blood* **118**, 6769–71.

- 189. Hagenbeek, A, Gadeberg, O, Johnson, P, Pedersen, L. M, Walewski, J, Hellmann, A, Link, B. K, Robak, T, Wojtukiewicz, M, Pfreundschuh, M, Kneba, M, Engert, A, Sonneveld, P, Flensburg, M, Petersen, J, Losic, N, & Radford, J. (2008) First clinical use of ofatumumab, a novel fully human anti-cd20 monoclonal antibody in relapsed or refractory follicular lymphoma: results of a phase 1/2 trial. *Blood* **111**, 5486–95.
- 190. Heinzel, A. S, Grotzke, J. E, Lines, R. A, Lewinsohn, D. A, McNabb, A. L, Streblow, D. N, Braud, V. M, Grieser, H. J, Belisle, J. T, & Lewinsohn, D. M. (2002) HLA-E-dependent presentation of Mtb-derived antigen to human CD8+ T cells. *Journal of Experimental Medicine* 196, 1473–1481.
- 191. Gold, M. C, Cerri, S, Smyk-Pearson, S, Cansler, M. E, Vogt, T. M, Delepine, J, Winata, E, Swarbrick, G. M, Chua, W.-J, Yu, Y. Y. L, Lantz, O, Cook, M. S, Null, M. D, Jacoby, D. B, Harriff, M. J, Lewinsohn, D. A, Hansen, T. H, & Lewinsohn, D. M. (2010) Human mucosal associated invariant T cells detect bacterially infected cells. *PLoS Biology* 8, e1000407.
- 192. Porcelli, S, Brenner, M. B, Greenstein, J. L, Balk, S. P, Terhorst, C, & Bleicher, P. A. (1989) Recognition of cluster of differentiation 1 antigens by human CD4-CD8-cytolytic T lymphocytes. *Nature* **341**, 447–450.
- 193. Han, M, Hannick, L. I, DiBrino, M, & Robinson, M. A. (1999) Polymorphism of human CD1 genes. *Tissue Antigens* 54, 122–127.
- 194. Porcelli, S. A & Modlin, R. L. (1999) The CD1 system: antigen-presenting molecules for T cell recognition of lipids and glycolipids. *Annual Review of Immunology* **17**, 297–329.
- 195. Montamat-Sicotte, D. J, Millington, K. A, Willcox, C. R, Hingley-Wilson, S, Hackforth, S, Innes, J, Kon, O. M, Lammas, D. A, Minnikin, D. E, Besra, G. S, Willcox, B. E, & Lalvani, A. (2011) A mycolic acid–specific CD1-restricted T cell population contributes to acute and memory immune responses in human tuberculosis infection. *Journal of Clinical Investigation* **121**, 2493–2503.
- 196. Felio, K, Nguyen, H, Dascher, C. C, Choi, H.-J, Li, S, Zimmer, M. I, Colmone, A, Moody, D. B, Brenner, M. B, & Wang, C.-R. (2009) CD1-restricted adaptive immune responses to Mycobacteria in human group 1 CD1 transgenic mice. *Journal of Experimental Medicine* **206**, 2497–2509.
- 197. Van Rhijn, I, Koets, A. P, Im, J. S, Piebes, D, Reddington, F, Besra, G. S, Porcelli, S. A, van Eden, W, & Rutten, V. P. M. G. (2006) The bovine CD1 family contains group 1 CD1 proteins, but no functional CD1d. *Journal of Immunology* **176**, 4888–4893.
- 198. De Jong, A, Peña-Cruz, V, Cheng, T.-Y, Clark, R. A, Van Rhijn, I, & Moody, D. B. (2010) CD1a-autoreactive T cells are a normal component of the human $\alpha\beta$ T cell repertoire. *Nature Immunology* **11**, 1102–1109.
- 199. Moody, D. B, Young, D. C, Cheng, T.-Y, Rosat, J.-P, Roura-Mir, C, O'Connor, P. B, Zajonc, D. M, Walz, A, Miller, M. J, Levery, S. B, Wilson, I. A, Costello, C. E, &

Brenner, M. B. (2004) T cell activation by lipopeptide antigens. *Science* **303**, 527–531.

- 200. Russano, A. M, Bassotti, G, Agea, E, Bistoni, O, Mazzocchi, A, Morelli, A, Porcelli, S. A, & Spinozzi, F. (2007) CD1-restricted recognition of exogenous and self-lipid antigens by duodenal gammadelta+ T lymphocytes. *Journal of Immunology* **178**, 3620–3626.
- 201. Sieling, P. A, Chatterjee, D, Porcelli, S. A, Prigozy, T. I, Mazzaccaro, R. J, Soriano, T, Bloom, B. R, Brenner, M. B, Kronenberg, M, & Brennan, P. J. (1995) CD1-restricted T cell recognition of microbial lipoglycan antigens. *Science* 269, 227–230.
- 202. Cui, Y, Kang, L, Cui, L, & He, W. (2009) Human gammadelta T cell recognition of lipid A is predominately presented by CD1b or CD1c on dendritic cells. *Biology Direct* **4**, 47.
- 203. Moody, D. B, Ulrichs, T, Muhlecker, W, Young, D. C, Gurcha, S. S, Grant, E, Rosat, J. P, Brenner, M. B, Costello, C. E, Besra, G. S, & Porcelli, S. A. (2000) CD1c-mediated T-cell recognition of isoprenoid glycolipids in Mycobacterium tuberculosis infection. *Nature* 404, 884–888.
- 204. Godfrey, D. I, Pellicci, D. G, Patel, O, Kjer-Nielsen, L, Mccluskey, J, & Rossjohn, J. (2010) Antigen recognition by CD1d-restricted NKT T cell receptors. *Seminars in Immunology* 22, 61–67.
- 205. Kawano, T, Cui, J, Koezuka, Y, Toura, I, Kaneko, Y, Motoki, K, Ueno, H, Nakagawa, R, Sato, H, Kondo, E, Koseki, H, & Taniguchi, M. (1997) CD1d-restricted and TCR-mediated activation of valpha14 NKT cells by glycosylceramides. *Science* 278, 1626–1629.
- 206. Chang, D. H, Deng, H, Matthews, P, Krasovsky, J, Ragupathi, G, Spisek, R, Mazumder, A, Vesole, D. H, Jagannath, S, & Dhodapkar, M. V. (2008) Inflammationassociated lysophospholipids as ligands for CD1d-restricted T cells in human cancer. *Blood* **112**, 1308–1316.
- 207. de la Salle, H, Mariotti, S, Angenieux, C, Gilleron, M, Garcia-Alles, L.-F, Malm, D, Berg, T, Paoletti, S, Maître, B, Mourey, L, Salamero, J, Cazenave, J. P, Hanau, D, Mori, L, Puzo, G, & Libero, G. D. (2005) Assistance of microbial glycolipid antigen processing by CD1e. *Science* **310**, 1321–1324.
- 208. Berzins, S. P, Kyparissoudis, K, Pellicci, D. G, Hammond, K. J, Sidobre, S, Baxter, A, Smyth, M. J, Kronenberg, M, & Godfrey, D. I. (2004) Systemic NKT cell deficiency in NOD mice is not detected in peripheral blood: implications for human studies. *Immunology and Cell Biology* **82**, 247–252.
- 209. Godfrey, D. I, MacDonald, H. R, Kronenberg, M, Smyth, M. J, & Van Kaer, L. (2004) NKT cells: what's in a name? *Nature Reviews Immunology* **4**, 231–237.
- 210. Molling, J. W, Kölgen, W, van der Vliet, H. J. J, Boomsma, M. F, Kruizenga, H, Smorenburg, C. H, Molenkamp, B. G, Langendijk, J. A, Leemans, C. R, von Blomberg, B. M. E, Scheper, R. J, & van den Eertwegh, A. J. M. (2005) Peripheral blood IFN-gamma-secreting Valpha24+Vbeta11+ NKT cell numbers are decreased

in cancer patients independent of tumor type or tumor load. *International Journal of Cancer* **116**, 87–93.

- 211. Sandberg, J. K, Bhardwaj, N, & Nixon, D. F. (2003) Dominant effector memory characteristics, capacity for dynamic adaptive expansion, and sex bias in the innate Valpha24 NKT cell compartment. *European Journal of Immunology* **33**, 588–596.
- 212. Kenna, T, Golden-Mason, L, Porcelli, S. A, Koezuka, Y, Hegarty, J. E, O'farrelly, C, Doherty, D. G, & Mason, L. G. (2003) NKT cells from normal and tumor-bearing human livers are phenotypically and functionally distinct from murine NKT cells. *Journal of Immunology* **171**, 1775–1779.
- 213. Terabe, M & Berzofsky, J. A. (2007) NKT cells in immunoregulation of tumor immunity: a new immunoregulatory axis. *Trends in Immunology* **28**, 491–496.
- 214. Exley, M. A, Tahir, S. M, Cheng, O, Shaulov, A, Joyce, R, Avigan, D, Sackstein, R, & Balk, S. P. (2001) A major fraction of human bone marrow lymphocytes are Th2-like CD1d-reactive T cells that can suppress mixed lymphocyte responses. *Journal of Immunology* **167**, 5531–5534.
- 215. Uemura, Y, Suzuki, M, Liu, T.-Y, Narita, Y, Hirata, S, Ohyama, H, Ishihara, O, & Matsushita, S. (2008) Role of human non-invariant NKT lymphocytes in the maintenance of type 2 T helper environment during pregnancy. *International Immunology* 20, 405–412.
- 216. Pang, D. J, Neves, J. F, Sumaria, N, & Pennington, D. J. (2012) Understanding the complexity of gammadelta t-cell subsets in mouse and human. *Immunology* 136, 283–90.
- 217. Le Bourhis, L, Guerri, L, Dusseaux, M, Martin, E, Soudais, C, & Lantz, O. (2011) Mucosal-associated invariant T cells: unconventional development and function. *Trends in Immunology* **32**, 212–8.
- 218. Montoya, C. J, Pollard, D, Martinson, J, Kumari, K, Wasserfall, C, Mulder, C. B, Rugeles, M. T, Atkinson, M. A, Landay, A. L, & Wilson, S. B. (2007) Characterization of human invariant natural killer T subsets in health and disease using a novel invariant natural killer T cell-clonotypic monoclonal antibody, 6B11. *Immunology* **122**, 1–14.
- 219. Lee, P. T, Benlagha, K, Teyton, L, & Bendelac, A. (2002) Distinct functional lineages of human V(alpha)24 natural killer T cells. *Journal of Experimental Medicine* **195**, 637–641.
- 220. Gumperz, J. E, Miyake, S, Yamamura, T, & Brenner, M. B. (2002) Functionally distinct subsets of CD1d-restricted natural killer T cells revealed by CD1d tetramer staining. *Journal of Experimental Medicine* **195**, 625–636.
- 221. Rolf, J, Berntman, E, Stenstrom, M, Smith, E. M, Mansson, R, Stenstad, H, Yamagata, T, Agace, W, Sigvardsson, M, & Cardell, S. L. (2008) Molecular profiling reveals distinct functional attributes of CD1d-restricted natural killer (NK) T cell subsets. *Molecular Immunology*.
- 222. Crowe, N. Y, Coquet, J. M, Berzins, S. P, Kyparissoudis, K, Keating, R, Pellicci, D. G,

Hayakawa, Y, Godfrey, D. I, & Smyth, M. J. (2005) Differential antitumor immunity mediated by NKT cell subsets in vivo. *Journal of Experimental Medicine* **202**, 1279–1288.

- 223. Jing, Y, Gravenstein, S, Chaganty, N. R, Chen, N, Lyerly, K. H, Joyce, S, & Deng, Y. (2007) Aging is associated with a rapid decline in frequency, alterations in subset composition, and enhanced Th2 response in CD1d-restricted NKT cells from human peripheral blood. *Experimental Gerontology* **42**, 719–732.
- 224. van Der Vliet, H. J, Nishi, N, de Gruijl, T. D, von Blomberg, B. M, van den Eertwegh, A. J, Pinedo, H. M, Giaccone, G, & Scheper, R. J. (2000) Human natural killer T cells acquire a memory-activated phenotype before birth. *Blood* **95**, 2440–2442.
- 225. Vilarinho, S, Ogasawara, K, Nishimura, S, Lanier, L. L, & Baron, J. L. (2007) Blockade of NKG2D on NKT cells prevents hepatitis and the acute immune response to hepatitis B virus. *Proceedings of the National Academy of Sciences of the United States of America* **104**, 18187–18192.
- 226. Kawamura, T, Takeda, K, Kaneda, H, Matsumoto, H, Hayakawa, Y, Raulet, D. H, Ikarashi, Y, Kronenberg, M, Yagita, H, Kinoshita, K, Abo, T, Okumura, K, & Smyth, M. J. (2009) NKG2A inhibits invariant NKT cell activation in hepatic injury. *Journal of Immunology* **182**, 250–258.
- 227. Exley, M, Porcelli, S, Furman, M, Garcia, J, & Balk, S. (1998) CD161 (NKR-P1A) costimulation of CD1d-dependent activation of human T cells expressing invariant V alpha 24 J alpha Q T cell receptor alpha chains. *Journal of Experimental Medicine* **188**, 867–876.
- 228. Patterson, S, Chaidos, A, Neville, D. C. A, Poggi, A, Butters, T. D, Roberts, I. A. G, & Karadimitris, A. (2008) Human invariant NKT cells display alloreactivity instructed by invariant TCR-CD1d interaction and killer Ig receptors. *Journal of Immunology* 181, 3268–3276.
- 229. Metelitsa, L. S, Weinberg, K. I, Emanuel, P. D, & Seeger, R. C. (2003) Expression of CD1d by myelomonocytic leukemias provides a target for cytotoxic NKT cells. *Leukemia* **17**, 1068–1077.
- 230. Cox, D, Fox, L, Tian, R, Bardet, W, Skaley, M, Mojsilovic, D, Gumperz, J, & Hildebrand, W. (2009) Determination of cellular lipids bound to human CD1d molecules. *PLoS One* **4**, e5325.
- 231. Venkataswamy, M. M & Porcelli, S. A. (2010) Lipid and glycolipid antigens of CD1drestricted natural killer T cells. *Seminars in Immunology* **22**, 68–78.
- Park, J.-E, Wu, D. Y, Prendes, M, Lu, S. X, Ragupathi, G, Schrantz, N, & Chapman, P. B. (2008) Fine specificity of natural killer T cells against GD3 ganglioside and identification of GM3 as an inhibitory natural killer T-cell ligand. *Immunology* 123, 145–155.
- 233. Jahng, A, Maricic, I, Aguilera, C, Cardell, S, Halder, R. C, & Kumar, V. (2004) Prevention of autoimmunity by targeting a distinct, noninvariant CD1d-reactive T cell population reactive to sulfatide. *Journal of Experimental Medicine* 199, 947–957.

- 234. Li, X, Fujio, M, Imamura, M, Wu, D, Vasan, S, Wong, C.-H, Ho, D. D, & Tsuji, M. (2010) Design of a potent CD1d-binding NKT cell ligand as a vaccine adjuvant. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 13010–13015.
- 235. Silk, J. D, Salio, M, Reddy, B. G, Shepherd, D, Gileadi, U, Brown, J, Masri, S. H, Polzella, P, Ritter, G, Besra, G. S, Jones, E. Y, Schmidt, R. R, & Cerundolo, V. (2008) Cutting edge: nonglycosidic CD1d lipid ligands activate human and murine invariant NKT cells. *Journal of Immunology* **180**, 6452–6456.
- 236. Oki, S, Chiba, A, Yamamura, T, & Miyake, S. (2004) The clinical implication and molecular mechanism of preferential IL-4 production by modified glycolipid-stimulated NKT cells. *Journal of Clinical Investigation* **113**, 1631–1640.
- 237. Bricard, G, Venkataswamy, M. M, Yu, K. O, Im, J. S, Ndonye, R. M, Howell, A. R, Veerapen, N, Illarionov, P. A, Besra, G. S, Li, Q, Chang, Y. T, & Porcelli, S. A. (2010) Alpha-galactosylceramide analogs with weak agonist activity for human inkt cells define new candidate anti-inflammatory agents. *PLoS ONE* **5**, e14374.
- 238. Kumar, V, Bhardwaj, V, Soares, L, Alexander, J, Sette, A, & Sercarz, E. (1995) Major histocompatibility complex binding affinity of an antigenic determinant is crucial for the differential secretion of interleukin 4/5 or interferon gamma by t cells. *Proceedings of the National Academy of Sciences of the United States of America* **92**, 9510–4.
- 239. Pfeiffer, C, Stein, J, Southwood, S, Ketelaar, H, Sette, A, & Bottomly, K. (1995) Altered peptide ligands can control cd4 t lymphocyte differentiation in vivo. *Journal of Experimental Medicine* **181**, 1569–74.
- 240. Matsuda, J. L, Mallevaey, T, Scott-Browne, J, & Gapin, L. (2008) CD1d-restricted iNKT cells, the 'Swiss-Army knife' of the immune system. *Current Opinion in Immunology* **20**, 358–368.
- 241. Coquet, J. M, Chakravarti, S, Kyparissoudis, K, McNab, F. W, Pitt, L. A, McKenzie, B. S, Berzins, S. P, Smyth, M. J, & Godfrey, D. I. (2008) Diverse cytokine production by NKT cell subsets and identification of an IL-17-producing CD4-NK1.1- NKT cell population. *Proceedings of the National Academy of Sciences of the United States of America* 105, 11287–11292.
- 242. Kitamura, H, Iwakabe, K, Yahata, T, Nishimura, S, Ohta, A, Ohmi, Y, Sato, M, Takeda, K, Okumura, K, Van Kaer, L, Kawano, T, Taniguchi, M, & Nishimura, T. (1999) The natural killer T (NKT) cell ligand alpha-galactosylceramide demonstrates its immunopotentiating effect by inducing interleukin (IL)-12 production by dendritic cells and IL-12 receptor expression on NKT cells. *Journal of Experimental Medicine* **189**, 1121–1128.
- 243. Hayakawa, Y, Takeda, K, Yagita, H, Van Kaer, L, Saiki, I, & Okumura, K. (2001) Differential regulation of Th1 and Th2 functions of NKT cells by CD28 and CD40 costimulatory pathways. *Journal of Immunology* **166**, 6012–6018.
- 244. Brigl, M, Bry, L, Kent, S. C, Gumperz, J. E, & Brenner, M. B. (2003) Mechanism of

CD1d-restricted natural killer T cell activation during microbial infection. *Nature Immunology* **4**, 1230–1237.

- 245. Im, J. S, Tapinos, N, Chae, G. T, Illarionov, P. A, Besra, G. S, DeVries, G. H, Modlin, R. L, Sieling, P. A, Rambukkana, A, & Porcelli, S. A. (2006) Expression of CD1d molecules by human schwann cells and potential interactions with immunoregulatory invariant NK T cells. *Journal of Immunology* **177**, 5226–5235.
- 246. Bezbradica, J. S, Stanic, A. K, Matsuki, N, Bour-Jordan, H, Bluestone, J. A, Thomas, J. W, Unutmaz, D, Van Kaer, L, & Joyce, S. (2005) Distinct roles of dendritic cells and B cells in Va14Ja18 natural T cell activation in vivo. *Journal of Immunology* **174**, 4696–4705.
- 247. Sakuishi, K, Oki, S, Araki, M, Porcelli, S. A, Miyake, S, & Yamamura, T. (2007) Invariant NKT cells biased for IL-5 production act as crucial regulators of inflammation. *Journal of Immunology* **179**, 3452–3462.
- 248. Molano, A, Illarionov, P. A, Besra, G. S, Putterman, C, & Porcelli, S. A. (2008) Modulation of invariant natural killer T cell cytokine responses by indoleamine 2,3dioxygenase. *Immunology Letters* **117**, 81–90.
- 249. Michel, M. L, Keller, A. C, Paget, C, Fujio, M, Trottein, F, Savage, P. B, Wong, C. H, Schneider, E, Dy, M, & Leite-de Moraes, M. C. (2007) Identification of an IL-17-producing NK1.1(neg) iNKT cell population involved in airway neutrophilia. *Journal of Experimental Medicine* 204, 995–1001.
- 250. Yoshiga, Y, Goto, D, Segawa, S, Ohnishi, Y, Matsumoto, I, Ito, S, Tsutsumi, A, Taniguchi, M, & Sumida, T. (2008) Invariant NKT cells produce IL-17 through IL-23-dependent and -independent pathways with potential modulation of Th17 response in collagen-induced arthritis. *International Journal of Molecular Medicine* **22**, 369–374.
- 251. Snyder-Cappione, J. E, Tincati, C, Eccles-James, I. G, Cappione, A. J, Ndhlovu, L. C, Koth, L. L, & Nixon, D. F. (2010) A Comprehensive Ex Vivo Functional Analysis of Human NKT Cells Reveals Production of MIP1- α and MIP1- β a Lack of IL-17, and a Th1-Bias in Males. *PLoS One* **5**, e15412.
- 252. Rachitskaya, A. V, Hansen, A. M, Horai, R, Li, Z, Villasmil, R, Luger, D, Nussenblatt, R. B, & Caspi, R. R. (2008) Cutting edge: NKT cells constitutively express IL-23 receptor and RORgammat and rapidly produce IL-17 upon receptor ligation in an IL-6-independent fashion. *Journal of Immunology* **180**, 5167–5171.
- 253. Gansert, J. L, Kiessler, V, Engele, M, Wittke, F, Röllinghoff, M, Krensky, A. M, Porcelli, S. A, Modlin, R. L, & Stenger, S. (2003) Human NKT cells express granulysin and exhibit antimycobacterial activity. *Journal of Immunology* **170**, 3154–3161.
- 254. Wingender, G, Krebs, P, Beutler, B, & Kronenberg, M. (2010) Antigen-specific cytotoxicity by invariant NKT cells in vivo is CD95/CD178-dependent and is correlated with antigenic potency. *Journal of Immunology* **185**, 2721–2729.
- 255. Gansuvd, B, Hagihara, M, Yu, Y, Inoue, H, Ueda, Y, Tsuchiya, T, Masui, A, Ando, K, Nakamura, Y, Munkhtuvshin, N, Kato, S, Thomas, J. M, & Hotta, T. (2002) Hu-

man umbilical cord blood NK T cells kill tumors by multiple cytotoxic mechanisms. *Human Immunology* **63**, 164–175.

- 256. Nakagawa, R, Nagafune, I, Tazunoki, Y, Ehara, H, Tomura, H, Iijima, R, Motoki, K, Kamishohara, M, & Seki, S. (2001) Mechanisms of the antimetastatic effect in the liver and of the hepatocyte injury induced by alpha-galactosylceramide in mice. *Journal of Immunology* **166**, 6578–6584.
- 257. Nieda, M, Nicol, A, Koezuka, Y, Kikuchi, A, Lapteva, N, Tanaka, Y, Tokunaga, K, Suzuki, K, Kayagaki, N, Yagita, H, Hirai, H, & Juji, T. (2001) TRAIL expression by activated human CD4(+) Valpha24 NKT cells induces in vitro and in vivo apoptosis of human acute myeloid leukemia cells. *Blood* **97**, 2067–2074.
- 258. Hermans, I. F, Silk, J. D, Gileadi, U, Salio, M, Mathew, B, Ritter, G, Schmidt, R, Harris, A. L, Old, L, & Cerundolo, V. (2003) NKT cells enhance CD4+ and CD8+ T cell responses to soluble antigen in vivo through direct interaction with dendritic cells. *Journal of Immunology* **171**, 5140–5147.
- 259. Fujii, S.-I, Liu, K, Smith, C, Bonito, A. J, & Steinman, R. M. (2004) The linkage of innate to adaptive immunity via maturing dendritic cells in vivo requires CD40 ligation in addition to antigen presentation and CD80/86 costimulation. *Journal of Experimental Medicine* **199**, 1607–1618.
- 260. Fujii, S.-I, Shimizu, K, Smith, C, Bonifaz, L, & Steinman, R. M. (2003) Activation of natural killer T cells by alpha-galactosylceramide rapidly induces the full maturation of dendritic cells in vivo and thereby acts as an adjuvant for combined CD4 and CD8 T cell immunity to a coadministered protein. *Journal of Experimental Medicine* 198, 267–279.
- 261. Vincent, M. S, Leslie, D. S, Gumperz, J. E, Xiong, X, Grant, E. P, & Brenner, M. B. (2002) CD1-dependent dendritic cell instruction. *Nature Immunology* 3, 1163– 1168.
- 262. Hegde, S, Jankowska-Gan, E, Roenneburg, D. A, Torrealba, J, Burlingham, W. J, & Gumperz, J. E. (2009) Human NKT cells promote monocyte differentiation into suppressive myeloid antigen-presenting cells. *Journal of Leukocyte Biology* 86, 757– 768.
- 263. Chen, Y.-G, Choisy-Rossi, C.-M, Holl, T. M, Chapman, H. D, Besra, G. S, Porcelli, S. A, Shaffer, D. J, Roopenian, D, Wilson, S. B, & Serreze, D. V. (2005) Activated NKT cells inhibit autoimmune diabetes through tolerogenic recruitment of dendritic cells to pancreatic lymph nodes. *Journal of Immunology* 174, 1196–1204.
- 264. Monteiro, M, Almeida, C. F, Caridade, M, Ribot, J. C, Duarte, J, Agua-Doce, A, Wollenberg, I, Silva-Santos, B, & Graca, L. (2010) Identification of regulatory Foxp3+ invariant NKT cells induced by TGF-beta. *Journal of Immunology* 185, 2157–2163.
- 265. Cava, A, Kaer, L, & Fudongshi. (2006) CD4+CD25+ Tregs and NKT cells: regulators regulators. *Trends in Immunology* **27**, 322–327.
- 266. Galli, G, Pittoni, P, Tonti, E, Malzone, C, Uematsu, Y, Tortoli, M, Maione, D, Volpini, G, Finco, O, Nuti, S, Tavarini, S, Dellabona, P, Rappuoli, R, Casorati, G, & Abrig-

nani, S. (2007) Invariant NKT cells sustain specific B cell responses and memory. *Proceedings of the National Academy of Sciences of the United States of America* **104**, 3984–3989.

- 267. Lang, G. A, Devera, T. S, & Lang, M. L. (2008) Requirement for CD1d expression by B cells to stimulate NKT cell-enhanced antibody production. *Blood* **111**, 2158–2162.
- 268. Leadbetter, E. A, Brigl, M, Illarionov, P, Cohen, N, Luteran, M. C, Pillai, S, Besra, G. S, & Brenner, M. B. (2008) NK T cells provide lipid antigen-specific cognate help for B cells. *Proceedings of the National Academy of Sciences of the United States of America* 105, 8339–8344.
- 269. Bosma, A, Abdel-Gadir, A, Isenberg, D. A, Jury, E. C, & Mauri, C. (2012) Lipidantigen presentation by cd1d(+) b cells is essential for the maintenance of invariant natural killer t cells. *Immunity* **36**, 477–90.
- 270. Bialecki, E, Paget, C, Fontaine, J, Capron, M, Trottein, F, & Faveeuw, C. (2009) Role of marginal zone b lymphocytes in invariant nkt cell activation. *Journal of Immunology* **182**, 6105–13.
- 271. Smyth, M. J, Crowe, N. Y, Pellicci, D. G, Kyparissoudis, K, Kelly, J. M, Takeda, K, Yagita, H, & Godfrey, D. I. (2002) Sequential production of interferon-gamma by NK1.1(+) T cells and natural killer cells is essential for the antimetastatic effect of alpha-galactosylceramide. *Blood* **99**, 1259–1266.
- Smyth, M. J. (2005) Sequential activation of NKT cells and NK cells provides effective innate immunotherapy of cancer. *Journal of Experimental Medicine* 201, 1973– 1985.
- 273. Eberl, G & MacDonald, H. R. (2000) Selective induction of NK cell proliferation and cytotoxicity by activated NKT cells. *European Journal of Immunology* **30**, 985–992.
- 274. Coquet, J. M, Kyparissoudis, K, Pellicci, D. G, Besra, G, Berzins, S. P, Smyth, M. J, & Godfrey, D. I. (2007) IL-21 is produced by NKT cells and modulates NKT cell activation and cytokine production. *Journal of Immunology* **178**, 2827–2834.
- 275. Smyth, M. J, Wallace, M. E, Nutt, S. L, Yagita, H, Godfrey, D. I, & Hayakawa, Y. (2005) Sequential activation of NKT cells and NK cells provides effective innate immunotherapy of cancer. *Journal of Experimental Medicine* **201**, 1973–1985.
- 276. Moreno, M, Molling, J. W, von Mensdorff-Pouilly, S, Verheijen, R. H. M, Hooijberg, E, Kramer, D, Reurs, A. W, van den Eertwegh, A. J. M, von Blomberg, B. M. E, Scheper, R. J, & Bontkes, H. J. (2008) IFN-gamma-producing human invariant NKT cells promote tumor-associated antigen-specific cytotoxic T cell responses. *Journal of Immunology* 181, 2446–2454.
- 277. Leite-de Moraes, M. C, Lisbonne, M, Arnould, A, Machavoine, F, Herbelin, A, Dy, M, & Schneider, E. (2002) Ligand-activated natural killer T lymphocytes promptly produce IL-3 and GM-CSF in vivo: relevance to peripheral myeloid recruitment. *European Journal of Immunology* **32**, 1897–1904.
- 278. Lee, K. A, Kang, M. H, Lee, Y. S, Kim, Y. J, Kim, D. H, Ko, H. J, & Kang, C. Y. (2008) A distinct subset of natural killer T cells produces IL-17, contributing to airway

infiltration of neutrophils but not to airway hyperreactivity. *Cellular Immunology* **251**, 50–55.

- 279. Hwang, S. J, Kim, S, Park, W. S, & Chung, D. H. (2006) IL-4-secreting NKT cells prevent hypersensitivity pneumonitis by suppressing IFN-gamma-producing neutrophils. *Journal of Immunology* 177, 5258–5268.
- 280. Umetsu, D. T & Dekruyff, R. H. (2010) Natural killer T cells are important in the pathogenesis of asthma: the many pathways to asthma. *Journal of Allergy and Clinical Immunology* **125**, 975–979.
- 281. Brooks, C. R, Weinkove, R, Hermans, I. F, van Dalen, C. J, & Douwes, J. (2010) Invariant natural killer T cells and asthma: immunologic reality or methodologic artifact? *Journal of Allergy and Clinical Immunology* **126**, 882–885.
- 282. Song, L, Asgharzadeh, S, Salo, J, Engell, K, Wu, H.-w, Sposto, R, Ara, T, Silverman, A. M, DeClerck, Y. A, Seeger, R. C, & Metelitsa, L. S. (2009) Valpha24-invariant NKT cells mediate antitumor activity via killing of tumor-associated macrophages. *Journal of Clinical Investigation* **119**, 1524–1536.
- 283. De Santo, C, Salio, M, Masri, S. H, Lee, L. Y.-H, Dong, T, Speak, A. O, Porubsky, S, Booth, S, Veerapen, N, Besra, G. S, Gröne, H.-J, Platt, F. M, Zambon, M, & Cerundolo, V. (2008) Invariant NKT cells reduce the immunosuppressive activity of influenza A virus-induced myeloid-derived suppressor cells in mice and humans. *Journal of Clinical Investigation* **118**, 4036–4048.
- 284. Kinjo, Y & Kronenberg, M. (2005) V alpha14 i NKT cells are innate lymphocytes that participate in the immune response to diverse microbes. *Journal of Clinical Immunology* **25**, 522–533.
- 285. Montoya, C. J, Cataño, J. C, Ramirez, Z, Rugeles, M. T, Wilson, S. B, & Landay, A. L. (2008) Invariant NKT cells from HIV-1 or Mycobacterium tuberculosis-infected patients express an activated phenotype. *Clinical Immunology* **127**, 1–6.
- 286. Im, J, Kang, T, Lee, S, Kim, C, Lee, S, Venkataswamy, M, Serfass, E, Chen, B, Illarionov, P, & Besra, G. (2008) Alteration of the relative levels of iNKT cell subsets is associated with chronic mycobacterial infections. *Clinical Immunology* **127**, 214– 224.
- 287. Chattree, V, Khanna, N, Bisht, V, & Rao, D. N. (2008) Inhibition of apoptosis, activation of NKT cell and upregulation of CD40 and CD40L mediated by M. leprae antigen(s) combined with Murabutide and Trat peptide in leprosy patients. *Molecular and Cellular Biochemistry* **309**, 87–97.
- 288. Chang, J. H, Lee, J. M, Youn, H. J, Lee, K. A, Chung, Y, Lee, A. Y, Kweon, M. N, Kim, H. Y, Taniguchi, M, & Kang, C. Y. (2008) Functional maturation of lamina propria dendritic cells by activation of NKT cells mediates the abrogation of oral tolerance. *European Journal of Immunology* **38**, 2727–2739.
- 289. Akbari, O, Stock, P, Meyer, E, Kronenberg, M, Sidobre, S, Nakayama, T, Taniguchi, M, Grusby, M. J, DeKruyff, R. H, & Umetsu, D. T. (2003) Essential role of NKT cells producing IL-4 and IL-13 in the development of allergen-induced airway hyperreactivity. *Nature Medicine* **9**, 582–588.

- 290. Mattner, J, Savage, P. B, Leung, P, Oertelt, S. S, Wang, V, Trivedi, O, Scanlon, S. T, Pendem, K, Teyton, L, Hart, J, Ridgway, W. M, Wicker, L. S, Gershwin, M. E, & Bendelac, A. (2008) Liver autoimmunity triggered by microbial activation of natural killer T cells. *Cell Host & Microbe* **3**, 304–315.
- 291. Miellot-Gafsou, A, Biton, J, Bourgeois, E, Herbelin, A, Boissier, M. C, & Bessis, N. (2010) Early activation of invariant natural killer T cells in a rheumatoid arthritis model and application to disease treatment. *Immunology* **130**, 296–306.
- 292. Gober, M. D, Fishelevich, R, Zhao, Y, Unutmaz, D, & Gaspari, A. A. (2008) Human natural killer T cells infiltrate into the skin at elicitation sites of allergic contact dermatitis. *Journal of Investigative Dermatology* **128**, 1460–1469.
- 293. Miyamoto, K, Miyake, S, & Yamamura, T. (2001) A synthetic glycolipid prevents autoimmune encephalomyelitis by inducing TH2 bias of natural killer T cells. *Nature* **413**, 531–534.
- 294. Fletcher, M. T & Baxter, A. G. (2009) Clinical application of NKT cell biology in type I (autoimmune) diabetes mellitus. *Immunology and Cell Biology* **87**, 315–323.
- 295. Grajewski, R. S, Hansen, A. M, Agarwal, R. K, Kronenberg, M, Sidobre, S, Su, S. B, Silver, P. B, Tsuji, M, Franck, R. W, Lawton, A. P, Chan, C.-C, & Caspi, R. R. (2008) Activation of invariant NKT cells ameliorates experimental ocular autoimmunity by a mechanism involving innate IFN-gamma production and dampening of the adaptive Th1 and Th17 responses. *Journal of Immunology* **181**, 4791–4797.
- 296. Jacques, P, Venken, K, Van Beneden, K, Hammad, H, Seeuws, S, Drennan, M. B, Deforce, D, Verbruggen, G, Apostolaki, M, Kollias, G, Lambrecht, B. N, De Vos, M, & Elewaut, D. (2010) Invariant natural killer T cells are natural regulators of murine spondylarthritis. *Arthritis & Rheumatism* **62**, 988–999.
- 297. Onzuka, T, Tomita, Y, Shimizu, I, Okano, S, Yamada, H, Yoshikai, Y, & Tominaga, R. (2008) Role of the cytokine profiles produced by invariant natural killer T cells in the initial phase of cyclophosphamide-induced tolerance. *Transplantation* **86**, 1301–1310.
- 298. Liu, Y. P, Li, Z, Nador, R. G, & Strober, S. (2008) Simultaneous protection against allograft rejection and graft-versus-host disease after total lymphoid irradiation: role of natural killer T cells. *Transplantation* **85**, 607–614.
- 299. Pillai, A. B, George, T. I, Dutt, S, & Strober, S. (2009) Host natural killer T cells induce an IL-4 dependent expansion of donor CD4+CD25+Foxp3+ Tregs that protects against graft-versus-host disease. *Blood* pp. 1–11.
- 300. Kuns, R. D, Morris, E. S, Macdonald, K. P. A, Markey, K. A, Morris, H. M, Raffelt, N. C, Banovic, T, Don, A. L. J, Rowe, V, Burman, A. C, Clouston, A. D, Farah, C, Besra, G. S, Illarionov, P. A, Smyth, M. J, Porcelli, S. A, & Hill, G. R. (2009) Invariant natural killer T cell-natural killer cell interactions dictate transplantation outcome after alpha-galactosylceramide administration. *Blood* **113**, 5999–6010.
- 301. Smyth, M. J, Thia, K. Y, Street, S. E, Cretney, E, Trapani, J. A, Taniguchi, M, Kawano, T, Pelikan, S. B, Crowe, N. Y, & Godfrey, D. I. (2000) Differential tumor surveillance

by natural killer (NK) and NKT cells. *Journal of Experimental Medicine* **191**, 661–668.

- 302. Bellone, M, Ceccon, M, Grioni, M, Jachetti, E, Calcinotto, A, Napolitano, A, Freschi, M, Casorati, G, & Dellabona, P. (2010) inkt cells control mouse spontaneous carcinoma independently of tumor-specific cytotoxic t cells. *PLoS ONE* **5**, e8646.
- 303. Giaccone, G, Punt, C. J. A, Ando, Y, Ruijter, R, Nishi, N, Peters, M, von Blomberg, B. M. E, Scheper, R. J, van der Vliet, H. J. J, van den Eertwegh, A. J. M, Roelvink, M, Beijnen, J, Zwierzina, H, & Pinedo, H. M. (2002) A phase I study of the natural killer T-cell ligand alpha-galactosylceramide (KRN7000) in patients with solid tumors. *Clinical Cancer Research* 8, 3702–3709.
- 304. Tahir, S. M, Cheng, O, Shaulov, A, Koezuka, Y, Bubley, G. J, Wilson, S. B, Balk, S. P, & Exley, M. A. (2001) Loss of IFN-gamma production by invariant NK T cells in advanced cancer. *Journal of Immunology* **167**, 4046–4050.
- 305. Najera Chuc, A. E. Cervantes, L. A. Retiguin, F. P. Ojeda, J. V. & Maldonado, E. R. (2012) Low number of invariant nkt cells is associated with poor survival in acute myeloid leukemia. *Journal of Cancer Research and Clinical Oncology* 138, 1427–32.
- 306. Lynch, L, O'shea, D, Winter, D. C, Geoghegan, J, Doherty, D. G, & O'farrelly, C. (2009) Invariant NKT cells and CD1d(+) cells amass in human omentum and are depleted in patients with cancer and obesity. *European Journal of Immunology* 39, 1893–1901.
- 307. Dhodapkar, M. V, Geller, M. D, Chang, D. H, Shimizu, K, Fujii, S.-I, Dhodapkar, K. M, & Krasovsky, J. (2003) A reversible defect in natural killer T cell function characterizes the progression of premalignant to malignant multiple myeloma. *Journal of Experimental Medicine* **197**, 1667–1676.
- 308. Molling, J. W, Langius, J. A, Langendijk, J. A, Leemans, C. R, Bontkes, H. J, Van Der Vliet, H. J, Von Blomberg, B. M. E, Scheper, R. J, & Van Den Eertwegh, A. J. (2007) Low Levels of Circulating Invariant Natural Killer T Cells Predict Poor Clinical Outcome in Patients With Head and Neck Squamous Cell Carcinoma. *Journal* of Clinical Oncology 25, 862–868.
- 309. Haraguchi, K, Takahashi, T, Nakahara, F, Matsumoto, A, Kurokawa, M, Ogawa, S, Oda, H, Hirai, H, & Chiba, S. (2006) CD1d expression level in tumor cells is an important determinant for anti-tumor immunity by natural killer T cells. *Leukemia & Lymphoma* 47, 2218–2223.
- 310. Bagnara, D, Ibatici, A, Corselli, M, Sessarego, N, Tenca, C, De Santanna, A, Mazzarello, A, Daga, A, Corvo, R, De Rossi, G, Frassoni, F, Ciccone, E, & Fais, F. (2009) Adoptive immunotherapy mediated by ex vivo expanded natural killer T cells against CD1d-expressing lymphoid neoplasms. *Haematologica* **94**, 967–974.
- 311. Hermans, I. F, Silk, J. D, Gileadi, U, Masri, S. H, Shepherd, D, Farrand, K. J, Salio, M, & Cerundolo, V. (2007) Dendritic cell function can be modulated through cooperative actions of TLR ligands and invariant NKT cells. *Journal of Immunology* 178, 2721–2729.

- 312. Pilones, K. A, Kawashima, N, Yang, A. M, Babb, J. S, Formenti, S. C, & Demaria, S. (2009) Invariant natural killer T cells regulate breast cancer response to radiation and CTLA-4 blockade. *Clinical Cancer Research* 15, 597–606.
- Renukaradhya, G. J, Sriram, V, Du, W, Gervay-Hague, J, Van Kaer, L, & Brutkiewicz, R. R. (2006) Inhibition of antitumor immunity by invariant natural killer T cells in a T-cell lymphoma model in vivo. *International Journal of Cancer* **118**, 3045–3053.
- Berzofsky, J. A & Terabe, M. (2008) A novel immunoregulatory axis of NKT cell subsets regulating tumor immunity. *Cancer Immunology, Immunotherapy* 57, 1679– 1683.
- 315. Ambrosino, E, Terabe, M, Halder, R. C, Peng, J, Takaku, S, Miyake, S, Yamamura, T, Kumar, V, & Berzofsky, J. A. (2007) Cross-regulation between type I and type II NKT cells in regulating tumor immunity: a new immunoregulatory axis. *Journal of Immunology* **179**, 5126–5136.
- 316. Berzofsky, J. A & Terabe, M. (2009) The contrasting roles of NKT cells in tumor immunity. *Current Molecular Medicine* **9**, 667–672.
- 317. Hix, L. M, Shi, Y. H, Brutkiewicz, R. R, Stein, P. L, Wang, C.-R, & Zhang, M. (2011) CD1d-Expressing Breast Cancer Cells Modulate NKT Cell-Mediated Antitumor Immunity in a Murine Model of Breast Cancer Metastasis. *PLoS One* 6, e20702.
- 318. Ben-Bassat, I, Raanani, P, & Gale, R. P. (2007) Graft-versus-leukemia in chronic lymphocytic leukemia. *Bone Marrow Transplantation* **39**, 441–446.
- 319. Pavletic, S. Z, Zhou, G, Sobocinski, K, Marti, G, Doney, K, DiPersio, J, Feremans, W, Foroni, L, Goodman, S, Prentice, G, LeMaistre, C, Bandini, G, Ferrant, A, Jacobsen, N, Khouri, I, Gale, R. P, Wiestner, A, Giralt, S, Montserrat, E, Chan, W. C, Bredeson, C, Center for International Blood and Marrow Transplant Research (CIB-MTR), & Medical College of Wisconsin, Milwaukee. (2007) Genetically identical twin transplantation for chronic lymphocytic leukemia. *Leukemia* 21, 2452–2455.
- 320. Del Giudice, I, Chiaretti, S, Tavolaro, S, De Propris, M. S, Maggio, R, Mancini, F, Peragine, N, Santangelo, S, Marinelli, M, Mauro, F. R, Guarini, A, & Foa, R. (2009) Spontaneous regression of chronic lymphocytic leukemia: clinical and biologic features of 9 cases. *Blood* **114**, 638–646.
- 321. Thomas, R, Ribeiro, I, Shepherd, P, Johnson, P, Cook, M, Lakhani, A, Kaczmarski, R, Carrington, P, & Catovsky, D. (2002) Spontaneous clinical regression in chronic lymphocytic leukaemia. *British Journal of Haematology* **116**, 341–345.
- 322. Wajima, T. (2000) Spontaneous regression of chronic lymphocytic leukemia and simultaneous development of autoimmune hemolytic anemia and autoimmune thrombocytopenia. *American Journal of Hematology* **65**, 88–89.
- 323. Gitelson, E, Hammond, C, Mena, J, Lorenzo, M, Buckstein, R, Berinstein, N. L, Imrie, K, & Spaner, D. E. (2003) Chronic lymphocytic leukemia-reactive T cells during disease progression and after autologous tumor cell vaccines. *Clinical Cancer Research* 9, 1656–1665.
- 324. Wierda, W. G, Cantwell, M. J, Woods, S. J, Rassenti, L. Z, Prussak, C. E, & Kipps,

T. J. (2000) CD40-ligand (CD154) gene therapy for chronic lymphocytic leukemia. *Blood* **96**, 2917–2924.

- 325. Hus, I, Roliński, J, Tabarkiewicz, J, Wojas, K, Bojarska-Junak, A, Greiner, J, Giannopoulos, K, Dmoszyńska, A, & Schmitt, M. (2005) Allogeneic dendritic cells pulsed with tumor lysates or apoptotic bodies as immunotherapy for patients with earlystage B-cell chronic lymphocytic leukemia. *Leukemia* **19**, 1621–1627.
- 326. Ianni, M. D, Moretti, L, Terenzi, A, Bazzucchi, F, Papa, B. D, Bazzucchi, M, Ciurnelli, R, Lucchesi, A, Sportoletti, P, Rosati, E, Marconi, P. F, Falzetti, F, & Tabilio, A. (2009) Activated autologous T cells exert an anti-B-cell chronic lymphatic leukemia effect in vitro and in vivo. *Cytotherapy* 11, 86–96.
- 327. Nagorsen, D & Baeuerle, P. A. (2011) Immunomodulatory therapy of cancer with T cell-engaging BiTE antibody blinatumomab. *Experimental Cell Research* pp. 1–6.
- 328. Porter, D. L, Levine, B. L, Kalos, M, Bagg, A, & June, C. H. (2011) Chimeric Antigen Receptor-Modified T Cells in Chronic Lymphoid Leukemia. *New England Journal of Medicine* 365, 725–733.
- 329. Gabrilovich, D. I & Nagaraj, S. (2009) Myeloid-derived suppressor cells as regulators of the immune system. *Nature Reviews Immunology* **9**, 162–174.
- 330. Nishio, M, Endo, T, Tsukada, N, Ohata, J, Kitada, S, Reed, J. C, Zvaifler, N. J, & Kipps, T. J. (2005) Nurselike cells express BAFF and APRIL, which can promote survival of chronic lymphocytic leukemia cells via a paracrine pathway distinct from that of SDF-1alpha. *Blood* **106**, 1012–1020.
- 331. Seiffert, M, Schulz, A, Ohl, S, Döhner, H, Stilgenbauer, S, & Lichter, P. (2010) Soluble CD14 is a novel monocyte-derived survival factor for chronic lymphocytic leukemia cells, which is induced by CLL cells in vitro and present at abnormally high levels in vivo. *Blood* **116**, 4223–4230.
- 332. D'Arena, G, Laurenti, L, Minervini, M. M, Deaglio, S, Bonello, L, De Martino, L, De Padua, L, Savino, L, Tarnani, M, De Feo, V, & Cascavilla, N. (2011) Regulatory T-cell number is increased in chronic lymphocytic leukemia patients and correlates with progressive disease. *Leukemia Research* **35**, 363–368.
- 333. Vuillier, F, Maloum, K, Thomas, E. K, Jouanne, C, Dighiero, G, & Scott-Algara, D. (2001) Functional monocyte-derived dendritic cells can be generated in chronic lymphocytic leukaemia. *British Journal of Haematology* **115**, 831–844.
- 334. Riches, J. C, Ramsay, A. G, & Gribben, J. G. (2010) T-cell function in chronic lymphocytic leukaemia. *Seminars in Cancer Biology* **20**, 431–438.
- 335. Kremer, J. P, Reisbach, G, Nerl, C, & Dörmer, P. (1992) B-cell chronic lymphocytic leukaemia cells express and release transforming growth factor-beta. *British Journal of Haematology* **80**, 480–487.
- 336. Menier, C, Rouas-Freiss, N, Favier, B, LeMaoult, J, Moreau, P, & Carosella, E. D. (2010) Recent advances on the non-classical major histocompatibility complex class I HLA-G molecule. *Tissue A* **75**, 201–206.

- 337. Maki, G, Hayes, G. M, Naji, A, Tyler, T, Carosella, E. D, Rouas-Freiss, N, & Gregory, S. A. (2008) NK resistance of tumor cells from multiple myeloma and chronic lymphocytic leukemia patients: implication of HLA-G. *Leukemia* 22, 998–1006.
- 338. Nückel, H, Switala, M, Sellmann, L, Horn, P. A, Dürig, J, Dührsen, U, Küppers, R, Grosse-Wilde, H, & Rebmann, V. (2010) The prognostic significance of soluble NKG2D ligands in B-cell chronic lymphocytic leukemia. *Leukemia* 24, 1152–1159.
- 339. Naylor, K, Li, G, Vallejo, A. N, Lee, W.-W, Koetz, K, Bryl, E, Witkowski, J, Fulbright, J, Weyand, C. M, & Goronzy, J. J. (2005) The influence of age on T cell generation and TCR diversity. *Journal of Immunology* **174**, 7446–7452.
- 340. Rezvany, M.-R. (2003) Leukemia-associated monoclonal and oligoclonal TCR-BV use in patients with B-cell chronic lymphocytic leukemia. *Blood* **101**, 1063–1070.
- 341. Rezvany, M. R, Tehrani, M. J, Karlsson, C, Lundin, J, Rabbani, H, Österborg, A, & Mellstedt, H. (2006) Reconstitution of the T-cell repertoire following treatment with alemtuzumab (anti-CD52 monoclonal antibody) in patients with B-cell chronic lymphocytic leukaemia. *British Journal of Haematology* **135**, 475–485.
- 342. Dreger, P, Viehmann, K, von Neuhoff, N, Krüss, D, Glass, B, Kneba, M, Mitsky, P, Jopp, P, Rautenberg, P, Mills, B, & Schmitz, N. (2000) A prospective study of positive/negative ex vivo B-cell depletion in patients with chronic lymphocytic leukemia. *Experimental Hematology* **28**, 1187–1196.
- 343. Corbière, V, Chapiro, J, Stroobant, V, Ma, W, Lurquin, C, Lethé, B, van Baren, N, Van den Eynde, B. J, Boon, T, & Coulie, P. G. (2011) Antigen spreading contributes to MAGE vaccination-induced regression of melanoma metastases. *Cancer Research* 71, 1253–1262.
- 344. Tabarkiewicz, J & Giannopoulos, K. (2010) Definition of a target for immunotherapy and results of the first Peptide vaccination study in chronic lymphocytic leukemia. *Transplantation Proceedings* **42**, 3293–3296.
- 345. Zirlik, K. M, Zahrieh, D, Neuberg, D, & Gribben, J. G. (2006) Cytotoxic T cells generated against heteroclitic peptides kill primary tumor cells independent of the binding affinity of the native tumor antigen peptide. *Blood* **108**, 3865–3870.
- 346. Giannopoulos, K, Li, L, Bojarska-Junak, A, Rolinski, J, Dmoszynska, A, Hus, I, Greiner, J, Renner, C, Döhner, H, & Schmitt, M. (2006) Expression of RHAMM/CD168 and other tumor-associated antigens in patients with B-cell chronic lymphocytic leukemia. *International Journal of Oncology* 29, 95–103.
- 347. Mayr, C, Bund, D, Schlee, M, Moosmann, A, Kofler, D. M, Hallek, M, & Wendtner, C.-M. (2005) Fibromodulin as a novel tumor-associated antigen (TAA) in chronic lymphocytic leukemia (CLL), which allows expansion of specific CD8+ autologous T lymphocytes. *Blood* **105**, 1566–1573.
- 348. Rezvany, M.-R, Jeddi-Tehrani, M, Rabbani, H, Ruden, U, Hammarstrom, L, Osterborg, A, Wigzell, H, & Mellstedt, H. (2000) Autologous T lymphocytes recognize the tumour-derived immunoglobulin VH-CDR3 region in patients with B-cell chronic lymphocytic leukaemia. *British Journal of Haematology* **111**, 230–238.

- 349. Trojan, A, Schultze, J. L, Witzens, M, Vonderheide, R. H, Ladetto, M, Donovan, J. W, & Gribben, J. G. (2000) Immunoglobulin framework-derived peptides function as cytotoxic T-cell epitopes commonly expressed in B-cell malignancies. *Nature Medicine* 6, 667–672.
- 350. Kokhaei, P, Palma, M, Hansson, L, Osterborg, A, Mellstedt, H, & Choudhury, A. (2007) Telomerase (hTERT 611–626) serves as a tumor antigen in B-cell chronic lymphocytic leukemia and generates spontaneously antileukemic, cytotoxic T cells. *Experimental Hematology* **35**, 297–304.
- 351. Fukuda, T, Chen, L, Endo, T, Tang, L, Lu, D, Castro, J. E, Widhopf, G. F, Rassenti, L. Z, Cantwell, M. J, Prussak, C. E, Carson, D. A, & Kipps, T. J. (2008) Antisera induced by infusions of autologous Ad-CD154-leukemia B cells identify ROR1 as an oncofetal antigen and receptor for Wnt5a. *Proceedings of the National Academy of Sciences of the United States of America* **105**, 3047–3052.
- 352. Daneshmanesh, A. H, Mikaelsson, E, Jeddi-Tehrani, M, Bayat, A. A, Ghods, R, Ostadkarampour, M, Akhondi, M, Lagercrantz, S, Larsson, C, Osterborg, A, Shokri, F, Mellstedt, H, & Rabbani, H. (2008) Ror1, a cell surface receptor tyrosine kinase is expressed in chronic lymphocytic leukemia and may serve as a putative target for therapy. *International Journal of Cancer* **123**, 1190–1195.
- 353. Bund, D, Mayr, C, Kofler, D. M, Hallek, M, & Wendtner, C.-M. (2006) Human Ly9 (CD229) as novel tumor-associated antigen (TAA) in chronic lymphocytic leukemia (B-CLL) recognized by autologous CD8+ T cells. *Experimental Hematology* 34, 860–869.
- 354. Mayr, C, Bund, D, Schlee, M, Bamberger, M, Kofler, D. M, Hallek, M, & Wendtner, C.-M. (2006) MDM2 is recognized as a tumor-associated antigen in chronic lymphocytic leukemia by CD8+ autologous T lymphocytes. *Experimental Hematology* **34**, 44–53.
- 355. Bund, D, Mayr, C, Kofler, D. M, Hallek, M, & Wendtner, C.-M. (2007) CD23 is recognized as tumor-associated antigen (TAA) in B-CLL by CD8+ autologous T lymphocytes. *Experimental Hematology* **35**, 920–930.
- 356. Hus, I, Kawiak, J, Tabarkiewicz, J, Radej, S, Hoser, G, Bojarska-Junak, A, Schmitt, M, Giannopoulos, K, Dmoszynska, A, & Rolinski, J. (2008) Immunotherapy with irradiated autologous leukemic cells in patients with B-CLL in early stages. *Oncology Reports* **20**, 443–451.
- 357. Spaner, D. E, Miller, R. L, Mena, J, Grossman, L, Sorrenti, V, & Shi, Y. (2005) Regression of lymphomatous skin deposits in a chronic lymphocytic leukemia patient treated with the Toll-like receptor-7/8 agonist, imiquimod. *Leukemia & Lymphoma* 46, 935–939.
- 358. Dazzi, F, D'Andrea, E, Biasi, G, De Silvestro, G, Gaidano, G, Schena, M, Tison, T, Vianello, F, Girolami, A, & Caligaris-Cappio, F. (1995) Failure of B cells of chronic lymphocytic leukemia in presenting soluble and alloantigens. *Clinical Immunology and Immunopathology* **75**, 26–32.
- 359. Łuczyński, W, Stasiak-Barmuta, A, Piszcz, J, Iłendo, E, Kowalczuk, O, & Krawczuk-

Rybak, M. (2007) B-cell chronic lymphocytic leukemia-derived dendritic cells stimulate allogeneic T-cell response and express chemokines involved in T-cell migration. *Neoplasma* **54**, 527–535.

- 360. Buhmann, R, Nolte, A, Westhaus, D, Emmerich, B, & Hallek, M. (1999) CD40activated B-cell chronic lymphocytic leukemia cells for tumor immunotherapy: stimulation of allogeneic versus autologous T cells generates different types of effector cells. *Blood* **93**, 1992–2002.
- Krackhardt, A. M, Harig, S, Witzens, M, Broderick, R, Barrett, P, & Gribben, J. G. (2002) T-cell responses against chronic lymphocytic leukemia cells: implications for immunotherapy. *Blood* 100, 167–173.
- 362. Kato, K, Cantwell, M. J, Sharma, S, & Kipps, T. J. (1998) Gene transfer of CD40ligand induces autologous immune recognition of chronic lymphocytic leukemia B cells. *Journal of Clinical Investigation* **101**, 1133–1141.
- 363. Biagi, E, Dotti, G, Yvon, E, Lee, E, Pule, M, Vigouroux, S, Gottschalk, S, Popat, U, Rousseau, R, & Brenner, M. (2005) Molecular transfer of CD40 and OX40 ligands to leukemic human B cells induces expansion of autologous tumor-reactive cytotoxic T lymphocytes. *Blood* **105**, 2436–2442.
- 364. Kater, A. P, Evers, L. M, Remmerswaal, E. B, Jaspers, A, Oosterwijk, M. F, van Lier, R. A, van Oers, M. H, & Eldering, E. (2004) CD40 stimulation of B-cell chronic lymphocytic leukaemia cells enhances the anti-apoptotic profile, but also Bid expression and cells remain susceptible to autologous cytotoxic T-lymphocyte attack. *British Journal of Haematology* 127, 404–415.
- 365. Wierda, W. G, Castro, J. E, Aguillon, R, Sampath, D, Jalayer, A, McMannis, J, Prussak, C. E, Keating, M, & Kipps, T. J. (2010) A phase I study of immune gene therapy for patients with CLL using a membrane-stable, humanized CD154. *Leukemia* 24, 1893–1900.
- 366. Biagi, E, Rousseau, R, Yvon, E, Schwartz, M, Dotti, G, Foster, A, Havlik-Cooper, D, Grilley, B, Gee, A, Baker, K, Carrum, G, Rice, L, Andreeff, M, Popat, U, & Brenner, M. (2005) Responses to human CD40 ligand/human interleukin-2 autologous cell vaccine in patients with B-cell chronic lymphocytic leukemia. *Clinical Cancer Research* 11, 6916–6923.
- 367. Litzinger, M. T, Foon, K. A, Tsang, K.-Y, Schlom, J, & Palena, C. (2010) Comparative analysis of MVA-CD40L and MVA-TRICOM vectors for enhancing the immunogenicity of chronic lymphocytic leukemia (CLL) cells. *Leukemia Research* **34**, 1351–1357.
- 368. Spaner, D. E, Shi, Y, White, D, Shaha, S, He, L, Masellis, A, Wong, K, & Gorczynski, R. (2010) A phase I/II trial of TLR-7 agonist immunotherapy in chronic lymphocytic leukemia. *Leukemia* 24, 222–226.
- 369. Spaner, D. E & Masellis, A. (2007) Toll-like receptor agonists in the treatment of chronic lymphocytic leukemia. *Leukemia* **21**, 53–60.
- Muzio, M, Scielzo, C, Bertilaccio, M. T. S, Frenquelli, M, Ghia, P, & Caligaris-Cappio,
 F. (2009) Expression and function of toll like receptors in chronic lymphocytic leukaemia cells. *British Journal of Haematology* 144, 507–516.

- 371. Rožková, D, Novotná, L, Pytlík, R, Hochová, I, Kozák, T, Bartůňková, J, & Spíšek, R. (2009) Toll-like receptors on B-CLL cells: Expression and functional consequences of their stimulation. *International Journal of Cancer* **126**, 1132–1143.
- 372. Dubovsky, J. A, Wang, D, Powers, J. J, Berchmans, E, Smith, M. A, Wright, K. L, Sotomayor, E. M, & Pinilla-Ibarz, J. A. (2011) Restoring the functional immunogenicity of chronic lymphocytic leukemia using epigenetic modifiers. *Leukemia Research* 35, 394–404.
- 373. Kronenberger, K, Nössner, E, Frankenberger, B, Wahl, U, Dreyling, M, Hallek, M, & Mocikat, R. (2008) A polyvalent cellular vaccine induces T-cell responses against specific self-antigens overexpressed in chronic lymphocytic B-cell leukemia. *Journal of Immunotherapy* **31**, 723–730.
- 374. Goddard, R. V, Prentice, A. G, Copplestone, J. A, & Kaminski, E. R. (2003) In vitro dendritic cell-induced T cell responses to B cell chronic lymphocytic leukaemia enhanced by IL-15 and dendritic cell-B-CLL electrofusion hybrids. *Clinical and Experimental Immunology* **131**, 82–89.
- 375. Hus, I, Schmitt, M, Tabarkiewicz, J, Radej, S, Wojas, K, Bojarska-Junak, A, Schmitt, A, Giannopoulos, K, Dmoszyńska, A, & Roliński, J. (2008) Vaccination of B-CLL patients with autologous dendritic cells can change the frequency of leukemia antigenspecific CD8+ T cells as well as CD4+CD25+FoxP3+ regulatory T cells toward an antileukemia response. *Leukemia* **22**, 1007–1017.
- 376. Ritchie, D, Hermans, I, Yang, J, Walton, J, Matthews, K, Carter, J, Findlay, M, Dady, P, Rawson, P, & Ronchese, F. (2006) Autologous dendritic cells pulsed with eluted peptide as immunotherapy for advanced B-cell malignancies. *Leukemia & Lymphoma* 47, 675–682.
- 377. Kokhaei, P, Choudhury, A, Mahdian, R, Lundin, J, Moshfegh, A, Österborg, A, & Mellstedt, H. (2004) Apoptotic tumor cells are superior to tumor cell lysate, and tumor cell RNA in induction of autologous T cell response in B-CLL. *Leukemia* **18**, 1810–1815.
- 378. Kokhaei, P, Rezvany, M.-R, Virving, L, Choudhury, A, Rabbani, H, Osterborg, A, & Mellstedt, H. (2003) Dendritic cells loaded with apoptotic tumour cells induce a stronger T-cell response than dendritic cell-tumour hybrids in B-CLL. *Leukemia* **17**, 894–899.
- 379. Palma, M, Adamson, L, Hansson, L, Kokhaei, P, Rezvany, R, Mellstedt, H, Osterborg, A, & Choudhury, A. (2008) Development of a dendritic cell-based vaccine for chronic lymphocytic leukemia. *Cancer Immunology, Immunotherapy* 57, 1705–1710.
- 380. Adamson, L, Palma, M, Choudhury, A, Eriksson, I, Näsman-Glaser, B, Hansson, M, Hansson, L, Kokhaei, P, Osterborg, A, & Mellstedt, H. (2009) Generation of a dendritic cell-based vaccine in chronic lymphocytic leukaemia using CliniMACS platform for large-scale production. *Scandinavian Journal of Immunology* **69**, 529–536.
- 381. Castro, J. E, Wierda, W. G, Kipps, T. J, Keating, M. J, Bole, J, Anderson, B, Meyer, J, Bonyhadi, M, Berenson, R. J, & Frohlich, M. W. (2004) A Phase I/II Trial of

Xcellerated T Cells[™] in Patients with Chronic Lymphocytic Leukemia. *Blood (ASH Annual Meeting Abstracts)* **104**, 2508.

- 382. Porter, D. L, Levine, B. L, Bunin, N, Stadtmauer, E. A, Luger, S. M, Goldstein, S, Loren, A, Phillips, J, Nasta, S, Perl, A, Schuster, S, Tsai, D, Sohal, A, Veloso, E, Emerson, S, & June, C. H. (2006) A phase 1 trial of donor lymphocyte infusions expanded and activated ex vivo via CD3/CD28 costimulation. *Blood* 107, 1325–1331.
- 383. Abrahamsen, I. W, Stronen, E, Ichli, S. W. a, Johansen, J. N, Kjellevoll, S, Kumari, S, Komada, M, Gaudernack, G, Tjonnfjord, G, Toebes, M, Schumacher, T. N, Lund-Johansen, F, & Olweus, J. (2010) Targeting B cell leukemia with highly specific allogeneic T cells with a public recognition motif. *Leukemia* 24, 1901–1909.
- 384. Xue, S.-A, Gao, L, Hart, D, Gillmore, R, Qasim, W, Thrasher, A, Apperley, J, Engels, B, Uckert, W, Morris, E, & Stauss, H. (2005) Elimination of human leukemia cells in NOD/SCID mice by WT1-TCR gene-transduced human T cells. *Blood* 106, 3062– 3067.
- 385. Robbins, P. F, Morgan, R. A, Feldman, S. A, Yang, J. C, Sherry, R. M, Dudley, M. E, Wunderlich, J. R, Nahvi, A. V, Helman, L. J, Mackall, C. L, Kammula, U. S, Hughes, M. S, Restifo, N. P, Raffeld, M, Lee, C. C, Levy, C. L, Li, Y. F, El-Gamil, M, Schwarz, S. L, Laurencot, C, & Rosenberg, S. A. (2011) Tumor regression in patients with metastatic synovial cell sarcoma and melanoma using genetically engineered lymphocytes reactive with NY-ESO-1. *Journal of Clinical Oncology* **29**, 917–24.
- 386. Morgan, R. A, Dudley, M. E, Wunderlich, J. R, Hughes, M. S, Yang, J. C, Sherry, R. M, Royal, R. E, Topalian, S. L, Kammula, U. S, Restifo, N. P, Zheng, Z, Nahvi, A, de Vries, C. R, Rogers-Freezer, L. J, Mavroukakis, S. A, & Rosenberg, S. A. (2006) Cancer regression in patients after transfer of genetically engineered lymphocytes. *Science* **314**, 126–129.
- 387. Schumacher, T. N. M. (2002) T-cell-receptor gene therapy. *Nature Reviews Immunology* **2**, 512–519.
- 388. Foster, A. E., Brenner, M. K., & Dotti, G. (2008) Adoptive T-cell immunotherapy of chronic lymphocytic leukaemia. *Best Practice & Research Clinical Haematology* 21, 375–389.
- 389. Hudecek, M, Schmitt, T. M, Baskar, S, Lupo-Stanghellini, M. T, Nishida, T, Yamamoto, T. N, Bleakley, M, Turtle, C. J, Chang, W.-C, Greisman, H. A, Wood, B, Maloney, D. G, Jensen, M. C, Rader, C, & Riddell, S. R. (2010) The B-cell tumor-associated antigen ROR1 can be targeted with T cells modified to express a ROR1-specific chimeric antigen receptor. *Blood* **116**, 4532–4541.
- 390. Kochenderfer, J. N, Feldman, S. A, Zhao, Y, Xu, H, Black, M. A, Morgan, R. A, Wilson, W. H, & Rosenberg, S. A. (2009) Construction and preclinical evaluation of an anti-CD19 chimeric antigen receptor. *Journal of Immunotherapy* **32**, 689–702.
- 391. Giordano Attianese, G. M, Marin, V, Hoyos, V, Savoldo, B, Pizzitola, I, Tettamanti, S, Agostoni, V, Parma, M, Ponzoni, M, Bertilaccio, M. T. S, Ghia, P, Biondi, A, Dotti, G, & Biagi, E. (2011) In vitro and in vivo model of a novel immunotherapy approach

for chronic lymphocytic leukemia by anti-CD23 chimeric antigen receptor. *Blood* **117**, 4736–4745.

- 392. Jensen, M. C, Popplewell, L, Cooper, L. J, DiGiusto, D, Kalos, M, Ostberg, J. R, & Forman, S. J. (2010) Antitransgene rejection responses contribute to attenuated persistence of adoptively transferred CD20/CD19-specific chimeric antigen receptor redirected T cells in humans. *Biology of Blood and Marrow Transplantation* **16**, 1245–1256.
- 393. Kochenderfer, J. N, Wilson, W. H, Janik, J. E, Dudley, M. E, Stetler-Stevenson, M, Feldman, S. A, Maric, I, Raffeld, M, Nathan, D. A, Lanier, B. J, Morgan, R. A, & Rosenberg, S. A. (2010) Eradication of B-lineage cells and regression of lymphoma in a patient treated with autologous T cells genetically engineered to recognize CD19. *Blood* **116**, 4099–4102.
- 394. Ly, L. V, Sluijter, M, Versluis, M, Luyten, G. P. M, van der Burg, S. H, Melief, C. J. M, Jager, M. J, & van Hall, T. (2010) Peptide vaccination after T-cell transfer causes massive clonal expansion, tumor eradication, and manageable cytokine storm. *Cancer Research* **70**, 8339–8346.
- 395. Brentjens, R, Yeh, R, Bernal, Y, Riviere, I, & Sadelain, M. (2009) Treatment of Chronic Lymphocytic Leukemia With Genetically Targeted Autologous T Cells: Case Report of an Unforeseen Adverse Event in a Phase I Clinical Trial. *Molecular Therapy* 18, 666–668.
- 396. Howe, S. J, Mansour, M. R, Schwarzwaelder, K, Bartholomae, C, Hubank, M, Kempski, H, Brugman, M. H, Pike-Overzet, K, Chatters, S. J, de Ridder, D, Gilmour, K. C, Adams, S, Thornhill, S. I, Parsley, K. L, Staal, F. J. T, Gale, R. E, Linch, D. C, Bayford, J, Brown, L, Quaye, M, Kinnon, C, Ancliff, P, Webb, D. K, Schmidt, M, von Kalle, C, Gaspar, H. B, & Thrasher, A. J. (2008) Insertional mutagenesis combined with acquired somatic mutations causes leukemogenesis following gene therapy of SCID-X1 patients. *Journal of Clinical Investigation* **118**, 3143–3150.
- 397. Almåsbak, H, Rian, E, Hoel, H. J, Pule, M, Wälchli, S, Kvalheim, G, Gaudernack, G, & Rasmussen, A.-M. (2011) Transiently redirected T cells for adoptive transfer. *Cytotherapy* 13, 629–640.
- 398. Amos, S. M, Duong, C. P. M, Westwood, J. A, Ritchie, D. S, Junghans, R. P, Darcy, P. K, & Kershaw, M. H. (2011) Autoimmunity associated with immunotherapy of cancer. *Blood* **118**, 499–509.
- 399. Johnson, L. A, Morgan, R. A, Dudley, M. E, Cassard, L, Yang, J. C, Hughes, M. S, Kammula, U. S, Royal, R. E, Sherry, R. M, Wunderlich, J. R, Lee, C. C. R, Restifo, N. P, Schwarz, S. L, Cogdill, A. P, Bishop, R. J, Kim, H, Brewer, C. C, Rudy, S. F, VanWaes, C, Davis, J. L, Mathur, A, Ripley, R. T, Nathan, D. A, Laurencot, C. M, & Rosenberg, S. A. (2009) Gene therapy with human and mouse T-cell receptors mediates cancer regression and targets normal tissues expressing cognate antigen. *Blood* **114**, 535–546.
- 400. Bargou, R, Leo, E, Zugmaier, G, Klinger, M, Goebeler, M, Knop, S, Noppeney, R, Viardot, A, Hess, G, Schuler, M, Einsele, H, Brandl, C, Wolf, A, Kirchinger, P, Klappers, P, Schmidt, M, Riethmüller, G, Reinhardt, C, Baeuerle, P. A, & Kufer, P. (2008)

Tumor regression in cancer patients by very low doses of a T cell-engaging antibody. *Science* **321**, 974–977.

- 401. Mous, R, Savage, P, Remmerswaal, E. B. M, van Lier, R. A. W, Eldering, E, & van Oers, M. H. J. (2006) Redirection of CMV-specific CTL towards B-CLL via CD20-targeted HLA/CMV complexes. *Leukemia* 20, 1096–1102.
- 402. Moreno, M, Molling, J. W, von Mensdorff-Pouilly, S, Verheijen, R. H. M, von Blomberg, B. M. E, van den Eertwegh, A. J. M, Scheper, R. J, & Bontkes, H. J. (2008) In vitro expanded human invariant natural killer T-cells promote functional activity of natural killer cells. *Clinical Immunology* **129**, 145–154.
- 403. Morita, M, Motoki, K, Akimoto, K, Natori, T, Sakai, T, Sawa, E, Yamaji, K, Koezuka, Y, Kobayashi, E, & Fukushima, H. (1995) Structure-activity relationship of alphagalactosylceramides against B16-bearing mice. *Journal of Medicinal Chemistry* **38**, 2176–2187.
- 404. Woltman, A. M, Ter Borg, M. J, Binda, R. S, Sprengers, D, von Blomberg, B. M, Scheper, R. J, Hayashi, K, Nishi, N, Boonstra, A, van der Molen, R, & Janssen, H. L. (2009) Alpha-galactosylceramide in chronic hepatitis B infection: results from a randomized placebo-controlled Phase I/II trial. *Antiviral Therapy* 14, 809–818.
- 405. Parekh, V. V, Wilson, M. T, Olivares-Villagómez, D, Singh, A. K, Wu, L, Wang, C.-R, Joyce, S, & Van Kaer, L. (2005) Glycolipid antigen induces long-term natural killer T cell anergy in mice. *Journal of Clinical Investigation* **115**, 2572–2583.
- 406. Burdin, N, Brossay, L, & Kronenberg, M. (1999) Immunization with alphagalactosylceramide polarizes CD1-reactive NK T cells towards Th2 cytokine synthesis. *European Journal of Immunology* 29, 2014–2025.
- 407. Fujii, S.-I, Shimizu, K, Kronenberg, M, & Steinman, R. M. (2002) Prolonged IFNgamma-producing NKT response induced with alpha-galactosylceramide-loaded DCs. *Nature Immunology* **3**, 867–874.
- 408. Toura, I, Kawano, T, Akutsu, Y, Nakayama, T, Ochiai, T, & Taniguchi, M. (1999) Cutting edge: inhibition of experimental tumor metastasis by dendritic cells pulsed with alpha-galactosylceramide. *Journal of Immunology* **163**, 2387–2391.
- 409. Kunii, N, Horiguchi, S, Motohashi, S, Yamamoto, H, Ueno, N, Yamamoto, S, Sakurai, D, Taniguchi, M, Nakayama, T, & Okamoto, Y. (2009) Combination therapy of in vitro-expanded natural killer T cells and alpha-galactosylceramide-pulsed antigen-presenting cells in patients with recurrent head and neck carcinoma. *Cancer Science* 100, 1092–1098.
- 410. Ishikawa, A, Motohashi, S, Ishikawa, E, Fuchida, H, Higashino, K, Otsuji, M, Iizasa, T, Nakayama, T, Taniguchi, M, & Fujisawa, T. (2005) A phase I study of alphagalactosylceramide (KRN7000)-pulsed dendritic cells in patients with advanced and recurrent non-small cell lung cancer. *Clinical Cancer Research* **11**, 1910–1917.
- 411. Motohashi, S, Nagato, K, Kunii, N, Yamamoto, H, Yamasaki, K, Okita, K, Hanaoka, H, Shimizu, N, Suzuki, M, Yoshino, I, Taniguchi, M, Fujisawa, T, & Nakayama, T. (2009) A phase I-II study of alpha-galactosylceramide-pulsed IL-2/GM-CSF-cultured

peripheral blood mononuclear cells in patients with advanced and recurrent nonsmall cell lung cancer. *Journal of Immunology* **182**, 2492–2501.

- 412. Nieda, M, Okai, M, Tazbirkova, A, Lin, H, Yamaura, A, Ide, K, Abraham, R, Juji, T, Macfarlane, D. J, & Nicol, A. J. (2004) Therapeutic activation of Valpha24+Vbeta11+ NKT cells in human subjects results in highly coordinated secondary activation of acquired and innate immunity. *Blood* **103**, 383–389.
- 413. Chang, D. H, Osman, K, Connolly, J, Kukreja, A, Krasovsky, J, Pack, M, Hutchinson, A, Geller, M, Liu, N, Annable, R, Shay, J, Kirchhoff, K, Nishi, N, Ando, Y, Hayashi, K, Hassoun, H, Steinman, R. M, & Dhodapkar, M. V. (2005) Sustained expansion of NKT cells and antigen-specific T cells after injection of alpha-galactosyl-ceramide loaded mature dendritic cells in cancer patients. *Journal of Experimental Medicine* **201**, 1503–1517.
- 414. Uchida, T, Horiguchi, S, Tanaka, Y, Yamamoto, H, Kunii, N, Motohashi, S, Taniguchi, M, Nakayama, T, & Okamoto, Y. (2007) Phase I study of α -galactosylceramidepulsed antigen presenting cells administration to the nasal submucosa in unresectable or recurrent head and neck cancer. *Cancer Immunology, Immunotherapy* **57**, 337–345.
- 415. Richter, J, Neparidze, N, Zhang, L, Nair, S, Monesmith, T, Sundaram, R, Miesowicz, F, Dhodapkar, K. M, & Dhodapkar, M. V. (2012) Clinical regressions and broad immune activation following combination therapy targeting human NKT cells in myeloma. *Blood* Oct 24 [Epub ahead of print].
- 416. Takahashi, T, Haraguchi, K, Chiba, S, Yasukawa, M, Shibata, Y, & Hirai, H. (2003) Valpha24+ natural killer T-cell responses against T-acute lymphoblastic leukaemia cells: implications for immunotherapy. *British Journal of Haematology* 122, 231– 239.
- 417. Fais, F, Morabito, F, Stelitano, C, Callea, V, Zanardi, S, Scudeletti, M, Varese, P, Ciccone, E, & Grossi, C. E. (2004) CD1d is expressed on B-chronic lymphocytic leukemia cells and mediates alpha-galactosylceramide presentation to natural killer T lymphocytes. *International Journal of Cancer* **109**, 402–411.
- 418. Motohashi, S, Ishikawa, A, Ishikawa, E, Otsuji, M, Iizasa, T, Hanaoka, H, Shimizu, N, Horiguchi, S, Okamoto, Y, Fujii, S.-I, Taniguchi, M, Fujisawa, T, & Nakayama, T. (2006) A phase I study of in vitro expanded natural killer T cells in patients with advanced and recurrent non-small cell lung cancer. *Clinical Cancer Research* **12**, 6079–6086.
- 419. Stober, D, Jomantaite, I, Schirmbeck, R, & Reimann, J. (2003) NKT cells provide help for dendritic cell-dependent priming of MHC class I-restricted CD8+ T cells in vivo. *Journal of Immunology* **170**, 2540–2548.
- 420. Petersen, T. R, Sika-Paotonu, D, Knight, D. A, Dickgreber, N, Farrand, K. J, Ronchese, F, & Hermans, I. F. (2010) Potent anti-tumor responses to immunization with dendritic cells loaded with tumor tissue and an NKT cell ligand. *Immunology and Cell Biology* **88**, 596–604.
- 421. Chung, Y, Kim, B.-S, Kim, Y.-J, Ko, H.-J, Ko, S.-Y, Kim, D.-H, & Kang, C.-Y. (2006)

CD1d-restricted T cells license B cells to generate long-lasting cytotoxic antitumor immunity in vivo. *Cancer Research* **66**, 6843–6850.

- 422. Shimizu, K, Kurosawa, Y, Taniguchi, M, Steinman, R. M, & Fujii, S.-I. (2007) Crosspresentation of glycolipid from tumor cells loaded with alpha-galactosylceramide leads to potent and long-lived T cell mediated immunity via dendritic cells. *Journal of Experimental Medicine* **204**, 2641–2653.
- 423. Shimizu, K, Goto, A, Fukui, M, Taniguchi, M, & Fujii, S.-I. (2007) Tumor cells loaded with alpha-galactosylceramide induce innate NKT and NK cell-dependent resistance to tumor implantation in mice. *Journal of Immunology* **178**, 2853–2861.
- 424. Chung, Y, Qin, H, Kang, C.-Y, Kim, S, Kwak, L. W, & Dong, C. (2007) An NKTmediated autologous vaccine generates CD4 T-cell dependent potent antilymphoma immunity. *Blood* **110**, 2013–2019.
- 425. Liu, K, Idoyaga, J, Charalambous, A, Fujii, S.-I, Bonito, A, Mordoh, J, Wainstok, R, Bai, X.-F, Liu, Y, & Steinman, R. M. (2005) Innate NKT lymphocytes confer superior adaptive immunity via tumor-capturing dendritic cells. *Journal of Experimental Medicine* **202**, 1507–1516.
- 426. Fujii, S. i, Goto, A, & Shimizu, K. (2009) Antigen mRNA-transfected, allogeneic fibroblasts loaded with NKT-cell ligand confer antitumor immunity. *Blood* 113, 4262–4272.
- 427. Gonzalez-Aseguinolaza, G, Van Kaer, L, Bergmann, C. C, Wilson, J. M, Schmieg, J, Kronenberg, M, Nakayama, T, Taniguchi, M, Koezuka, Y, & Tsuji, M. (2002) Natural killer T cell ligand alpha-galactosylceramide enhances protective immunity induced by malaria vaccines. *Journal of Experimental Medicine* **195**, 617–624.
- 428. Lindqvist, M, Persson, J, Thörn, K, & Harandi, A. M. (2009) The mucosal adjuvant effect of alpha-galactosylceramide for induction of protective immunity to sexually transmitted viral infection. *Journal of Immunology* **182**, 6435–6443.
- 429. Guillonneau, C, Mintern, J. D, Hubert, F.-X, Hurt, A. C, Besra, G. S, Porcelli, S, Barr, I. G, Doherty, P. C, Godfrey, D. I, & Turner, S. J. (2009) Combined NKT cell activation and influenza virus vaccination boosts memory CTL generation and protective immunity. *Proceedings of the National Academy of Sciences of the United States of America* 106, 3330–3335.
- 430. Huang, Y, Chen, A, Li, X, Chen, Z, Zhang, W, Song, Y, Gurner, D, Gardiner, D, Basu, S, Ho, D. D, & Tsuji, M. (2008) Enhancement of HIV DNA vaccine immunogenicity by the NKT cell ligand, alpha-galactosylceramide. *Vaccine* 26, 1807–1816.
- 431. Stirnemann, K, Romero, J. F, Baldi, L, Robert, B, Cesson, V, Besra, G. S, Zauderer, M, Wurm, F, Corradin, G, Mach, J.-P, MacDonald, H. R, & Donda, A. (2008) Sustained activation and tumor targeting of NKT cells using a CD1d-anti-HER2-scFv fusion protein induce antitumor effects in mice. *Journal of Clinical Investigation* **118**, 994–1005.
- 432. Kohrt, H. E, Pillai, A. B, Lowsky, R, & Strober, S. (2010) NKT cells, Treg, and their interactions in bone marrow transplantation. *European Journal of Immunology* **40**, 1862–1869.

- 433. Dellabona, P, Casorati, G, de Lalla, C, Montagna, D, & Locatelli, F. (2011) On the use of donor-derived iNKT cells for adoptive immunotherapy to prevent leukemia recurrence in pediatric recipients of HLA haploidentical HSCT for hematological malignancies. *Clinical Immunology* **140**, 152–159.
- 434. Salio, M, Speak, A. O, Shepherd, D, Polzella, P, Illarionov, P. A, Veerapen, N, Besra, G. S, Platt, F. M, & Cerundolo, V. (2007) Modulation of human natural killer T cell ligands on TLR-mediated antigen-presenting cell activation. *Proceedings of the National Academy of Sciences of the United States of America* 104, 20490–20495.
- 435. Durgan, K, Ali, M, Warner, P, & Latchman, Y. E. (2011) Targeting NKT cells and PD-L1 pathway results in augmented anti-tumor responses in a melanoma model. *Cancer Immunology, Immunotherapy* **60**, 547–558.
- 436. Storkus, W. J, Alexander, J, Payne, J. A, Dawson, J. R, & Cresswell, P. (1989) Reversal of natural killing susceptibility in target cells expressing transfected class I HLA genes. *Proceedings of the National Academy of Sciences of the United States of America* **86**, 2361–2364.
- 437. Cheson, B. D, Bennett, J. M, Grever, M, Kay, N, Keating, M. J, O'Brien, S, & Rai, K. R. (1996) National Cancer Institute-sponsored Working Group guidelines for chronic lymphocytic leukemia: revised guidelines for diagnosis and treatment. *Blood* **87**, 4990–4997.
- 438. Konjević, G, Jurisić, V, & Spuzić, I. (1997) Corrections to the original lactate dehydrogenase (LDH) release assay for the evaluation of NK cell cytotoxicity. *Journal of Immunological Methods* **200**, 199–201.
- 439. Hermans, I. F, Silk, J. D, Yang, J, Palmowski, M. J, Gileadi, U, McCarthy, C, Salio, M, Ronchese, F, & Cerundolo, V. (2004) The VITAL assay: a versatile fluorometric technique for assessing CTL- and NKT-mediated cytotoxicity against multiple targets in vitro and in vivo. *Journal of Immunological Methods* **285**, 25–40.
- 440. Scrivener, S, Goddard, R, Kaminski, E, & Prentice, A. (2003) Abnormal T-cell Function in B-cell Chronic Lymphocytic Leukaemia. *Leukemia & Lymphoma* **44**, 383–389.
- 441. Peralbo, E, DelaRosa, O, Gayoso, I, Pita, M. L, Tarazona, R, & Solana, R. (2006) Decreased frequency and proliferative response of invariant Valpha24Vbeta11 natural killer T (iNKT) cells in healthy elderly. *Biogerontology* **7**, 483–492.
- 442. Chan, A. C, Neeson, P, Leeansyah, E, Tainton, K, Quach, H, Prince, H. M, Godfrey, D. I, Ritchie, D, & Berzins, S. P. (2010) Testing the NKT cell hypothesis in lenalidomide-treated myelodysplastic syndrome patients. *Leukemia* **24**, 592–600.
- 443. Metelitsa, L. (2004) Flow cytometry for natural killer T cells: multi-parameter methods for multifunctional cells. *Clinical Immunology* **110**, 267–276.
- 444. Christopoulos, P, Pfeifer, D, Bartholome, K, Follo, M, Timmer, J, Fisch, P, & Veelken, H. (2011) Definition and characterization of the systemic T-cell dysregulation in untreated indolent B-cell lymphoma and very early CLL. *Blood* 117, 3836–3846.
- 445. Brigl, M & Brenner, M. B. (2004) CD1: antigen presentation and T cell function. *Annual Review of Immunology* **22**, 817–890.

- 446. Spanoudakis, E, Hu, M, Naresh, K, Terpos, E, Melo, V, Reid, A, Kotsianidis, I, Abdalla, S, Rahemtulla, A, & Karadimitris, A. (2009) Regulation of multiple myeloma survival and progression by CD1d. *Blood* **113**, 2498–2507.
- 447. Dzionek, A, Fuchs, A, Schmidt, P, Cremer, S, Zysk, M, Miltenyi, S, Buck, D. W, & Schmitz, J. (2000) BDCA-2, BDCA-3, and BDCA-4: three markers for distinct subsets of dendritic cells in human peripheral blood. *Journal of Immunology* **165**, 6037–6046.
- 448. Smed-Sorensen, A, Moll, M, Cheng, T.-Y, Lore, K, Norlin, A.-C, Perbeck, L, Moody, D. B, Spetz, A.-L, & Sandberg, J. K. (2008) IgG regulates the CD1 expression profile and lipid antigen-presenting function in human dendritic cells via Fc RIIa. *Blood* **111**, 5037–5046.
- 449. Flieger, D & Ziegler-Heitbrock, H. W. (1988) Abnormal blood monocytes in chronic lymphocytic leukemia. *Cancer Research* **48**, 4812–4816.
- 450. Ziegler-Heitbrock, H. W, Fingerle, G, Strobel, M, Schraut, W, Stelter, F, Schutt, C, Passlick, B, & Pforte, A. (1993) The novel subset of CD14+/CD16+ blood monocytes exhibits features of tissue macrophages. *European Journal of Immunology* **23**, 2053–2058.
- 451. Belge, K. U, Dayyani, F, Horelt, A, Siedlar, M, Frankenberger, M, Frankenberger, B, Espevik, T, & Ziegler-Heitbrock, L. (2002) The proinflammatory CD14+CD16+DR++ monocytes are a major source of TNF. *Journal of Immunology* 168, 3536–3542.
- 452. Fingerle, G, Pforte, A, Passlick, B, Blumenstein, M, Strobel, M, & Ziegler-Heitbrock, H. W. (1993) The novel subset of CD14+/CD16+ blood monocytes is expanded in sepsis patients. *Blood* 82, 3170–3176.
- 453. Feng, A. L, Zhu, J. K, Sun, J. T, Yang, M. X, Neckenig, M. R, Wang, X. W, Shao, Q. Q, Song, B. F, Yang, Q. F, Kong, B. H, & Qu, X. (2011) CD16+ monocytes in breast cancer patients: expanded by monocyte chemoattractant protein-1 and may be useful for early diagnosis. *Clinical and Experimental Immunology* **164**, 57–65.
- 454. Subimerb, C, Pinlaor, S, Lulitanond, V, Khuntikeo, N, Okada, S, McGrath, M. S, & Wongkham, S. (2010) Circulating CD14(+) CD16(+) monocyte levels predict tissue invasive character of cholangiocarcinoma. *Clinical and Experimental Immunology* **161**, 471–479.
- 455. Sánchez-Torres, C, García-Romo, G. S, Cornejo-Cortés, M. A, Rivas-Carvalho, A, & Sánchez-Schmitz, G. (2001) CD16+ and CD16- human blood monocyte subsets differentiate in vitro to dendritic cells with different abilities to stimulate CD4+ T cells. *International Immunology* **13**, 1571–1581.
- 456. Rivas-Carvalho, A, Meraz-Rios, M. A, Santos-Argumedo, L, Bajana, S, Soldevila, G, Moreno-Garcia, M. E, & Sánchez-Torres, C. (2004) CD16+ human monocytederived dendritic cells matured with different and unrelated stimuli promote similar allogeneic Th2 responses: regulation by pro- and anti-inflammatory cytokines. *International Immunology* 16, 1251–1263.

- 457. Saleh, M. N, Goldman, S. J, LoBuglio, A. F, Beall, A. C, Sabio, H, McCord, M. C, Minasian, L, Alpaugh, R. K, Weiner, L. M, & Munn, D. H. (1995) CD16+ monocytes in patients with cancer: spontaneous elevation and pharmacologic induction by recombinant human macrophage colony-stimulating factor. *Blood* **85**, 2910–2917.
- 458. Podhorecka, M, Dmoszynska, A, Rolinski, J, & Wasik, E. (2002) T type 1/type 2 subsets balance in B-cell chronic lymphocytic leukemia–the three-color flow cytometry analysis. *Leukemia Research* **26**, 657–660.
- 459. Rossmann, E. D, Lewin, N, Jeddi-Tehrani, M, Österborg, A, & Mellstedt, H. (2002) Intracellular T cell cytokines in patients with B cell chronic lymphocytic leukaemia (B-CLL). *European Journal of Haematology* **68**, 299–306.
- 460. Podhorecka, M, Dmoszynska, A, & Rolinski, J. (2004) Intracellular IFN-gamma expression by CD3+/CD8+ cell subset in B-CLL patients correlates with stage of the disease. *European Journal of Haematology* **73**, 29–35.
- 461. O'Reilly, V, Zeng, S. G, Bricard, G, Atzberger, A, Hogan, A. E, Jackson, J, Feighery, C, Porcelli, S. A, & Doherty, D. G. (2011) Distinct and overlapping effector functions of expanded human CD4+, CD8alpha+ and CD4-CD8alpha- invariant natural killer T cells. *PLoS ONE* 6, e28648.
- 462. Chowdhury, D & Lieberman, J. (2008) Death by a Thousand Cuts: Granzyme Pathways of Programmed Cell Death. *Annual Review of Immunology* **26**, 389–420.
- 463. Lin, H, Nieda, M, & Nicol, A. J. (2004) Differential proliferative response of NKT cell subpopulations to in vitro stimulation in presence of different cytokines. *European Journal of Immunology* **34**, 2664–2671.
- 464. Skov, S, Bonyhadi, M, Odum, N, & Ledbetter, J. A. (2000) IL-2 and IL-15 regulate CD154 expression on activated CD4 T cells. *Journal of Immunology* **164**, 3500–3505.
- 465. Wong, G. H, Clark-Lewis, I, McKimm-Breschkin, J. L, & Schrader, J. W. (1982) Interferon-gamma-like molecule induces Ia antigens on cultured mast cell progenitors. *Proceedings of the National Academy of Sciences of the United States of America* **79**, 6989–6993.
- 466. Halper, J. P, Fu, S. M, Gottlieb, A. B, Winchester, R. J, & Kunkel, H. G. (1979) Poor mixed lymphocyte reaction stimulatory capacity of leukemic B cells of chronic lymphocytic leukemia patients despite the presence of Ia antigens. *Journal of Clinical Investigation* **64**, 1141–1148.
- 467. Han, T, Bloom, M. L, Dadey, B, Bennett, G, Minowada, J, Sandberg, A. A, & Ozer, H. (1982) Lack of autologous mixed lymphocyte reaction in patients with chronic lymphocytic leukemia: evidence for autoreactive T-cell dysfunction not correlated with phenotype, karyotype, or clinical status. *Blood* 60, 1075–1081.
- 468. Patterson, S, Kotsianidis, I, Almeida, A, Politou, M, Rahemtulla, A, Matthew, B, Schmidt, R. R, Cerundolo, V, Roberts, I. A. G, & Karadimitris, A. (2005) Human invariant NKT cells are required for effective in vitro alloresponses. *Journal of Immunology* 175, 5087–5094.

- 469. Chiron, D, Bekeredjian-Ding, I, Pellat-Deceunynck, C, Bataille, R, & Jego, G. (2008) Toll-like receptors: lessons to learn from normal and malignant human B cells. *Blood* 112, 2205–2213.
- 470. Spaner, D. E, Shi, Y, White, D, Mena, J, Hammond, C, Tomic, J, He, L, Tomai, M. A, Miller, R. L, Booth, J, & Radvanyi, L. (2006) Immunomodulatory effects of Toll-like receptor-7 activation on chronic lymphocytic leukemia cells. *Leukemia* 20, 286–295.
- 471. Gross, H. J, Verwer, B, Houck, D, & Recktenwald, D. (1993) Detection of rare cells at a frequency of one per million by flow cytometry. *Cytometry* **14**, 519–526.
- 472. Croudace, J. E, Curbishley, S. M, Mura, M, Willcox, C. R, Illarianov, P. A, Besra, G. S, Adams, D. H, & Lammas, D. A. (2008) Identification of distinct human invariant natural killer T-cell response phenotypes to alpha-galactosylceramide. *BMC Immunology* 9, 71.
- 473. Bernardo, D, van Hoogstraten, I. M, Verbeek, W. H, Pena, A. S, Mearin, M. L, Arranz, E, Garrote, J. A, Scheper, R. J, Schreurs, M. W, Bontkes, H. J, Mulder, C. J, & von Blomberg, B. M. (2008) Decreased circulating iNKT cell numbers in refractory coeliac disease. *Clinical Immunology* **126**, 172–179.
- 474. van Der Vliet, H. J, von Blomberg, B. M, Nishi, N, Reijm, M, Voskuyl, A. E, van Bodegraven, A. A, Polman, C. H, Rustemeyer, T, Lips, P, van den Eertwegh, A. J, Giaccone, G, Scheper, R. J, & Pinedo, H. M. (2001) Circulating V(alpha24+) Vbeta11+ NKT cell numbers are decreased in a wide variety of diseases that are characterized by autoreactive tissue damage. *Clinical Immunology* 100, 144–148.
- 475. Vyth-Dreese, F. A, Sein, J, Van De Kasteele, W, Dellemijn, T. A. M, Van Den Bogaard, C, Nooijen, W. J, De Gast, G. C, Haanen, J. B. A. G, & Bex, A. (2010) Lack of anti-tumour reactivity despite enhanced numbers of circulating natural killer T cells in two patients with metastatic renal cell carcinoma. *Clinical and Experimental Immunology* 162, 447–459.
- 476. Dhodapkar, K. M, Cirignano, B, Chamian, F, Zagzag, D, Miller, D. C, Finlay, J. L, & Steinman, R. M. (2004) Invariant natural killer T cells are preserved in patients with glioma and exhibit antitumor lytic activity following dendritic cell-mediated expansion. *International Journal of Cancer* **109**, 893–899.
- 477. Song, W, van der Vliet, H. J. J, Tai, Y.-T, Prabhala, R, Wang, R, Podar, K, Catley, L, Shammas, M. A, Anderson, K. C, Balk, S. P, Exley, M. A, & Munshi, N. C. (2008) Generation of Antitumor Invariant Natural Killer T Cell Lines in Multiple Myeloma and Promotion of Their Functions via Lenalidomide: A Strategy for Immunotherapy. *Clinical Cancer Research* 14, 6955–6962.
- 478. Pang, D. J, Hayday, A. C, & Bijlmakers, M.-J. (2007) CD8 Raft localization is induced by its assembly into CD8alpha beta heterodimers, Not CD8alpha alpha homodimers. *Journal of Biological Chemistry* **282**, 13884–13894.
- 479. Garrido, F, Festenstein, H, & Schirrmacher, V. (1976) Further evidence for depression of H-2 and Ia-like specificities of foreign haplotypes in mouse tumour cell lines. *Nature* **261**, 705–707.

- 480. Garrido, F, Cabrera, T, & Aptsiauri, N. (2010) "Hard" and "soft" lesions underlying the HLA class I alterations in cancer cells: implications for immunotherapy. *International Journal of Cancer* **127**, 249–256.
- 481. Verheyden, S, Ferrone, S, Mulder, A, Claas, F. H, Schots, R, De Moerloose, B, Benoit, Y, & Demanet, C. (2009) Role of the inhibitory KIR ligand HLA-Bw4 and HLA-C expression levels in the recognition of leukemic cells by Natural Killer cells. *Cancer Immunology, Immunotherapy* **58**, 855–865.
- 482. Bourguin-Plonquet, A, Rouard, H, Roudot-Thoraval, F, Bellanger, C, Marquet, J, Delfau-Larue, M.-H, Diviné, M, & Farcet, J.-P. (2002) Severe decrease in peripheral blood dendritic cells in hairy cell leukaemia. *British Journal of Haematology* **116**, 595–597.
- 483. Ronger-Savle, S, Valladeau, J, Claudy, A, Schmitt, D, Peguet-Navarro, J, Dezutter-Dambuyant, C, Thomas, L, & Jullien, D. (2005) TGFbeta inhibits CD1d expression on dendritic cells. *Journal of Investigative Dermatology* **124**, 116–118.
- 484. Beum, P. V, Lindorfer, M. A, & Taylor, R. P. (2008) Within peripheral blood mononuclear cells, antibody-dependent cellular cytotoxicity of rituximab-opsonized Daudi cells is promoted by NK cells and inhibited by monocytes due to shaving. *Journal of Immunology* **181**, 2916–2924.
- 485. Dahlke, E, Schlag, R, Langenmayer, I, Frankenberger, M, Käfferlein, E, Subkowski, T, Emmerich, B, & Ziegler-Heitbrock, H. W. (1995) Decreased production of TNF and IL-6 in whole blood of CLL patients. *American Journal of Hematology* 49, 76–82.
- 486. Ziegler-Heitbrock, H. W. (1996) Heterogeneity of human blood monocytes: the CD14+ CD16+ subpopulation. *Immunology Today* **17**, 424–8.
- 487. Weber, C, Belge, K. U, von Hundelshausen, P, Draude, G, Steppich, B, Mack, M, Frankenberger, M, Weber, K. S, & Ziegler-Heitbrock, H. W. (2000) Differential chemokine receptor expression and function in human monocyte subpopulations. *Journal of Leukocyte Biology* **67**, 699–704.
- 488. Ancuta, P, Rao, R, Moses, A, Mehle, A, Shaw, S. K, Luscinskas, F. W, & Gabuzda, D. (2003) Fractalkine preferentially mediates arrest and migration of CD16+ monocytes. *Journal of Experimental Medicine* **197**, 1701–7.
- Hart, K. M, Bak, S. P, Alonso, A, & Berwin, B. (2009) Phenotypic and functional delineation of murine CX(3)CR1 monocyte-derived cells in ovarian cancer. *Neoplasia* 11, 564–73.
- 490. Heimbeck, I, Hofer, T. P. J, Eder, C, Wright, A. K, Frankenberger, M, Marei, A, Boghdadi, G, Scherberich, J, & Ziegler-Heitbrock, L. (2010) Standardized single-platform assay for human monocyte subpopulations: Lower CD14+CD16++ monocytes in females. *Cytometry* **77A**, 823–830.
- 491. Fernandez-Calotti, P, Gamberale, R, Costas, M, Sanchez Avalos, J, Geffner, J, & Giordano, M. (2006) Fludarabine induces pro-inflammatory activation of human monocytic cells through a MAPK/ERK pathway. *International Immunopharmacology* 6, 715–23.

- 492. Matsuda, J. L, Gapin, L, Baron, J. L, Sidobre, S, Stetson, D. B, Mohrs, M, Locksley, R. M, & Kronenberg, M. (2003) Mouse V alpha 14i natural killer T cells are resistant to cytokine polarization in vivo. *Proceedings of the National Academy of Sciences of the United States of America* **100**, 8395–400.
- 493. Matsuda, J. L, Naidenko, O. V, Gapin, L, Nakayama, T, Taniguchi, M, Wang, C. R, Koezuka, Y, & Kronenberg, M. (2000) Tracking the response of natural killer t cells to a glycolipid antigen using cd1d tetramers. *Journal of Experimental Medicine* **192**, 741–54.
- 494. Foster, B, Prussin, C, Liu, F, Whitmire, J. K, & Whitton, J. L. (2007) Detection of intracellular cytokines by flow cytometry. *Current Protocols in Immunology* Chapter 6, Unit 6 24.
- 495. Stetson, D. B, Mohrs, M, Reinhardt, R. L, Baron, J. L, Wang, Z. E, Gapin, L, Kronenberg, M, & Locksley, R. M. (2003) Constitutive cytokine mRNAs mark natural killer (NK) and NK T cells poised for rapid effector function. *Journal of Experimental Medicine* **198**, 1069–76.
- 496. Reyes, E, Prieto, A, Carrión, F, Garcia-Suarez, J, Esquivel, F, Guillén, C, & Alvarez-Mon, M. (1998) Altered pattern of cytokine production by peripheral blood CD2+ cells from B chronic lymphocytic leukemia patients. *American Journal of Hematology* 57, 93–100.
- 497. Kawano, T, Nakayama, T, Kamada, N, Kaneko, Y, Harada, M, Ogura, N, Akutsu, Y, Motohashi, S, Iizasa, T, Endo, H, Fujisawa, T, Shinkai, H, & Taniguchi, M. (1999) Antitumor cytotoxicity mediated by ligand-activated human V alpha24 NKT cells. *Cancer Research* **59**, 5102–5105.
- 498. Pallasch, C. P, Ulbrich, S, Brinker, R, Hallek, M, Uger, R. A, & Wendtner, C.-M. (2009) Disruption of T cell suppression in chronic lymphocytic leukemia by CD200 blockade. *Leukemia Research* **33**, 460–464.
- 499. Lindqvist, C. A, Christiansson, L. H, Simonsson, B, Enblad, G, Olsson-Strömberg, U, & Loskog, A. S. I. (2010) T regulatory cells control T-cell proliferation partly by the release of soluble CD25 in patients with B-cell malignancies. *Immunology* 131, 371–376.
- 500. van der Vliet, H. J. J, Molling, J. W, Nishi, N, Masterson, A. J, Kölgen, W, Porcelli, S. A, van den Eertwegh, A. J. M, von Blomberg, B. M. E, Pinedo, H. M, Giaccone, G, & Scheper, R. J. (2003) Polarization of Valpha24+ Vbeta11+ natural killer T cells of healthy volunteers and cancer patients using alpha-galactosylceramide-loaded and environmentally instructed dendritic cells. *Cancer Research* 63, 4101–4106.
- 501. Webb, T. J, Bieler, J. G, Schneck, J. P, & Oelke, M. (2009) Ex vivo induction and expansion of natural killer T cells by CD1d1-Ig coated artificial antigen presenting cells. *Journal of Immunological Methods* **346**, 38–44.
- 502. Zemmour, J, Little, A. M, Schendel, D. J, & Parham, P. (1992) The HLA-A,B "negative" mutant cell line C1R expresses a novel HLA-B35 allele, which also has a point mutation in the translation initiation codon. *Journal of Immunology* **148**, 1941– 1948.

- 503. Pawelec, G, Wagner, W, Adibzadeh, M, & Engel, A. (1999) T cell immunosenescence in vitro and in vivo. *Experimental Gerontology* **34**, 419–429.
- 504. Lenardo, M. J. (1991) Interleukin-2 programs mouse alpha beta T lymphocytes for apoptosis. *Nature* **353**, 858–861.
- 505. Refaeli, Y, Van Parijs, L, London, C. A, Tschopp, J, & Abbas, A. K. (1998) Biochemical mechanisms of IL-2-regulated Fas-mediated T cell apoptosis. *Immunity* **8**, 615–623.
- 506. Caserta, S, Alessi, P, Basso, V, & Mondino, A. (2010) IL-7 is superior to IL-2 for ex vivo expansion of tumour-specific CD4(+) T cells. *European Journal of Immunology* **40**, 470–479.
- 507. Harada, Y, Imataki, O, Heike, Y, Kawai, H, Shimosaka, A, Mori, S.-i, Kami, M, Tanosaki, R, Ikarashi, Y, Iizuka, A, Yoshida, M, Wakasugi, H, Saito, S, Takaue, Y, Takei, M, & Kakizoe, T. (2005) Expansion of alpha-galactosylceramide-stimulated Valpha24+ NKT cells cultured in the absence of animal materials. *Journal of Immunotherapy* 28, 314–321.
- 508. Nishi, N, van Der Vliet, H. J, Koezuka, Y, von Blomberg, B. M, Scheper, R. J, Pinedo, H. M, & Giaccone, G. (2000) Synergistic effect of KRN7000 with interleukin-15, -7, and -2 on the expansion of human V alpha 24+V beta 11+ T cells in vitro. *Human Immunology* **61**, 357–365.
- 509. Exley, M. A, Wilson, B, & Balk, S. P. (2010) Isolation and functional use of human NKT cells. *Current Protocols in Immunology* **Chapter 14**, Unit 14.11.
- 510. Yanagisawa, K, Seino, K.-i, Ishikawa, Y, Nozue, M, Todoroki, T, & Fukao, K. (2002) Impaired proliferative response of V alpha 24 NKT cells from cancer patients against alpha-galactosylceramide. *Journal of Immunology* **168**, 6494–6499.
- 511. van der Vliet HJ, Nishi, N, Koezuka, Y, von Blomberg BM, van den Eertwegh AJ, Porcelli, S. A, Pinedo, H. M, Scheper, R. J, & Giaccone, G. (2001) Potent expansion of human natural killer T cells using alpha-galactosylceramide (KRN7000)-loaded monocyte-derived dendritic cells, cultured in the presence of IL-7 and IL-15. *Journal of Immunological Methods* **247**, 61–72.
- 512. Liu, T.-Y, Uemura, Y, Suzuki, M, Narita, Y, Hirata, S, Ohyama, H, Ishihara, O, & Matsushita, S. (2008) Distinct subsets of human invariant NKT cells differentially regulate T helper responses via dendritic cells. *European Journal of Immunology* **38**, 1012–1023.
- 513. Kotsianidis, I, Silk, J. D, Spanoudakis, E, Patterson, S, Almeida, A, Schmidt, R. R, Tsatalas, C, Bourikas, G, Cerundolo, V, Roberts, I. A. G, & Karadimitris, A. (2006) Regulation of hematopoiesis in vitro and in vivo by invariant NKT cells. *Blood* **107**, 3138–3144.
- 514. Rogers, P. R, Matsumoto, A, Naidenko, O, Kronenberg, M, Mikayama, T, & Kato, S. (2004) Expansion of human Valpha24+ NKT cells by repeated stimulation with KRN7000. *Journal of Immunological Methods* **285**, 197–214.
- 515. Shimizu, K, Hidaka, M, Kadowaki, N, Makita, N, Konishi, N, Fujimoto, K, Uchiyama, T, Kawano, F, Taniguchi, M, & Fujii, S.-I. (2006) Evaluation of the function of

human invariant NKT cells from cancer patients using alpha-galactosylceramide-loaded murine dendritic cells. *Journal of Immunology* **177**, 3484–3492.

- 516. Exley, M. A, Hou, R, Shaulov, A, Tonti, E, Dellabona, P, Casorati, G, Akbari, O, Akman, H. O, Greenfield, E. A, Gumperz, J. E, Boyson, J. E, Balk, S. P, & Wilson, S. B. (2008) Selective activation, expansion, and monitoring of human iNKT cells with a monoclonal antibody specific for the TCR alpha-chain CDR3 loop. *European Journal of Immunology* **38**, 1756–1766.
- 517. Cosmi, L, De Palma, R, Santarlasci, V, Maggi, L, Capone, M, Frosali, F, Rodolico, G, Querci, V, Abbate, G, Angeli, R, Berrino, L, Fambrini, M, Caproni, M, Tonelli, F, Lazzeri, E, Parronchi, P, Liotta, F, Maggi, E, Romagnani, S, & Annunziato, F. (2008) Human interleukin 17-producing cells originate from a CD161+CD4+ T cell precursor. *Journal of Experimental Medicine* 205, 1903–1916.
- 518. Hameed, A, Lowrey, D. M, Lichtenheld, M, & Podack, E. R. (1988) Characterization of three serine esterases isolated from human IL-2 activated killer cells. *Journal of Immunology* **141**, 3142–3147.
- 519. Decker, T & Lohmann-Matthes, M. L. (1988) A quick and simple method for the quantitation of lactate dehydrogenase release in measurements of cellular cytotoxicity and tumor necrosis factor (TNF) activity. *Journal of Immunological Methods* **115**, 61–69.
- 520. Kitamura, H, Ohta, A, Sekimoto, M, Sato, M, Iwakabe, K, Nakui, M, Yahata, T, Meng, H, Koda, T, Nishimura, S, Kawano, T, Taniguchi, M, & Nishimura, T. (2000) alphagalactosylceramide induces early B-cell activation through IL-4 production by NKT cells. *Cellular Immunology* **199**, 37–42.
- 521. Ranheim, E. A & Kipps, T. J. (1993) Activated T cells induce expression of B7/BB1 on normal or leukemic B cells through a CD40-dependent signal. *Journal of Experimental Medicine* **177**, 925–35.
- 522. Galli, G, Nuti, S, Tavarini, S, Galli-Stampino, L, De Lalla, C, Casorati, G, Dellabona, P, & Abrignani, S. (2003) CD1d-restricted help to B cells by human invariant natural killer T lymphocytes. *Journal of Experimental Medicine* **197**, 1051–7.
- 523. Veuillen, C, Aurran-Schleinitz, T, Castellano, R, Rey, J, Mallet, F, Orlanducci, F, Pouyet, L, Just-Landi, S, Coso, D, Ivanov, V, Carcopino, X, Bouabdallah, R, Collette, Y, Fauriat, C, & Olive, D. (2012) Primary B-CLL resistance to NK cell cytotoxicity can be overcome in vitro and in vivo by priming NK cells and monoclonal antibody therapy. *Journal of Clinical Immunology* **32**, 632–46.
- 524. Suresh, K, Fraser, G, Scheid, E, Leber, B, Gauldie, J, & Foley, R. (2006) Generation of in vitro B-CLL specific HLA class I restricted CTL responses using autologous dendritic cells pulsed with necrotic tumor lysate. *Leukemia & Lymphoma* **47**, 297–306.
- 525. Kornacker, M, Moldenhauer, G, Herbst, M, Weilguni, E, Tita-Nwa, F, Harter, C, Hensel, M, & Ho, A. D. (2006) Cytokine-induced killer cells against autologous CLL: direct cytotoxic effects and induction of immune accessory molecules by interferon-gamma. *International Journal of Cancer* **119**, 1377–1382.

- 526. Chu, P, Wierda, W. G, & Kipps, T. J. (2000) CD40 activation does not protect chronic lymphocytic leukemia B cells from apoptosis induced by cytotoxic T lymphocytes. *Blood* **95**, 3853–3858.
- 527. Kollgaard, T, Petersen, S. L, Hadrup, S. R, Masmas, T. N, Seremet, T, Andersen, M. H, Madsen, H. O, Vindeløv, L, & thor Straten, P. (2005) Evidence for involvement of clonally expanded CD8+ T cells in anticancer immune responses in CLL patients following nonmyeloablative conditioning and hematopoietic cell transplantation. *Leukemia* **19**, 2273–2280.
- 528. Dutton, R. W & Eady, J. D. (1964) An in Vitro System for the Study of the Mechanism of Antigenic Stimulation in the Secondary Response. *Immunology* **7**, 40–53.
- 529. Lyons, A. B & Parish, C. R. (1994) Determination of lymphocyte division by flow cytometry. *Journal of Immunological Methods* **171**, 131–137.
- 530. Fulcher, D & Wong, S. (1999) Carboxyfluorescein succinimidyl ester-based proliferative assays for assessment of T cell function in the diagnostic laboratory. *Immunology and Cell Biology* **77**, 559–564.
- 531. Meyer, E. H, Wurbel, M. A, Staton, T. L, Pichavant, M, Kan, M. J, Savage, P. B, DeKruyff, R. H, Butcher, E. C, Campbell, J. J, & Umetsu, D. T. (2007) iNKT cells require CCR4 to localize to the airways and to induce airway hyperreactivity. *Journal of Immunology* **179**, 4661–4671.
- 532. Burger, J. A. (2010) Chemokines and chemokine receptors in chronic lymphocytic leukemia (CLL): From understanding the basics towards therapeutic targeting. *Seminars in Cancer Biology* **20**, 424–430.
- 533. Geissmann, F, Cameron, T. O, Sidobre, S, Manlongat, N, Kronenberg, M, Briskin, M. J, Dustin, M. L, & Littman, D. R. (2005) Intravascular immune surveillance by CXCR6+ NKT cells patrolling liver sinusoids. *PLoS Biology* 3, e113.
- 534. Kyriakakis, E, Cavallari, M, Andert, J, Philippova, M, Koella, C, Bochkov, V, Erne, P, Wilson, S. B, Mori, L, Biedermann, B. C, Resink, T. J, & De Libero, G. (2010) Invariant natural killer T cells: linking inflammation and neovascularization in human atherosclerosis. *European Journal of Immunology* **40**, 3268–3279.
- 535. Rosenberg, S. A, Yang, J. C, Sherry, R. M, Kammula, U. S, Hughes, M. S, Phan, G. Q, Citrin, D. E, Restifo, N. P, Robbins, P. F, Wunderlich, J. R, Morton, K. E, Laurencot, C. M, Steinberg, S. M, White, D. E, & Dudley, M. E. (2011) Durable complete responses in heavily pretreated patients with metastatic melanoma using T-cell transfer immunotherapy. *Clinical Cancer Research* **17**, 4550–7.
- 536. Xu, X. J, Zhao, H. Z, & Tang, Y. M. (2012) Efficacy and safety of adoptive immunotherapy using anti-CD19 chimeric antigen receptor transduced T-cells: a systematic review of phase I clinical trials. *Leukemia & Lymphoma*.
- 537. van Ravenswaay Claasen, H. H, Kluin, P. M, & Fleuren, G. J. (1992) Tumor infiltrating cells in human cancer. on the possible role of CD16+ macrophages in antitumor cytotoxicity. *Laboratory Investigation* **67**, 166–74.

- 538. Klebanoff, C. A, Acquavella, N, Yu, Z, & Restifo, N. P. (2011) Therapeutic cancer vaccines: are we there yet? *Immunological Reviews* **239**, 27–44.
- 539. Gilham, D. E, Debets, R, Pule, M, Hawkins, R. E, & Abken, H. (2012) CAR-T cells and solid tumors: tuning T cells to challenge an inveterate foe. *Trends in Molecular Medicine* **18**, 377–84.
- 540. Or, R, Shapira, M. Y, Resnick, I, Amar, A, Ackerstein, A, Samuel, S, Aker, M, Naparstek, E, Nagler, A, & Slavin, S. (2003) Nonmyeloablative allogeneic stem cell transplantation for the treatment of chronic myeloid leukemia in first chronic phase. *Blood* **101**, 441–445.
- 541. Garban, F. (2006) Prospective comparison of autologous stem cell transplantation followed by dose-reduced allograft (IFM99-03 trial) with tandem autologous stem cell transplantation (IFM99-04 trial) in high-risk de novo multiple myeloma. *Blood* **107**, 3474–3480.
- 542. Khouri, I. F, Bassett, R, Poindexter, N, O'Brien, S, Bueso-Ramos, C. E, Hsu, Y, Ferrajoli, A, Keating, M. J, Champlin, R, & Fernandez-Vina, M. (2011) Nonmyeloablative allogeneic stem cell transplantation in relapsed/refractory chronic lymphocytic leukemia: Long-Term Follow-Up, Prognostic Factors, and Effect of Human Leukocyte Histocompatibility Antigen Subtype on Outcome. *Cancer* doi: 10.1002/cncr.26091. [Epub ahead of print].
- 543. Blaise, D, Bay, J. O, Faucher, C, Michallet, M, Boiron, J.-M, Choufi, B, Cahn, J.-Y, Gratecos, N, Sotto, J.-J, François, S, Fleury, J, Mohty, M, Chabannon, C, Bilger, K, Gravis, G, Viret, F, Braud, A. C, Bardou, V. J, Maraninchi, D, & Viens, P. (2004) Reduced-intensity preparative regimen and allogeneic stem cell transplantation for advanced solid tumors. *Blood* **103**, 435–441.
- 544. Childs, R, Chernoff, A, Contentin, N, Bahceci, E, Schrump, D, Leitman, S, Read, E. J, Tisdale, J, Dunbar, C, Linehan, W. M, Young, N. S, & Barrett, A. J. (2000) Regression of metastatic renal-cell carcinoma after nonmyeloablative allogeneic peripheralblood stem-cell transplantation. *New England Journal of Medicine* **343**, 750–758.
- 545. Doney, K, Hägglund, H, Leisenring, W, Chauncey, T, Appelbaum, F. R, & Storb, R. (2003) Predictive factors for outcome of allogeneic hematopoietic cell transplantation for adult acute lymphoblastic leukemia. *Biology of Blood and Marrow Transplantation* **9**, 472–481.
- 546. Koh, H, Nakamae, H, Hagihara, K, Nakane, T, Manabe, M, Hayashi, Y, Nishimoto, M, Umemoto, Y, Nakamae, M, Hirose, A, Inoue, E, Inoue, A, Yoshida, M, Bingo, M, Okamura, H, Aimoto, R, Aimoto, M, Terada, Y, Koh, K.-R, Yamane, T, Ohsawa, M, & Hino, M. (2011) Factors that contribute to long-term survival in patients with leukemia not in remission at allogeneic hematopoietic cell transplantation. *Journal of Experimental & Clinical Cancer Research* **30**, 36.
- 547. Fox, E. J, Salk, J. J, & Loeb, L. A. (2009) Cancer Genome Sequencing–An Interim Analysis. *Cancer Research* **69**, 4948–4950.
- 548. Lanzavecchia, A, Roosnek, E, Gregory, T, Berman, P, & Abrignani, S. (1988) T cells

can present antigens such as HIV gp120 targeted to their own surface molecules. *Nature* **334**, 530–532.

- 549. Male, D. K, Pryce, G, & Hughes, C. C. (1987) Antigen presentation in brain: MHC induction on brain endothelium and astrocytes compared. *Immunology* **60**, 453–459.
- 550. Taflin, C, Charron, D, Glotz, D, & Mooney, N. (2011) Immunological function of the endothelial cell within the setting of organ transplantation. *Immunology Letters* **139**, 1–6.
- 551. Herkel, J, Jagemann, B, Wiegard, C, Lazaro, J. F. G, Lueth, S, Kanzler, S, Blessing, M, Schmitt, E, & Lohse, A. W. (2003) MHC class II-expressing hepatocytes function as antigen-presenting cells and activate specific CD4 T lymphocyutes. *Hepatology* **37**, 1079–1085.
- 552. Zaloudik, J, Moore, M, Ghosh, A. K, Mechl, Z, & Rejthar, A. (1988) DNA content and MHC class II antigen expression in malignant melanoma: clinical course. *Journal of Clinical Pathology* **41**, 1078–1084.
- 553. Meijer, K, de Vries, M, Al-Lahham, S, Bruinenberg, M, Weening, D, Dijkstra, M, Kloosterhuis, N, van der Leij, R. J, van der Want, H, Kroesen, B.-J, Vonk, R, & Rezaee, F. (2011) Human Primary Adipocytes Exhibit Immune Cell Function: Adipocytes Prime Inflammation Independent of Macrophages. *PLoS One* 6, e17154.
- 554. Dissanayake, S. K, Thompson, J. A, Bosch, J. J, Clements, V. K, Chen, P. W, Ksander, B. R, & Ostrand-Rosenberg, S. (2004) Activation of tumor-specific CD4(+) T lymph-ocytes by major histocompatibility complex class II tumor cell vaccines: a novel cell-based immunotherapy. *Cancer Research* 64, 1867–1874.
- 555. Vigouroux, S, Michallet, M, Porcher, R, Attal, M, Ades, L, Bernard, M, Blaise, D, Tabrizi, R, Garban, F, Cassuto, J.-P, Chevalier, P, Facon, T, Ifrah, N, Renaud, M, Tilly, H, Vernant, J.-P, Kuentz, M, Bourhis, J.-H, Bordigoni, P, Deconinck, E, Lioure, B, Socié, G, Milpied, N, & French Society of Bone Marrow Graft Transplantation and Cellular Therapy (SFGM-TC). (2007) Long-term outcomes after reduced-intensity conditioning allogeneic stem cell transplantation for low-grade lymphoma: a survey by the French Society of Bone Marrow Graft Transplantation and Cellular Therapy (SFGM-TC). *Haematologica* **92**, 627–634.
- 556. Schwab, R, Szabo, P, Manavalan, J. S, Weksler, M. E, Posnett, D. N, Pannetier, C, Kourilsky, P, & Even, J. (1997) Expanded CD4+ and CD8+ T cell clones in elderly humans. *Journal of Immunology* **158**, 4493–4499.
- 557. Yager, E. J, Ahmed, M, Lanzer, K, Randall, T. D, Woodland, D. L, & Blackman, M. A. (2008) Age-associated decline in T cell repertoire diversity leads to holes in the repertoire and impaired immunity to influenza virus. *Journal of Experimental Medicine* **205**, 711–723.
- 558. Rosenberg, S. A, Sherry, R. M, Morton, K. E, Scharfman, W. J, Yang, J. C, Topalian, S. L, Royal, R. E, Kammula, U, Restifo, N. P, Hughes, M. S, Schwartzentruber, D, Berman, D. M, Schwarz, S. L, Ngo, L. T, Mavroukakis, S. A, White, D. E, & Steinberg, S. M. (2005) Tumor progression can occur despite the induction of very high levels

of self/tumor antigen-specific CD8+ T cells in patients with melanoma. *Journal of Immunology* **175**, 6169–76.

- 559. Aarntzen, E. H, de Vries, I. J, Lesterhuis, W. J, Schuurhuis, D, Jacobs, J. F, Bol, K, Schreibelt, G, Mus, R, De Wilt, J. H, Haanen, J. B, Schadendorf, D, Croockewit, S, Blokx, W. A, van Rossum, M. M, Kwok, W. W, Adema, G. J, Punt, C. J, & Figdor, C. G. (2012) Targeting CD4+ T helper cells improves the induction of anti-tumor responses in dendritic cell based vaccination. *Cancer Research* [Epub ahead of print].
- 560. Friedman, K. M, Prieto, P. A, Devillier, L. E, Gross, C. A, Yang, J. C, Wunderlich, J. R, Rosenberg, S. A, & Dudley, M. E. (2012) Tumor-specific CD4+ melanoma tumor-infiltrating lymphocytes. *Journal of Immunotherapy* **35**, 400–8.
- 561. Marmont, A. M, Horowitz, M. M, Gale, R. P, Sobocinski, K, Ash, R. C, van Bekkum, D. W, Champlin, R. E, Dicke, K. A, Goldman, J. M, & Good, R. A. (1991) T-cell depletion of HLA-identical transplants in leukemia. *Blood* 78, 2120–2130.
- 562. Kollman, C, Howe, C. W, Anasetti, C, Antin, J. H, Davies, S. M, Filipovich, A. H, Hegland, J, Kamani, N, Kernan, N. A, King, R, Ratanatharathorn, V, Weisdorf, D, & Confer, D. L. (2001) Donor characteristics as risk factors in recipients after transplantation of bone marrow from unrelated donors: the effect of donor age. *Blood* 98, 2043–2051.
- 563. Wu, R, Forget, M. A, Chacon, J, Bernatchez, C, Haymaker, C, Chen, J. Q, Hwu, P, & Radvanyi, L. G. (2012) Adoptive T-cell therapy using autologous tumor-infiltrating lymphocytes for metastatic melanoma: current status and future outlook. *Cancer Journal* 18, 160–75.
- 564. Williams, K. M, Hakim, F. T, & Gress, R. E. (2007) T cell immune reconstitution following lymphodepletion. *Seminars in Immunology* **19**, 318–330.
- 565. Dudley, M. E, Yang, J. C, Sherry, R, Hughes, M. S, Royal, R, Kammula, U, Robbins, P. F, Huang, J, Citrin, D. E, Leitman, S. F, Wunderlich, J, Restifo, N. P, Thomasian, A, Downey, S. G, Smith, F. O, Klapper, J, Morton, K, Laurencot, C, White, D. E, & Rosenberg, S. A. (2008) Adoptive cell therapy for patients with metastatic melanoma: evaluation of intensive myeloablative chemoradiation preparative regimens. *Journal of Clinical Oncology* **26**, 5233–5239.
- 566. Rech, A. J & Vonderheide, R. H. (2009) Clinical Use of Anti-CD25 Antibody Daclizumab to Enhance Immune Responses to Tumor Antigen Vaccination by Targeting Regulatory T cells. *Annals of the New York Academy of Sciences* **1174**, 99–106.
- 567. Vincent, J, Vincent, J, Mignot, G, Mignot, G, Chalmin, F, Chalmin, F, Ladoire, S, Ladoire, S, Bruchard, M, Bruchard, M, Chevriaux, A, Chevriaux, A, Martin, F, Martin, F, Apetoh, L, Apetoh, L, Rebe, C, Rebe, C, Ghiringhelli, F, & Ghiringhelli, F. (2010) 5-Fluorouracil Selectively Kills Tumor-Associated Myeloid-Derived Suppressor Cells Resulting in Enhanced T Cell-Dependent Antitumor Immunity. *Cancer Research* **70**, 3052–3061.
- 568. Peggs, K. S, Quezada, S. A, & Allison, J. P. (2008) Cell intrinsic mechanisms of T-cell inhibition and application to cancer therapy. *Immunological Reviews* **224**, 141–165.
- 569. Miller, J. S, Soignier, Y, Panoskaltsis-Mortari, A, McNearney, S. A, Yun, G. H, Fautsch, S. K, McKenna, D, Le, C, Defor, T. E, Burns, L. J, Orchard, P. J, Blazar, B. R, Wagner, J. E, Slungaard, A, Weisdorf, D. J, Okazaki, I. J, & McGlave, P. B. (2005) Successful adoptive transfer and in vivo expansion of human haploidentical NK cells in patients with cancer. *Blood* 105, 3051–3057.
- 570. Gale, R. P. Horowitz, M. M, Ash, R. C, Champlin, R. E, Goldman, J. M, Rimm, A. A, Ringdén, O, Stone, J. A, & Bortin, M. M. (1994) Identical-twin bone marrow transplants for leukemia. *Annals of Internal Medicine* **120**, 646–652.
- 571. Juliusson, G, Theorin, N, Karlsson, K, Frödin, U, & Malm, C. (2006) Subcutaneous alemtuzumab vs ATG in adjusted conditioning for allogeneic transplantation: influence of Campath dose on lymphoid recovery, mixed chimerism and survival. *Bone Marrow Transplantation* **37**, 503–510.
- 572. Kroger, N, Shaw, B, Iacobelli, S, Zabelina, T, Peggs, K, Shimoni, A, Nagler, A, Binder, T, Eiermann, T, Madrigal, A, Schwerdtfeger, R, Kiehl, M, Sayer, H. G, Beyer, J, Bornhauser, M, Ayuk, F, Zander, A. R, Marks, D. I, & the Clinical Trial Committee of the British Society of Blood and Marrow Transplantation and the German Cooperative Transplant Group. (2005) Comparison between antithymocyte globulin and alemtuzumab and the possible impact of KIR-ligand mismatch after dose-reduced conditioning and unrelated stem cell transplantation in patients with multiple myeloma. *British Journal of Haematology* **129**, 631–643.
- 573. Ishii, K. J & Akira, S. (2007) Toll or Toll-Free Adjuvant Path Toward the Optimal Vaccine Development. *Journal of Clinical Immunology* **27**, 363–371.
- 574. Draube, A, Klein-Gonzalez, N, Mattheus, S, Brillant, C, Hellmich, M, Engert, A, & von Bergwelt-Baildon, M. (2011) Dendritic cell based tumor vaccination in prostate and renal cell cancer: a systematic review and meta-analysis. *PLoS ONE* **6**, e18801.
- 575. Hobohm, U. (2001) Fever and cancer in perspective. *Cancer Immunology, Immunotherapy* **50**, 391–396.
- 576. Vosika, G. J, Schmidtke, J. R, Goldman, A, Ribi, E, Parker, R, & Gray, G. R. (1979) Intralesional immunotherapy of malignant melanoma with mycobacterium smegmatis cell wall skeleton combined with trehalose dimycolate (P3). *Cancer* **44**, 495–503.
- 577. Adams, S. (2009) Toll-like receptor agonists in cancer therapy. *Immunotherapy* **1**, 949–964.
- 578. Trof, R. J, Beishuizen, A, Wondergem, M. J, & Strack van Schijndel, R. J. M. (2007) Spontaneous remission of acute myeloid leukaemia after recovery from sepsis. *Netherlands Journal of Medicine* **65**, 259–262.
- 579. Speiser, D. E, Lienard, D, Rufer, N, Rubio-Godoy, V, Rimoldi, D, Lejeune, F, Krieg, A. M, Cerottini, J. C, & Romero, P. (2005) Rapid and strong human CD8+ t cell responses to vaccination with peptide, IFA, and CpG oligodeoxynucleotide 7909. *Journal of Clinical Investigation* **115**, 739–46.
- 580. Rosenberg, S. A, Yang, J. C, Kammula, U. S, Hughes, M. S, Restifo, N. P, Schwarz, S. L, Morton, K. E, Laurencot, C. M, & Sherry, R. M. (2010) Different adjuvanticity of incomplete Freund's adjuvant derived from beef or vegetable components in

melanoma patients immunized with a peptide vaccine. *Journal of Immunotherapy* **33**, 626–9.

- 581. van de Laar, L, Coffer, P. J, & Woltman, A. M. (2012) Regulation of dendritic cell development by GM-CSF: molecular control and implications for immune homeostasis and therapy. *Blood* **119**, 3383–93.
- 582. Kantoff, P. W, Higano, C. S, Shore, N. D, Berger, E. R, Small, E. J, Penson, D. F, Redfern, C. H, Ferrari, A. C, Dreicer, R, Sims, R. B, Xu, Y, Frohlich, M. W, & Schellhammer, P. F. (2010) Sipuleucel-T immunotherapy for castration-resistant prostate cancer. *New England Journal of Medicine* **363**, 411–22.
- 583. Kantoff, P. W, Schuetz, T. J, Blumenstein, B. A, Glode, L. M, Bilhartz, D. L, Wyand, M, Manson, K, Panicali, D. L, Laus, R, Schlom, J, Dahut, W. L, Arlen, P. M, Gulley, J. L, & Godfrey, W. R. (2010) Overall survival analysis of a phase II randomized controlled trial of a Poxviral-based PSA-targeted immunotherapy in metastatic castration-resistant prostate cancer. *Journal of Clinical Oncology* 28, 1099–105.
- 584. Yi, Q, Szmania, S, Freeman, J, Qian, J, Rosen, N. A, Viswamitra, S, Cottler-Fox, M, Barlogie, B, Tricot, G, & van Rhee, F. (2010) Optimizing dendritic cell-based immunotherapy in multiple myeloma: intranodal injections of idiotype-pulsed CD40 ligand-matured vaccines led to induction of type-1 and cytotoxic T-cell immune responses in patients. *British Journal of Haematology* **150**, 554–64.
- 585. Wilson, J, Cullup, H, Lourie, R, Sheng, Y, Palkova, A, Radford, K. J, Dickinson, A. M, Rice, A. M, Hart, D. N, & Munster, D. J. (2009) Antibody to the dendritic cell surface activation antigen CD83 prevents acute graft-versus-host disease. *Journal of Experimental Medicine* 206, 387–98.
- 586. Wierda, W. (2005) Chemoimmunotherapy With Fludarabine, Cyclophosphamide, and Rituximab for Relapsed and Refractory Chronic Lymphocytic Leukemia. *Journal* of Clinical Oncology 23, 4070–4078.
- 587. Badoux, X. C, Keating, M. J, Wang, X, O'Brien, S. M, Ferrajoli, A, Faderl, S, Burger, J, Koller, C, Lerner, S, Kantarjian, H, & Wierda, W. G. (2011) Fludarabine, cyclophosphamide, and rituximab chemoimmunotherapy is highly effective treatment for relapsed patients with CLL. *Blood* **117**, 3016–3024.
- 588. Rosenberg, S. A & Dudley, M. E. (2009) Adoptive cell therapy for the treatment of patients with metastatic melanoma. *Current Opinion in Immunology* **21**, 233–240.
- 589. Gassner, F. J, Weiss, L, Geisberger, R, Hofbauer, J. P, Egle, A, Hartmann, T. N, Greil, R, & Tinhofer, I. (2011) Fludarabine modulates composition and function of the T cell pool in patients with chronic lymphocytic leukaemia. *Cancer Immunology, Immunotherapy* **60**, 75–85.
- 590. Aquino, V. M, Douek, D. C, Berryman, B, Johnson, M, Jain, V. K, & Collins, R. H. (2003) Evaluation of thymic output by measurement of T-cell-receptor gene rearrangement excisional circles (TREC) in patients who have received fludarabine. *Leukemia & Lymphoma* **44**, 343–8.
- 591. Tam, C. S, O'Brien, S, Wierda, W, Kantarjian, H, Wen, S, Do, K. A, Thomas, D. A, Cortes, J, Lerner, S, & Keating, M. J. (2008) Long-term results of the fludarabine,

cyclophosphamide, and rituximab regimen as initial therapy of chronic lymphocytic leukemia. *Blood* **112**, 975–80.

- 592. Tam, C. (2012) *New developments in CLL therapy*. (HAA Annual Scientific Meeting, Melbourne, Australia).
- 593. Bottcher, S, Stilgenbauer, S, Busch, R, Bruggemann, M, Raff, T, Pott, C, Fischer, K, Fingerle-Rowson, G, Dohner, H, Hallek, M, Kneba, M, & Ritgen, M. (2009) Standardized MRD flow and ASO IGH RQ-PCR for MRD quantification in CLL patients after rituximab-containing immunochemotherapy: a comparative analysis. *Leukemia* 23, 2007–17.
- 594. Rawstron, A. C, Bottcher, S, Letestu, R, Villamor, N, Fazi, C, Kartsios, H, de Tute, R. M, Shingles, J, Ritgen, M, Moreno, C, Lin, K, Pettitt, A. R, Kneba, M, Montserrat, E, Cymbalista, F, Hallek, M, Hillmen, P, & Ghia, P. (2012) Improving efficiency and sensitivity: European Research Initiative in CLL (ERIC) update on the international harmonised approach for flow cytometric residual disease monitoring in CLL. *Leukemia* doi: 10.1038/leu.2012.216. [Epub ahead of print].
- 595. Bottcher, S, Ritgen, M, Fischer, K, Stilgenbauer, S, Busch, R. M, Fingerle-Rowson, G, Fink, A. M, Buhler, A, Zenz, T, Wenger, M. K, Mendila, M, Wendtner, C. M, Eichhorst, B. F, Dohner, H, Hallek, M. J, & Kneba, M. (2012) Minimal residual disease quantification is an independent predictor of progression-free and overall survival in chronic lymphocytic leukemia: a multivariate analysis from the randomized gcllsg cll8 trial. *Journal of Clinical Oncology* **30**, 980–8.
- 596. Young, M. H & Gapin, L. (2011) Group 1 CD1-restricted T cells take center stage. *European Journal of Immunology* **41**, 592–4.

Appendix A

Study recruitment materials

26 January 2009

***** ******

Dear ** ******,

Re: Invitation to participate in a study of chronic lymphocytic leukaemia

We are writing to invite you to take part in a study of chronic lymphocytic leukaemia. This study is taking place at the Malaghan Institute of Medical Research, in conjunction with Wellington Blood & Cancer Centre. Our hospital clinic records indicated that you might be eligible for this study, and we are contacting you with the permission of your GP.

The aim of this study is to look at immune system function in people with and without chronic lymphocytic leukaemia. Taking part in the study involves completing a questionnaire, having a brief examination for enlarged lymph nodes, and giving a blood sample at the Malaghan Institute of Medical Research at Victoria University in Kelburn. The whole process takes about 30 minutes. The study involves one visit only, and it does not involve any treatments. The Central Regional Ethics Committee has approved this study.

If you are interested in taking part in the study and would like more information, please complete the attached form and send it back to us in the envelope provided. Taking part in the study is entirely voluntary (your choice), and whether you participate or not, the study does not affect your medical care.

Yours faithfully,

Dr Robert Weinkove Haematology Registrar, Malaghan Institute Asst Professor John Carter Consultant Haematologist, CCDHB

Please indicate your response by ticking one of the boxes below.

□ **Yes**, I am interested in the chronic lymphocytic leukaemia study. Please contact me with further details.

My telephone number is:

or: My email address is:

No, I would rather not take part in this study.

Please post this form to:	Dr Robert Weinkove, Clinical Research Fellow,
	Malaghan Institute of Medical Research,
	PO Box 7060, Wellington
Or email:	rweinkove@malaghan.org.nz
Telephone:	04 499 6914 extension 851 (24h answerphone)

MALAGHAN IN OF MEDICAL R	NSTITUTE BESEARCH RESEARCH
Could you	u help with leukaemia research?
Researchers at the Malages system in people with an new treatments for CLL	ghan Institute of Medical Research are conducting a study comparing the immune nd without chronic lymphocytic leukaemia (CLL). This study may help us to design and similar conditions.
	We are looking for: healthy volunteers aged 50 and over , and people with chronic lymphocytic leukaemia
The study involves:	A short questionnaire A blood sample A brief examination to look for lymph nodes (for people with CLL only)
If you are interested in p	oarticipating, please contact:
	Dr Robert Weinkove Clinical Research Fellow, Malaghan Institute of Medical Research Telephone: 04 499 6914 extension 851 (24h answerphone) Email: rweinkove@malaghan.org.nz
This study has been as	and by the Control Decised Lthice Committee

I his study has been approved by the Central Regional Ethics Committee

v 3, 05.08.2009

Appendix B

Participant information sheets

iNKT cells in CLL study. Information	 What are the aims of the study? To find out if people with CLL have different numbers of iNKT cells in their blood to people without CLL, and to see if the cells behave differently in the laboratory. Using tests within the laboratory, to find out whether substances which stimulate the iNKT cells can help the immune system to fight leukaemia 	We think that iNKT cells are able to direct the immune system.	CLL causes changes to the body's immune system that make it difficult for the immune system to fight the leukaemia. We want to study a particular part of the immune system in people with and without CUT. The type of blood	study. You may have a triend, tamily or whanau support to help you understand this study and any other explanation you may require. CLL is the most common leukaemia in New Zealand. Many people with CLL do not need treatment for many years, and some never need treatment at all. Sometimes CLL is more aggressive and requires treatment with chemotherapy or other methods.	You are invited to take part a study looking at the immune system in chronic lymphocytic leukaemia (CLL). This study is voluntary (your choice). You do not have to take part in this study. You may withdraw from the study at any time without affecting to your health care now or in the future. You will be given as much time as you need to decide if you wish to take part in this	iNKT cells in Chronic Lymphocytic Leukaemia Information Sheet for Patients	MALAGHAN INSTITUTE OF MEDICAL RESEARCH UNIVERSITY of OTAGO
sheet for natients version 6. 05/08/2009	 investigator can help you to complete this. A brief medical examination by a doctor to check the size of any lymph glands in your neck, armpits and groin, and to see if your liver and spleen are enlarged. You may request a chaperone if you wish. A sample of blood (up to 120ml) from a vein in the arm. Your paper and electronic medical records will be reviewed to check information relevant to the study and to check for health problems that might prevent you from participating in the study. 	 What does the study involve? Your permission. After explaining the study, we shall ask you to complete and sign a consent form. A short questionnaire (10 minutes long) for you to complete. The study 	How long is the study? The study will take about 45 minutes of your time. The study will be completed in December 2010.	Where is the study held? Blood samples will be collected at the Malaghan Institute of Medical Research in Kelburn or at the Blood and Cancer Centre at Wellington Hospital. The blood samples will be tested in the laboratories of the Malaghan Institute of Medical Research in Kelburn.	If you have ever had chemotherapy for a condition other than CLL, if you have ever been given medicines that affect the immune system (such as azathioprine or ciclosporin), or if you have an infection at the moment, you should not enter the study.	Who can be in the study? People with CLL and healthy volunteers of a similar age. This study is for adults (over 18 years) only.	We hope that the knowledge we gain will help us to develop new ways to treat CLL and similar diseases.

This study is voluntary (your choice). You may withdraw from the study at any time without affecting your health care now or in the future.

choot for notionts vorsion 6 05/08/2000	initamenta III in CIII et allo
A blood test can be uncomfortable. A temporary bruise may develop. Rarely, people faint after blood drawing. Very rarely, the vein in which the needle has been inserted may become inflamed or infected, which can be treated.	You may choose to have your samples disposed of using appropriate karakia at the end of the study. If you would like this, please tell a study investigator and circle this choice on the consent form.
near wellington blood and Cancer Centre (at the Cancer Society). There is no payment or reimbursement for participation. If travel costs might prevent you from taking part please let the researchers know, as reimbursement of your travel costs could be provided.	How will my blood sample be disposed of? Your blood samples will be disposed of by incineration in accordance with standard safe practice.
You will need to visit the Malaghan Institute of Medical Research, on the campus of Victoria University in Kelburn, or the Wellington Blood and Cancer Centre at Wellington Hospital to have the blood sample taken. Free parking will be available at the Malaghan Institute of Medical Research and	the study show unexpected abnormalities (such as significant anaemia, very low platelets or an very high white blood cell count), the study investigators will contact you and ask you to visit your family doctor or a hospital doctor for a repeat of the blood test.
What are the risks and inconveniences?	We expect to see some abnormalities of your blood count because you have chronic lymphocytic leukaemia. However, if the blood tests we do as part of
The study will help researchers understand how CLL affects the immune system. The study might help researchers to develop new ways of treating CLL which make use of patients' own immune systems.	traits or for any genes that can be inherited. What if my blood test results are abnormal?
What are the benefits of the study?	researchers have shown that CLL can behave differently depending on which genetic changes the leukaemia cells have. We shall not test for any genetic
BENEFITS AND RISKS	Health Board and to Canterbury Health Laboratories to analyse genetic changes inside the leukaemia cells. This is important because other
blood sample and for any data to be destroyed. If you would like to do this, please contact one of the study investigators (see page 3 of this leaflet for contact details).	(immune globulin levels). Anonymous samples of your blood will be sent to Capital and Coast District
Can I change my mind and withdraw from the study? Yes. You are free to withdraw from the study at any time, without giving a reason and without affecting your future healthcare. You can ask for your	An anonymous sample of your blood will be sent to Capital and Coast District Health Board to check the blood count (the number of white and red blood cells and platelets) and to measure markers of immune function
Can my blood sample be returned to me? Unfortunately we cannot return any blood samples to participants. This is because the samples will have been treated with chemicals and substances that might be dangerous.	What will happen to my blood sample? Most of your blood sample will be analysed at the research laboratory in the Malaghan Institute of Medical Research where the numbers and activity of iNKT cells and other immune system cells will be measured.

Will I have to give any more samples?

or if your leukaemia progresses while the study is still under way, we might ask you to give a second sample of blood. You do not have to do so, and any However, if there is a technical problem with analysing your blood sample, further involvement in the study is entirely your choice. We would like each participant to attend and give a blood sample once only.

What will the study cost me?

There is no cost to you apart from the time that you give

Confidentiality

confidential manner. study will be kept for 10 years by the investigators, and then destroyed in a only. Study records will be stored on a password-protected computer and in a this study. Your blood samples will be labelled with a study code number No material that could personally identify you will be used in any reports of facility, and will only be used by the study researchers. Records from the locked filing cabinet. Computers and files are kept in a restricted-entry

Results of the study

request a copy of the study results by indicating your choice on the consent published in scientific journals and presented at research meetings. You can Investigator. form. If you need further information please contact the Principal The overall results of the study (from which you cannot be identified) will be

Ethical approval

Committee. This study has received ethical approval from the Central Regional Ethics

The study team thank you for participating.

telephone: in this study you may wish to contact a Health and Disability Advocate, If you have any questions or concerns about your rights as a participant

0800 42 36 38 (0800 4 ADNET)

understand, or with questions regarding this study: You may contact the researchers to ask about anything you do not

Principal Investigator or	Co-Inv
Dr Robert Weinkove	Profess
Clinical Research Fellow	Consul
Malaghan Institute of Medical Research	Blood
PO Box 7060	Welling
Wellington	
Phone: (04) 499 6914 ext 851	Welling
Mobile: 021 2160135	Phone:

& Cancer Centre tant Haematologist gton Hospital sor John Carter /estigator

lgton (04) 385 5999 ext 6756

Email: <u>rweinkove@malaghan.org.nz</u>

Compensation

might not get any compensation. This depends on a number of factors such as study, you may be covered by ACC under the Injury Prevention, Rehabilitation and you have ACC cover, generally this will affect your right to sue the investigators. reimbursement of costs and expenses and there may be no lump sum compensation assessed by ACC according to the provisions of the 2002 Injury Prevention Compensation Act. ACC cover is not automatic and your case will need to be In the unlikely event of a physical injury as a result of your participation in this payable. There is no cover for mental injury unless it is a result of physical injury. If whether you are an earner or non-earner. ACC usually provides only partial Rehabilitation and Compensation Act. If your claim is accepted by ACC, you still

investigator If you have any questions about ACC, contact your nearest ACC office or the

MALAGHAN INSTITUTE OF MEDICAL RESEARCH	UNIVERSITY _{of} OTAGO te whare wannga o otágo	Who can be in the study? Healthy volunteers and people with CLL. This study is for adults (over 18
iNKT cells in CLI	L study	years) only.
Information Sheet for	Volunteers	If you have ever had chemotherapy before, if you have ever been given medicines that affect the immine system (such as azathionrine or
You are invited to take part a study comparing volunteers with that of patients with chronic ly	g the immune system of healthy ymphocytic leukaemia (CLL).	ciclosporin), or if you have an infection at the moment, you should not enter the study.
This study is voluntary (your choice). You do study. You may withdraw from the study at an your health care now or in the future. You will need to decide if you wish to take part in this s family or whanau support to help you understa explanation you may require.	not have to take part in this ny time without affecting to I be given as much time as you study. You may have a friend, and this study and any other	Where is the study held? Blood samples will be collected at the Malaghan Institute of Medical Research in Wellington. The blood samples will be tested in the laboratories of the Malaghan Institute of Medical Research and at Capital and Coast District Health Board.
CLL is the most common leukaemia in New Z the body's immune system that make it diffi fight the leukaemia. We want to study a partic	Zealand. CLL causes changes to icult for the immune system to sular part of the immune system	How long is the study? The study will take about 45 minutes of your time. The whole study will be completed in December 2010.
in people with and without CLL. The type c called the 'invariant natural killer T cell' (iN cells are able to direct the immune system.	of blood cell we shall study is VKT cell). We think that iNKT	What does the study involve? • Your permission. After explaining the study, we shall ask you to complete
ABOUT THE STU	UDY	• A short questionnaire (10 minutes long) for you to complete. The study investigator can belo you to complete this
What are the aims of the study?		• A sample of blood (up to 120ml) from a vein in the arm.
• To find out if people with CLL have different blood to healthy volunteers, and to see if thes	t numbers of iNKT cells in their se cells behave differently in the	What will happen to my blood sample?
 Iaboratory. Using tests within the laboratory, to find stimulate the iNKT cells can help the immune 	out whether substances which e system to fight leukaemia	Research for special tests to measure the numbers and activity of iNKT cells and other immune system cells. Anonymous samples of your blood will be
We hope that the knowledge we gain will help CLL and similar diseases.	us to develop new ways to treat	sent to Capital and Coast District freatin Board to eneck your blood count (the numbers of white and red blood cells and platelets) and to measure markers of immune function (immune globulin levels).
iNF This study is voluntary (your ch	KT cells in CLL study. Information sh hoice). You may withdraw from the st	1 neet for Volunteers version 5, 08/06/2009 udy at any time without affecting your health care now or in the future.

How will my blood sample be disposed of?

Your blood samples will be disposed of by incineration in accordance with standard safe practice.

You may choose to have your samples disposed of using appropriate karakia at the end of the study. If you would like this, please tell a study investigator and circle this choice on the consent form.

Can my blood sample be returned to me?

Unfortunately we cannot return any blood samples to participants. This is because the samples will have been treated with chemicals and substances that might be dangerous.

Can I change my mind and withdraw from the study?

Yes. You are free to withdraw from the study at any time, without giving a reason and without affecting your future healthcare. You can ask for your blood sample and for any data to be destroyed. If you would like to do this, please contact one of the study investigators (see page 3 of this leaflet for contact details).

BENEFITS AND RISKS

What are the benefits of the study?

The study will help researchers understand how CLL affects the immune system. The study might help researchers to develop new ways of treating CLL which make use of patients' own immune systems. This study will not be of direct benefit to you.

What are the risks and inconveniences?

You will need to visit the Malaghan Institute of Medical Research, on the campus of Victoria University in Kelburn to have the blood sample taken. Free parking will be available. There is no payment or reimbursement for participation. If travel costs might prevent you from taking part please let the researchers know, as reimbursement of your travel costs could be provided.

A blood test can be uncomfortable. A temporary bruise may develop. Rarely, people faint after blood drawing. Very rarely, the vein in which the needle has been inserted may become inflamed or infected, which can be treated.

Will I have to give any more samples?

We would like each participant to attend and give a blood sample once only. However, if there is a technical problem with analysing your blood sample, we might ask you to give a second sample of blood. You do not have to do so, and any further involvement in the study is entirely your choice.

What if my blood results are abnormal?

We have asked you to enter the study as a healthy volunteer. However, there is a small chance that the blood count we perform shows an abnormal result. For instance, it might show that you have anaemia (low number of red blood cells), or that you have an abnormal number of white blood cells or platelets. If this happens we shall contact you and ask you to visit your family doctor or an emergency doctor for a repeat blood test.

What will the study cost me?

There is no cost to you apart from the time that you give.

Confidentiality No material that could personally identify you will be used in any reports of this study. Your blood samples will be labelled with a study code number	If you have any questions or concerns ab in this study you may wish to contact a F telephone:	oout your rights as a participant Health and Disability Advocate,
only. Study records will be stored on a password-protected computer and in a locked filing cabinet. Computers and files are kept in a restricted-entry facility, and will only be used by the study researchers. Records from the study will be kept for 10 years by the investigators, and then destroyed in a	0800 42 36 38 (0800	(4 ADNET)
confidential manner. Bosints of the study.	You may contact the researchers to ask a understand, or with questions regarding	about anything you do not ç this study:
The overall results of the study (from which you cannot be identified) will be published in scientific journals and presented at research meetings. You can request a copy of the study results by indicating your choice on the consent form. If you need further information please contact the Principal Investigator.	Principal Investigator or Dr Robert Weinkove Clinical Research Fellow Malaghan Institute of Medical Research PO Box 7060	Co-Investigator Professor John Carter Consultant Haematologist Blood & Cancer Centre Wellington Hospital
Ethical approval This study has received ethical approval from the Central Regional Ethics Committee.	Wellington Phone: (04) 499 6914 ext 851 Mobile: 021 2160135 Email: <u>rweinkove@malaghan.org.nz</u>	Wellington Phone: (04) 385 5999 ext 6756
The study team thank you for participating.	Compensation In the unlikely event of a physical injury as study, you may be covered by ACC under the	a result of your participation in this Injury Prevention, Rehabilitation and
	Compensation Act. ACC cover is not autom assessed by ACC according to the provisi Rehabilitation and Compensation Act. If your might not get any compensation. This deper whether you are an earner or non-earner. reimbursement of costs and expenses and there payable. There is no cover for mental injury un	natic and your case will need to be ons of the 2002 Injury Prevention c claim is accepted by ACC, you still nds on a number of factors such as ACC usually provides only partial e may be no lump sum compensation alless it is a result of physical injury. If
	If you have any questions about ACC, cont investigator.	tact your nearest ACC office or the
		3

Appendix C

Participant questionnaire





Questionnaire

iNKT cells in Chronic Lymphocytic Leukaemia (CLL) study

Name:		
Telephone: Date of birth: Email:	H: M:	
Please circle yo	ur answer to each question.	
Q1. Do you hav	ve CLL (chronic lymphocytic leukaemia)?	Yes / No
Q2. Do you hav	ve or have you ever had cancer (except CLL)?	Yes / No
Q3. Have you e	ever received chemotherapy or radiotherapy?	Yes / No
Q4. Are you tal medicines?	king or have you ever taken any of the following	g
Steroid tablets of hydrocortisone)	or injections (e.g. prednisolone, dexamethasone,	Yes / No
Other immune s tacrolimus, myc	suppression medicines (e.g. azathioprine, ciclospo ophenolate, cyclophosphamide)	rin, Yes / No
Q5. Do you hav	ve an infection at the moment?	Yes / No
This section of t	he questionnaire is for participants with CLL only:	
Q6. When was	your CLL diagnosed?	
Q7. Do you exp	perience drenching night sweats?	Yes / No
Q8. Have you h	ad unexplained weight loss in the past 6 mont	hs?
Q9. Do you exp	perience unexplained fevers?	Yes / No
Investigator use only:		

Appendix D

Consent forms





CONSENT FORM (patients)

iNKT cells in Chronic Lymphocytic Leukaemia (CLL) study

	Request for interpreter: Please circle your response below		•
English	I wish to have an interpreter.	Yes	No
Maori	E hiahia ana ahau ki tetahi kaiwhakamaori/kaiwhaka pakeha korero.	Ae	Kao
Cook Island	Ka inangaro au i tetai tangata uri reo.	Ae	Kare
Fijian	Au gadreva me dua e vakadewa vosa vei au.	lo	Sega
Niuean	Fia manako au ke fakaaoga e taha tagata fakahokohoko kupu.	Е	Nakai
Samoan	Ou te mana'o ia i ai se fa'amatala upu.	loe	Leai
Tokelaun	Ko au e fofou ki he tino ke fakaliliu te gagana Peletania ki na gagana o na motu o te Pahefika	loe	Leai
Tongan	Oku ou fiema'u ha fakatonulea.	lo	Ikai

Consent: Please circle your responses below		
I have read and understand the information sheet dated 08/06/2009 for patients taking part in the iNKT cells in CLL study. I have had the opportunity to discuss this study. I am satisfied with the answers I have been given.	Yes	No
I have had the opportunity to use whanau support or a friend to help me ask questions and understand the study.	Yes	No
I have had time to consider whether I will take part in this study.	Yes	No
I understand that taking part in this study is voluntary, that I may withdraw from the study at any time and that this would not affect my future healthcare.	Yes	No
I understand that my participation in this study is confidential and that no material that could identify me will be used in any reports on this study.	Yes	No
I know who to contact if I have any side effects after having my blood taken or have any questions about the study.	Yes	No
I agree to the researchers storing a specimen of my blood for its use in this study, or for extensions to this study approved by the Central Regional Ethics Committee.	Yes	No
I agree to any remaining samples being destroyed at the end of the study.	Yes	No
I understand that I should tell the researcher if I would like any remaining samples disposed with appropriate karakia at the end of the study.	Yes	No
I agree to the researchers sending an anonymous sample of my blood to Capital & Coast District Health Board to check my blood count	Yes	No
I agree to the researchers contacting me if my blood count result is unexpectedly abnormal.	Yes	No
I agree to the researchers sending anonymous samples of my blood to Capital & Coast District Health Board and Canterbury Health Laboratories to measure markers of immune function and test for genetic changes in the leukaemia cells.	Yes	No
I agree to the researchers reviewing relevant medical records to check that I am suitable for the study and to record information about the stage of my CLL.	Yes	No
I wish to receive a copy of the results of the study.	Yes	No





Date:	
Participant's signature:	
Study explained by:	
Study role:	
Signature:	
Date:	

Karakia requested:	
--------------------	--

Yes	
No	Γ

Thank you for your participation in this study

For further information please contact:

Dr Robert Weinkove: 04 499 6914 extension 851 Clinical Research Fellow Malaghan Institute of Medical Research, PO Box 7060, Wellington Email: <u>rweinkove@malaghan.org.nz</u>

Professor John Carter: 04 385 5999 extension 6756 Consultant Haematologist, Wellington Blood and Cancer Centre Wellington Hospital, Private Bag 7902, Wellington

If you have any queries or concerns regarding your rights as a participant in this study you may wish to contact a Health and Disability Advocate, telephone 0800 42 36 38 (0800 4 ADNET).

You may request a copy of this consent form for your own records.





CONSENT FORM (volunteers)

iNKT cells in Chronic Lymphocytic Leukaemia (CLL) study

Request for interpreter: Please circle your response below			
English	I wish to have an interpreter.	Yes	No
Maori	E hiahia ana ahau ki tetahi kaiwhakamaori/kaiwhaka pakeha korero.	Ae	Kao
Cook Island	Ka inangaro au i tetai tangata uri reo.	Ae	Kare
Fijian	Au gadreva me dua e vakadewa vosa vei au.	lo	Sega
Niuean	Fia manako au ke fakaaoga e taha tagata fakahokohoko kupu.	Е	Nakai
Samoan	Ou te mana'o ia i ai se fa'amatala upu.	loe	Leai
Tokelaun	Ko au e fofou ki he tino ke fakaliliu te gagana Peletania ki na gagana o na motu o te Pahefika	loe	Leai
Tongan	Oku ou fiema'u ha fakatonulea.	lo	Ikai

Consent: Please circle your responses below		
I have read and understand the information sheet dated 08/06/2009 for volunteers taking part in the iNKT cells in CLL study. I have had the opportunity to discuss this study. I am satisfied with the answers I have been given.	Yes	No
I have had the opportunity to use whanau support or a friend to help me ask questions and understand the study.	Yes	No
I have had time to consider whether I will take part in this study.	Yes	No
I understand that taking part in this study is voluntary, that I may withdraw from the study at any time and that this would not affect my future healthcare.	Yes	No
I understand that my participation in this study is confidential and that no material that could identify me will be used in any reports on this study.	Yes	No
I know who to contact if I have any side effects after having my blood taken or have any questions about the study.	Yes	No
I agree to the researchers storing a specimen of my blood for its use as a part of this study or extensions to this study approved by the Regional Ethics Committee.	Yes	No
I agree to any remaining samples being destroyed at the end of the study.	Yes	No
I understand that I should tell the researcher if I would like any remaining samples disposed with appropriate karakia at the end of the study.	Yes	No
I agree to the researchers sending an anonymous sample of my blood to Capital & Coast District Health Board to check my blood count and to measure markers of immune function.	Yes	No
I agree to the researchers contacting me if my blood count result is unexpectedly abnormal.	Yes	No
I wish to receive a copy of the results of the study.	Yes	No





Date:	
Participant's signature:	
Study explained by:	
Study role:	
Signature:	
Date:	

Karakia requested:

Yes	
No	

Thank you for your participation in this study

For further information please contact:

Dr Robert Weinkove: 04 499 6914 extension 851 Clinical Research Fellow Malaghan Institute of Medical Research, PO Box 7060, Wellington Email: <u>rweinkove@malaghan.org.nz</u>

Professor John Carter: 04 385 5999 extension 6756 Consultant Haematologist, Wellington Blood and Cancer Centre Wellington Hospital, Private Bag 7902, Wellington

If you have any queries or concerns regarding your rights as a participant in this study you may wish to contact a Health and Disability Advocate, telephone 0800 42 36 38 (0800 4 ADNET).

One copy of this consent form is given to you for your own records.

Appendix E

Publications

Publications resulting from work presented in this thesis:

Weinkove R, Brooks CR, Carter JM, Hermans IF, Ronchese F. Functional invariant natural killer T cell and CD1d axis in chronic lymphocytic leukemia: implications for immunotherapy. *Haematologica* 2012 **97**:xxx [*Epub ahead of print*] doi: 10.3224/haematol.2012.072835

Publications contributed to during the PhD studies:

Hunn MK, Farrand KJ, Broadley KWR, **Weinkove R**, Ferguson P, Miller RJ, Field CS, Petersen T, McConnell MJ, Hermans IF. Vaccination with irradiated tumor cells pulsed with an adjuvant that stimulates NKT cells is an effective treatment for glioma. *Clinical Cancer Research* 2012; **18(23)**: 1-14.

Dangerfield EM, Cheng JM, Knight DA, **Weinkove R**, Dunbar PR, Hermans IF, Timmer MS, Stocker BL. Species-specific activity of glycolipid ligands for invariant NKT cells. *Chembiochem* 2012;**13(9)**:1359-56.

Brooks CR, **Weinkove R**, Hermans IF, van Dalen CJ, Douwes J. Invariant natural killer T cells and asthma: immunologic reality or methodologic artifact? *Journal of Allergy and Clinical Immunology* 2010;**126(4)**:882-885.